Improving our understanding of the use of online food delivery services to access food prepared out-of-home

Matthew George Keeble

Wolfson College

MRC Epidemiology Unit, University of Cambridge

November 2022

This thesis is submitted for the degree of Doctor of Philosophy





Declaration

This thesis is the result of my own work under the supervision of Dr Thomas Burgoine and Professor Jean Adams and includes nothing which is the outcome of work done in collaboration except as declared in the preface and specified in the text. This thesis is not substantially the same as any work that has already been submitted before for any degree or other qualification except as declared in the preface and specified in the text. Finally, it does not exceed the prescribed word limit for the Clinical Medicine and Veterinary Medicine Degree Committee.

Matthew George Keeble

22 November 2022

Abstract

Food prepared out-of-home is typically energy-dense and nutrient-poor. More frequent consumption of this food is associated with poorer overall dietary patterns and obesity. Previous research has tended to focus on access to food prepared out-of-home in the physical food environment, which is often greatest in more deprived areas. However, this food can now also be accessed through online food delivery services. Although online food delivery services are globally established, there is limited public health knowledge about the prevalence of their use and factors influencing this, the sociodemographic characteristics of customers, and the extent to which the opportunity to use them is socioeconomically patterned. The aim of my thesis was to help better understand the use of online food delivery services to access food prepared outof-home.

First, I analysed survey data collected in 2018 to identify the prevalence of online food delivery service use and the sociodemographic characteristics of adult customers across Australia, Canada, Mexico, the United Kingdom, and the United States of America. From 19,378 respondents, around one in six had used an online food delivery service in the past week. Respondents who were male, younger, more highly educated, living with children and those who identified with an ethnic minority had greater odds of online food delivery service use. These patterns were similar in each country but the strength of associations varied. Notably, respondents with a high versus low level of education had greater odds of online food delivery service use service use in all countries except the United Kingdom.

Second, I conducted telephone interviews with 22 adults living in England who were frequent online food delivery service customers. Participants reported that they could access a higher number of food outlets and a broader range of cuisines through online food delivery services compared with the physical food environment. Additionally, these services allowed participants to access exclusive price-promotions and use streamlined purchasing processes, which were seen as unique advantages. Participants reported that they believed the food available through online food delivery services was mostly unhealthy. Nevertheless, this food met their expectations about `takeaway food`. Despite reported drawbacks of online food delivery services, which included over-convenience and, paradoxically, access to unhealthy food, participants reported no intention to discontinue use because it was normal to purchase food prepared out-of-home in this manner. Informed by existing evidence and my qualitative research, I investigated associations between access to food prepared out-of-home through online food delivery services and online food delivery service use, and body weight. In my individual-level data-linkage study of 3067 adults living in Great Britain, the number of food outlets accessible online was positively associated with online food delivery service use in the past week. However, it was not associated with body weight. Despite the perspective of frequent customers, the number of cuisine types accessible online was not associated with online food delivery service use.

Finally, I examined access to food prepared out-of-home through online food delivery services in England and variation according to area-level socioeconomic position. In a cross-sectional study, I found that the number of food outlets accessible online was highest in the most deprived areas of England in November 2019. I followed this with a repeat-cross sectional study that assessed changes in the number of food outlets accessible online over time, within the context of the Coronavirus Disease 2019 (COVID-19) pandemic. I found that it was only in the most deprived areas of England that the number of food outlets accessible online in March 2022 surpassed the number from November 2019. Taken together, these findings indicate that online access to food prepared out-of-home in England is socioeconomically patterned, and that inequalities therein widened over time.

The findings from my thesis help better understand multiple aspects of online food delivery service use. The number of food outlets accessible online emerged as being particularly important. A higher number was positively associated with online food delivery service use, which suggests it can influence food purchasing practices. Indeed, this was supported by the views of frequent online food delivery service customers in my qualitative research. Additionally, the number of food outlets accessible online was highest in the most deprived areas of England in November 2019, and increasingly so over time compared with less deprived areas. Thus, the opportunity for online food delivery service use is unequal across the socioeconomic spectrum in England. Future research should seek to understand how food purchased through online food delivery services contributes to overall dietary patterns and health, and the need for public health intervention.

List of relevant publications, contributions, and dissemination

Manuscripts included in my thesis and dissemination of findings

Chapter 2 is published in the *International Journal of Environmental Research and Public Health* as: Keeble, M., Adams, J., Sacks, G., Vanderlee, L., White, C.M., Hammond, D., Burgoine, T. (2020), Use of online food delivery services to order food prepared away-from-home and associated sociodemographic characteristics: a cross-sectional, multi-country analysis, https://doi.org/10.3390/ijerph17145190. I presented the findings from this research at the 2020 International Society of Behavioral Nutrition and Physical Activity Annual Congress during an online symposium: "New ways of purchasing (fast) food: are we transitioning away from tradition?".

Chapter 3 is published in *BMC Public Health* as: Keeble, M., Adams, J., Burgoine, T. (2022), Investigating experiences of frequent online food delivery service use: a qualitative study in UK adults, https://doi.org/10.1186/s12889-022-13721-9. An abstract for this chapter is also published in *The Journal of Epidemiology and Community Health* as: Keeble, M., Burgoine, T., Adams, J. (2022), P14 Investigating frequent online food delivery service use: a qualitative study in UK adults, https://dx.doi.org/10.1136/jech-2022-SSMabstracts.113. I shared the findings from this research during an oral presentation at the 2021 UK Society for Behavioural Medicine Annual Scientific Meeting (held online), and a poster presentation at the 2022 Society for Social Medicine Annual Scientific Meeting, held in-person in Exeter, United Kingdom.

Chapter 4 is published in *BMC Public Health* as: Keeble, M., Adams, J., Hammond, D., Vanderlee, L., Burgoine, T. (2021), Associations between online food outlet access and online food delivery service use amongst adults in the UK: a cross-sectional analysis of linked data, https://doi.org/10.1186/s12889-021-11953-9. I shared the findings from this research during an oral presentation at the 2022 International Medical Geography Symposium, held in-person in Edinburgh, United Kingdom.

Chapter 5 is published in *Applied Geography* as: Keeble, M., Adams, J., Bishop, T.R.P., Burgoine, T. (2021), Socioeconomic inequalities in food outlet access through an online food delivery service in England: a cross-sectional descriptive analysis,

https://doi.org/10.1016/j.apgeog.2021.102498. I shared the findings from this research during oral presentations at the 2021 American Association of Geographers Annual Meeting (held online), the 2021 Public Health Research and Science Conference (held online), and the 2022 International Medical Geography Symposium, held in-person in Edinburgh, United Kingdom. **Chapter 6** is under review at *JMIR Public Health and Surveillance* as: Keeble, M., Adams, J., Burgoine, T. Changes in online food access during the COVID-19 pandemic and associations with deprivation: a repeat-cross sectional analysis. I shared the findings from this research during an oral presentation at the 2022 International Medical Geography Symposium, held inperson in Edinburgh, United Kingdom, and a poster presentation at the 2022 UK Public Health Science National Conference, held in-person in Glasgow, United Kingdom. For the latter, the abstract is published in *The Lancet* as: Keeble, M., Adams, J., Burgoine, T. (2022), Changes in online food access during the COVID-19 pandemic and associations with deprivation: a longitudinal analysis, https://doi.org/10.1016/S0140-6736(22)02264-4. I will also share the findings from this research during an oral presentation at the 2023 American Association of Geographers Annual Meeting, held in-person in Denver, United States of America.

Contributions to publications not included in my thesis

I have contributed to the following research that is under review or published, as well as published grey-literature:

- I. Bennett, R., **Keeble, M.**, Zorbas, C., Sacks, G., Driessen, C., Grigsby-Duffy, L., Adams, J., Burgoine, T., Backholer, K. The potential influence of the digital food retail environment on diet and health: a systematic scoping review of the literature. Under review at *Obesity Reviews*.
- II. Goodman, S., Armendariz, G., Corkum, A., Arellano, L., Jauregui, A., Keeble, M., Marshall, J., Sacks, G., Thrasher, J., Vanderlee, L., White, C.M., Hammond, D. (2021), Recall of government healthy eating campaigns by consumers in five countries, *Public Health Nutrition*, https://doi.org/10.1017/S1368980021001415.
- III. Yau, A., Singh-Lalli, H., Forde, H., Keeble, M., White, M., Adams, J. (2021), Newspaper coverage of food insecurity in UK, 2016-2019: a multi-method analysis, *BMC Public Health*, https://doi.org/10.1186/s12889-021-11214-9.
- IV. Food Foundation (2021 and 2022): "Broken Plate Report 2021". Available from: https://foodfoundation.org.uk/wp-content/uploads/2021/07/FF-Broken-Plate-2021.pdf, and "Broken Plate Report 2022". Available from: https://foodfoundation.org.uk/publication/broken-plate-2022.
- V. **Obesity Health Alliance** (2021): "Turning the Tide: A 10-year Healthy Weight Strategy". Available from: https://aso.org.uk/news/turning-tide-10-year-healthy-weight-strategy.
- VI. **Public Health England** (2020): "Using the planning system to promote healthy weight environments. Guidance and supplementary planning document template for local authority public health and planning teams". Available from: https://www.gov.uk/government/publications/healthy-weight-environments-using-the-planning-system.

Invited speaker

I was invited to speak at the following online events to present the findings from the research included in my thesis:

- I. Nutrition and Obesity Policy Research and Evaluation Network (NOPREN) (2021): "Online fast-food retail in the UK: research progress and future challenges".
- II. Public Health England Healthy Places Spatial Planning and Obesity Symposium (2021): "Evaluation of planning policy, planning appeals, exclusion zones and online food delivery services".
- III. The Association for the Study of Obesity Webinar: Obesity and the Environment- where are we now? (2021): "Investigating online access to takeaway food".
- IV. Cambridge Global Food Security Coffee Break Seminar Series (2021): "New ways of buying takeaway food: what do we know about online food delivery services?".

Awards and collaborations

- I. Winner of the National Institute for Health and Care Research (NIHR) School for Public Health Research three-minute thesis video competition.
- II. Collaboration established with Early Career Researchers at the University of Newcastle, University of Hertfordshire, and Lancaster University to apply for NIHR Applied Research Collaboration funding. In November 2022, we submitted a proposal to investigate consumer perspectives on the appropriateness and acceptability of a Takeaway Masterclass intervention.
- III. Collaboration established with policy and practice partners from the United Kingdom and the United States of America to host a workshop at the 2023 American Planning Association National Planning Conference: "Spatial planning and community development for healthier diets".

Acknowledgements

Science is a team sport. I would not be presenting this thesis without the team I have had behind me over the past four years.

I am extremely grateful to my supervisors, Tom Burgoine and Jean Adams. Under their supervision, I have been provided with innumerable opportunities to develop as a researcher and an individual. Their patience, direction, and energy made my PhD journey stimulating and immensely enjoyable. I would also like to thank them for teaching me to be ambitious and to persevere, especially when we were adapting to a new way of working and studying during a global pandemic.

To the multiple researchers that I have had the opportunity to collaborate with: Tom Bishop, David Hammond, Gary Sacks, Lana Vanderlee, and Christine White, thank you for allowing me to learn from each of you. I would also like to thank colleagues in the Population Health Interventions group at the MRC Epidemiology Unit for entertaining my presentations and poster designs, and for showing a genuine interest in my research. CEDAR (as it will always be in my mind) is a truly special place to work and study.

Aspects of my thesis rely on already collected data. I would like to thank the many participants who provided it and the researchers who helped to collect it. Thank you also to the participants in the primary research I conducted.

Finally, thank you to my friends and family for their ongoing support. I want to give special thanks to my sisters, Anna and Charlotte, for humouring my weekly phone calls, being as excited about the small wins as I have been and acting as the voice of reason; and to my parents for their unconditional love and the sacrifices they have made for me.

Funding

The research presented in my thesis was supported by the National Institute for Health and Care Research (NIHR) School for Public Health Research [grant number PD_SPH_2015] and the Medical Research Council [grant number MC_UU_00006_7].

Table of contents

Chapter 1: Introduction	1
1.1 The link between food, diet, obesity, and health, and inequalities therein	1
1.2 Food prepared out-of-home	3
1.2.1 Definitions of food prepared out-of-home	3
1.2.2 The popularity of food prepared out-of-home	4
1.2.3 The energy and nutrient content of food prepared out-of-home	4
1.2.4 Public health concerns about the consumption of food prepared out-of-home	5
1.2.4.1 Associations between the consumption of food prepared out-of-home and diet and body weight	5
1.2.5 Factors proposed to influence the consumption of food prepared out-of-home.	6
1.2.5.1 The socioecological model	6
1.2.5.2 A conceptual framework of food access	7
1.2.5.3 Socioeconomic position	8
1.2.5.4 Individual-level sociodemographic characteristics1	0
1.3 The digital food environment1	2
1.3.1 A definition for the digital food environment1	2
1.3.2 Previous research in the context of the digital food environment	2
1.4 Online food delivery services1	3
1.4.1 A definition for online food delivery services	3
1.4.2 The global presence and popularity of online food delivery services	3
1.4.3 Purchasing food prepared out-of-home through online food delivery services 1	4
1.4.4 A proposed pathway between the use of online food delivery services and bod weight	/ 6
1.4.5 Online access to food prepared out-of-home1	7
1.4.5.1 Access to food prepared out-of-home through online food delivery services and inequalities therein	; 8
1.4.5.2 Online food delivery service use and sociodemographic characteristics of customers2	0
1.4.5.3 Online food delivery service use and body weight	1
1.5 Summary of the Introduction2	2
1.6 Thesis aims2	2
1.7 Thesis structure2	3
Chapter 2: Multi-country prevalence of online food delivery service use and	
sociodemographic characteristics of customers	4
2.1 Details of author contributions, publication and dissemination2	4
2.2 Abstract	5
2.3 Introduction2	6
2.3.1 Study aims2	7
2.4 Methods2	7
2.4.1 Data collection2	7
2.4.2 Measures	7
2.4.2.1 Purchasing format use2	7
2.4.2.2 Sociodemographic characteristics	8
2.4.3 Study sample	9
2.4.4 Statistical analysis2	9

2 4 4 1 Descriptive analyses	20
2.4.4.2 Inferential analyses	20
2.5 Results	30
2.5.1 Sociodemographic characteristics	30
2.5.2 Meals nurchased out-of-home	31
2.5.2 Nears purchased out of nome manner food delivery service use	34
2.5.5 Sociodemographic correlates of online food derivery service use	25
2.6 Discussion	
2.6 1 Summary of findings	40
2.6.2 Interpretation of findings	40
2.6.2 Interpretation of findings	41 42
	44
Chapter 3: Experiences and perspectives of frequent online food delive customers in the United Kingdom	ery service 45
3.1 Details of author contributions, publication and dissemination	45
3.2 Abstract	46
3.3 Introduction	47
3.3.1 Study aims	48
3.4 Methods	48
3.4.1 Methodological orientation	48
3.4.2 Participants and recruitment	48
3.4.3 Data collection	
3.4.3.1 Before data collection	
3.4.3.2 Throughout data collection	
3.4.3.3 After data collection	
3.4.4 Data analysis	
3.5 Results	
3.5.1 Participant and data overview	
3.5.2 Summary and structure of findings	
3 5 3 Overarching concepts	54
3 5 3 1 Place Time Situation	54
3 5 3 2 Perceived advantages outweigh recognised drawbacks	56
3 5 4 Analytic themes	57
3 5 4 1 The importance of takeaway food	57
3.5.4.2 Less effort for more convenience	58
3.5.4.2 Less enorthor more convenience	60
3.5.4.4 Online food delivery service normalisation	62
3.5.4.4 Online food derivery service normalisation	02 62
2.6 Discussion	02 65
2.6.1 Summary of findings	
2.6.2 Interpretation of findings	סכ دר
2.6.2 Describle implications for public health and future research	כס
2.6.4 Strongthe and limitations	

4.1 Details of author contributions, publication and dissemination	70
4.2 Abstract	71
4.3 Introduction	72
4.3.1 Study aims	73
4.4 Methods	73
4.4.1 Study design	73
4.4.2 Study sample	73
4.4.3 Measures	73
4.4.3.1 Exposures: online food outlet access and unique type of cuis	ine access73
4.4.3.2 Primary outcome: online food delivery service use	74
4.4.3.3 Secondary outcomes: body mass index and weight status	75
4.4.3.4 Potential confounders: sociodemographic characteristics and neighbourhood food outlet access	75
4.4.4 Exclusion criteria	76
4.4.5 Statistical analysis	77
4.4.5.1 Sensitivity analyses	77
4.4.5.2 Secondary outcomes	78
4.5 Results	78
4.5.1 Sample characteristics	78
4.5.2 Associations between online food outlet access and online food d use in the past week	elivery service 80
4.5.3 Interactions between online food outlet access and sociodemogra characteristics	aphic 81
4.5.4 Sensitivity analyses	83
4.5.5 Associations between unique types of cuisine accessible online ar delivery service use	nd online food
4.5.6 Secondary analyses: associations between online food outlet acc	ess and body
weight	84
4.6 Discussion	84
4.6.1 Summary of findings	84
4.6.2 Interpretation of findings	
4.6.3 Possible implications for public health and future research	
4.6.4 Strengths and limitations	
4.7 Conclusions	87
Chapter 5: The socioeconomic patterning of online access to food p of-home in England	repared out- 89
5.1 Details of author contributions, publication and dissemination	89
5.2 Abstract	90
5.3 Introduction	91
5.3.1 Study aims	93
5.4 Methods	93
5.4.1 Study setting and analytic scale	93
5.4.2 Exposure measure	93
5.4.3 Outcome measures	94
5.4.3.1 Online food access	94
5.4.3.2 Percentage of food outlets registered to accept orders online	95
5.4.3.3 Percentage of neighbourhood food outlets accessible online	95
5.4.4 Covariates	96

5.4.4.1 Physical food outlet access	96
5.4.4.2 Online food outlet access	96
5.4.4.3 Rural Urban classification and population density	96
5.4.5 Statistical analysis	98
5.4.5.1 Sensitivity analyses	98
5.5 Results	99
5.5.1 Food outlet access across England	99
5.5.2 Association between deprivation and the percentage of food outlets	registered
to accept orders online	102
5.5.3 Association between deprivation and online food outlet access	
5.5.4 Association between deprivation and unique cuisine type access	
5.5.5 Association between deprivation and the percentage of neighbourhor outlets accessible online	ood food
5.5.6 Sensitivity analyses	106
5.6 Discussion	106
5.6.1 Summary of findings	106
5.6.2 Interpretation of findings	106
5.6.3 Possible implications for public health and future research	
5.6.4 Strengths and limitations	
5.7 Conclusions	109
Chapter 6: Changes in online access to food prepared out-of-home in over time and associations with socioeconomic position	England 110
6.1 Details of author contributions, publication status and dissemination	
6.2 Abstract	111
6.3 Introduction	113
6.3.1 Access to food prepared out-of-home during the first two years of t 19 pandemic	he COVID- 113
6.3.2 Possible implications for access to food prepared out-of-home throu food delivery services	igh online
6 3 3 Study aims	117
6 4 Methods	117
6.4.1 Study setting, period, and analytic scale	
6.4.2 Exposure measure	
6.4.3 Outcome measures: data source and collection	
6.4.3.1 Number of food outlets registered to accept orders online	
6.4.3.2 Number of food outlets accessible online	
6.4.3.3 Percentage of food outlets registered to accept orders online	118
6.4.4 Covariates	119
6.4.4.1 Index of multiple deprivation	120
6.4.4.2 Rural Urban classification and population density	120
6.4.5 Statistical analysis	120
6.4.5.1 Descriptive statistics	121
6.4.5.2 Inferential statistics	121
6.4.5.3 Associations with deprivation	121
6.5 Results	100
6.5.1 Number of food outlets registered to accept orders online	

6.5.3 Percentage of food outlets registered to accept orders online	. 127
6.6 Discussion	. 128
6.6.1 Summary of findings	. 128
6.6.2 Interpretation of findings	. 129
6.6.3 Possible implications for public health and future research	. 131
6.6.4 Strengths and limitations	. 132
6.7 Conclusions	. 132
Chapter 7: Discussion	134
7.1 Summary and interpretation of principal findings	. 134
7.2 Methodological considerations	. 136
7.2.1 Confounding and reverse causality	. 136
7.2.2 Generalisability	. 137
7.2.3 Investigating multiple domains of food access	. 138
7.2.4 Approach to, and scale of, data collection	. 138
7.2.5 Purposeful exposure to online food delivery services	. 139
7.3 Possible implications for public health and policy	. 140
7.3.1 The interdependency of the physical and digital food environments	. 140
7.3.2 A public health intervention involving multiple stakeholders	. 141
7.4 Future research agenda	. 142
7.4.1 Cumulative access to food prepared out-of-home	. 143
7.4.2 Supplementation or substitution of alternative purchasing formats and food	d
sources	.144
7.4.3 Moving beyond the frequency of online food delivery service use	. 144
7.4.4 Online food delivery service diversification	. 145
7.4.5 Changes to the location of food preparation	. 146
Chapter 8: Conclusions	147

List of tables

Table 1.1 : Name and definition of the domains of access in relation to the physicalfood environment.7
Table 1.2 : Name and definition of the domains of food access in the context ofonline food delivery services.18
Table 2.1 : Sociodemographic characteristics of online food delivery servicecustomers amongst the analytic sample (n=19,378) from the 2018 InternationalFood Policy Study.32
Table 2.2 : Purchasing formats used to acquire meals prepared out-of-home in thepast week amongst the analytic sample (n=19,378) from the 2018 InternationalFood Policy Study.33
Table 2.3 : Associations between country and prevalence of any online fooddelivery service use in the past week amongst the analytic sample ($n=19,378$) fromthe 2018 International Food Policy Study
Table 3.1 : Sociodemographic characteristics and online food delivery service useamongst frequent customers (n=22) interviewed between June and August 2020 53
Table 3.2 : Examples of how frequent online food delivery service customers(n=22) compared the advantages and drawbacks of features of the online fooddelivery service business model, identified from data collected between June andAugust 2020.64
Table 4.1 : Prevalence of online food delivery service use, exposure to foodprepared out-of-home, and sociodemographic characteristics of the analytic sample(n=3067) from the 2019 International Food Policy Study
Table 5.1 : Summary of measures and the exposure: outcome relationshipsinvestigated.97
Table 5.2 : Summary of measures for the online food delivery service and thephysical food environment across postcode districts in England (n=2118), stratifiedby deprivation decile.100
Table 6.1 : Descriptive summary of online access to food prepared out-of-homeacross postcode districts in England ($n=2118$), at baseline (November 2019) andthe end of the study period (March 2022), stratified by deprivation quintile

List of figures

Figure 1.1: Differences in the consumption of fruits and vegetables across age groups and income quintiles in England
Figure 1.2: Differences in the prevalence of adult obesity in England across income quintile
Figure 1.3: Socioecological model of the proposed determinants of health
Figure 1.4 : Relationship between Local Authority level of deprivation and the density of fast-food outlets
Figure 1.5 : The role of individual online food delivery services, food businesses, and customers in the process of purchasing food prepared out-of-home through the online food delivery service business model
Figure 1.6 : Proposed causal pathway between online food access (access to food prepared out-of-home through online food delivery services) and body weight
Figure 2.1 : Associations between prevalence of any online food delivery service use in the past week and sociodemographic characteristics, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study
Figure 2.2 : Associations between prevalence of any online food delivery service use in the past week and age, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study
Figure 2.3 : Associations between prevalence of any online food delivery service use in the past week and living with children aged under 18 years, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study
Figure 2.4 : Associations between prevalence of any online food delivery service use in the past week and sex, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study
Figure 2.5 : Associations between prevalence of any online food delivery service use in the past week and education level, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study
Figure 3.1 : Inclusion criteria used during frequent online food delivery service customer recruitment between June and August 2020
Figure 4.1 : Odds of any online food delivery service use in the past week per quarter of online food outlet access amongst the analytic sample (n=3067) from the 2019 International Food Policy Study
Figure 4.2 : Odds of any online food delivery service use in the past week per quarter of online food outlet access amongst analytic sample (n=3067) from the 2019 International Food Policy Study, stratified by sociodemographic
characteristics
Figure 4.3 : Odds of any online food delivery service use in the past week per quarter of online access to unique types of cuisine amongst the analytic sample (n=3067) from the 2019 International Food Policy Study
Figure 5.1 : Deciles of the number (count) of food outlets accessible online across postcode districts (n=2118) in England in November 2019101
Figure 5.2 : Percentage of food outlets registered to accept orders online across postcode districts (n=2084) in England in 2019102
Figure 5.3: Number of food outlets accessible online across postcode districts (n=2088) in England in 2019
Figure 5.4 : Number of unique cuisine types accessible online across postcode districts (n=2088) in England in 2019

Figure 5.5 : Number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood across postcode districts (n=2076) in England in 2019
Figure 6.1 : Details of selected emergency regulations introduced, action taken, and broader events in England in relation to the COVID-19 pandemic and food outlets selling food prepared out-of-home
Figure 6.2 : Count of food outlets registered to accept orders online in England between baseline (November 2019) and the end of the study period (March 2022)122
Figure 6.3 : Estimated mean number (count) of food outlets registered to accept orders online in England between November 2019 and March 2022, stratified by deprivation quintile
Figure 6.4 : Estimated mean number (count) of food outlets accessible online in England between November 2019 and March 2022, stratified by deprivation quintile
Figure 6.5 : Estimated mean percentage (%) of food outlets registered to accept orders online in England between November 2019 and March 2022, stratified by derivation quintile
Figure 7.1 : Contribution to knowledge for the proposed causal pathway between online food access (access to food prepared out-of-home through online food delivery services) and body weight

List of appendices

Appendix A: Supplementary material for Chapter 2	178
Appendix B: Supplementary material for Chapter 3	
Appendix C: Supplementary material for Chapter 4	185
Appendix D: Supplementary material for Chapter 5	191
Appendix E: Supplementary material for Chapter 6	197

List of abbreviations

BMI	Body mass index
CI	Confidence intervals
COVID-19	Coronavirus Disease 2019
GEE	Generalised estimating equation
GIS	Geographic information system
IFPS	International Food Policy Study
IMD	Indices of Multiple Deprivation
IQR	Interquartile range
IRR	Incidence rate ratios
LSOA	Lower super output area
MAUP	Modifiable areal unit problem
OR	Odds ratio
OS POI	Ordnance Survey Points of Interest
SD	Standard deviation
UK	United Kingdom
USA	United States of America
WHO	World Health Organization

Chapter 1: Introduction

In my thesis, I will present research on the use of online food delivery services to access food prepared out-of-home. Before defining these services and discussing why they may be a concern for public health, I provide relevant background and context for my research. I begin this chapter by describing the relationship between food and diet, obesity, and health. Next, I define food prepared out-of-home and discuss its energy and nutrient content, established public health concerns about the purchase and consumption of this food, and factors that plausibly influence this. After defining the digital food environment and discussing existing evidence, I focus on the use of online food delivery services. In the sections that follow, I outline why knowledge about the use of these services is important in the context of food prepared out-of-home and public health. I conclude this chapter by stating the overall aims and structure of my thesis.

1.1 The link between food, diet, obesity, and health, and inequalities therein

All foods, including those that can be prepared and cooked at home and those that are sold ready-to-consume, often purchased outside of the home, contribute to overall dietary patterns (1, 2). Overall dietary patterns contribute to health, with greater consumption of micronutrient rich foods, such as fruits, vegetables, legumes, pulses, and nuts linked to overall dietary patterns that can be health protecting (3, 4), and lead to less weight gain over time (5). However, poorer overall dietary patterns have been associated with the onset of cardiovascular diseases and several non-communicable diseases and cancers (6-8). Promoting healthier food consumption is therefore a recognised public health priority (9, 10). This is especially important because population-wide dietary patterns often fail to meet published dietary guidelines (11). Additionally, over 600 million adults globally and 28% of adults in England were estimated to be living with obesity between 2015 and 2019 (12, 13).

Individuals with less access to social and economic resources, and lower income and levels of education consistently have poorer diets and health compared with those who have greater resource access, and higher income and levels of education (14, 15). Figure 1.1, published by the National Food Strategy (an independent review of the United Kingdom (UK) food system) using data from the National Diet and Nutrition Survey collected between 2014 and 2016, demonstrates that in both children and adults, those living in lower income households (i.e. more deprived) generally consume fewer fruits and vegetables (16). There are also differences in

obesity prevalence across the socioeconomic spectrum, with adults living in more deprived areas reported to have a higher body mass index (BMI) (17). Figure 1.2, also published by the National Food Strategy, shows that the proportion of adults with either overweight or obesity was highest amongst those in the most deprived areas of England in 2019 (16). Together, Figure 1.1 and Figure 1.2 indicate that differences in food consumption and obesity prevalence are unevenly distributed across the socioeconomic spectrum in England.



Figure 1.1: Differences in the consumption of fruits and vegetables across age groups and income quintiles in England.

Notes: Q = quintiles. F&V = fruits and vegetables. Reproduced from the National Food Strategy, published in 2022, based on data from the National Diet and Nutrition Survey (16).



Figure 1.2: Differences in the prevalence of adult obesity in England across income quintile. Notes: Reproduced from the National Food Strategy, published in 2022, based on data from the Health Survey for England and the English Indices of Multiple Deprivation (16).

Olstad and Kirkpatrick (18), amongst others (19, 20), propose that the selection of one type of food over another is a response to environmental factors and cultural contexts. In the physical food environment, food can be purchased from retail outlets such as convenience stores, supermarkets, restaurants, and hot food takeaway outlets, each of which operate with customer-facing premises that often co-locate (21-23). The focus of my thesis is the purchase of food prepared out-of-home, specifically, through online food delivery services, which I will discuss in Section 1.4.3.

1.2 Food prepared out-of-home

In the next sections, I will first define food prepared out-of-home. I will then discuss the popularity of this food (Section 1.2.2), its energy and nutrient content (Section 1.2.3), and public health concerns regarding its consumption (Section 1.2.4).

1.2.1 Definitions of food prepared out-of-home

Heterogeneity in definitions of food prepared out-of-home has contributed to it being a poorly defined food category (24). In one scoping review, no consistent definition was used across 57 included articles (25), with further reviews reporting similar findings (26, 27). For my thesis, I have considered food prepared out-of-home to be a broad category that includes meals (rather than snacks) served ready-to-consume for either on-premises or off-premises consumption. This

food is sold by many retailers in the physical food environment, including hot food takeaway outlets and restaurants (28, 29).

1.2.2 The popularity of food prepared out-of-home

The number of chain and independent outlets selling food prepared out-of-home has increased over time in the UK (30). In one area of England (Norfolk), the number of hot food takeaway outlets increased by 45% between 1990 (n=265) and 2008 (n=385) (31). This evidence was from a single area and for a single type of food outlet selling food prepared out-of-home, which does not capture the full range of outlets from which this food can be accessed. Nevertheless, between 2004 and 2018, the number of a broader range of food outlets selling this food increased at the national level in the Netherlands (32), and Mexico (33), and over large subnational geographies in Australia (34), and New Zealand (35).

The increased supply of food prepared out-of-home is partly a reflection of increased demand. In many countries, it is popular to purchase and consume this food (36), with greater consumption and expenditure over time reported in the United States of America (USA) (37), and Australia (19). For the UK, estimated revenue for out-of-home food retail exceeded £20 billion in 2022, compared with around £18 billion in 2012 (38). Further growth has also been forecast (39). In addition, national dietary surveillance data collected between 2008 and 2012 indicated that over one fifth of adults consumed food prepared out-of-home inside a food outlet (27%), or at home (21%), once per week or more (40).

1.2.3 The energy and nutrient content of food prepared out-of-home

Food prepared out-of-home is often high in calories, with meals sold in portion sizes that exceed public health recommendations for single-meal energy content (41, 42). Moreover, a single meal can provide almost all of, or more than, the average recommended daily energy requirements for adults in the UK (43). Food prepared out-of-home also tends to be high in total fat, saturated fat, salt and in some cases, sugar (44). Amongst comparable menu items from chain restaurants in the UK and the USA, 97% and 96%, respectively, exceeded public health recommendations for at least one of salt, fat, saturated fat, or sugars (45). Moreover, there is evidence that it would be difficult to consume food prepared out-of-home and adhere to the recommended upper-limit for daily salt intake in the UK (6 g for adults) (46). Although the majority of food prepared out-of-home is energy-dense and nutrient-poor (47, 48), in some instances foods lower in calories and fat and salt are available (41). Nevertheless, availability does not necessarily equate to purchasing. In one study, only a minority of meals purchased out-of-home over a two-week period were considered healthy according to the author's

classification (meals containing less than 693 calories, less than 2% saturated fat, virtually no trans-fat, and at least 75 g of vegetables) (49).

1.2.4 Public health concerns about the consumption of food prepared out-of-home

There is evidence from systematic and narrative reviews that more frequent consumption of food prepared out-of-home is associated with poorer overall dietary patterns and health, and living with a higher body weight (29, 50-53). This evidence has contributed to the publication of documents that outline public health concerns regarding the consumption of this food (54-56). In the next section, I will briefly outline evidence for diet and body weight.

1.2.4.1 Associations between the consumption of food prepared out-of-home and diet and body weight

Across four extensively validated indicators (the Healthy Eating Index, Diet Quality Index, Healthy Diet Indicator and Mediterranean Diet Score), overall trends indicate that more frequent consumption of food prepared out-of-home is associated with poorer overall dietary patterns (57-60). For the Healthy Eating Index, Barnes and colleagues reported that consumption of food prepared out-of-home was inversely associated with intake of vegetables, whole grains, and fibre and positively associated with overall daily energy intake (58). Similar trends for energy intake have been reported amongst adolescents living in the UK (59). Evidence that the consumption of food prepared out-of-home contributes to poorer overall dietary patterns is important given the link between diet and health that I discussed in Section 1.1.

Evidence of an association between the consumption of food prepared out-of-home and body weight is inconsistent (61). A mixed evidence base partly reflects that previous research in this area has often been cross-sectional. This study design provides a limited opportunity to understand the temporality of associations (62). The importance of study design was demonstrated in the findings of one systematic review that included 16 studies (63). A positive association between more frequent consumption of food prepared out-of-home (`fast food` in this instance) and BMI was reported across seven prospective cohort studies, but rarely across six cross-sectional studies (63). Although several other systematic reviews have reiterated that the evidence base taken as a whole is equivocal, they also report a trend between more frequent and greater overall consumption of food prepared out-of-home and elevated body weight (36, 52, 64). While recognising that further research is required, it has been suggested that there is sufficient evidence of an association with body weight to raise public health concerns about the consumption of food prepared out-of-home (29, 63, 65).

1.2.5 Factors proposed to influence the consumption of food prepared out-of-home

Moving on from public health concerns about food prepared out-of-home, in the next sections, I discuss factors that are proposed to influence its purchase and consumption.

1.2.5.1 The socioecological model

Socioecological models provide a framework that can help us to understand factors that act at different levels to influence health (66). Figure 1.3 is a widely-cited model reproduced from Dahlgren and Whitehead (67). This model illustrates proposed determinants of health that overlap and interact with one another, including area-level determinants such as socioeconomic and environmental factors, and individual-level determinants such as sociodemographic factors (68). Given that socioecological models can help to understand the complexity of health determinants, it has been reproduced and applied to the context of food purchasing and consumption (69-72). I use the Dahlgren and Whitehead model as a framework to consider factors that plausibly contribute to the purchase and consumption of food prepared out-of-home.



Figure 1.3: Socioecological model of the proposed determinants of health. Note: Reproduced from Dahlgren and Whitehead (67).

1.2.5.2 A conceptual framework of food access

Penchansky and Thomas described how healthcare access encapsulates multiple overlapping domains: availability, accessibility, affordability, accommodation, and acceptability (73). Away from healthcare, Charreire and colleagues (27), and others (74), have suggested that these domains are important for food-related practices. It is hypothesised that greater food access in the physical food environment will be associated with greater overall and more frequent purchase and consumption of food sold. In part, this is due to individuals receiving purchasing cues and having the opportunity for their needs to be met within each of the aforementioned domains (61, 75). Thus, the five domains defined by Penchansky and Thomas have been adapted for food access in the physical food environment, as shown in Table 1.1, and used within research examining the purchase and consumption of fruits and vegetables (76), and food prepared out-of-home (50).

Table 1.1: Name and definition of the domains of access in relation to the physical food
environment.

Domain	Definition ^a
Availability	The presence of certain types of food outlet
Accessibility	The location of food supply and the ease of getting to that location
Affordability	Food prices and people's perceptions of worth relative to cost
Accommodation	People's attitudes about how well a given food supply accepts and adapts to their individual needs
Acceptability	People's attitudes about attributes of their local food environment, and whether or not the given supply of products meets their personal standards

^a definitions reproduced from Caspi et al. (77).

One systematic review reported that almost two-thirds of 432 articles published between 2007 and 2015 examined aspects of food availability and accessibility through geospatial analysis (78). However, despite being theoretically plausible, evidence of an association between these domains of food access and food-related practices is mixed, with numerous reviews reporting positive, null and negative associations (25, 29, 36, 61, 64, 79, 80). A further systematic review from 2012 synthesised evidence according to the domains of food access defined in Table 1.1 (77). Across 38 included studies, more than two-thirds (n=26) conducted geospatial analysis. Of these studies, 20 investigated food outlet availability, measured in terms of presence or count. Focusing on food prepared out-of-home, of the five studies that investigated a relationship between a spatial food environment exposure through geospatial analysis (availability or accessibility domains of food access in this instance) and the purchase and consumption of food sold (60, 81-84), only one found a positive association (83). In their study, Thornton and colleagues reported that access to one additional brand of fast-food chain increased the odds of monthly fast food purchasing by 13%. Moreover, there was some evidence from unadjusted and partially adjusted analyses that the density (i.e. count) of food outlets was associated with monthly and weekly fast food purchasing. However, this association was attenuated after adjustment for individual-level sociodemographic characteristics.

Although previous research has tended to use geospatial analysis to investigate the availability and accessibility domains of food access, the remaining domains (i.e. affordability, accommodation, and acceptability) are not necessarily spatially dependent. In part, an inconsistent evidence base might reflect that these domains have been investigated less frequently than more spatially dependent domains (77). This is important since the price and type of food prepared out-of-home, and sociocultural norms therein, can influence purchasing practices (85-87), especially when multiple food outlets are accessible (76, 88). Indeed, failing to consider that food purchasing practices are complex and influenced by multiple overlapping factors is a recognised limitation of previous research in this area (27, 74). Further methodological variation across previous research, including how neighbourhood food environments were characterised in terms of their shape and size, the spatial scale of analyses, data used, and the specific relationships investigated may have also attenuated evidence of a positive association (26, 27, 89-91). Moreover, it is necessary to consider that most of the previous research regarding access to food prepared out-of-home has been conducted in the context of the physical food environment. Food prepared out-of-home can now also be accessed through purchasing formats (i.e. the way that food is acquired) that are part of food retail within the digital food environment. I will discuss this further in Section 1.4.3.

1.2.5.3 Socioeconomic position

In Section 1.1, I indicated that there are socioeconomic inequalities in both overall dietary patterns and obesity prevalence. Physical food environments that provide abundant access to food prepared out-of-home have been referred to as obesogenic (92), and obesity-promoting (93). Within the socioecological model (see Section 1.2.5.1: Figure 1.3), socioeconomic and environmental factors are included as plausible determinants of health. These factors can influence the spatial distribution of food outlets in the physical food environment. As such, a possible association between area-level socioeconomic position and measures of food access has been a consistent focus of previous research (51, 53, 94). Much of the current evidence is

from the USA (95, 96), and differences in the measurement of socioeconomic position and the spatial distribution of food outlets in the physical food environment makes comparisons between countries difficult (97). Elsewhere, there is evidence that the number of food outlets selling food prepared out-of-home in the physical food environment is higher in more deprived areas of the UK (31, 98, 99). As shown in Figure 1.4, using 2017 Local Authority data from the Food Standards Agency, Public Health England reported that fast-food outlet density per 100,000 population was positively associated with deprivation at the Local Authority level in England (100). Populations living in these areas might receive a higher number of cues to purchase foods sold, which can influence food purchasing practices (2), and shape sociocultural norms about the appropriateness and acceptability of this food (101). Accordingly, unequal access to food in the physical food environment might contribute to overall dietary patterns, and related inequalities. I will discuss evidence on these inequalities within the context of online food delivery services in Section 1.4.5.1.





Notes: Reproduced from Public Health England (with modification), published in 2018 (100). Fast-food outlets predominantly sell food prepared out-of-home. Data are from the Food Standards Agency (2017) and the English Indices of Multiple Deprivation (2015).

Moving away from area-level socioeconomic and environmental determinants of health, the socioecological model shown in Figure 1.3 (see Section 1.2.5.1) also suggests that individuallevel factors are a determinant of health. Individual-level socioeconomic position is often characterised in terms of education, income, and occupation (102, 103), however, there is no clear consensus about the most appropriate measure to use (104). In the context of food, each measure provides a different indication about how purchase and consumption practices might be influenced, including the ability to interpret food labels, the availability of finances for food purchasing and time for meal preparation (102, 103). Although education, income and occupation are related, they are imperfectly correlated (105), and heterogeneity in their characterisation can make comparisons between them difficult (106, 107). Nevertheless, there is evidence that purchasing and consuming food prepared out-of-home varies based on measures of education, income, and occupation (50, 61, 107). To exemplify, amongst adults living in Australia, those with the highest level of education consumed more food prepared out-of-home compared with those who had lower levels of education (108). In England (109), and the USA (110), however, adults with the highest level of education consumed the least food prepared out-of-home. Furthermore, there is evidence that the purchase and consumption of food prepared out-of-home varies by location, with more highly educated individuals accessing this food through restaurants rather than hot food takeaway outlets (111), and that they purchase healthier foods (112). Taken together, these findings suggest that country-specific context and food characteristics interact with individual-level factors to influence purchasing practices. This is reflected in qualitative research that reported on the complexities of takeaway food consumption (113-115). Moving on to income, amongst females living in Australia, those in the lowest income group (less than 500 Australian Dollars per week) consumed food prepared outof-home more frequently than those in higher income groups (104). Elsewhere, however, household income was not associated with overall consumption of food prepared out-of-home (112). Finally, for occupation, purchasing and consuming food prepared out-of-home inside of food outlets was more common amongst adults in higher (e.g. managerial) rather than lower (e.g. manual) employment classifications (40, 116).

1.2.5.4 Individual-level sociodemographic characteristics

In the previous section, I discussed how the purchase and consumption of food prepared outof-home varies by individual-level socioeconomic position. Further individual-level sociodemographic characteristics including age, sex and ethnicity are also suggested to influence these food-related practices (50, 107, 117, 118). Next, I briefly discuss evidence related to each of these sociodemographic characteristics in turn, with acknowledgement that they intersect.

Although there is evidence that food prepared out-of-home is consumed by individuals of all ages, consumption tends to be highest in younger populations (36). For instance, the proportion of adults in the UK that purchased food prepared out-of-home once per week or more between 2008 and 2012 was over 40% amongst those aged 19 to 29 years, and 20% for those aged between 60 and 69 years (40). Elsewhere, frequency of consumption was inversely associated with age amongst adults in the USA (119). There are multiple reasons why younger adults might be more inclined to consume food prepared out-of-home and visit food outlets in-person. Doing so offers convenience compared with cooking at home, can form part of routines and habits, and adolescents have reported that food outlets are a safe and social place to meet (120-122).

Regarding sex, males have been reported to consume food prepared out-of-home more frequently than females (111, 123). To some extent, differences in consumption might reflect that males prepare food for themselves less frequently (124, 125), and spend less time cooking at home (126, 127), compared with females who are traditionally placed into roles responsible for food purchasing and preparation (128). These differences are not necessarily biologically driven, and it is recognised that food practices are influenced by sociocultural norms (129). Since social, environmental and cultural, factors can influence the gender to which a person selfidentifies (130), the aforementioned differences might be a reflection of gender rather than sex. However, sex assigned at birth is commonly measured in research (131).

Finally, there is evidence that consumption of food prepared out-of-home varies across ethnicity groups (132). However, evidence in the context of Western Europe is limited. In part, this is because ethnic minorities are under-represented in research (133, 134), and because multiple ethnic groups are often aggregated into a single category, which limits the opportunity for a more granular understanding of differences (135). Making comparisons between countries in terms of ethnic differences in the consumption of food prepared out-of-home is also difficult due to variations in historical and cultural influences on food practices (136). Focusing on evidence from the UK, the consumption of food prepared out-of-home was most prevalent and frequent amongst Black and South Asian adolescents aged 13-15 years living in London (137). Furthermore, adults that identified with a White ethnicity had lower odds of consuming food prepared out-of-home more than twice per week, compared with those who identified with a non-white ethnic group (111). Finally, females that identified with an ethnic minority reportedly consumed food prepared out-of-home ween time was limited (138).

11

1.3 The digital food environment

So far in this chapter, I have discussed the importance of a healthy diet, that food prepared outof-home is typically high in calories, total fat, saturated fat and salt, and evidence that consumption of this food can have implications for overall dietary patterns, body weight and health. To reiterate, there is established public health concern about the consumption of food prepared out-of-home. Most of the previous research in relation to this food was conducted in the context of the physical food environment. However, food prepared out-of-home can now also be accessed through online food delivery services, which form part of the digital food environment.

1.3.1 A definition for the digital food environment

The internet is used by two-thirds of the global population (139), and many aspects of daily life have become partially, if not fully, digitalised (140). Widespread internet use, a demand for greater convenience within changing sociocultural norms and technological advances have contributed to the development of the digital food environment, which acts in parallel with the physical food environment (141-143). Although the digital food environment has existed for over two decades, it is not well defined (144). According to the European region of the World Health Organization (WHO), the digital food environment is *"a range of elements, including social media, digital health promotion interventions, digital food marketing and online food retail"* (145). In the digital food environment, factors that plausibly influence food-related practices are present and interacted with in online settings (146). While I recognise that the digital food environment is complex and multifaceted, of particular interest to my thesis is `online food retail`, and specifically, the use of online food delivery services to purchase food prepared out-of-home. I will discuss the use of these services in Section 1.4.3. Next, however, I consider existing evidence on the digital food environment.

1.3.2 Previous research in the context of the digital food environment

A systematic scoping review from 2022 synthesised evidence on the digital food environment published between 2000 and 2019 (147). Although 357 studies were included, and the number of publications increased from 2012 onwards, there was limited public health focused research. In part, this reflects that research was more frequently published in journals focused on marketing, business or economics (n=82 studies), or food and nutrition (n=93 studies), rather than medicine or health (n=80 studies). Moreover, although online food retail was the most investigated setting (n=95 studies), research did not typically focus on food prepared out-of-home.

Elsewhere, within public health focused research, limited elements of the digital food environment have been studied (148). Digital food marketing is recognised for its influence on food purchasing practices and preferences (149, 150). Various features of social media have been associated with food-related sociocultural norms (140). And, it has been suggested that the digital food environment and purchasing formats therein challenge multiple Sustainable Development Goals related to health and wellbeing that were adopted by the United Nations in 2015 (151, 152). Specifically for online food retail, although multiple types of food can be purchased, research to date has often focused on food from supermarkets rather than other types of food (153). For example, there has been limited research into the purchase of food prepared out-of-home through internet-based purchasing formats such as online food delivery services (148).

1.4 Online food delivery services

Delivery of food prepared out-of-home is not new. Historically, ordering food prepared out-ofhome for delivery would have involved contacting food outlets (i.e. businesses) directly by telephone. Since the emergence of online food delivery services, however, this is no longer necessarily the case (154), with these services referred to elsewhere as `disruptors` to traditional food retail (155). In the sections that follow, I will discuss why a better understanding of online food delivery service use is important from a public health perspective.

1.4.1 A definition for online food delivery services

The online food delivery service business model has been referred to by others as *"meal delivery services"* (156), *"food delivery apps"* (141), *"digital food ordering platforms"* (157), and *"internet based ordering and delivery"* (158). Despite this variation, each definition refers to the same type of service. That is, one that aggregates information about food outlets (for example, menus, price-promotions, contact details, previous customer reviews) onto a single digital platform and provides customers with the opportunity to purchase food prepared out-of-home before having it delivered. In the absence of an established or agreed upon definition, I have used *"online food delivery services"* in my thesis, which is similar to how one internationally established service and the market leader in the UK (Just Eat) refers to themselves.

1.4.2 The global presence and popularity of online food delivery services

International operations of online food delivery services including Just Eat and Uber Eats has led to a presence in each continent (141, 142, 156, 159-162). In addition, there are also small-scale online food delivery services that facilitate access to food prepared out-of-home across local geographies, such as within a single city. In some cases, however, internationally established online food delivery services have acquired small-scale services. In turn, this has contributed to a concentration of online food delivery services operating across multiple markets under the branding of different subsidiaries. For example, Just Eat has a portfolio that includes SkipTheDishes in Canada and Menulog in Australia (163).

In part, the aforementioned global presence reflects an increased popularity of online food delivery service use (143, 146, 148). For instance, Just Eat has operated in the UK since around 2006, however, demand for this type of service has increased only more recently as customers reportedly seek a convenient and time efficient way to order and receive food prepared out-of-home within a digital and on-demand society (24, 39). There may be further unique aspects of the online food delivery service business model that have contributed to this popularity, which I investigate from the perspective of customers in Chapter 3. In the next section, I describe the process of online food delivery service use.

1.4.3 Purchasing food prepared out-of-home through online food delivery services

In Section 1.4, I stated that historically, having food prepared out-of-home delivered would have typically involved contacting food outlets (i.e. businesses) directly by telephone to place orders. Food businesses would then prepare and deliver meals. This is dissimilar to the business model of online food delivery services, which primarily involves three groups of stakeholders: online food delivery services, food businesses, and customers. When the research in my thesis was conceived, online food delivery services primarily aggregated information about food businesses onto a single platform and allowed customers to place their orders. Once customers had placed their orders, food businesses would prepare and deliver meals. However, the business model of these services has evolved, and in some instances, online food delivery services now facilitate meal delivery. The focus of my thesis is the delivery of food prepared out-of-home when ordered through online food delivery services, irrespective of who is responsible for delivery.

Figure 1.5 illustrates the process of online food delivery service use. Below, I summarise key points from this figure:

 Before food prepared out-of-home can be purchased through online food delivery services, a given online food delivery service must operate in a given area, food businesses must be registered to accept orders through this service, and individuals must be registered as customers.

- Online food delivery services and food businesses continuously interact with one another to provide a range of information that customers can use to inform their purchasing practices.
- Food ordered through online food delivery services is typically prepared inside of existing customer-facing food outlets located in the physical food environment. Thus, the online food delivery service business model is predicated on the location of food outlets in the physical food environment.

	ONLINE FOOD DELIVERY SERVICES	FOOD BUSINESSES	CUSTOMERS
	Operate in a given place (i.e. country, city) *	Register to accept orders through a given online food delivery service *	Register with a given online food delivery service *
PURCHASING PROCESS	Aggregate information about registered food businesses on digital platforms (i.e. websites and smartphone applications) Advertise the availability of their service (e.g. through physical or digital advertising or by providing branding materials to registered food businesses) Provide and advertise centralised price promotions (e.g. seasonal or time limited discounts)	Provide online food delivery services with a business description, and information about menu items, promotional offers, and the places to which meals can be delivered Provide unique price promotions (e.g. free food or drinks with orders)	 On a given online food delivery service platform (i.e. website or smartphone application): 1. enter a location for meal delivery 2. receive and review information about 'accessible' food businesses (i.e. those that will deliver to the entered location) 3. select a food business and menu items, and place order
		Prepare meals in facilities of existing (customer-facing) food outlets located in the physical food environment	
	Deliver meals through a centralised network of couriers	Deliver meals independently (when not being completed centrally by online food delivery services)	Wait for meal delivery
			Receive and consume meal

* prerequisite for individual-level access to, and use of, online food delivery services

Figure 1.5: The role of individual online food delivery services, food businesses, and customers in the process of purchasing food prepared out-of-home through the online food delivery service business model.

1.4.4 A proposed pathway between the use of online food delivery services and body weight

Figure 1.6 is a proposed causal pathway showing factors that plausibly influence online food delivery service use. Since I use Figure 1.6 to highlight the research focus of the upcoming empirical chapters, I have not illustrated the multiple preceding or proceeding factors that could influence those included and that I will investigate. Nevertheless, as evidenced elsewhere (36, 50), I acknowledge that bodyweight, for example, is determined by multiple factors that extend beyond the physical food environment or online food access. For example, physical activity levels and overall dietary intake. Figure 1.6 was informed by the evidence discussed in the previous sections; particularly the socioecological model (see Section 1.2.5.1) and the domains of food access (see Section 1.2.5.2). I propose that online food delivery service use depends on individuals having `online food access` (i.e. access to food prepared out-of-home through online food delivery services). This online food access might be influenced by area-level socioeconomic position due to the location of food outlets in the physical food environment. However, there are also broader social and economic factors including labour workforce, industrial development, digital connectivity and area-specific food related traditions and values, which may also be influenced and informed by area-level socioeconomic position. Moreover, online food access could also be influenced by individual-level characteristics, since some populations, for example, those who are younger, more highly educated and living in urban areas, have higher levels of digital literacy, which could lead to them spending more time using internetenabled devices and receiving prompts for food prepared out-of-home (164). Finally, online food delivery service use might be associated with body weight further along the causal pathway.



Figure 1.6: Proposed causal pathway between online food access (access to food prepared out-of-home through online food delivery services) and body weight. Note: Dashed lines indicate hypothesised links.

1.4.5 Online access to food prepared out-of-home

It is a core tenet of my thesis that the domains of food access discussed in Section 1.2.5.2 (availability, accessibility, affordability, accommodation, and acceptability) are relevant to the use of online food delivery services. As such, I have adapted the definitions of these domains to reflect food access through online food delivery services, as shown in Table 1.2. I will discuss current evidence about these domains and elements of the proposed causal pathway (see Section 1.4.4: Figure 1.6) in the next sections. Doing so will help to identify knowledge gaps and provide insight into the potential need for public health concern about access to food through, and the use of, online food delivery services.

Domain	Original definition ^a	Definition adapted for online food delivery services
Availability	The presence of certain types of food outlet	Online food delivery services present and operating in a given area and individual food businesses registered to accept orders ^b
Accessibility	The location of food outlets and the ease of getting to that location	The interplay between the location of customer-facing outlets of food businesses registered to accept orders ^b , their delivery areas, and the location of a given individual that has internet access, an appropriate internet-enabled device, and who is registered as a customer
Affordability	Food prices and people's perceptions of worth relative to cost	The price of food sold and any additional costs incurred when using online food delivery services, and people's perceptions of worth relative to cost
Accommodation	People's attitudes about how well a given food outlets accepts and adapts to their individual needs	People's attitudes about how well online food delivery services and individual food businesses accepts and adapts to their individual needs
Acceptability	People's attitudes about attributes of their local food environment, and how well a given supply of foods meets their personal standards	People's attitudes about attributes of online food delivery services and accessible food businesses, and how well a given supply meets their personal standards

Table 1.2: Name and definition of the domains of food access in the context of online food delivery services.

Notes: ^a original definitions were for the physical food environment, reproduced from Caspi et al. (77). ^b registered to accept orders through the online food delivery service operating in a given area.

1.4.5.1 Access to food prepared out-of-home through online food delivery services and inequalities therein

The domains of food access adapted for online food delivery services that I presented in Table 1.2 have not been *explicitly* investigated to date. Nevertheless, there is some relevant evidence related to access to food prepared out-of-home through these services and variation according to area-level socioeconomic position. In terms of availability and accessibility, one study reported that the number of food outlets that could be ordered from through an internationally established online food delivery service was lower in the most, versus least, deprived areas of one city in Australia (477 outlets versus 682 outlets) and one city in the Netherlands (443 outlets versus 1053 outlets) (156). However, the number of food outlets was higher in the most, versus least, deprived areas of one city in the USA (859 outlets versus 809 outlets). Other research has reported that the number of food outlets registered to accept orders through a market leading online food delivery service was higher in less deprived areas across 13 local areas in Sydney,
Australia, yet not different across Auckland, New Zealand (165). Finally, across 24 randomly selected postcodes in Canada's most populated region, the number of food outlets that could be ordered from through a leading online food delivery service ranged from 33 to 472 (166). Although stratification by area-level socioeconomic position was not investigated, the authors noted heterogeneity in online food outlet access within and between cities. Together, this evidence indicates that there is variation in the availability of and accessibility to food prepared out-of-home through online food delivery services, and that this is socioeconomically patterned, albeit not always in the expected direction. Beyond this research, knowledge about online food access in other countries is limited to information that online food delivery services are available and accessible, and variation according to area-level socioeconomic position, is not yet known. This represents a substantial gap in knowledge since online food access is likely a necessity for online food delivery service use (as I proposed in Section 1.4.4). In Chapter 5, I report the findings from research that helps to address this gap in knowledge for the UK.

In addition, despite the aforementioned evidence that online food delivery services are available and accessible, the extent to which measures of online food access are related to the use of online food delivery services is unclear. I investigate this association in Chapter 4.

Further related to the availability and accessibility domains of food access, the location of food outlets in the physical food environment is important for the business model of online food delivery services. As I discussed in Section 1.2.5.3, the number of food outlets in the physical food environment tends to be higher in more deprived areas of England. Online food delivery services offer an alternative purchasing format without necessarily replacing others (142). As such, food prepared out-of-home can be accessed in multiple ways within both the physical and digital food environments. In turn, variation in online food delivery service availability and accessibility according to socioeconomic position might increase overall access to food prepared out-of-home. This increase might contribute to diet-related inequalities. However, links between the physical food environment and online food access have not yet been investigated. I report the findings from research that investigated plausible links in Chapters 5 and 6.

The density and concentration of food outlets in the physical food environment has increased over time (32, 34, 35). In turn, this might influence online food access as a higher number of food outlets could plausibly register to accept orders through online food delivery services. Given that the popularity of using these services is forecast to increase (39), it is reasonable to suggest that more food outlets will register to accept orders online in the future as a way to

capitalise on perceived demand. Since this is untested, further research is warranted. A better understanding of changes in online food access over time and the extent to which this is socioeconomically patterned would provide insight into market trends, possible implications for customer interactions with the physical food environment, and the potential for food practices and inequalities therein to be influenced. I report the findings from research that investigated changes in online food access over time in Chapter 6.

Finally, unlike the availability and accessibility domains of food access presented in Table 1.2 (see Section 1.4.5), to my knowledge there is no published evidence related to the domains of affordability, accommodation and acceptability. Unique features of online food delivery services, encapsulated within these domains, plausibly influence the use of this purchasing format by ensuring that an individual's standards and needs are met. It has been suggested that online food delivery services are used for their convenience and because they expand food access beyond the traditional physical food environment (148). Nevertheless, experiences of online food delivery service use from the customer perspective are unknown. The research I present in Chapter 3 helps to address this knowledge gap.

1.4.5.2 Online food delivery service use and sociodemographic characteristics of customers

Within the proposed causal pathway that I presented in Figure 1.6 (see Section 1.4.4), I suggest that individual-level sociodemographic characteristics influence online food access and, in turn, online food delivery service use. As it stands, there is limited empirical evidence on the use of online food delivery services and the sociodemographic characteristics of customers. In one study, 28% of a sample of adults in living Australia (n=2010) had used an online food delivery service at least once in the month prior to data collection in 2018 (135). Online food delivery service use was highest amongst younger adults. Moreover, individuals with a higher (university degree) versus lower (compulsory) level of education, and a higher income (compared to less than \$9999 per year) had greater odds of online food delivery service use. Despite this research, the prevalence of online food delivery service use in the UK and many other countries, and the sociodemographic characteristics of customers therein, has not been established. This knowledge is vital for a better understanding about individual-level factors that might influence online food access and ultimately, online food delivery service use. I examine the prevalence of online food delivery service use and the sociodemographic characteristics of customers across multiple countries in Chapter 2, and in Great Britain in Chapter 4.

1.4.5.3 Online food delivery service use and body weight

Evidence from multiple countries indicates that food sold and marketed through online food delivery services is high in calories, total fat, saturated fat and salt (159, 162, 165, 167, 168). As shown in Figure 1.5 (see Section 1.4.3), food sold through these services is typically prepared inside of existing customer-facing food outlets located in the physical food environment. Thus, the aforementioned findings might be expected. Nevertheless, this research included only a limited number of the most popular food outlets and most frequently purchased menu items, which possibly misrepresents the full scope of foods sold. It is certainly plausible that by increasing the number and diversity of food outlets that can be ordered from, online food delivery services facilitate access to healthier foods prepared out-of-home than in the physical food environment (141, 148, 169). However, this remains untested. Based on current evidence about the types of food available to purchase through online food delivery services, the association between consumption of food prepared out-of-home and body weight that I discussed in Section 1.2.4.1 might be present. Given that the use of online food delivery services might be a substitute for other ways of purchasing food prepared out-of-home, this is especially important to understand. Therefore, in Chapter 4, I investigate the possible relationship between online food access and body weight.

1.5 Summary of the Introduction

In summary, poor diet is a contributor to poor health and there are socioeconomic inequalities in both. The purchase and consumption of food prepared out-of-home is recognised as a public health concern due to its energy-dense and nutrient-poor composition. Although this food remains accessible in the physical food environment, it is now also accessible through online food delivery services, which are an aspect of food retail within the digital food environment. However, the prevalence of online food delivery service use, and by whom, has not yet been established in the UK and many other countries. Moreover, possible reasons for selecting one purchasing format over another are unclear.

There is evidence that access to food prepared out-of-home in the physical food environment is higher in more deprived areas of the UK, and that this contributes to diet-related inequalities. Food sold through online food delivery services is typically prepared inside of existing customerfacing food outlets located in the physical food environment. As such, access to these services may increase overall access to food prepared out-of-home, especially in more deprived areas, which could contribute to diet-related inequalities. Therefore, research to help better understand access to food prepared out-of-home through online food delivery services, and the extent to which this is socioeconomically patterned and possibly influences health, is required.

1.6 Thesis aims

The aim of my thesis was to provide an improved understanding of the use of online food delivery services to access food prepared out-of-home, specifically with regard to the:

- 1. Prevalence of online food delivery service use and sociodemographic characteristics of customers.
- Experiences of online food delivery service use from the perspective of customers.
- Associations between measures of access to food prepared out-of-home through online food delivery services and online food delivery service use and body weight.
- 4. Extent to which access to food prepared out-of-home through online food delivery services is socioeconomically patterned, and how this access and any inequality changed over time.

1.7 Thesis structure

After this introductory chapter, I present a series of complementary empirical chapters. Together, these chapters help to deliver the specific aims outlined in the previous section and provide evidence for multiple elements of the proposed causal pathway presented in Figure 1.6 (see Section 1.4.4).

Chapters 2 to 6 are based on research that I led, with support and contributions from coauthors. The research I present in Chapters 2 to 5 has been peer-reviewed and published. The order in which I present the research in these chapters does not reflect the order of publication. Because of this difference, at points I refer to published research despite it not yet being presented in my thesis. For the most part, in Chapters 2 to 5, I present the published manuscripts. As a result, there is overlap across chapters in terms of background content and methods. However, I have made minor amendments to minimise repetition, to provide further information and rationale where appropriate, and to present a coherent structure to my thesis. I present the supplementary material for each empirical chapter in the appendices.

Regarding the aims outlined in Section 1.6, in Chapter 2, I address aim 1 by describing the prevalence of online food delivery service use and sociodemographic characteristics of customers amongst adults living across Australia, Canada, Mexico, the UK, and the USA. In Chapter 3, I address aim 2 by reporting experiences of online food delivery service use from the perspective of customers living in the UK. In Chapter 4, I address aim 3 by providing evidence on the association between online access to food prepared out-of-home and online food delivery service use and body weight amongst adults living in Great Britain. In Chapters 5 and 6, I address aim 4 by examining online access to food prepared out-of-home across England and variation according to socioeconomic position at one time point and over time. Finally, in Chapter 7, I discuss the findings from each of the empirical chapters as a whole, summarise the possible implications for public health and policy, and outline an agenda for future research.

Chapter 2:

Multi-country prevalence of online food delivery service use and sociodemographic characteristics of customers

2.1 Details of author contributions, publication and dissemination

For the research in this chapter, I developed the research questions and conceptualised data analysis with Jean Adams (JA) and Thomas Burgoine (TB), after consultation with David Hammond (DH), Gary Sacks (GS), and Lana Vanderlee (LV). Christine White (CW) provided data management support. I led data analysis. All authors read and provided critical comments on the manuscript that I first drafted and agreed to the final version.

This chapter is published as: Keeble, M., Adams, J., Sacks, G., Vanderlee, L., White, C.M., Hammond, D., Burgoine, T. (2020). Use of online food delivery services to order food prepared away-from-home and associated sociodemographic characteristics: a cross-sectional, multicountry analysis. International Journal of Environmental Research and Public Health. 17(4), 5190. https://doi.org/10.3390/ijerph17145190.

I shared the findings from the research in this chapter during a symposium at the 2020 International Society of Behavioral Nutrition and Physical Activity Annual Congress (held online): "New ways of purchasing (fast) food: are we transitioning away from tradition?".

2.2 Abstract

Background

Online food delivery services like Just Eat and Grubhub facilitate online ordering and home delivery of food prepared out-of-home. It is poorly understood how these services are used and by whom. I aimed to investigate the prevalence of online food delivery service use and sociodemographic characteristics of customers in Australia, Canada, Mexico, the United Kingdom (UK), and the United States of America (USA).

Methods

I used online survey data (n=19,378) from the International Food Policy Study, conducted in 2018. I identified respondents who reported any online food delivery service use in the past week and calculated the frequency of use and number of meals ordered. I used adjusted logistic regression to investigate whether odds of any online food delivery service use in the past week differed by sociodemographic characteristics.

Results

Overall, 15% of respondents (n=2929) reported online food delivery service use, with the greatest prevalence amongst respondents in Mexico (n=895; 26%). Online food delivery services had most frequently been used once and the median number of meals purchased in this way was two. Odds of any online food delivery service use in the past week were lower per additional year of age (odds ratio (OR): 0.95; 95% confidence intervals (CI): 0.94, 0.95) and greater for respondents who were male (OR: 1.50; 95% CI: 1.35, 1.66), that identified with an ethnic minority (OR: 1.57; 95% CI: 1.38, 1.78), were highly educated (OR: 1.66; 95% CI: 1.46, 1.90), or living with children (OR: 2.71; 95% CI: 2.44, 3.01).

Conclusions

Further research is required to investigate how online food delivery service use may influence diet and health.

2.3 Introduction

According to global estimates, 11% of males and 15% of females were living with obesity in 2016 (170, 171). Although the drivers of obesity are complex, excess calorie intake through the consumption of food prepared out-of-home is a recognised contributor (172-174). Food prepared out-of-home is typically served ready-to-consume, is often energy-dense, high in fat and salt, and less healthy than food prepared at home, with more frequent consumption of this food associated with elevated body weight (46, 175-177). Traditionally, this food may have been purchased through `conventional` purchasing formats whereby customers visit food outlets inperson or contact food outlets directly to place orders before collection or delivery. Although food prepared out-of-home can still be purchased in this manner, third-party platforms that facilitate online ordering and delivery, referred to as `online food delivery services`, provide an alternative purchasing format that coexists with conventional purchasing formats (142). Online food delivery services operate as intermediaries between customers and food outlets (39). Customers place orders through online platforms, their orders are forwarded to food outlets where meals are prepared, and once ready, meals are delivered to customers by couriers working for the food outlet or for the online food delivery service (142, 178).

In 2020, prominent online food delivery services such as Just Eat (including subsidiaries such as Menulog in Australia), Uber Eats, and Deliveroo operated internationally, whilst Grubhub was established in many cities across the United States of America (USA) (179-182). Online food delivery service availability and accessibility has been forecast to increase, possibly leading to greater use. In turn, this could increase the consumption of food prepared out-of-home (183). There is currently limited understanding of the nutritional quality of food items sold through online food delivery services. Nonetheless, given that the food sold through these services is typically prepared inside of existing customer-facing food outlets located in the physical food environment (39), it is likely to have a similar nutrient profile to food prepared out-of-home ordered in conventional ways. As such, online food delivery services could contribute to excess calorie intake and poor health (41, 46, 175). Accordingly, interventions to reduce online food delivery service use, or to improve the quality of available food, may be called for in the future.

Previous research into online food delivery services from a public health perspective is limited. A public health focused narrative review concluded that convenience and food outlet variety were potential drivers of online food delivery service use (144). To date, however, aspects of online food delivery service use that might influence diet and health, including the prevalence and frequency, have not been investigated. This knowledge about online food delivery service use will help to establish a baseline against which future patterns of use can be compared and serve

as an indicator of potential public health harm. Moreover, the sociodemographic characteristics of online food delivery service customers are poorly understood. Knowledge about these characteristics might provide vital intelligence for public health intervention development, if required in the future.

2.3.1 Study aims

In my study, I aimed to describe the prevalence and frequency of online food delivery service use, investigate associations between online food delivery service use and sociodemographic characteristics of customers, and describe how online food delivery service customers used other purchasing formats to access food prepared out-of-home in five upper-middle or highincome countries.

2.4 Methods

2.4.1 Data collection

I used cross-sectional data from the International Food Policy Study (IFPS), which is an ongoing annual repeat cross-sectional survey conducted in Australia, Canada, Mexico, the United Kingdom (UK), and the USA. Data collection methods have been described in full elsewhere (184). Briefly, between November and December 2018, data were collected via self-completed online surveys from adults aged 18 years or over, recruited through Nielsen Consumer Insights Global Panel and their partners' panels. Panellists were screened for eligibility and quota requirements based on device type, age, and sex. Email invitations containing links to an online survey in national languages were sent to a random sample of eligible panellists in each country. Respondents provided consent prior to survey completion. The IFPS was reviewed by and received ethics clearance through a University of Waterloo (Canada) Research Ethics Committee (ORE# 21460). Data collection in the UK was approved by the University of Cambridge Humanities and Social Science Research Ethics Committee (Reference: 19/225).

2.4.2 Measures

2.4.2.1 Purchasing format use

In the IFPS survey, all respondents were asked: *"During the past 7-days, how many meals did you get that were prepared away-from-home in places such as restaurants, fast food or takeaway places, food stands, or from vending machines?"*. A similar question has been asked in previous research (185, 186). Respondents who had purchased at least one meal prepared out-of-home were asked to report the number of meals ordered: *"using a food delivery service (e.g.,*

[country-specific examples]) and delivered", "directly from a restaurant and delivered", "at a restaurant/food outlet within 5 minutes of your home" and "at a restaurant/food outlet more than 5 minutes away from your home". Country-specific examples of online food delivery services available in each country included Uber Eats (all countries), Just Eat (Canada, Mexico, UK), Deliveroo (Australia, UK), Foodora (Australia), SkipTheDishes (Canada), and Grubhub (USA). I used answers to the second question for my analyses. I collapsed responses for "at a restaurant/food outlet within 5 minutes of your home" and "at a restaurant/food outlet more than 5 minutes away from your home" into a single category for analyses: "directly from food outlets in-person".

2.4.2.2 Sociodemographic characteristics

In the IFPS survey, all respondents were asked to self-report sociodemographic characteristics. I included sex, age, ethnicity, education, body mass index (BMI), and living with children aged under 18 years as independent variables. Sex at birth was reported as male or female. I treated this as a binary variable for analysis. Food purchasing practices (and plausibly, as a result, online food delivery service use) are a reflection of social norms and expectations (187). This might mean that online food delivery service use is a gendered practice rather than being driven by biological differences. Nevertheless, there is evidence of a correlation between sex at birth and gender in some instances (188). Age was reported in years (continuous). Ethnicity was reported as the group that best described racial or ethnic background, with different categories provided for respondents in each country. I dichotomised respondents into `majority` (selection of a White' category for respondents living in the USA or the UK, predominantly speaking English at home for those living in Australia, selection of 'European' rather than any other category for respondents living in Canada, and `not Indigenous` for those living in Mexico) and `minority` (all other responses, which varied across each studied country). Education was reported as the highest level completed. I categorised respondents as having: 'low' (high school completion or lower), `medium` (some post-high school qualifications) or `high` (university degree or higher) levels of education, and used this variable as a marker of socioeconomic position (189). Height and weight were reported in either metric or imperial units. I calculated BMI (kg/m²) and grouped respondents by World Health Organization categories: `underweight` (BMI <18.5), `normal weight` (BMI 18.5-24.9), `overweight` (BMI 25.0-29.9) or `obesity` (BMI > 30.0) (190). I collapsed the `underweight` and `normal weight` categories into a `not overweight` category (BMI ≤24.9). As individuals living with a higher BMI may not always report their height and weight (191), I included respondents with missing data for this variable. Living with children aged under 18 years was reported as a binary (yes/no) measure.

2.4.3 Study sample

In total, data were available for 22,824 respondents. I removed respondents with missing data for any variable of interest (except BMI), when the total number of meals purchased out-of-home and the number of meals purchased through each purchasing format summed did not match, or when the total number of meals purchased out-of-home in the past week exceeded 21 (I did not consider this to be plausible based on the maximum consumption of three meals per day). The final analytical sample included 19,378 respondents.

2.4.4 Statistical analysis

Following data collection, sample weights for the IFPS were constructed using population estimates from the census in each country based on age, sex, region, ethnicity (except in Canada) and education (except in Mexico) (184, 192). I rescaled sample weights to reflect the number of participants included in the analytic sample. Unless specified, I report weighted findings.

2.4.4.1 Descriptive analyses

For each country, I determined the prevalence of online food delivery service use by identifying respondents who reported that they had used an online food delivery service at least once in the past week. For these `online food delivery service customers`, I identified the frequency of online food delivery service use. I also calculated the number and proportion of all meals purchased out-of-home using each purchasing format (i.e. `online food delivery services`, `directly from food outlets for delivery` and `directly from food outlets in-person`). I also identified respondents who had purchased at least one meal prepared out-of-home directly from food outlets for delivery or in-person but had not used an online food delivery service. For these `non-online food delivery service customers`, I calculated the number and proportion of all meals purchased out-of-home `directly from food outlets for delivery service from food outlets for delivery service customers`, I calculated the number and proportion of all meals purchased out-of-home `directly from food outlets for delivery service customers`, I calculated the number and proportion of all meals purchased out-of-home `directly from food outlets for delivery from food outlets for delivery service customers`, I calculated the number and proportion of all meals purchased out-of-home `directly from food outlets for delivery` and `directly from food outlets in-person`.

2.4.4.2 Inferential analyses

I used online food delivery service use as the primary outcome measure. As data were not normally distributed and highly right-skewed, I dichotomised respondents into any online food delivery service use in the past week or not. I used Pearson's χ^2 to compare differences in sociodemographic characteristics of online food delivery service customers in each country. I used logistic regression following a sequential modelling strategy to investigate associations between my primary outcome and each of: sex, age, ethnicity, education, BMI, and living with children aged under 18 years (across all countries combined). Model 0 was unadjusted, Model 1 was adjusted for all independent variables except education to investigate variation by individual-level socioeconomic position, and finally, Model 2 was as Model 1 but with education included (193, 194).

I used separate logistic regression models, adjusted according to Model 2, with each country as the reference category, to investigate differences in the prevalence of online food delivery service use between countries.

I investigated differences in the prevalence of online food delivery service use and independent variables between countries by adding a two-way interaction term (country x independent variable) to separate logistic regression models (adjusted according to Model 2). I used post-estimation Wald tests to determine interaction term significance. When interaction terms were significant, I stratified analyses by country. I used Stata version 15.1 (StataCorp LLC., College Station, TX, USA) to complete analyses with a significance threshold of p<0.05 used throughout.

2.5 Results

Amongst the analytic sample, 78% (n=15,093) had purchased at least one meal prepared out-ofhome in the past week; 63% (n=12,163) had purchased food prepared out-of-home directly from food outlets for delivery or in-person but had not used an online food delivery service, whereas 15% (n=2929) had used an online food delivery service at least once. The greatest prevalence of online food delivery service use was amongst respondents in Mexico (n=895; 26%) (see Table 2.1).

2.5.1 Sociodemographic characteristics

Table S1 (see Appendix A) shows the sociodemographic characteristics of the analytic sample. Overall, more than half of the analytic sample were female or identified with an ethnic majority, the median age was 47 years, over 40% had a low level of education, over 20% were living with obesity, and less than 30% lived with children aged under 18 years. Table S2 (see Appendix A) shows the sociodemographic characteristics of non-online food delivery service customers. These respondents were similar in terms of sociodemographic characteristics to the analytic sample.

Table 2.1 shows the sociodemographic characteristics of online food delivery service customers. Overall, more than half of online food delivery service customers were male, identified with an ethnic majority, were highly educated, or living with children aged under 18 years. Around 40% were living with overweight or obesity, and the median age was 33 years (interquartile range (IQR): 26, 41).

2.5.2 Meals purchased out-of-home

Of the respondents that reported any online food delivery service use, around half had used this purchasing format to purchase one meal, meaning that they had purchased food prepared outof-home in this way only once in the past week (see Appendix A: Figure S1).

Table 2.2 reports the purchase of meals prepared out-of-home through each purchasing format for online food delivery service customers and non-online food delivery service customers, respectively. Overall, online food delivery service customers ordered a median of 2.0 (IQR: 1.0, 3.0) meals through an online food delivery service, which represented 36% of all meals prepared out-of-home. Online food delivery service customers ordered a median of 1.0 (IQR: 0.0, 2.0) meal directly from food outlets for delivery and 2.0 (IQR: 0.0, 3.0) meals directly from food outlets in-person. Overall, the median number of meals that non-online food delivery service customers ordered directly from food outlets for delivery and from food outlets in-person was 2.0. **Table 2.1**: Sociodemographic characteristics of online food delivery service customers amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

	Au	ıstralia	C	anada	N	1exico		UK		USA	Т	otal	p value for
	(n=	=3578)	(n:	=3698)	(n:	=3515)	(n=	=4694)	(n:	=3893)	(n=	19378)	difference ^a
Online food delivery service customers ^{b c}	498	(13.9)	327	(8.8)	895	(25.5)	747	(15.9)	461	(11.8)	2929	(15.1)	p>0.0001
Variable													
Sex													p>0.0001
Male	305	(61.1)	196	(60.0)	433	(48.3)	422	(56.4)	273	(59.2)	1629	(55.6)	
Ethnicity													p>0.0001
Majority	310	(32.1)	201	(61.3)	662	(73.9)	570	(76.3)	259	(56.1)	2001	(68.3)	
Age (years)													
Median (IQR)	31	(25-40)	33	(26-41)	34	(27-42)	32	(25-41)	33	(26-38)	33	(26-41)	
Education													p>0.0001
Low	136	(27.4)	95	(28.9)	119	(13.3)	320	(42.8)	177	(38.3)	846	(28.9)	
Medium	133	(26.6)	110	(33.7)	90	(10.1)	171	(22.9)	35	(7.6)	538	(18.4)	
High	229	(46.0)	122	(37.4)	686	(76.6)	257	(34.4)	250	(54.4)	1545	(52.7)	
BMI (kg/m²)													p>0.0001
Not overweight (≤24.9)	255	(51.2)	164	(50.2)	420	(46.9)	321	(42.9)	206	(44.6)	1366	(46.6)	
Overweight (25.0-29.9)	118	(23.7)	77	(23.4)	265	(29.6)	150	(20.1)	135	(29.2)	744	(25.4)	
Obesity (≥30.0)	52	(10.3)	60	(18.2)	145	(16.1)	106	(14.2)	77	(16.7)	439	(15.0)	
Missing	73	(14.7)	27	(8.2)	66	(7.4)	170	(22.8)	44	(9.5)	380	(13.0)	
Child <18 years in home													p>0.0001
Yes	226	(45.4)	131	(40.0)	639	(71.4)	364	(48.7)	240	(51.9)	1600	(54.6)	

Notes: ^a p values from Pearson's χ^2 test represents difference between categories within variable.

^b Online food delivery service customers had purchased at least one meal prepared out-of-home through an online food delivery service in the past week.

^c Unless specified, data reported as n (%).

Table 2.2: Purchasing formats used to acquire meals prepared out-of-home in the past week amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

Purchasing format	Australia (n=3578)	Canada (n=3698)	Mexico (n=3515)	UK (n=4694)	USA (n=3893)	Total (n=19378)	p value for difference ^a
Online food delivery service customers b	400 (12 0)	227 (0.0)		747 (15 0)		2020 (15.1)	
Number (%)	498 (13.9)	327 (8.8)	895 (25.5)	747 (15.9)	461 (11.8)	2929 (15.1)	
Online food delivery services ^c							p>0.0001
Number of meals	2.0 (1.0-2.0)	1.0 (1.0-2.0)	2.0 (1.0-3.0)	1.0 (1.0-2.0)	2.0 (1.0-3.0)	2.0 (1.0-3.0)	
Proportion (%)	40.0 (50.0-57.1)	40.0 (25.0-66.7)	33.3 (23.1-50.0)	50.0 (33.3-100.0)	33.3 (25.0-50.0)	35.7 (25.0-50.0)	
Directly from food outlets for							p>0.0001
delivery ^c							
Number of meals	0.0 (0.0-1.0)	0.0 (0.0-1.0)	1.0 (0.0-2.0)	0.0 (0.0-1.0)	1.0 (0.0-2.0)	1.0 (0.0-2.0)	
Proportion (%)	0.0 (0.0-28.6)	0.0 (0.0-28.6)	25.0 (0.0-40.0)	0.0 (0.0-25.0)	20.0 (0.0-33.3)	16.7 (0.0-33.3)	
Directly from food outlets in-person ^c							p>0.0001
Number of meals	2.0 (0.0-3.0)	1.0 (0.0-3.0)	2.0 (1.0-4.0)	1.0 (0.0-2.0)	2.0 (1.0-4.0)	2.0 (0.0-3.0)	
Proportion (%)	40.0 (0.0-60.0)	40.0 (0.0-54.5)	40.0 (20.0-52.4)	33.3 (0.0-50.0)	40.0 (20.0-57.1)	40.0 (0.0-50.0)	
Non-online food delivery service							
<i>customers</i> ^d							
Number (%)	2188 (61.2)	2420 (65.4)	2396 (68.2)	2439 (52.0)	2721 (69.9)	12163 (62.8)	
Directly from food outlets for							p>0.0001
delivery ^c							
Number of meals	1.0 (1.0-2.0)	1.0 (1.0-2.0)	2.0 (1.0-4.0)	1.0 (1.0-2.0)	2.0 (1.0-3.0)	2.0 (1.0-3.0)	
Proportion (%)	66.7 (50.0-100.0)	100.0 (40.0-100.0)	75.0 (50.0-100.0)	100.0 (50.0-100.0)	100.0 (40.0-100.0)	83.3 (50.0-100.0)	
Directly from food outlets in-person ^c							p>0.0001
Number of meals	2.0 (1.0-3.0)	2.0 (1.0-3.0)	3.0 (2.0-4.0)	2.0 (1.0-3.0)	2.0 (1.0-4.0)	2.0 (1.0-3.0)	
Proportion (%)	100.0 (100.0-100.0)	100.0 (100.0-100.0)	100.0 (66.7-100.0)	100.0 (100.0-100.0)	100.0 (100.0-100.0)	100.0 (100.0-100.0)	

Notes: ^a p value from Pearson's χ^2 test represents difference between categories within variable.

^b Online food delivery service customers had purchased at least one meal prepared out-of-home through an online food delivery service in the past week.

^c Data reported as median (IQR) number of meals, and median (IQR) proportion of all meals purchased out-of-home, per person.

^d Non-online food delivery service customers had purchased at least one meal prepared out-of-home directly from food outlets but not through an online food delivery service, in the past week.

2.5.3 Sociodemographic correlates of online food delivery service use

Table S3 (see Appendix A) reports sociodemographic correlates of online food delivery service use from unadjusted (Model 0) and partially adjusted (Model 1) analyses.

Figure 2.1 reports findings from maximally adjusted analyses (Model 2). Overall, there were greater odds of any online food delivery service use in the past week amongst respondents who were male rather than female (odds ratio (OR): 1.50; 95% confidence intervals (CI): 1.35, 1.66), those that identified with an ethnic minority rather than an ethnic majority (OR: 1.57; 95% CI: 1.38, 1.78), who lived with children (OR: 2.71; 95% CI: 2.44, 3.01) or had a high rather than low level of education (OR: 1.66; 95% CI: 1.46, 1.90). Odds of online food delivery service use were lower per additional year of age (OR: 0.95; 95% CI: 0.94, 0.95). There were no differences by BMI category.



Figure 2.1: Associations between prevalence of any online food delivery service use in the past week and sociodemographic characteristics, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

Note: Data analysed using logistic regression adjusted for independent variables (sex, ethnicity, age, living with children aged under 18 years at home, education, and body mass index (BMI)). Reference groups: ethnicity = majority, education level = low, BMI category = not overweight.

2.5.4 Between country variation

Table 2.3 reports between country variation in the prevalence of online food delivery service use. Respondents in Canada had lower odds of any online food delivery service use in the past week compared with respondents in all other countries. Respondents in the UK and Mexico typically had greater odds compared with respondents in other countries.

Amongst online food delivery service customers in Australia, Mexico and the USA, the median number of meals ordered through online food delivery services per person was 2.0, and in Canada and the UK, the median number per person was 1.0 (with minimal variation in respective IQR values) (see Table 2.2).

Table S4 (see Appendix A) shows that there were significant between-country interactions. The association between online food delivery service use and each of: age (p<0.0001), living with children aged under 18 years (p=0.037), sex (p<0.0001) and education (p<0.0001) varied between countries. Figures 2.2 to 2.5 show country-stratified findings for each of: age, living with children aged under 18 years, sex and education. Odds of online food delivery service use were lower per additional year of age (and relatively consistent) amongst respondents in all countries. Respondents who lived with children aged under 18 years had greater odds of online food delivery service use in all countries, with the strongest association observed amongst respondents in the USA (OR: 3.22; 95% CI: 2.49, 4.20). There was no difference in odds of online food delivery service use by sex amongst respondents in Mexico (OR: 1.02; 95% CI: 0.85, 1.23), whereas males in all other countries had greater odds. Respondents with high (versus low) education had greater odds of online food delivery service use in all countries had greater odds. Respondents with high (versus low)

Country ^a	OR ^b	95% CI °		
Australia (ref)	-	-	-	
Canada	0.65	0.54	0.78	
Mexico	1.21	1.03	1.43	
UK	1.39	1.18	1.64	
USA	0.85	0.72	1.02	
Australia	1.55	1.29	1.87	
Canada (ref)	-	-		
Mexico	1.88	1.58	2.25	
UK	2.15	1.79	2.57	
USA	1.32	1.10	1.59	
Australia	0.82	0.69	0.97	
Canada	0.53	0.45	0.63	
Mexico (ref)	-	-	-	
UK	1.14	0.98	1.33	
USA	0.70	0.60	0.82	
Australia	0.72	0.61	0.85	
Canada	0.47	0.39	0.56	
Mexico	0.88	0.75	1.02	
UK (ref)	-	-	-	
USA	0.61	0.52	0.73	
Australia	1.17	0.98	1.40	
Canada	0.76	0.63	0.91	
Mexico	1.43	1.22	1.67	
UK	1.63	1.37	1.93	
USA (ref)	-	-	-	

Table 2.3: Associations between country and prevalence of any online food delivery service use in the past week amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

Notes: ^a Each country used as reference (ref) in separate adjusted logistic regression models.

^b Odds ratio.

^c 95% confidence intervals.



Figure 2.2: Associations between prevalence of any online food delivery service use in the past week and age, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study. Note: Data analysed using country-stratified logistic regression adjusted for independent variables (sex, ethnicity, living with children aged under 18 years at home, education, and body mass index (BMI)).



Figure 2.3: Associations between prevalence of any online food delivery service use in the past week and living with children aged under 18 years, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

Note: Data analysed using country-stratified logistic regression adjusted for independent variables (sex, ethnicity, education, and body mass index (BMI)).



Figure 2.4: Associations between prevalence of any online food delivery service use in the past week and sex, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study. Note: Data analysed using country-stratified logistic regression adjusted for independent variables (ethnicity, living with children aged under 18 years at home, education, and body mass index (BMI)).



Figure 2.5: Associations between prevalence of any online food delivery service use in the past week and education level, amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study.

Note: Data analysed using country-stratified logistic regression adjusted for independent variables (sex, ethnicity, living with children aged under 18 years at home, and body mass index (BMI)).

2.6 Discussion

2.6.1 Summary of findings

My study included almost 20,000 adults living across Australia, Canada, Mexico, the UK, and the USA. Overall, more than 60% had purchased food prepared out-of-home directly from food outlets in the past week but had not used an online food delivery service, however, 15% reported that they had used an online food delivery service in the past week. Online food delivery services were mostly used once in the past week, however, on occasion they were used more frequently. Overall, the median number of meals purchased through online food delivery services was 2, and the median proportion of all meals purchased out-of-home ordered in this manner exceeded 30%. Adults who were male, younger, with higher rather than lower levels of education, those who lived with children, or identified with an ethnic minority had greater odds

of online food delivery service use. Sociodemographic correlates of online food delivery service use were similar in each country but the strength of associations varied.

2.6.2 Interpretation of findings

As the first study to investigate the prevalence and frequency of online food delivery service use in and across studied countries, I am unable to conclude that identified levels are relatively high or low. As such, my findings provide novel insight into the use of these services. My findings also provide insight into how multiple purchasing formats coexist. Importantly, purchasing food prepared out-of-home in multiple ways may lead to greater total consumption of food sold. In turn, this might lead to increased risk of excess weight and adverse health outcomes. When having food delivered, IFPS respondents who reported any online food delivery service use in the past week appeared to favour this purchasing format. In contrast, non-online food delivery service customers continued to order directly from food outlets. This observation could support the suggestion that online food delivery services have the capacity to displace conventional purchasing formats within out-of-home food retail (155). Nevertheless, some online food delivery service customers reported that they also visited food outlets in-person, indicating that individuals continued to interact with the physical food environment regardless of online food delivery service use.

In my research, males had greater odds of online food delivery service use compared with females. Consistent with my finding, males reportedly purchase more food prepared out-of-home more and cook at home less frequently than females (111, 112). It is unclear how reasons for purchasing food prepared out-of-home might differ based on the purchasing format used, and how these reasons vary by sex. Given that broader social, cultural and environmental factors can influence both food purchasing practices and the gender to which an individual identifies (195), factors other than biology likely contribute. Qualitative research could help better understand reasons for online food delivery service use and factors that influence this purchasing practice.

Additionally, adults that identified with an ethnic minority had greater odds of online food delivery service use. Analyses of data from the National Health and Nutrition Examination Survey in the USA reported that respondents who identified with being Black consumed more fast food than respondents that identified with other ethnic groups (196). However, further research from the USA (197), and from the UK (198), concluded that individuals who identified with an ethnic minority group allocated more time to home food preparation and consumed more home cooked food than individuals that identified with an ethnic majority group. There are differences

in racial and ethnic factors across countries (136), including unique cultures, traditions, beliefs, history and experiences (199). Accordingly, the meaning of identifying with a certain ethnic or racial group is unlikely to be similar for populations living in the studied countries. Since this meaning and possible differences therein would not have been captured in the current study due to category aggregation, future research could seek to better understand how identifying with a certain racial or ethnic group is associated with online food delivery service use on a more granular level.

Online food delivery service use could reduce the prevalence and frequency of home cooking, which in turn, might have implications for overall dietary patterns. Although it is possible to meet dietary guidelines by consuming food prepared out-of-home, it is difficult and more expensive compared with consuming food prepared at home (198, 200), and bound by the types of food outlet that can be accessed (201). The latter makes food outlet access in both the physical and digital food environments particularly important to consider in future research.

In my research, online food delivery service customers were younger, had higher rather than lower levels of education, and often lived with children. For age, one explanation is that older individuals are not inclined to order food online due to technology unfamiliarity and a loyalty towards conventional purchasing formats (202). In contrast, younger individuals have plausibly never experienced purchasing food prepared out-of-home in ways that are not internet-based (i.e. `conventional` purchasing formats). Moreover, both parents and those who are more highly educated often report having limited time, and may purchase food prepared out-of-home to offset pressure from limited time resources (84, 203, 204).

Online food delivery service use did not vary by respondent weight status. This may be due to the use of cross-sectional data and simultaneous exposure and outcome measurement. Further research is necessary to develop a better understanding about plausible associations between online food delivery service use and body weight. Analysis of the National Diet and Nutrition Survey in the UK identified that greater consumption of food from fast-food outlets but not restaurants or cafés was associated with living with obesity (205). However, in my study, it was not possible to disaggregate online food delivery service use by food outlet type during analyses. Moreover, it is necessary to consider the direction of association between online food delivery service use and body weight.

The prevalence of online food delivery service use and the number of meals purchased directly from food outlets in-person by non-online food delivery service customers, were both greatest amongst respondents in Mexico. Together, these findings may reflect cultural norms in Mexico aligned with the frequent purchase of food prepared out-of-home (206). However, online food delivery service use was apparent across all studied countries supporting that these services are internationally established (144). Individuals with greater access to food outlets through online food delivery services could be inclined to use them more frequently. Indeed, my finding that respondents from Canada had lower odds of online food delivery service use compared with respondents in other countries could indicate that there is currently limited access to food outlets through this purchasing format. This conclusion would help to explain plans from Just Eat, branded as SkipTheDishes, to increase the number of food outlets in Canada registered to accept orders through their platform (181).

Sociodemographic characteristics of online food delivery service customers were similar between countries, however, the strength of associations varied. Notably, a higher (versus lower) level of education was associated with greater odds of online food delivery service use in all countries except the UK. The type of food available through online food delivery services in the UK is currently unclear. Elsewhere, amongst food outlets registered to accept orders through an online food delivery service in Australia, the Netherlands and the USA, common food labels used to describe the type of food sold included `Burgers`, `Pizza` and `Italian`, with `Healthy` food labels less commonly used (156). Although labels appointed by food outlets may not always reflect the food they sell, given the apparent lack of `healthy` options, it is plausible that online food delivery services in the UK do not sell food that accommodates the needs of individuals with higher education, plausibly limiting their use. Further work to validate self-appointed labels against the types of food sold, and the nutritional quality of this food, is warranted.

2.6.3 Strengths and limitations

My research represents the most comprehensive description of online food delivery service use to date. Nonetheless, my findings are subject to limitations, including those common to surveybased research. Respondents in the IFPS were recruited using nonprobability-based sampling, which means that my findings are not necessarily nationally representative. I applied poststratification sample weights to improve representativeness, however, respondents in Mexico had higher levels of education than census estimates, and average BMI scores were lower than national averages for respondents in all countries (184). Recruitment may have also been biased towards individuals with internet access. However, internet penetration rates across studied countries in 2016 were between 67% (Mexico) and 93% (Australia), with rates of 88% or higher in Canada, the UK and the USA (207). My analyses were based on cross-sectional data, limiting the ability to draw strong causal inferences. Additionally, data were self-reported. Social desirability bias may have led to the number of meals purchased out-of-home, online food delivery service use, and body weight being under-reported. The risk of this may have been reduced by online data collection, which offers respondents a sense of anonymity when reporting sensitive information (208, 209). Finally, I used education as a marker of socioeconomic position, which may not have been comparable across studied countries (102, 103, 111).

2.7 Conclusions

I found that 15% of a large sample of adults living across Australia, Canada, Mexico, the UK, and the USA had purchased food prepared out-of-home through online food delivery services in the past week. Online food delivery service use was associated with being male, identifying with an ethnic minority, being younger, or having a high level of education, with the strongest associations observed for those living with children. Although sociodemographic characteristics of online food delivery service customers were consistent across countries, there was variation in the strength of associations. Norms surrounding the purchase of food prepared out-of-home, stressors on time that plausibly limit the opportunity for cooking at home, and the number and type of food outlets that can be accessed through online food delivery services may vary internationally and could help to explain observed differences between countries. Further research is needed to understand the extent to which online food delivery service use supplements or substitutes other purchasing formats, the reasons for using these services when alternative options are seemingly available, and implications for diet and diet-related health.

Chapter 3:

Experiences and perspectives of frequent online food delivery service customers in the United Kingdom

3.1 Details of author contributions, publication and dissemination

For the research in this chapter, I conceptualised the research questions and study design along with JA and TB. I was responsible for participant recruitment and data collection and led data analysis. I led data interpretation with support from JA and TB. All authors read and provided critical comments on the initial manuscript and agreed to the final version.

This chapter is published as: Keeble, M., Adams, J., Burgoine, T. (2022). Investigating experiences of frequent online food delivery service use: a qualitative study in UK adults. BMC Public Health. 22, 1365. https://doi.org/10.1186/s12889-022-13721-9.

I shared the findings from the research in this chapter during an oral presentation at the 2021 UK Society for Behavioural Medicine Annual Scientific Meeting (held online), and a poster presentation at the 2022 Society for Social Medicine Annual Scientific Meeting (held in-person, Exeter, United Kingdom). For the latter, the meeting abstract is published as: Keeble, M., Burgoine, T., Adams, J. (2022). P14 Investigating frequent online food delivery service use: a qualitative study in UK adults. Journal of Epidemiology Community Health. 76, A54-A55. https://dx.doi.org/10.1136/jech-2022-SSMabstracts.113.

3.2 Abstract

Background

Food prepared out-of-home is typically energy-dense and nutrient-poor. This food can be purchased from multiple retailers, including restaurants and takeaway food outlets. Using online food delivery services to purchase food prepared out-of-home is increasing in popularity. This may lead to more frequent consumption, which is positively associated with poor diet and living with obesity. Understanding possible reasons for using online food delivery services might contribute to the development of future public health interventions, if deemed necessary. This knowledge would be best obtained by engaging with individuals who use online food delivery services as part of established routines. Therefore, I aimed to investigate customer experiences of using online food delivery services to understand their reasons for purchasing food prepared out-of-home in this way, including any advantages and drawbacks.

Methods and results

In 2020, I conducted telephone interviews with 22 adults living in the United Kingdom (UK) who had used online food delivery services on at least a monthly basis over the previous year. Through codebook thematic analysis, I generated five themes: `The importance of takeaway food', 'Less effort for more convenience', 'Saving money and reallocating time', 'Online food delivery service normalisation` and `Maintained home food practices`. Two concepts were overarching throughout: 'Place. Time. Situation.' and 'Perceived advantages outweigh recognised drawbacks`. After considering the context of their location and the time of day, participants reported that they typically selected online food delivery services to purchase food prepared out-of-home instead of other purchasing formats. Participants reported that they did not use online food delivery services to purchase healthy food. Participants considered online food delivery service use to be a normal practice that involves little effort due to optimised purchasing processes. As a result, these services were seen to offer convenient access to food aligned with sociocultural expectations. Participants reported that this convenience was often an advantage but could be a drawback. Although participants were price-sensitive, they were willing to pay delivery fees for the opportunity to complete other tasks whilst waiting for meal delivery. Furthermore, participants valued price-promotions and concluded that receiving them justified their online food delivery service use. Despite online food delivery service use, participants considered home cooking to be irreplaceable.

Conclusions

Future public health interventions might seek to increase the healthiness of food available online whilst maintaining sociocultural expectations. Extending restrictions adopted in other retail settings to online food delivery services could also be explored.

3.3 Introduction

Purchasing food that is prepared out-of-home and served ready-to-consume (often referred to as 'takeaway food') is prevalent across the world (29). The physical food environment includes food outlets where individuals can purchase and consume takeaway food, with an increased number of outlets selling this food plausibly contributing to a normalisation of its consumption (210). Purchasing formats represent ways to buy takeaway food. Although the opportunity to purchase this food was once limited to 'conventional' purchasing formats like visiting food outlets in-person or placing orders directly with food outlets by telephone, additional options such as online food delivery services now exist (145). Online food delivery services are an aspect of food retail within the digital food environment. On a single online platform, customers receive aggregated information about food outlets they can order from based on their location. Customers then select a food outlet, and place and pay for their order. These orders are forwarded to food outlets with a customer-facing premises in the physical food environment where meals are prepared before being delivered (147). Widespread internet and smartphone access is reported to have increased online food delivery service use, with market revenue in the United Kingdom (UK) estimated at around £11.5 billion in 2021 (211).

Food prepared out-of-home is typically nutrient-poor and served in portion sizes that exceed public health recommendations for energy content (41, 175). More frequent consumption of this food has been associated with poor diet and elevated body weight over time (110). Although it is currently unclear, using online food delivery services might lead to more frequent and greater overall consumption of food prepared out-of-home. In turn, this could lead to increased risk of elevated body weight. Since an estimated 67% of males and 60% of females in the UK were considered overweight or obese in 2019 (13), the possibility that using online food delivery services increases overall consumption of food prepared out-of-home is a public health concern that has been recognised by the European region of the World Health Organization (145, 212).

With respect to the physical food environment, food outlet accessibility (number and proximity (i.e. distance to nearest)) and availability (presence of variety), as well as attitudinal dimensions (acceptability, accommodation and affordability) contribute to food purchasing practices (77). Each of these domains plausibly influences food access through online food delivery services. In 2019, the number of food outlets accessible through the leading online food delivery service (Just Eat) was 50% greater in the most deprived areas of England compared with the least deprived areas (213) (see Chapter 5). Furthermore, adults living in the UK with the highest number of food outlets accessible online had greater odds of online food delivery service use in the past week compared with those who had the lowest number (214) (see Chapter 4). However,

attitudinal dimensions of online food delivery service use are poorly understood. Given the complexity of food purchasing practices, there are likely to be unique and specific reasons for using online food delivery services. Understanding these reasons from the perspective of customers could contribute to more informed public health policies, which is important since public health interventions that include online food delivery services may be necessary as their use becomes increasingly popular (141, 212).

3.3.1 Study aims

In my study, I investigated experiences of using online food delivery services from the perspective of adults living in the UK who use them frequently. I aimed to understand their reasons for using these services, the possible advantages and drawbacks of doing so, and how it coexists with other food-related practices.

3.4 Methods

Between June and August 2020, I used semi-structured telephone interviews to study experiences of using online food delivery services from the perspective of adults living in the UK. I used the Consolidated Criteria for Reporting Qualitative Research (COREQ) checklist to guide study development and reporting (215). The University of Cambridge Humanities and Social Science Research Ethics Committee provided ethical approval (Reference: 19/220).

3.4.1 Methodological orientation

I used a qualitative description methodological orientation to investigate my study aims. Qualitative description has been framed as less interpretative than other orientations as it is often used to explore a phenomenon in its natural state whilst relying on the spoken word of participants during data interpretation (216). Nevertheless, in line with my study aims, qualitative description is theoretically and epistemologically flexible and can facilitate a rich description of perspectives (217).

3.4.2 Participants and recruitment

I used convenience sampling to recruit adults that used online food delivery services frequently. For the purpose of my study, I defined frequent customers as those who had used online food delivery services on at least a monthly basis over the previous year. I believed this level of use would make participants positioned to provide their experiences of using this purchasing format within established purchasing practices, rather than less frequent or only recent use. Guided by the findings from Chapter 2 (218), and from others elsewhere (219), I specifically wanted to recruit younger rather than older adults. I did not initially specify further individual-level sociodemographic characteristics as inclusion criteria. However, as data collection progressed, I introduced criteria so that participants were not only highly educated (see Figure 3.1).



Figure 3.1: Inclusion criteria used during frequent online food delivery service customer recruitment between June and August 2020.

Note: * introduced after 12 participants had been recruited so that the study sample included participants with different levels of education.

I used two social media platforms (Twitter and Reddit) to recruit participants. Participant recruitment through social media platforms can be fast and efficient (220-222). If targeted advertising is not used (as in my study), participant recruitment in this way is also free. This approach was appropriate because my aims were related to understanding experiences of online food delivery service use, which requires internet access. For Twitter, users can publish and re-publish information, images, videos, and links to external sites. During recruitment through this platform, I published study information using my personal account and relied on existing connections to re-publish i. For Reddit, users can publish information, images and videos, and discuss topics within focused forums known as 'Subreddits'. For recruitment through this platform, I created an alias account (I did not have a personal account at the time of my fieldwork) and published study information in Subreddits for cities in the UK with large populations according to the 2011 UK census, those related to online food delivery services, and those that discuss topics relevant to the UK (223). Box S1 (see Appendix B) reports a complete list of Subreddits.

Published study information asked interested individuals to contact me by email. When contacted, I responded by email with screening questions that asked individuals to self-report their frequency of online food delivery service use over the past year, and their age and level of education (only after 12 participants had been recruited). When eligibility was confirmed, I provided further information about the study by email. This information included the study aims, details about researchers involved, the offer of a £20.00 electronic high street shopping

voucher, and a formal invitation to participate. After five business days with no response to my invitation, I sent a further email. After another five business days, I classified individuals that did not respond as `non-respondents`.

3.4.3 Data collection

3.4.3.1 Before data collection

Before starting data collection, I planned to complete a maximum of 25 interviews. I considered this number realistic within the constraints of conducting research as part of my PhD studentship. I did not target data saturation, partly to reflect that I pre-specified the maximum number of telephone interviews that I would complete. Additionally, food practices, including purchasing and consuming food prepared out-of-home, are highly individual and informed by previous experiences, cultural backgrounds, and preferences (224). As such, I also felt that it would be difficult to conclude that data saturation had been achieved (225, 226). Instead, I prioritised conceptual depth and information strength. This approach was aligned with the qualitative description methodological orientation of my study (see Section 3.4.1) (226).

I wanted to investigate experiences of online food delivery service use from before the Coronavirus Disease 2019 (COVID-19) pandemic, which was a time when there were no restrictions on access to multiple purchasing formats and the opportunity for on-premises food consumption. Therefore, I pre-specified that I would stop data collection if it became difficult for participants to refer to the time before March 2020, which is when COVID-19 pandemic related travel and food outlet access restrictions were first introduced in the UK. I piloted an initial protocol with an eligible individual to confirm this would be possible and then made amendments based on their feedback.

Before starting data collection, I reflected on my position as a public health researcher and my previous training and experience in qualitative research (227). I also reflected on my own consumption of food prepared out-of-home and previous online food delivery service use. As of the start of my fieldwork in June 2020, I consumed this food infrequently and had previously placed one order with an online food delivery service. Although I was not a frequent customer according to my classification, I was familiar with online food delivery services operating in the UK. I concluded that despite understanding why online food delivery services might be used, I could not use my own experiences to provide detailed reasons for favouring this purchasing format over alternative options.

3.4.3.2 Throughout data collection

I completed one-off semi-structured telephone interviews with participants at a convenient time selected by them. At the start of the interview process, I confirmed the rationale for my study, gave participants the opportunity to ask clarifying questions, and asked them to provide verbal consent. I used a topic guide that was developed based on *a priori* knowledge, pilot interview feedback and previous research related to food prepared out-of-home and online food delivery services (36, 50, 218). I amended the topic guide as data collection progressed so that points I had not initially considered could be discussed in future interviews. Interview questions focused on reasons for using online food delivery services, the perceived advantages and drawbacks of using these services, and how their use coexisted with the use of alternative purchasing formats and other food-related practices. Box S2 (see Appendix B) shows the final topic guide.

Although I completed interviews in the context of the COVID-19 pandemic, I did not ask questions related to this. I also asked participants to think about the time before March 2020 so that we could discuss their experiences from before the COVID-19 pandemic. I digitally recorded interview audio and made field notes to track points for discussion within each respective interview.

3.4.3.3 After data collection

Immediately after each interview, I reflected on topics discussed with participants, data collection progress, possible links with existing theory and the ability of participants to think about the time before the COVID-19 pandemic. I used my post-interview reflections to help decide when to stop data collection.

3.4.4 Data analysis

A professional company transcribed interview audio verbatim. Whilst listening to the corresponding audio, I quality assured and anonymised each transcript. Participants did not review their transcripts.

I used codebook thematic analysis. When using this analytic approach, researchers develop a codebook based on the final topic guide used during data collection and data familiarity, which is achieved by reviewing collected data (228, 229). Codebook thematic analysis is aligned with qualitative description methodological orientations as it allows researchers to remain close to the data and facilitates an understanding of a topic through the spoken word of participants (230). In practice, I developed an initial codebook. I then reviewed three transcripts (a 10% sample) alongside co-authors. This number was manageable and allowed us to discuss a sample

of collected data (231). After discussion with co-authors, I refined the initial codebook to collapse codes that overlapped and to add new codes. In doing so, I formed the final codebook. I coded each transcript with the final codebook and reviewed reflections written after each interview. I then studied the coded data to generate themes that were discussed and finalised with co-authors. In the context of my study, themes summarise experiences of using online food delivery services from the perspective of participants. I also identified that across the generated themes, there were overarching concepts. Concepts should be seen to offer an overall and consistent structure that capture the common and overlapping elements of each of the generated themes (232).

I used NVivo (version 12) to manage the data and facilitate interpretation.

3.5 Results

3.5.1 Participant and data overview

I conducted telephone interviews with 22 frequent online food delivery service customers between June and August 2020. Interviews lasted between 35 and 61 minutes. Table 3.1 reports the sociodemographic characteristics of participants and their online food delivery service use. There were 12 males, 13 participants were aged between 20 and 29 years, and 15 participants had completed University (i.e. higher education). Since initial adoption, online food delivery services had mostly been used on a weekly or monthly basis, and in one instance, on a daily basis. Participants consistently referred to using the three largest online food delivery services operating in the UK (Just Eat, Deliveroo, and Uber Eats).

During the 19th interview, conducted in August 2020, I identified that it seemed difficult for the participant to think about the time before the onset of the COVID-19 pandemic in March 2020. I completed three further interviews and then concluded that this difficulty was consistent so stopped data collection. I included data from all completed interviews in analyses since experiences and perspectives of participants from before March 2020 were gathered after appropriate prompting. In addition to the completed interviews, three interviews were scheduled but cancelled by individuals without providing a reason, and there were nine non-respondents.

Participant number	Gender	Age	Highest education ^a	Online food delivery service(s) used	Typical usage frequency ^b	Time since adoption
		(years)				(years)
1	Female	20-29	Higher	Deliveroo & Just Eat	Monthly	≤5
2	Male	20-29	Higher	Deliveroo, Just Eat & Uber Eats	Fortnightly	>5
3	Female	20-29	Higher	Deliveroo	Weekly	Unknown
4	Female	30-39	Higher	Deliveroo & Just Eat	Weekly	≤5
5	Female	30-39	Higher	Deliveroo & Just Eat	Weekly	≤5
6	Female	20-29	Higher	Deliveroo & Just Eat	Monthly	Unknown
7	Female	20-29	Higher	Deliveroo	Weekly	≤5
8	Female	40-49	Higher	Just Eat	Monthly	≤5
9	Male	30-39	Compulsory	Deliveroo, Just Eat & Uber Eats	Weekly	>5
10	Female	20-29	Further	Deliveroo	Monthly	≤5
11	Male	30-39	Higher	Just Eat	Monthly	>5
12	Female	20-29	Higher	Deliveroo, Just Eat & Uber Eats	Fortnightly	>5
13	Male	20-29	Compulsory	Deliveroo, Just Eat & Uber Eats	Weekly	≤5
14	Female	20-29	Higher	Just Eat	Monthly	>5
15	Male	40-49	Further	Just Eat	Monthly	≤5
16	Male	20-29	Higher	Deliveroo & Uber Eats	Fortnightly	≤5
17	Male	30-39	Higher	Deliveroo	Monthly	>5
18	Male	20-29	Higher	Just Eat	Fortnightly	>5
19	Male	30-39	Further	Deliveroo	Weekly	≤5
20	Male	20-29	Compulsory	Deliveroo & Just Eat	Fortnightly	≤5
21	Male	20-29	Compulsory	Deliveroo & Uber Eats	Daily	≤5
22	Male	30-39	Higher	Deliveroo & Uber Eats	Weekly	>5

Table 3.1: Sociodemographic characteristics and online food delivery service use amongst frequent customers (n=22) interviewed between June and August 2020.

Notes: ^a Highest level achieved or underway. `Compulsory` = High school, `Further` = Education after high school, not including a university degree, `Higher` = University degree. ^b Since initial adoption.

3.5.2 Summary and structure of findings

Two concepts were overarching throughout the data: `Place. Time. Situation.` and `Perceived advantages outweigh recognised drawbacks`. Within these overarching concepts, there were five generated themes: `The importance of takeaway food`, `Less effort for more convenience`, `Saving money and reallocating time`, `Online food delivery service normalisation` and `Maintained home food practices`.

The sections that follow are organised so that I present the findings for each of the overarching concepts, followed by each of the themes. Whilst I discuss each concept and theme individually, elements of each were present throughout the data and should be thought of as dynamic, overlapping, and non-hierarchical. For example, participants consistently reflected on features of online food delivery services according to the context of their location at a specific time. The conclusion of this process determined if a given feature was viewed as an advantage or a drawback and in some cases, if online food delivery services would be used. Table 3.2 at the end of the Results provides examples of this comparison process.

3.5.3 Overarching concepts

3.5.3.1 Place. Time. Situation.

Participants described how their location and the time of day influenced their ability to access different types of food. When choosing one type of food over another, participants had a multi-factorial thought and decision-making process that considered their food at home, their immediate finances available for food, and the food they had already eaten that day.

Although my interview questions focused on the consumption of food prepared out-of-home, which I referred to as `takeaway food` throughout, participants were clear that purchasing this type of food was not always appropriate. As participant 10 (Female: 20-29 years) stated; *"I don't always just go and get a takeaway; sometimes I'll walk to the shop, get some food, and make something"*. This view was shared by participant 11 (Male: 30-39 years); *"some days I'll decide that it's too expensive and I'll either get something else directly from the restaurant or go to the supermarket and then make food"*.

Nonetheless, participants indicated that purchasing takeaway food was preferable in many situations. For example, when acting spontaneously, when meals had not been previously planned for, or when other types of food could not satisfy their needs, then takeaway food was appropriate.
"I think you're more likely to get delivery and order online when it's unplanned and you need a pick-me-up, or you need something quick, or you don't have something and you're really hungry." Participant 15 (Male: 40-49 years)

When participants decided to purchase takeaway food, they recognised that their location and the time of day dictated the purchasing formats they could access and potentially use. Access to multiple purchasing formats created a second decision-making process. Participants considered the cuisine they wanted, delivery times estimated by online food delivery services versus the time it would take for them to travel to a food outlet, the weather, their willingness to leave home, and previous experience with food outlets that would deliver to them. Alongside these factors, choosing one purchasing format over another was often based on what was most convenient.

"If I'm out and about, on the way home and I'm passing via an outlet, then I'll pick it up. If I'm at home and just kind of, don't want to leave the house, I'll order via an app or online, because it's just convenient." Participant 2 (Male: 20-29 years)

Despite having apparently decided how they would purchase takeaway food, participants stated that they were willing to change their mind. In the case of online food delivery services, if estimated delivery times failed to meet their expectations then this purchasing format would not be appropriate and another option or type of food would be selected. Using a specific purchasing format also had to align with other routines and schedules. This was particularly clear when participant 8 (Female: 40-49 years) described that they used online food delivery services when they could *"relax on a Friday night with the whole evening free"*. However, if they did not have time to select a food outlet, place their order, and wait for delivery, then they *"normally just have some spaghetti because that takes 10 minutes"*.

Participants referred to online food delivery service marketing in their day-to-day environments. In the context of the physical food environment, this marketing included branded food outlet signs and equipment used by delivery couriers. Participants stated that these things did not always trigger immediate online food delivery service use, yet their omnipresence reminded them that these services were available.

"I don't know if I ever go onto Just Eat after seeing it advertised, I don't think that's ever directly led me to do it. But it certainly keeps it in your mind, it's certainly at the forefront of your mind whenever you think of takeaway food." Participant 11 (Male: 30-39 years)

3.5.3.2 Perceived advantages outweigh recognised drawbacks

Throughout the data, participants recognised that a single online food delivery service feature could be an advantage or a drawback based on their location and the time of day. This was clearest when participant 2 (Male: 20-29 years) discussed the number of food outlets accessible online compared with those accessible through alternative purchasing formats. There was value in having access to "20, 30, 40 food outlets" through online food delivery services as it meant there was greater choice, otherwise "you're more limited just by the virtue of where you are or what shops you're passing". However, access to a higher number of food outlets was a drawback when it meant that making a selection was difficult. The constant comparison of advantages and drawbacks prompted me to ask participants why they kept using online food delivery services. There was a consensus that these services had unique features that were, more often than not, advantageous. As participants continued to use online food delivery services to access unique features, this practice appears to be self-reinforcing, even if this means accepting that the same feature can sometimes be a drawback. Nevertheless, participants acknowledged that if the overall balance between advantages and drawbacks changed then they would purchase takeaway food in other ways. This solution emphasises that takeaway food can often be accessed through multiple purchasing formats, dependent on place and time. Despite this, as it stands, participants anticipated that they would continue to use online food delivery services indefinitely.

"I can't see any reason why I would [stop using online food delivery services], unless something went wrong with Just Eat, you know, the service had a massive problem, but at the moment I can't see any reason why I would." Participant 16 (Male: 20-29 years)

3.5.4 Analytic themes

Having discussed the two overarching concepts, I now present each of the five themes generated from analyses. As described, elements of each theme overlapped within the two overarching concepts presented to this point.

3.5.4.1 The importance of takeaway food

Participants emphasised that it was *"the food"* they most valued and led them towards online food delivery services.

"It's the food really, that leads me to use [online food delivery service] apps." Participant 10 (Female: 20-29 years)

Participants reported that they did not use online food delivery services with the intent of purchasing healthy food. In fact, they expected takeaway food and the food available to buy through online food delivery services to be unhealthy. This perspective influenced the types of food that participants were willing to purchase online. For example, pizza (seen as unhealthy) was appropriate but a salad (seen as healthy) was not. Moreover, participants recognised that if they wanted to consume healthy food, they would most likely cook for themselves.

Participants stated that takeaway food had social, cultural, and behavioural value. For many, purchasing and consuming takeaway food at the end of the working week signified the start of the weekend, which was seen as a time for relaxation and celebration. This tradition had been carried forward from childhood, with Friday night referred to as *"takeaway night"*. Using an online food delivery service allowed participants to maintain, yet digitalise, traditions.

"It's always a weekend thing, besides it being a convenient, really quick way of accessing food that is filling and tastes nice, for me, it marks the end of a work week." Participant 4 (Female: 30-39 years)

Participants reported that in some situations, consuming takeaway food as a group could be a way to socialise. This was especially the case during life transitions such as leaving home to start university.

"When you move out, you're concentrating on making friends, and getting a takeaway was quite an easy way for everyone to sit down around the table and socialise and to have drinks." Participant 14 (Female: 20-29 years)

Participants did not value online food delivery services to the same extent that they did takeaway food. This perspective reinforced that online food delivery services were primarily used to satisfy takeaway food purchasing needs.

"If Just Eat as an entity disappeared, or all online takeaways disappeared, I wouldn't be upset [...] it's a luxury, it makes life easier." Participant 9 (Male: 30-39 years)

3.5.4.2 Less effort for more convenience

Participants reported that it takes little effort to use online food delivery services because they receive information about each of the food outlets they can order from on a single platform. Participants particularly valued the opportunity to save their payment details, previous orders, and favourite food outlets for future use. Participants also reported that they could order from an increased number of food outlets and a more diverse range of foods compared with other purchasing formats. Due to the number of food outlets accessible online, the selection process was not always fast. Nonetheless, participants indicated that online food delivery services make purchasing takeaway food easier and more convenient than other purchasing formats where information is less readily available.

"You've got all of the different options laid out in front of you, it's like one resource where everything is there, and you can choose and make a decision, rather than having to pull out leaflets from a drawer or Google different takeaways in the area. It's all there and it's all uniform and it's in one place." Participant 3 (Female: 20-29 years) "I can pick through a whole wide selection rather than being limited to the few takeaways down on my road or having to drive somewhere." Participant 21 (Male: 20-29 years)

Participants emphasised that online food delivery service smartphone applications had been optimised to enhance their purchasing experience.

"I guess it's the convenience of just being able to open the app on my phone, and not having to go searching for menus or phone numbers and checking if places are open. So yeah, it's the convenience." Participant 15 (Male: 40-49 years)

"For me, it's just the ease of going on, clicking what you want, paying for it and it arriving. You don't have to move, you don't have to cook, you don't have to think, it's just there ready to go, someone's doing the hard work for you." Participant 1 (Female: 20-29 years)

However, greater convenience was not always advantageous. Some participants were concerned that convenient and easy access to takeaway food through online food delivery services might have negative consequences for health and other things.

"It's quite addictive in the way that it's just so convenient to order. I'm not making stuff fresh at home, and I'm eating unhealthier." Participant 21 (Male: 20-29 years) "I think it adds to a general kind of laziness that is not good for people really. If you actually got up and went for a walk to go and get this food, at least there's a slightly positive angle there." Participant 17 (Male: 30-39 years)

"The convenience is not necessarily a positive thing [because] *these apps can be abused because it's so easy to access foods."* Participant 10 (Female: 20-29 years)

3.5.4.3 Saving money and reallocating time

Participants were sensitive to the price of takeaway food and valued the opportunity to save money. When discussing financial aspects of online food delivery service use, participants referred to special offers they had received by email or through mobile device push notifications. Participants reported that direct discounts (e.g. 10% off), free items (e.g. free appetisers on orders over £20.00), free delivery (e.g. on orders over £30.00), or time-limited price-promotions (e.g. 40% off all orders for the next three-hours) justified their online food delivery service use.

"Getting a takeaway is always a treat, every time I do it I know I shouldn't but then basically I'm convinced to treat myself, if there's a discount I'm much more likely to do it because I don't feel like it's such a waste of money." Participant 18 (Male: 20-29 years)

Participants recognised takeaway food as being a distinct food category. Nevertheless, they appreciated that they could use online food delivery services to purchase other types of food, like food from restaurants. Since this food is usually accompanied by a complete dining experience that online food delivery services cannot replicate, participants expected to spend less when purchasing it online.

"Some restaurants deliver through Deliveroo, [these are places] *where you can sit down and have an experience, a dining experience,* [...] *well that's different because you might go there for the dining experience."* Participant 4 (Female: 30-39 years)

"Sometimes I'm deterred from using Uber Eats because I noticed that the restaurants increase their prices if you buy it through them rather than directly [...] I don't want to pay over £10 for a takeaway dish, whereas I would pay that if I ate at a restaurant." Participant 3 (Female: 20-29 years)

Although participants were sensitive to the price of takeaway food, they were willing to trade money for time. Participants compared the time they would spend cooking or travelling to customer-facing food outlets with the time taken to place orders through online food delivery services plus the tasks they could complete whilst waiting for meal delivery. Paying a delivery fee to have the opportunity to use time that would not have otherwise been available was acceptable.

"Yeah, it costs money but at the same time we're getting more time with the kids and more time to do other stuff, so it's absolutely fine as far as I'm concerned." Participant 9 (Male: 30-39 years)

However, some participants were unsure about the appropriateness of paying to have food delivered, as it might be unfair to delivery couriers.

"I don't feel like it's necessarily right to make a delivery courier drive two minutes up the road just because I can't be bothered to go and collect something that's not very far away." Participant 10 (Female: 20-29 years)

3.5.4.4 Online food delivery service normalisation

Participants consistently reported having positive previous experiences of using online food delivery services. These experiences influenced future custom and contributed to an overall sense that using this purchasing format was now a normal part of living in a digital society. Some participants referred to watching television online to exemplify this point. The normalisation of using online food delivery services was particularly evident when I asked participants for their first thoughts about the term `takeaway food`. Participants often referred to online food delivery services in the first instance and saw them as synonymous with this type of food.

"If you were to say `takeaway food` I'd pull out my phone and I'd open one of the apps and say `okay, what should we order`, I wouldn't say `oh let's go to this road`, or `let's go to that road`, I'd say `yeah, let's look on the app`." Participant 21 (Male: 20-29 years)

For participants in my study, using online food delivery services replaced purchasing takeaway food in other ways. This perspective was linked to habitual takeaway food purchasing and sociocultural values. Participants purchased takeaway food within set routines (for example only at the weekend) because they did not always think it was appropriate to do so at other times. As a result, participants reported that they had a limited number of opportunities to use multiple purchasing formats and thus increase existing levels of consumption.

3.5.4.5 Maintained home food practices

Most participants were responsible for cooking at home, enjoyed doing so, and said they were competent at it. Nonetheless, cooking at home required personal effort, and being *"lazy"*, *"tired"*, or *"having nothing in the cupboards"* justified online food delivery service use.

"I cook, when I'm not using these apps I cook and prepare food for myself, it's just on the odd occasion I might be feeling tired or want something different [...] *the rest of the time, I'm quite happy to cook."* Participant 10 (Female: 20-29 years) Despite the apparent normalisation of using online food delivery services, participants did not feel that these services would eliminate cooking at home. Most participants consumed home cooked food daily, whereas they consumed takeaway food less frequently. This contributed to the view that these two types of food were different. As a result, participants used online food delivery services to purchase food they could not, or would not, cook at home, for a break from normality, and as a *"cheat"* or *"treat"*.

Table 3.2: Examples of how frequent online food delivery service customers (n=22) compared the advantages and drawbacks of features of the online food delivery service business model, identified from data collected between June and August 2020.

Feature	Perceived advantage	Perceived drawback
Food outlet information and menus can be viewed, and orders placed, on one platform	Orders can be placed with little effort	It is <i>too easy</i> , and it takes no effort to purchase takeaway food
An increased number of food outlets are accessible compared with other purchasing formats	Food outlets, cuisines, and price points, including those not normally available, can be selected	Selecting a food outlet is difficult because there is <i>too much</i> choice
Unique promotional offers can be used	Money can be saved, additional food items can be received, and meals can be delivered for free	It is <i>too appealing</i> to place orders when promotional offers are available
Takeaway food can be purchased	The available food meets expectations	The available food is mostly unhealthy
Meals are delivered	Takeaway food can be received without leaving home	Having takeaway food delivered when the food outlet is nearby might be lazy
Delivery typically involves an additional fee	Paying a delivery fee is worth it to carry out other tasks whilst waiting	Delivery fees can be expensive

3.6 Discussion

3.6.1 Summary of findings

To my knowledge, this is the first research in the public health literature to investigate experiences of using online food delivery services from the perspective of frequent customers. Participants recognised that their location and the time of day meant that they often had access to different types of food through multiple purchasing formats at the same time. Participants stated that purchasing takeaway food was appropriate in many situations and typically favoured using online food delivery services. For many participants, using these services was now part of routines in their increasingly digital lives. As such, using online food delivery services appeared to be synonymous with takeaway food purchasing. This meant that participants expected food sold online to be unhealthy, with them reporting that they were not inclined to purchase healthy food in this manner. Participants consistently thought about how features of online food delivery services were an advantage or a drawback within the context of their location at a given point in time. This was a complex and dynamic process. Participants described how the advantages of these services were a strong enough reason to continue use, and that they outweighed drawbacks. Participants reported that using online food delivery services involved little effort as they were provided with food outlet information and menus, and payment facilities on one platform that had been optimised for their use. Moreover, although the cost of food was an important consideration for participants, they were willing to pay a fee in exchange for the opportunity to complete tasks whilst waiting for meal preparation and delivery. Finally, using online food delivery services substituted purchasing takeaway food in other ways. Nevertheless, participants reported that cooking at home was a distinct food practice that occurred more frequently and was irreplaceable.

3.6.2 Interpretation of findings

Participants described sociocultural values assigned to takeaway food. Purchasing this food formed part of weekend traditions and routines to celebrate the end of the working week. Sociocultural values are proposed to develop from previous experiences (86, 115), which might have influenced the perspective of participants. In the past, traditions and routines might have led to visiting food outlets in-person. However, online food delivery services can now be used. Since participants reported that takeaway food in and of itself was a reason for seeking out these services, it is reasonable to conclude that sociocultural values linked to this food are relevant across purchasing formats. The decision to purchase food has been recognised as being situational and made in the context of place and time (114, 233), with convenience reported as a consistent consideration (234). Participants reported that takeaway food was appropriate in many situations and acknowledged that it could often be accessed through multiple purchasing formats depending on their location at a given time. Selecting one purchasing format over another came after the consideration of multiple factors, including the level of effort required to find a suitable food outlet and to place orders. Using online food delivery services was often most convenient because doing so took little effort. However, participants were clear that although their decision had seemingly been made, it could be changed if an online food delivery service feature that was supposedly an advantage became a drawback. For example, if estimated delivery times were too long or delivery fees were too high then an alternative purchasing format would be considered. My findings support that the decision to purchase takeaway food is dynamic and influenced by place and time (36).

Food access has previously been summarised within the domains of availability, accessibility, affordability, accommodation, and acceptability (77). Broadly speaking, I investigated the acceptability of using online food delivery services, and participants made explicit reference to the domains of food accessibility, availability, and affordability. For example, one particularly valuable aspect of using online food delivery services was the ability to order from a higher number of food outlets compared with other purchasing formats. The experiences reported by participants in my study support the possibility that domains of food access are relevant to the decision to adopt, and maintain, online food delivery service use. Other features of these services, such as having information about each of the food outlets that could be ordered from on one platform, likely amplify the perceived benefit of increased food outlet access. Notably, access to an increased number of food outlets was not always advantageous, which is aligned with awareness about the negative aspects of takeaway food consumption reported by young adults in Australia and Canada (115, 235).

Participants also discussed how the price of food and delivery fees influenced their online food delivery service use. This reflects that food affordability contributes to food purchasing practices (36). This finding also provides insight into actions that retailers registered to accept orders through online food delivery services might take to attract customers. Customers can often select from multiple food outlets at the same time. As a result, food outlets might aim to compete with one another by lowering the price of food sold or by introducing price-promotions. The latter were particularly valued by participants. Elsewhere, price-promotions contribute to food purchasing practices (236, 237). Access to price-promotions through online

food delivery services has not been systematically documented. However, it is possible that their number is positively associated with the number of food outlets accessible online. Since both appear to influence online food delivery service use, the possibility of interaction between them is concerning for overall consumption of food prepared out-of-home, and subsequently, diet and diet-related health.

In some cases, participants reported that they used online food delivery services because they did not have time to cook at home. A number of tasks, including household chores, work, travel and childcare can limit the time available for, and take priority over, cooking at home (238). Using online food delivery services instead of cooking at home allowed participants to complete other tasks whilst waiting for meal preparation and delivery. This would not have necessarily been achievable when using more conventional purchasing formats. Due to sociocultural values and perceived `rules` about how frequently takeaway food `should` be purchased, participants did not see online food delivery services as a complete replacement for cooking at home. Nevertheless, even partial replacement has implications for diet and diet-related health, especially since purchased foods were acknowledged as unhealthy by participants in my study. This perception is aligned with evidence that food consumed at home is often healthier than food consumed out-of-home (239).

3.6.3 Possible implications for public health and future research

Participants reported that using online food delivery services had mostly substituted, not supplemented, the use of other purchasing formats. Given this perspective, food outlets could increasingly register to accept orders through these services to supply an apparent demand. Further research is required to understand the extent to which customer demand is driven by food outlet access through online food delivery services and vice versa. Relatedly, participants reported that despite using online food delivery services frequently, their overall takeaway food consumption had remained the same. It is unclear if this perception would be reflected in objective assessment of overall takeaway food consumption. Further research that quantifies the use of multiple purchasing formats and takeaway food consumption over time is required to understand the potential public health implications of using online food delivery services. Evidence from Australia suggests that food sold through online food delivery services tends to be high in calories, total fat, saturated fat and salt (165). This has not been established in the UK. However, the food is likely to be similar. Nevertheless, the type of food available does not necessarily reflect the balance of the food purchased. Assessment of the nutritional quality of foods available, and purchased, through online food delivery services in the UK would help to understand the extent to which public health concern is warranted.

Price-promotions justified online food delivery service use. Legislation to restrict access to volume-based price-promotions (e.g. buy-one-get-one-free, 50% extra free) on less healthy prepackaged food sold both in-store and online were due to be introduced in England in October 2022 (240). However, the introduction of this legislation was delayed, and at the time of writing had not been implemented. Regardless, hot food served ready-to-consume (i.e. food prepared out-of-home) was due to be excluded. Given what is known about the association between exposure to price-promotions and food purchasing practices (241), extension of these restrictions to hot food served ready-to-consume might be warranted. Understanding the extent to which price-promotions are associated with online food delivery service use represents a first step to understand the need for related public health intervention in the future.

3.6.4 Strengths and limitations

I recruited participants through two social media platforms, which means that those included were from a subset of all social media users. However, online recruitment was appropriate since I wanted to understand experiences of using a purchasing format that is an aspect of food retail within the digital food environment. The participants I recruited were mostly highly educated, which potentially reflects online food delivery service use amongst this socioeconomic group as I reported in Chapter 2 (although not for the UK) and as has been reported elsewhere (219, 242). I acknowledged this and adjusted my recruitment strategy after 12 telephone interviews to recruit a more balanced sample with respect to level of education. Nevertheless, future research should explore the perspectives of frequent online food delivery service customers or non-customers, as they would not have been in a position to provide information aligned with my study aims. However, since I have described experiences of using online food delivery services only from the perspective of frequent customers, future work should seek to understand the perspectives of others.

As the first study in the public health literature to investigate customer experiences of online food delivery service use, I used a descriptive methodological orientation. Doing so meant that I did not investigate the underlying meaning of the language used by participants. Nevertheless, my descriptive methodological orientation allowed me to use codebook thematic analysis and include co-authors in this process. Coding a 10% sample of interview transcripts and generating analytic themes together would have been less appropriate with reflexive approaches (228, 229, 243), but assisted with my interpretation of findings.

68

Finally, I conducted fieldwork during the early stages of the COVID-19 pandemic, which might have altered the recent experiences of online food delivery service use and participant perspectives. However, I asked participants to think about the time before the COVID-19 pandemic during interviews, and reflected on their ability to do so afterwards. My reflective practices are in line with established attempts to ensure qualitative rigour (216, 244), and allowed me to decide when it would be most appropriate to stop fieldwork.

3.7 Conclusions

I used telephone interviews with frequent online food delivery service customers to investigate experiences of using this purchasing format. I found that place and time influenced if and how takeaway food would be purchased. Participants often felt that online food delivery services were the most appropriate way to access this food. In part, this was due to the opportunity to use features not offered by other purchasing formats, including efficient ordering processes that had been optimised for customer convenience. Fundamentally, online food delivery services provide access to takeaway food. Although participants acknowledged that this food is unhealthy, it held strong sociocultural values. Participants were aware that some advantages of online food delivery services might also be drawbacks. Despite this, the drawbacks were not sufficient to stop their use. Finally, participants informed me that price-promotions justified their online food delivery service use. Public health interventions that seek to promote healthier food purchasing online may be increasingly warranted in the future. Interventions might include increasing the healthiness of the food available whilst maintaining sociocultural values and expectations. Restriction of price-promotions for food prepared out-of-home could also plausibly be extended to include online food delivery services.

Chapter 4:

Associations between online access to food prepared out-of-home and online food delivery service use in adults in the United Kingdom

4.1 Details of author contributions, publication and dissemination

For the research in this chapter, I developed the research questions and conceptualised data analysis with support from JA and TB, after consultation with DH and LV. Under the supervision of JA and TB, I led data preparation and analysis, and interpretation of findings. All authors read and provided critical comments on the initial manuscript and agreed to the final version.

This chapter is published as: Keeble, M., Adams, J., Hammond, D., Vanderlee, L., Burgoine, T. (2021). Associations between online food outlet access and online food delivery service use amongst adults in the UK: a cross-sectional analysis of linked data. BMC Public Health. 21, 1968. https://doi.org/10.1186/s12889-021-11953-9.

I shared the findings from the research in this chapter during an oral presentation at the 2022 International Medical Geography Symposium (held in-person, Edinburgh, United Kingdom).

4.2 Abstract

Background

Online food delivery services facilitate `online access` to food outlets that typically sell energydense nutrient-poor food. Greater online food outlet access might be related to online food delivery service use and living with excess body weight, however, this is not known. I aimed to investigate the association between aspects of online food outlet access and online food delivery service use, and differences according to customer sociodemographic characteristics, as well as the association between the number of food outlets accessible online and body weight.

Methods

I linked data for all food outlets in the United Kingdom (UK) registered with the leading online food delivery service, Just Eat (n=33,204), in 2019 with contemporaneous data on out-of-home food purchasing, body weight, and sociodemographic information collected through the International Food Policy Study (analytic sample n=3067). I used adjusted binomial logistic regression, linear regression, and multinomial logistic regression models to examine associations.

Results

Adults in the analytic sample had access to a median of 85 food outlets (interquartile range (IQR): 34, 181) and 85 unique types of cuisine (IQR: 64, 108) through the online food delivery service, and 15% reported online food delivery service use in the past week. Those with the highest number of food outlets accessible online (quarter four, 182–879) had 71% greater odds of online food delivery service use (odds ratio (OR): 1.71; 95% confidence intervals (CI): 1.09, 2.68) compared to those with the lowest number (quarter one, 0–34). The pattern of this association was evident amongst adults with a university degree (OR: 2.11; 95% CI: 1.15, 3.85), adults aged between 18 and 29 years (OR: 3.27, 95% CI: 1.59, 6.72), those living with children (OR: 1.94; 95% CI: 1.01; 3.75), and females at each increase in the number of food outlets accessible online. I found no association between the number of unique types of cuisine accessible online and online food delivery service use, or between the number of food outlets accessible online and body weight.

Conclusions

In conclusion, the number of food outlets accessible online was positively associated with online food delivery service use. Adults with the highest education, younger adults, those living with children, and females, were particularly susceptible to the greatest online food outlet access. Further research is required to investigate the possible implications of online food delivery service use on diet and diet-related health.

4.3 Introduction

Over the last two decades, the number of food outlets selling food prepared out-of-home has increased globally (31, 32, 34, 35). Whilst purchasing food prepared out-of-home is influenced by many factors (70), the number of food outlets selling this food that are physically accessible is thought to contribute and has been extensively researched (27, 66, 107). Evidence from cross-sectional and longitudinal studies demonstrates that individuals living in areas with greater food outlet access consume food prepared out-of-home more frequently and live with higher body weight and obesity (185, 245). These associations are stronger for those of a lower socioeconomic position (109), with further variation in the strength of associations based on sociodemographic characteristics including age, sex, and household composition (107).

In the past, food prepared out-of-home was conventionally purchased directly from food outlets. However, changing social-norms (142), technological advances (141), widespread internet availability (246), and a desire for greater convenience (247), have contributed to the emergence of alternative purchasing formats. Online food delivery services provide online access to food outlets selling food prepared out-of-home. Based on their location, customers receive aggregated information about the food outlets accessible online (i.e. those that will deliver to them). Customers select a food outlet and place their order through the platform of an online food delivery service. Orders are then forwarded to the food outlet where meals are prepared, and when ready, meals are delivered by couriers who work for the online food delivery service or independently for the food outlet.

Socioecological models propose that an interplay between physical food outlet access and individual-level characteristics influences the purchase of food prepared out-of-home (66, 72, 248, 249). However, little is known about the factors that plausibly influence online food delivery service use. Based on the findings of research investigating the role of physical food outlet access, it is reasonable to suggest that a higher number of food outlets accessible via online food delivery services is associated with more frequent use of this purchasing format, and that the influence of this exposure varies according to customer characteristics. Another possible explanation is that the characteristics of food outlets access to different types of food outlets that sell a range of cuisines (165). As taste preferences contribute to food practices (250), another possible contributor to online food delivery service use is the type of cuisine sold by food outlets. Finally, food available through online food delivery services is typically energy-dense and nutrient-poor (159, 165). Since consumption of such food has been associated with weight

gain over time (110), it is plausible that greater online food outlet access is associated with having elevated body weight on a distal basis.

4.3.1 Study aims

In my study, I aimed to investigate the association between the number of food outlets and the number of unique types of cuisine accessible online and online food delivery service use. Additionally, where an association was present, I aimed to examine differences according to known sociodemographic characteristics of online food delivery service customers. In secondary analyses, I aimed to investigate the association between the number of food outlets accessible online and body weight.

4.4 Methods

4.4.1 Study design

This was a cross-sectional data-linkage study.

4.4.2 Study sample

The International Food Policy Study (IFPS) is an ongoing annual repeat cross-sectional survey conducted in Australia, Canada, Mexico, the United Kingdom (UK), and the United States of America (USA). I used data collected between November and December 2019 from respondents living in the UK. I described data collection methods in Chapter 2, and they have also been described elsewhere (251). Briefly, respondents were recruited through Nielsen Consumer Insights Global Panel and their partners' panels. Panellists were screened for eligibility and quota requirements based on device type, age, and sex. Email invitations with links to an online survey were sent to a random sample of eligible panellists aged 18 years or over. Informed consent was obtained from all respondents prior to survey completion. The IFPS received ethics clearance through a University of Waterloo Research Ethics Committee (ORE# 21460). Data collection in the UK was approved by the University of Cambridge Humanities and Social Science Research Ethics Committee (Reference: 19/225).

4.4.3 Measures

4.4.3.1 Exposures: online food outlet access and unique type of cuisine access

Just Eat has been available in the UK since around 2006 and in 2020 was the online food delivery service market leader in terms of the number of food outlets registered to accept orders (around 35,000) and annual order volume (over 120 million) (252, 253). Unlike competitors that

tend to focus on operating only in large cities, Just Eat reports that it has outlets registered to accept orders through their service across all parts of the UK (254). My pilot work in one area in England, identified that 95% of food outlets registered to accept orders through the next largest online food delivery service operating in the UK (Deliveroo) were also registered with Just Eat (see Appendix C: Figure S1). Therefore, I used data from Just Eat (referred to as `the online food delivery service` hereafter) as a proxy for online food outlet access.

In November 2019, I used a web-browser extension to automate the collection of data on food outlets accessible online across England, Scotland, and Wales. First, on one weekday, I identified all food outlets registered to accept orders. Second, within 72-hours, I visited the profile of each food outlet on the online food delivery service website and collected information on their physical location, the types of cuisine sold (classifications self-determined by outlet owners), and their delivery area, which is all postcode districts to which they would deliver. A postcode district is the first half of a full postcode, for example, for the *postcode* `CB2 0QQ`, the *postcode district* is `CB2`. Postcode districts are used for mail and delivery routing purposes and have an average size of 33 mi² (255). Based on data from the 2011 census, the median postcode district population was 23,610 (interquartile range (IQR): 13,320-34,560) (223).

In the IFPS survey, respondents were asked to report their residential postcode. From this, I extracted the postcode district and identified the number of food outlets in data collected from the online food delivery service that listed the same postcode district in their delivery area. This was the number of food outlets accessible online. From these food outlets, I summed the number of unique types of cuisine that could be accessed.

4.4.3.2 Primary outcome: online food delivery service use

In the IFPS survey, all respondents were asked *"During the past 7-days, how many meals did you get that were prepared out-of-home in places such as restaurants, fast food or takeaway places, food stands, or from vending machines?"*. Respondents who had purchased at least one meal prepared out-of-home were then asked to report the number of meals that were *"ordered using a food delivery service (e.g., Uber Eats, Just Eat, Deliveroo) and delivered"*. I used answers to the second question for my primary outcome. As data were not normally distributed and right-skewed, I dichotomised respondents into those who reported any online food delivery service use in the past week and those who did not.

4.4.3.3 Secondary outcomes: body mass index and weight status

In the IFPS survey, all respondents were asked to self-report their height and weight in either metric or imperial units. I used this data to calculate body mass index (BMI; kg/m², continuous), and used World Health Organization BMI cut-offs to classify respondents as being: `not overweight` (BMI \leq 24.9 kg/m²); `overweight` (BMI 25.0-29.9 kg/m²); or `obese` (BMI \geq 30.0 kg/m²). I included respondents in a `not available` category when I was unable to calculate BMI due to body weight non-report, which is a possible reflection of social-desirability bias (256), or when calculated BMI was <14.0 kg/m² or >48.0 kg/m².

In accordance with findings from research investigating the relationship between food outlet access in the physical food environment and body weight (110, 245), it is plausible that greater online access to food prepared out-of-home is positively associated with BMI and weight status. However, the primary focus of my study was on the association between online food outlet access and online food delivery service use, which is more proximal and potentially less susceptible to bias from unmeasured confounding (107, 257). I have previously shown this potential pathway in Figure 1.6 (see Section 1.4.4). Given the above, I report the findings for my secondary outcomes in Appendix C.

4.4.3.4 Potential confounders: sociodemographic characteristics and neighbourhood food outlet access

In the IFPS survey, all respondents were asked to report sociodemographic information. I included potential confounders based on evidence regarding the purchase of food prepared out-of-home (64). Sex at birth was reported as male or female. I treated this as a binary variable for analysis. As I discussed in Chapter 2 and Section 1.2.5.4, although food purchasing practices are considered gendered, there is evidence of a correlation for responses to survey questions on sex and gender (188). Age was reported in years. Due to the possible non-linear influence on food purchasing, I grouped respondents into four age categories for analysis: 18-29 years, 30-44 years, 45-59 years, ≥60 years. Ethnicity was reported as the group that best described racial or ethnic background. I grouped respondents into a binary variable for analysis: `majority` (all `White` ethnicities) or `minority` (all other ethnicities, e.g. Asian, Black, Indian), which reflects that the majority of IFPS respondents in the UK identified with a White ethnic group (251). I used education and perceived income adequacy as markers of socioeconomic position (102, 103). Education was reported as the highest level completed. I categorised respondents as having a: `low` (high school completion or lower), `medium` (some post-high school qualifications), or `high` (university degree or higher) level of education for analysis. Perceived income adequacy

was reported based on how well total monthly incomes allowed needs to be met. I grouped respondents into two groups: `not easy` (don't know, refuse to answer, very difficult, difficult, or neither easy nor difficult responses) or `easy` (easy or very easy responses). Living with children under the age of 18 years and smoking status in the past 30 days were binary (yes/no) measures. I maintained the dichotomous nature of these measures for analysis.

Food sold through online food delivery services is typically prepared inside of existing customerfacing food outlets located in the physical food environment (39). Therefore, online food outlet access is predicated on the presence of food outlets in the physical food environment. I used Ordnance Survey Points of Interest (OS POI) data from June 2019 to determine the number of food outlets physically accessible. This is commercial data containing information about retailers across multiple sectors, collated from over 170 suppliers, is one of the most complete sources of food outlet location data available for the UK, and has been used in previous research investigating food outlet access in the physical food environment (258-260). I extracted information for the following categories as they include food outlets predominantly registered to accept orders through online food delivery services: `Fast food and takeaway outlets` (food outlets selling food prepared out-of-home for off-premises consumption), 'Fast food delivery services` (food outlets selling food prepared out-of-home for delivery, not explicitly through online food delivery services), 'Fish and Chip shops' (food outlets selling a traditional British cuisine, prepared out-of-home, typically for off-premises consumption) and `Restaurants` (food outlets selling food prepared out-of-home for on-premises consumption) (261). I used coordinates supplied in OS POI data, which have a stated accuracy of one metre (262), to map food outlets in a geographic information system (GIS) (ArcGIS version 10.7.1; ESRI Inc., Redlands, CA). I primarily used Doogal (a free web-based resource) to obtain coordinates for the postcodes of IFPS respondents (263). I used the GeoConvert tool (maintained by the UK Data Service) when this was not successful (264). I mapped obtained coordinates in the GIS and created a 1600 m (1-mile) Euclidean (straight-line) radius around postcode coordinates. I then counted the number of food outlets listed in OS POI data located within the 1-mile boundary to determine neighbourhood food outlet access (the number of physically accessible food outlets). A distance of 1600 m has been shown to reflect the spatial extent of an individual's typical shopping practices, and can reasonably be walked by an adult in less than 20 minutes (265, 266).

4.4.4 Exclusion criteria

Data were available for 4139 IFPS respondents living in the UK. I removed respondents with missing postcode information; covariate (except BMI and perceived income adequacy) or outcome data; when they lived in Northern Ireland (OS POI data does not include this country);

or when the total number of meals purchased out-of-home in the past week exceeded 21 (as I did not consider this to be plausible based on the maximum consumption of three meals per day). The final analytical sample included 3067 respondents (74% of all respondents).

4.4.5 Statistical analysis

I used Stata version 16.1 (StataCorp LLC., College Station, TX, USA) to complete statistical analysis with a significance threshold of p<0.05 throughout. I applied post-stratification sample weights constructed based on population estimates of age, sex, ethnicity and education from the 2011 UK census to reduce non-response and selection bias (251). I rescaled sample weights to reflect the number of participants included in the analytic sample. Unless specified, I report weighted findings.

Residuals for all measures were not normally distributed. Therefore, I categorised exposure measures (the number of food outlets accessible online and the number of unique types of cuisine accessible online) as quarters (Q). For each exposure measure, Q1 was the quarter with the lowest number and used as the reference category. I used binomial logistic regression models to investigate the association between each exposure and online food delivery service use in the past week (none or any). I completed unadjusted analyses and analyses adjusted for potential confounders (sex, age, education, perceived income adequacy, living with children and ethnicity, and neighbourhood food outlet access). Where the exposure was the number of unique types of cuisine accessible online, I additionally adjusted for the number of food outlets accessible online, due to a positive association.

I added a multiplicative interaction term (number of food outlets accessible online x respective sociodemographic characteristic) to separate adjusted binomial logistic regression models (not adjusted for the respective sociodemographic characteristic). Doing so allowed me to investigate if the association between the number of food outlets accessible online and online food delivery service use varied according to IFPS respondent education level, age, sex, or if they lived with children. I used post-estimation Wald Tests to determine interaction significance. When significant, I completed further analyses stratified by the respective sociodemographic characteristic.

4.4.5.1 Sensitivity analyses

In sensitivity analyses, I wanted to test that I had appropriately defined neighbourhood food outlet access. Therefore, when I constructed this potential confounder, I included additional categories from OS POI data alongside those initially included (`Fast food and takeaway outlets`,

`Fast food delivery services`, `Fish and Chip shops` and `Restaurants`). First, I included the number of supermarkets (as these outlets can provide access to food prepared out-of-home). Second, I included the number of: `Cafés, snack bars and tea rooms`, `Convenience stores`, `Supermarkets`, `Bakeries` and `Delicatessens`. As outlined in Section 4.4.3.4, I counted the number of food outlets in these categories that were within 1600 m of a given respondent's postcode and created quarters of exposure. I included these newly constructed measures of neighbourhood food outlet access in adjusted analyses.

4.4.5.2 Secondary outcomes

I used linear regression models to investigate the association between online food outlet access and BMI (continuous), and multinomial logistic regression models to investigate the association between online food outlet access and weight status. I completed unadjusted analyses and adjusted analyses that included potential confounders (sex, age, education, perceived income adequacy, living with children, ethnicity, and smoking status, which is negatively associated with body weight, and neighbourhood food outlet access) (267). Due to the exploratory nature of these analyses, I did not complete sensitivity analyses or explore interactions.

4.5 Results

4.5.1 Sample characteristics

Table 4.1 summarises sociodemographic characteristics, access to food prepared out-of-home and online food delivery service use for respondents in the analytic sample. The median number of food outlets and the median number of unique types of cuisine accessible online was 85.0 (IQR: 34.0, 181.0 and IQR: 64.0, 108.0, respectively). Around one in six respondents (15.1%) had used an online food delivery service in the past week. The average BMI of online food delivery service customers was 26.7 kg/m² and 48.1% were living with overweight or obesity. Table S1 and Table S2 (see Appendix C) report the sociodemographic characteristics, access to food prepared out-of-home and food purchasing practices of IFPS respondents not included in final analyses. These were not materially different to respondents in the analytic sample. **Table 4.1**: Prevalence of online food delivery service use, exposure to food prepared out-of-home, and sociodemographic characteristics of the analytic sample (n=3067) from the 2019 International Food Policy Study.

Measure	N ^a	%
Online food delivery service use in		
the past week		
None	2604	(84.9)
Any	464	(15.1)
Food exposures (count): median (IQR)		
Online		
Outlet number	85	(34.0-181.0)
Unique cuisine type number	85	(64.0-108.0)
Neighbourhood		
Outlet number	25	(9.0-57.0)
Sex		
Male	1513	(49.3)
Female	1554	(50.6)
Age		
18-29 years	479	(15.6)
30-44 years	716	(23.4)
45-59 years	823	(26.8)
<u>></u> 60 years	1048	(34.2)
Ethnicity		
Minority	267	(8.7)
Majority	2800	(91.3)
Education level		
Low	1570	(51.2)
Medium	652	(21.3)
High	845	(27.5)
Ability to make ends meet		
Not easy	1790	(58.4)
Easy	1277	(41.6)
Child at home		
No	2268	(74.0)
Yes	799	(26.0)
Regular smoker		
No	2417	(78.8)
Yes	650	(21.2)
BMI : mean (standard deviation) ^b	26.7	(5.3)
Weight Status (BMI: kg/m²)		
Not overweight (≤ 24.9)	1077	(35.1)
Overweight (25-29.9)	877	(28.6)
Obesity (≥ 30)	597	(19.5)
Not available	517	(16.9)

Notes: ^a Data presented as weighted number of respondents (%) unless stated. May not always equate to 3067 due to rounding.

^b 2551 respondents included in calculation.

4.5.2 Associations between online food outlet access and online food delivery service use in the past week

In the unadjusted model, the number of food outlets accessible online was positively associated with online food delivery service use in the past week, with suggestion of a dose response relationship (see Appendix C: Table S3). Figure 4.1 shows that associations were attenuated in the adjusted model, however, the positive association persisted for those with the greatest online food outlet access (Q4). Those with the highest number of food outlets accessible online (182-879 outlets) had 71% greater odds of online food delivery service use (odds ratio (OR): 1.71; 95% confidence intervals (CI): 1.09, 2.68) compared with those in Q1, who had the lowest number (0-34 outlets).



Figure 4.1: Odds of any online food delivery service use in the past week per quarter of online food outlet access amongst the analytic sample (n=3067) from the 2019 International Food Policy Study. Note: Data collected in 2019, analysed using adjusted binomial logistic regression. Analysis adjusted for the following potential confounders: neighbourhood food outlet access, sex, age, education level, perceived income adequacy, living with children and ethnicity. The number of food outlets accessible online for each quarter (Q) were; Q1: 0–34, Q2: 35–85, Q3: 86–181, Q4: 182–879.

4.5.3 Interactions between online food outlet access and sociodemographic characteristics

There was evidence that the association between the number of food outlets accessible online and online food delivery service use in the past week varied by sociodemographic characteristics of customers: education (p=0.0015), age (p<0.0001), living with children (p<0.0001) and sex (p<0.0001) (see Appendix C: Table S4). Figure 4.2 presents findings from stratified analyses. The positive association between the highest number of food outlets accessible online and online food delivery service use was evident amongst respondents with the highest education (OR: 2.11; 95% CI: 1.15, 3.85), those aged between 18-29 years (OR: 3.27; 95% CI: 1.59, 6.72), and those living with children (OR: 1.94; 95% CI: 1.01, 3.75), but not those in other strata of these variables. The positive association increased at each level of exposure for female respondents but was entirely absent in males.



Figure 4.2: Odds of any online food delivery service use in the past week per quarter of online food outlet access amongst analytic sample (n=3067) from the 2019 International Food Policy Study, stratified by sociodemographic characteristics.

Note: Data collected in 2019, analysed using separate adjusted logistic regression stratified by: A) education level; B) age; C) living with children; and D) sex. Analyses adjusted for the following potential confounders: neighbourhood food outlet access, sex, age, education level, perceived income adequacy, living with children and ethnicity. The number of food outlets accessible online for each quarter (Q) were; Q1: 0–34, Q2: 35–85, Q3: 86–181, Q4: 182–879.

4.5.4 Sensitivity analyses

Including neighbourhood access to supermarkets and a broader range of food outlets in the adjusted model revealed similar patterns to the main analyses (see Appendix C: Table S5).

4.5.5 Associations between unique types of cuisine accessible online and online food delivery service use

I found no evidence of an association between the number of unique types of cuisine accessible online and online food delivery service use in the past week in the unadjusted (see Appendix C: Table S6) and adjusted models (see Figure 4.3).



Figure 4.3: Odds of any online food delivery service use in the past week per quarter of online access to unique types of cuisine amongst the analytic sample (n=3067) from the 2019 International Food Policy Study.

Note: Data collected in 2019, analysed using adjusted binomial logistic regression. Analysis adjusted for the following potential confounders: neighbourhood food outlet access, online food outlet access, sex, age, education level, perceived income adequacy, living with children and ethnicity. The number of unique types of cuisine accessible online for each quarter (Q) were; Q1: 0–64, Q2: 65–85, Q3: 86–108, Q4: 109–148.

4.5.6 Secondary analyses: associations between online food outlet access and body weight

Table S7 (see Appendix C) reports findings from secondary analyses. In the unadjusted model, the number of food outlets accessible online was inversely associated with BMI for respondents with the highest number. Respondents with the highest number of food outlets accessible online also had lower odds of living with obesity compared with not being overweight. These associations were attenuated to extinction in adjusted models.

4.6 Discussion

4.6.1 Summary of findings

For the first time in the international published literature, I investigated the association between multiple measures of access to food prepared out-of-home through an online food delivery service and the use of this purchasing format. After adjustment for potential confounders, I found that adults living in the UK with the highest number of accessible food outlets (i.e. that they could order from) had 71% greater odds of reporting any online food delivery service use in the past week compared with those who had the lowest number. This positive association was evident amongst those who were more highly educated, those aged between 18-29 years, those living with children, and females. I found no evidence of an association between the number of unique types of cuisine accessible online and online food delivery service use, or between the number of food outlets accessible online and BMI or weight status.

4.6.2 Interpretation of findings

Amongst adults living in the UK, those with the highest number of food outlets accessible online (between 182 and 879 outlets) had greater odds of self-reporting online food delivery service use in the past week. As my study was the first investigation into the relationship between online food outlet access and online food delivery service use, there is no existing evidence with which to directly compare my findings. Nonetheless, socioecological models propose that exposure to food outlets is an environmental cue that can influence food purchasing practices (66, 268). Indeed, previous research has reported that access to a higher number of food outlets in the physical food environment is associated with purchasing food prepared out-of-home more frequently (52, 245). As food sold through online food delivery services is typically prepared inside of existing customer-facing food outlets, the number of food outlets accessible online is likely to be closely linked to the number of food outlets in the physical food environment (39). As I controlled for this in analyses, it is plausible that seeking out this purchasing format is a response to other environmental cues not necessarily captured by measures of food outlet density. These cues may include the presence of food delivery couriers (101), or digital cues from targeted marketing (269), both of which are likely to be more prominent when a higher number of food outlets are present in the physical food environment (270).

The positive relationship between online food outlet access and online food delivery service use was specific to adults with the highest education, younger adults, those living with children, and females. In Chapter 2, I reported that online food delivery services are used by individuals with these sociodemographic characteristics (218), with similar findings reported amongst Australian adults (219). Younger adults and those in higher socioeconomic positions spend more time using the internet (271), and have the greatest odds of self-reporting exposure to marketing from online food delivery services (272). Additionally, social roles suggest that females and parents seek out food for others (233, 273). As a result, these individuals in particular might be prone to receiving and acting on cues to visit online food delivery services. Intentionally seeking out this purchasing format reflects shifting sociocultural norms regarding the way that food prepared out-of-home is purchased (169, 246). Indeed, engagement with online food delivery services likely leads to and promotes exposure to multiple aspects of these services, including the food outlets that will deliver to a given location. In contrast, those who do not use these services are not exposed to, and thus influenced by, the number of food outlets that would deliver to them if they were a customer. Therefore, the decision to visit platforms provided by online food delivery services and to become a customer appears critical.

Future research might seek to understand why the number of unique types of cuisine accessible online was not associated with online food delivery service use. It seems intuitive that having access to a higher number of cuisines facilitates access to greater food choice (274). However, it is also plausible that access to a higher number of cuisines results in choice overload, which in turn could lead to the use of an alternative purchasing format (275). This was alluded to by frequent online food delivery service customers who reported that too much choice was seen as a drawback of these services (see Chapter 3) (242). 4.6.3 After adjustment for potential confounders, there were no significant associations between the number of food outlets accessible online and either BMI or weight status. Whilst an association is plausible, my findings might reflect that this relationship has not yet had time to develop. Although online food delivery services have been available in the UK since around 2006, their use is reported to have only become an established practice more recently (39). A relationship may emerge in the future. Furthermore, online food delivery services are one aspect of food retail within the digital food environment and the broader food system. Unmeasured factors such as food access through other purchasing formats, including within the neighbourhood food environment, lifestyle preferences, and established purchasing routines might have contributed to my findings, yet were not captured in my measure of food access. Possible implications for public health and future research

Individuals living across England, Scotland and Wales who had the highest education were particularly susceptible to greater online food outlet access in terms of online food delivery service use. As such, these services may have a negative influence on overall dietary patterns in population groups who currently tend to have the best (276), which is a public health concern.

Alternative purchasing formats that exist alongside online food delivery services include independent food outlet websites and small-scale online food delivery services. However, these are less prominent and offer access to a limited number and range of food outlets compared with internationally established online food delivery services. Nevertheless, their emergence suggests a normalisation towards accessing food prepared out-of-home in a digital manner. This normalisation is reflected in the forecasted increase in frequency of online food delivery service use and purchasing food in the digital food environment (141).

The food available through online food delivery services is typically energy-dense and nutrientpoor (159, 162). Public health interventions that aim to improve the nutritional quality of food prepared out-of-home have previously been adopted (277, 278). As these interventions are often implemented inside of customer-facing food outlets, which is where food sold through online food delivery services tends to be prepared, they are well placed to improve the nutritional quality of food accessible online.

4.6.4 Strengths and limitations

A major strength of my study was the use of automated data collection. This approach allowed unprecedented nationwide collection of exposure data contemporaneous with outcome data collected from a large sample of adults. The use of exposure and outcome data collected at different time points is common in research investigating associations between food outlet access and purchasing food prepared out-of-home (279). This temporal mismatch could result in exposure misclassification that was absent in my work. Moreover, I investigated online food delivery service use as my primary outcome. This outcome is most proximal to online food outlet access and improves the specificity of my investigations (280, 281). Finally, the IFPS survey was developed from existing measures that have been validated, are accurate, and are used in national surveys (251).

Nonetheless, my findings should be interpreted in light of methodological limitations. If food outlets were registered with the online food delivery service but not identified during data collection, exposure could have been underestimated and misrepresented (282). I identified that 33,204 food outlets were registered to accept orders online in November 2019. This is similar to the *"over 30,000"* reported as being registered to accept orders in contemporaneous reports from the data source (283), which provides confidence in the completeness of my data. Furthermore, my data were collected in November 2019, which pre-dates purported changes in food purchasing practices during the COVID-19 pandemic. I will further discuss this period of time in Chapter 6.

My cross-sectional analysis is unable to infer a strong causal relationship between the number of food outlets accessible online and online food delivery service use. Moreover, data for online food delivery service use were self-reported, which introduces the possibility of under-reporting due to social-desirability bias. These limitations are not unique to my research, yet could help understand my somewhat counterintuitive finding of no association between the number of food outlets accessible online and online food delivery service use.

Finally, I used 1600 m straight-line buffers to define the neighbourhood food environment of respondents. The use of this buffer size may have influenced the number of food outlets identified as being physically accessible. Although previous research has operationalised neighbourhood boundaries ranging from 400 m to 3200 m (26, 284), 1600 m buffers reflect the spatial extent of an individual's typical shopping practices, and this distance can be walked by an adult in around 20 minutes (265).

4.7 Conclusions

More frequent online food delivery service use could increase the consumption of food prepared out-of-home, which has known implications for diet and diet-related health. My study is the first to investigate the association between the number of food outlets and the number of unique types of cuisine accessible online and online food delivery service use. After adjusting for potential confounders, adults in the UK with the highest number of food outlets accessible online had greater odds of online food delivery service use compared with those who had the lowest number. This association was particularly evident in adults who were more highly educated, younger adults, those who lived with children, and amongst females. I did not find evidence that the number of unique types of cuisine accessible online was associated with online food delivery service use. Moreover, the number of food outlets accessible online was not associated with body weight.

Chapter 5:

The socioeconomic patterning of online access to food prepared out-of-home in England

5.1 Details of author contributions, publication and dissemination

For the research in this chapter, I developed the research questions alongside JA and TB. I developed data collection protocols, and collected data alongside Tom Bishop (TRPB). I led data analysis and interpretation of findings with support from JA and TB. All authors read and provided critical comments on the initial manuscript and agreed to the final version.

This chapter is published as: Keeble, M., Adams, J., Bishop, T.R.P., Burgoine, T. (2021). Socioeconomic inequalities in food outlet access through an online food delivery service in England: a cross-sectional descriptive analysis. Applied Geography. 133, 102498. https://doi.org/10.1016/j.apgeog.2021.102498.

I shared the findings from the research in this chapter during oral presentations at the 2021 American Association of Geographers Annual Meeting (held online), the 2021 Public Health Research and Science Conference (held online), and the 2022 International Medical Geography Symposium (held in-person, Edinburgh, United Kingdom).

5.2 Abstract

Background

Online food delivery services facilitate `online` access to food outlets selling food prepared outof-home. Systematic differences in online food outlet access could exacerbate existing health inequalities, which is a public health concern. However, this is not known since online food outlet access has not previously been investigated anywhere in England or across a whole country.

Methods

Across postcode districts in England (n=2118), I identified and described the number of food outlets and unique cuisine types accessible through the market leading online food delivery service (Just Eat). I investigated associations with area-level deprivation using adjusted negative binomial regression models. I also compared the number of food outlets accessible online with the number physically accessible in the neighbourhood (1600 m Euclidean buffers of postcode district geographic centroids) and investigated associations with deprivation using an adjusted general linear model. For each outcome, I predicted means and 95% confidence intervals (CI).

Results

In November 2019, 29,232 food outlets were registered to accept orders online. Overall, across England as a whole, the median number of food outlets accessible online per postcode district was 63.5 (interquartile range (IQR): 16.0, 156.0). For the number of food outlets accessible online expressed as a percentage of the number accessible within the neighbourhood, the median per postcode district was 63.4% (IQR: 35.6, 96.5). Analysis through adjusted negative binomial regression models showed that the number of food outlets accessible online was highest in the most deprived postcode districts (n=106.1; 95% CI: 91.9, 120.3). The number of food outlets accessible online was highest in the neighbourhood was highest in the least deprived postcode districts (n=86.2%; 95% CI: 78.6, 93.7).

Conclusions

In England, online access to food prepared out-of-home is socioeconomically patterned. Further research is required to understand how online food delivery service use varies according to area-level deprivation and implications for diet and health inequalities.
5.3 Introduction

In 2018, half of food expenditure in the United States of America (USA) was on food prepared out-of-home (285), and between 2008 and 2012, over one in four adults in the United Kingdom (UK) consumed at least one meal prepared out-of-home each week (40). Food available out-of-home is often characterised by high levels of energy, fat and salt, and on the whole, is less healthy than food prepared at home (46, 175, 286). Decisions related to when and where this food is purchased are complex and multifaceted (72, 174), with the physical food environment having a recognised influence (79). The number of physically accessible food outlets is a geographical measure of 'food access', whereby exposure to customer-facing premises is suggested to act as an environmental purchasing cue (73, 77). Accordingly, physical food environments that provide abundant access to food outlets selling food prepared out-of-home have been referred to as obesity promoting (287). Moreover, it has been consistently reported accross multiple countries that a higher number of food outlets selling food prepared out-of-home are located in more deprived areas (31, 99, 288-290), which may contribute to inequalities in diet and health.

A growing body of cross-sectional and longitudinal evidence from research investigating food outlet access at the individual level now exists (27, 77). Evidence of an association between physical food environment exposures and outcomes such as the purchase and consumption of food prepared out-of-home is mixed (29, 53). In part, this is a reflection of methodological heterogeneity across the existing evidence base, which includes the use of different geographical measures of food outlet access and different conceptualisations of neighbourhood food environments, as well as varying food environment contexts across countries (26, 291). Nonetheless, in two UK studies that used similar methods, exposure to fast-food outlets was positively associated with fast food consumption (109, 245). This food practice has been associated with excess weight gain over time (110). In addition to the aforementioned methodological heterogeneity, although acquiring food prepared out-of-home is no longer restricted to physical food outlet access, previous research has rarely considered alternative purchasing formats.

Online food delivery services facilitate online access to food outlets selling food prepared outof-home. Unlike visiting food outlets in-person, online food delivery service use is internet based (39). When using online food delivery services, customers receive information about each of the food outlets that will deliver to them based on their entered location. Customers then select a food outlet and place their order. Orders are then forwarded to individual food outlets where meals are prepared. When ready, meals are delivered by couriers who work for either the online food delivery service or the food outlet.

In 2020, prominent online food delivery services, Deliveroo and Uber Eats were available in multiple countries, and Grubhub was established in many cities across the USA (179-182). Just Eat (including subsidiaries such as Menulog in Australia) was available in 23 countries (253), and was the market leader in the UK in terms of the number of food outlets registered to accept orders (around 30,000) and the annual number of orders processed (almost 170 million) (252, 292). Like food outlet access in the physical food environment (248, 293), it is possible that access to a higher number of food outlets through online food delivery services leads to an increased number of opportunities to purchase food prepared out-of-home. Indeed, in Chapter 4, I reported that a higher number of accessible food outlets was positively associated with online food delivery service use (214). As mentioned, there are known differences in access to food outlets selling food prepared out-of-home in England, with outlets selling this food more prevalent in the physical food environments of more deprived areas (31). Food sold through online food delivery services is typically prepared inside of existing customer-facing food outlets located in the physical food environment (39). As such, differences in physical food outlet access may be reflected in online food outlet access. However, this has not been investigated in the UK. Moreover, research completed elsewhere has been conducted in a limited number of cities (156, 165). In turn, the full extent of nationwide variation in online food outlet access and potential differences across the full socioeconomic gradient that might only be observed across a whole country has not been investigated. This variation is important to understand since certain sociodemographic groups may be disproportionately influenced by greater cumulative food outlet access across multiple purchasing formats.

Other factors could also influence online food delivery service use beyond the number of food outlets accessible online (i.e. that can be ordered from). Broadly speaking, customers select food outlets based on the foods (i.e. cuisine) they sell (77, 294). Within the context of online food delivery services, access to a higher number of unique types of cuisine could facilitate more choice, resulting in customer needs being accommodated and in turn, online food delivery service use. In Chapter 3, I reported that access to a variety of food was important to frequent online food delivery service customers (242). However, in Chapter 4, I reported that the number of unique types of cuisine accessible online was not associated with online food delivery service use (214). Nevertheless, variation in access to types of cuisine through online food delivery services is important to understand given the differences in nutritional quality between them (295).

5.3.1 Study aims

In this cross-sectional, area-based study, I aimed to describe online access to food outlets and unique cuisines across England; compare online food outlet access with physical food outlet access in the neighbourhood; and examine whether and to what extent these measures were associated with socioeconomic position.

5.4 Methods

5.4.1 Study setting and analytic scale

The study setting was England. The analytic scale was the postcode district level. This analytic scale reflects how food outlets registered to accept orders through Just Eat delineate their `delivery area` (see Section 5.4.3.1). A postcode district is the first half of a full postcode and is formally known as the outward code (255). For the *postcode* `CB2 0QQ`, the *postcode district* is `CB2`. Postcode districts are used for mail and delivery routing purposes and have an average size of 33 mi² (255). Based on data from the 2011 census, the median postcode district population was 23,610 (interquartile range (IQR): 13,320-34,560) (223).

For analyses, I used boundary data from 2012, provided by the UK data service (296), to map postcode districts in England in a geographic information system (GIS) (ArcGIS version 10.7.1; ESRI Inc., Redlands, CA). I included 2118 postcode districts, which were those entirely within the border of England, as well as those in Scotland or Wales with an intersecting boundary (food outlets could be located in these countries yet deliver to locations in England).

5.4.2 Exposure measure

I used data from the 2019 Index of Multiple Deprivation (IMD) to measure relative deprivation. This is a compound measure of deprivation that includes metrics across seven domains: income deprivation, employment deprivation, crime levels, health deprivation and disability, education, skills and training deprivation, barriers to housing and associated services and living environment deprivation (297). Relative deprivation scores are available for lower super output areas (LSOAs) in England, which are administrative boundaries with a mean residential population of 1500 people (298). As LSOAs are typically geographically smaller than postcode districts, I aggregated LSOAs within and intersecting the boundary of each postcode district and calculated the mean IMD score (297). For analyses, I split postcode districts into deciles based on IMD score, with decile 10 containing the most deprived.

5.4.3 Outcome measures

Table 5.1 summarises each of the outcome measures I investigated in my study.

5.4.3.1 Online food access

Information about all food outlets registered to accept orders through the UK market leader (Just Eat), including their opening hours, menus, delivery fees, and customer reviews, is publicly available. As I stated in Chapter 4, my pilot work for one area in England identified that 95% of food outlets registered to accept orders through the next largest online food delivery service (Deliveroo) were also registered to accept orders through Just Eat (see Appendix C: Figure S1). Moreover, unlike competitors, food outlets registered to accept orders through Just Eat are reported to be accessible in almost all areas of England (254). Like Chapter 4, I used data from Just Eat as a proxy for online food outlet access. Given that the aims of my study were related to online food delivery services in a general sense, I refer to Just Eat as the `online food delivery service` hereafter. As I described in Chapter 4 (see Section 4.4.3.1), in November 2019, I used a web-browser extension to collect information about all food outlets registered to accept orders through the online food delivery service across England, Scotland and Wales (299). First, on one weekday, I identified all food outlets registered to accept orders. Second, within 72-hours, I visited the profile of each outlet on the online food delivery service website and collected information on their physical location, the types of cuisine sold (classifications self-determined by outlet owners), and their delivery area, which is a list of all postcode districts to which they would deliver.

For each food outlet registered to accept orders through the online food delivery service, I primarily used Doogal, which is a free web-based resource, to geocode the postcode of customer-facing premises in the physical food environment (263). When geocoding through Doogal was not successful, I used the GeoConvert tool, which is maintained by the UK Data Service (264). I was unable to geocode seven food outlets (0.02%). This left 29,232 food outlets. I used supplied coordinates to map the physical location of food outlets in the GIS.

I used the number of food outlets accessible online and the number of unique cuisine types accessible online as outcome measures. From the data I collected from the online food delivery service, I counted the number of food outlets registered to accept orders through the online food delivery service that listed each postcode district in their delivery area to identify the number of food outlets accessible online. For each postcode district, I counted the number of unique cuisine types used by accessible food outlets. Food outlets can select multiple cuisines to

describe the food they sell. As a result, the number of unique cuisine types accessible online could be greater than the number of food outlets.

5.4.3.2 Percentage of food outlets registered to accept orders online

For this outcome, I used Ordnance Survey Points of Interest (OS POI) data, which is commercial data containing information about retailers across multiple sectors, collated from over 170 suppliers (260). This dataset is one of the most complete sources of food outlet location data available for England (259), and has been used in previous research that investigated aspects of the physical food environment (258). I used data from June 2019 and extracted information for the following food outlet categories: 'Fast food and takeaway outlets' (food outlets selling food prepared out-of-home for off-premises consumption), 'Fast food delivery services' (food outlets selling food prepared out-of-home for delivery, not explicitly through online food delivery services), 'Fish and Chip shops' (food outlets selling a traditional British cuisine, prepared out-of-home, typically for off-premises consumption) and 'Restaurants' (food outlets selling food prepared out-of-home for on-premises consumption) (261). I selected these categories based on *a priori* knowledge that they included food outlets typically registered to accept orders through online food delivery services. I mapped the locations of food outlets using coordinates supplied in OS POI data that are reported to be accurate up to one metre (262), and then identified the postcode district in which they were located.

I compared the number of food outlets located in each postcode district that were registered to accept orders through the online food delivery service with the number of food outlets in each postcode district listed in OS POI data. Doing so allowed me to calculate the percentage of food outlets registered to accept orders online. I did not identify and match individual food outlets listed in both datasets. The number of food outlets registered to accept orders through the online food delivery service should not exceed the number of food outlets located in each postcode district, therefore, I used a bounded (between 0-100%) outcome measure in analyses.

5.4.3.3 Percentage of neighbourhood food outlets accessible online

I expressed the number of food outlets accessible online (see Section 5.4.3.1) as a percentage of the number listed in OS POI data that could be accessed in the neighbourhood food environment for a given (pseudo) population. I defined the neighbourhood food environment for each given population as a 1600 m (1-mile) Euclidean (straight-line) radius around the geographic centre of each postcode district. Food outlet access within this distance has been associated with food-related practices (265, 266). I did not identify and match individual food outlets listed in both datasets. The number of food outlets accessible online may exceed the

number physically accessible in the neighbourhood, therefore, for this outcome measure, the percentage could be greater than 100%.

Although I report the percentage of food outlets registered to accept orders online and the percentage of neighbourhood food outlets accessible online, I acknowledge that in the strictest sense these outcomes have not been calculated as such.

5.4.4 Covariates

5.4.4.1 Physical food outlet access

Food sold through online food delivery services is typically prepared inside of existing customerfacing food outlets located in the physical food environment. These outlets tend to concentrate in highly populated areas, which possibly reflects a perception of greater demand amongst business owners (34). As online food outlet access might be a function of physical food outlet access, I used the four categories from OS POI data described in Section 5.4.3.2 to identify the number of food outlets within the physical food environment of each postcode district. I included this number as a covariate when it was not used in the calculation of the outcome measure.

5.4.4.2 Online food outlet access

The number of unique cuisine types accessible online was positively associated with the number of food outlets accessible online. Therefore, I used the number of food outlets accessible online as a covariate when the outcome was the number of unique cuisine types accessible online.

5.4.4.3 Rural Urban classification and population density

I used the 2011 rural urban classification to categorise postcode districts as: `rural` when LSOAs within or intersecting their boundary were most frequently rural (populations less than 10,000 people within combined settlements, where the majority live in rural-related areas); `urban` when LSOAs were most frequently urban (populations greater than 10,000 people within combined settlements, where the majority live in urban-related areas); or `balanced` when the number of rural and urban LSOAs was equal (300). I included usual residential and workday population from the 2011 UK census (301, 302). The usual residential population is the number of individuals, including students and schoolchildren not living away from home during term-time, who usually reside in a postcode district. The usual workday population is the number of individuals working in a postcode district on a given day, in addition to residents who are unemployed. Data for population density were available for 2088 postcode districts.

Exposure	Outcome	Outcome description	Geography	Covariates added to adjusted model
	Percentage of food outlets registered to	The number of food outlets registered to accept orders online, expressed as a	Postcode district	Postcode district rural urban classification
Postcode district relative deprivation IMD score modelled as deciles: Decile 1 = least deprived.	accept orders online	percentage of the number of food outlets in a postcode district (bounded, 0-100%).		Postcode district population density: usual residential and usual workday
	Number of food outlets accessible	The number of food outlets accessible online based on a postcode district being	Postcode district	Postcode district rural urban classification
	online	listed in the delivery area of a food outlet registered to accept orders online.		Number of food outlets in postcode district
				Postcode district population density: usual residential and usual workday
	Number of unique cuisine types	The number of unique cuisine types accessible online from food outlets that	Postcode district	Postcode district rural urban classification
	accessible online	were accessible.		Number of food outlets in postcode district
				Postcode district population density: usual residential and usual workday
				Number of food outlets accessible online
	Percentage of neighbourhood food	The number of food outlets accessible online expressed as a percentage of the	1600 m Euclidean radius `neighbourhood` buffer of	Postcode district rural urban classification
	outlets accessible online	number physically accessible in the neighbourhood (unbounded, may exceed 100%).	postcode district geographic centroid	Postcode district population density: usual residential and usual workday

Table 5.1: Summary of measures and the exposure: outcome relationships investigated.

5.4.5 Statistical analysis

I used Stata version 16.1 (StataCorp LLC., College Station, TX, USA) to complete statistical analyses, with a significance threshold of p<0.05 throughout. For each exposure: outcome relationship investigated (see Table 5.1), I included postcode districts with complete data for all relevant variables. I completed unadjusted and adjusted analyses.

Data on the number (count) of food outlets and unique cuisine types accessible online were not normally distributed and were over-dispersed. Therefore, I used negative binomial regression to investigate associations with postcode district relative deprivation. Negative binomial regression reports incidence rate ratios (IRR) and 95% confidence intervals (CI). In the context of my research, IRR are the expected change in the outcome measure at each level of deprivation compared with the least deprived (decile 1). For analyses where the outcomes were the number of food outlets accessible online and the number of unique cuisine types accessible online, adjusted models included the number of food outlets in a postcode district, postcode district rural urban classification, and population density as covariates. When the number of unique cuisine types accessible online was the outcome, adjusted models additionally included the number of food outlets accessible online as a covariate. When the outcome was the percentage of food outlets registered to accept orders online or the percentage of neighbourhood food outlets accessible online, I used general linear models to investigate associations with postcode district deprivation. For these outcomes, model coefficients are the difference in the percentage at each level of deprivation compared with the least deprived. For these outcomes, adjusted models included postcode district rural urban classification and population density as covariates. I used the `margins` command to estimate the marginal means and 95% CI calculated from the coefficients or IRR from adjusted analyses. To aid interpretation of outcomes, I present these in the Results. I present the coefficients and IRR from unadjusted and adjusted analyses in Appendix D.

5.4.5.1 Sensitivity analyses

In sensitivity analyses, I wanted to test the sensitivity of my findings to the food outlet categories originally selected from OS POI data (`Fast food and takeaway outlets`, `Fast food delivery services`, `Fish and Chip shops`, `Restaurants`). Therefore, I used the same approach as described in Section 5.4.3.2 to construct measures that included the number of food outlets from additional categories listed in OS POI data: `Cafés, snack bars and tea rooms`, `Convenience stores`, `Supermarkets`, `Bakeries`, `Delicatessens`. I included the newly constructed measure in unadjusted and adjusted models.

5.5 Results

In November 2019, 29,232 food outlets across England were registered to accept orders through the online food delivery service (see Appendix D: Table S1).

5.5.1 Food outlet access across England

Table 5.2 provides descriptive statistics summarising online and physical food outlet access. Overall, the median number of food outlets accessible online per postcode district was 63.5 (IQR: 16.0, 156.0). Figure 5.1 shows that access to food outlets through the online food delivery service was widespread. I observed clusters of postcode districts in the North East, North West, West-Midlands regions and the Greater London area that had a high number of food outlets accessible online. Returning to Table 5.2, the median number of unique cuisine types accessible online was 39.0 (IQR: 16.0, 68.0). The median number of food outlets located in the physical food environment of postcode districts was 30.0 (IQR: 14.0, 52.0). The median number of food outlets registered to accept orders online and located in postcode districts expressed as a percentage of food outlets in postcode districts was 30.0% (IQR: 10.0, 40.0). When the number of food outlets accessible online was expressed as a percentage of the number physically accessible in the neighbourhood of a given population, the median was 63.4% (IQR: 35.6, 96.5). The median number of food outlets accessible online (186.0 outlets; IQR: 102.0, 294.0), and the percentage of food outlets in a postcode district registered to accept orders online (50.0%; IQR: 40.0, 60.0) were highest in postcode districts in decile 10 of deprivation. The median number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood of a given population was also highest in these postcode districts (77.4%; IQR: 62.2, 107.7).

Deprivation decile ^a											
	1 (least deprived)	2	3	4	5	6	7	8	9	10 (most deprived) All
Measure	n=214	n=210	n=213	n=211	n=211	n=212	n=212	n=212	n=212	n=211	n=2118
Online food delivery service											
Food outlets registered (count) ^b	3.0	3.0	3.0	5.0	5.0	6.0	9.5	14.5	21.5	24.0	7.0
	(1.0-7.0)	(0.0-10.0)	(1.0-9.0)	(1.0-13.0)	(0.0-14.0)	(1.0-18.0)	(1.0-23.5)	(5.0-26.0)	(10.0-37.5)	(14.0-37.0)	(1.0-21.0)
Accessible food outlets (count)	41.0	31.0	30.0	44.0	41.0	62.0	76.0	92.5	143.0	186.0	63.5
	(18.0-68.0)	(12.0-74.0)	(11.0-89.0)	(11.0-101.0)	(8.0-106.0)	(9.0-121.0)	(4.0-176.5)	(30.5-208.5)	(80.0-247.0)	(102.0-294.0)	(16.0-156.0)
Unique cuisine types accessible	31.0	26.5	27.0	32.0	29.0	36.0	44.0	49.0	59.5	71.0	39.0
(count)	(18.0-43.0)	(13.0-47.0)	(12.0-47.0)	(13.0-55.0)	(10.0-54.0)	(10.0-59.5)	(6.5-72.0)	(24.0-83.5)	(40.5-84.0)	(48.0-95.0)	(16.0-68.0)
Physical food environment											
Food outlets in postcode district	18.0	21.0	24.0	25.0	24.0	28.0	34.5	41.5	50.5	50.0	30.0
(count) ^c	(9.0-31.0)	(10.0-38.0)	(10.0-39.0)	(11.0-45.0)	(12.0-43.0)	(11.0-52.5)	(18.5-56.5)	(21.5-68.0)	(29.5-81.0)	(29.0-76.0)	(14.0-52.0)
Food outlets in neighbourhood	45.0	52.5	55.0	67.0	75.0	90.5	112.0	123.5	191.0	212.0	90.0
(count) ^c	(26.0-93.0)	(24.0-97.0)	(26.0-129.0)	(29.0-142.0)	(29.0-154.0)	(31.0-169.0)	(34.0-223.5)	(49.0-274.5)	(105.5-290.5)	(146.0-343.0)	(36.0-200.0)
Percentage registered (%) ^d	20.0	20.0	10.0	20.0	20.0	20.0	30.0	40.0	40.0	50.0	30.0
	(10.0-30.0)	(0.0-30.0)	(0.0-30.0)	(10.0-40.0)	(0.0-40.0)	(10.0-40.0)	(0.0-50.0)	(20.0-50.0)	(30.0-50.0)	(40.0-60.0)	(10.0-40.0)
Percentage accessible online (%) ^e	70.9	59.4	49.7	59.2	53.3	56.8	52.7	67.3	72.1	77.4	63.4
	(40.4-115.1)	(33.3-100.0)	(27.3-91.3)	(29.2-96.4)	(26.3-97.1)	(28.3-88.4)	(23.1-82.4)	(40.1-91.9)	(52.6-95.0)	(62.2-107.7)	(35.6-96.5)

Table 5.2: Summary of measures for the online food delivery service and the physical food environment across postcode districts in England (n=2118), stratified by deprivation decile.

Notes: ^a Decile 1 was least deprived. Decile 10 was most deprived. Data reported as median (IQR).

^b `Registered` = registered to accept orders online, through the online food delivery service.

^c Food outlet categories included: Fast food and takeaway outlets; Fast food delivery services; Fish and Chip shops; Restaurants. `Neighbourhood` = 1600 m Euclidean radius buffer of postcode district geographic centroid.

^d The number of food outlets registered to accept orders online expressed as a percentage of the number of food outlets in a postcode district.

^e The number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood.



Figure 5.1: Deciles of the number (count) of food outlets accessible online across postcode districts (n=2118) in England in November 2019.

5.5.2 Association between deprivation and the percentage of food outlets registered to accept orders online

In the adjusted model, I observed evidence suggestive of a positive dose-response association between deprivation and the percentage of food outlets located in postcode districts registered to accept orders online (see Appendix D: Table S2). Figure 5.2 reports predicted means with 95% CI, calculated from the coefficients of the adjusted model. Postcode districts in deciles 8-10 of deprivation (i.e. more deprived areas) had significantly higher percentages of food outlets registered to accept orders online than those in decile 1 (the least deprived). In the most deprived postcode districts, 42.9% (95% CI: 40.7, 45.1) of food outlets were predicted to be registered to accept orders online, compared with 22.8% (95% CI: 20.7, 25.0) in the least deprived.





Note: Data points are predicted means with 95% CI, calculated from coefficients estimated using a general linear model adjusted for rural urban classification and population density. IMD = Index of Multiple Deprivation.

5.5.3 Association between deprivation and online food outlet access

In the adjusted model, there was limited evidence of a trend in online food outlet access across deprivation deciles (see Appendix D: Table S3). Nevertheless, as shown in Figure 5.3, the predicted number of food outlets accessible online in the most deprived postcode districts (decile 10) was significantly higher compared with the least deprived postcode districts (106.1 outlets; 95% CI: 91.9, 120.3 and 70.4 outlets; 95% CI: 60.8, 80.1, respectively). Figure S1 (see Appendix D) shows the predicted number of food outlets accessible online across postcode districts in England, calculated from the IRR of the adjusted model. The clusters of postcode districts in the North East, North West, West-Midlands regions and in the Greater London with a high number of food outlets accessible online persisted.





Note: Data points are predicted means with 95% CI calculated from IRR estimated using negative binomial regression adjusted for rural urban classification, population density, and the number of food outlets in the physical food environment. IMD = Index of Multiple Deprivation.

5.5.4 Association between deprivation and unique cuisine type access

In the adjusted model, there was an inverse association between deprivation and the number of unique cuisine types accessible online (see Appendix D: Table S4). Figure 5.4 reports the predicted means with 95% CI, calculated from the IRR of the adjusted model. I observed evidence of a curvilinear relationship. The least deprived postcode districts had access to the greatest number of unique cuisine types (42.1; 95% CI: 39.1, 45.0).





Note: Data points are predicted means with 95% CI, calculated from IRR estimated using negative binomial regression adjusted for rural urban classification, the number of food outlets in the physical food environment, population density and the number of food outlets accessible online. IMD = Index of Multiple Deprivation.

5.5.5 Association between deprivation and the percentage of neighbourhood food outlets accessible online

In the adjusted model, I observed evidence of a curvilinear relationship between deprivation and the number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood (see Appendix D: Table S5). Figure 5.5 shows predicted means with 95% CI, calculated from the coefficients of the adjusted model. Postcode districts in deciles 3-9 of deprivation had a significantly lower percentage than the least deprived postcode districts. In the least deprived postcode districts, the number of food outlets accessible online as a percentage of the number of food outlets physically accessible in the neighbourhood was 86.2% (95% CI: 78.6, 93.7). This was greater than postcode districts in any other decile of deprivation.



Figure 5.5: Number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood across postcode districts (n=2076) in England in 2019. Note: Data points are predicted means with 95% CI, calculated from coefficients estimated using a general linear model adjusted for rural urban classification and population density. IMD = Index of Multiple Deprivation.

5.5.6 Sensitivity analyses

Tables S6-S9 (see Appendix D) show that the strength of associations in sensitivity analyses were either similar or slightly attenuated compared with the main analysis. The percentage of food outlets registered to accept orders online located in postcode districts continued to be positively associated with deprivation when additional food outlet types were included in the denominator. The number of food outlets and unique types of cuisine accessible online continued to be greatest in the most deprived postcode districts when I adjusted for additional food outlet types. Similarly, the curvilinear relationship between deprivation and the number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood persisted.

5.6 Discussion

5.6.1 Summary of findings

To the best of my knowledge, I have described online food outlet access across a whole country for the first time in the international published literature. I found that in 2019, almost 30,000 food outlets in England were registered to accept orders online, which was around a third of the number of outlets that predominantly sell food prepared out-of-home. A median of 64 food outlets and 39 unique cuisine types were accessible online per postcode district. Moreover, the median number of food outlets accessible online expressed as a percentage of the number physically accessible in the neighbourhood was 63%. I observed evidence of socioeconomic patterning. Although the number of food outlets accessible online was highest in the most deprived areas, it was widespread across England. Additionally, the number of food outlets registered to accept orders online, expressed as a percentage of the number in the physical food environment that predominantly sell food prepared out-of-home, increased with deprivation. The number of unique cuisine types accessible online and the number of food outlets accessible online and the number of food outlets accessible online and the number of food outlets accessible online areas. However, I observed evidence of a curvilinear relationship for these measures.

5.6.2 Interpretation of findings

The number of food outlets registered to accept orders online, expressed as a percentage of the number in the physical food environment that predominantly sell food prepared out-of-home, increased with deprivation. In the most deprived areas, the percentage was around two times greater than in the least deprived areas. As far as I am aware, possible reasons for registering

with online food delivery services from the perspective of food outlet owners, and differences in levels of registration according to neighbourhood socioeconomic position, have not been investigated. The physical food environment is vital to the business model of online food delivery services. Food outlets serving food prepared out-of-home cluster together in more deprived areas (31, 303), which perhaps reflects lower retail unit rental costs, or because food business owners believe there will be increased demand due to greater population density (304). In the context of my research, registering to accept orders through an online food delivery service may be one way to `compete` with others. Although business owners must pay initial registration fees and ongoing commission to online food delivery services (169), it seems that even in more deprived areas, this does not outweigh the possible benefits. For example, being registered to accept orders online likely leads to a larger potential customer base and a greater volume of orders. This would allow revenue to be maximised compared with relying on customers visiting customer-facing premises. However, if and how this is the case remains unclear. Regardless, the findings I have presented demonstrate that there remains considerable scope for growth in the number of food outlets registered to accept orders online, including in the most deprived areas.

The absolute number of food outlets accessible online was 50% greater in the most deprived areas of England, compared with the least deprived areas, with evidence of a dose-response association. Online food delivery services plausibly allow food prepared out-of-home to be ordered from food outlets not accessible through other purchasing formats. In turn, this might change perceived access to food outlets in the physical food environment and influence purchasing practices. In contrast to my finding, in one city in each of Australia, the Netherlands and the USA, the number of food outlets accessible through an online food delivery service was not related to area-level socioeconomic position (156). This previous research included 10 locations sampled from the least and most deprived areas of each city. I completed the research in the current chapter on a national scale and included all areas from across the socioeconomic gradient. In doing so, I provide a more comprehensive assessment of online food outlet access.

The number of food outlets accessible online compared with the number of food outlets physically accessible in the neighbourhood of a given population was similarly high in the least and most deprived areas of England. If online food outlet access represents a health risk in addition to that posed by physical food outlet access, my finding suggests that all areas might be affected. However, absolute numbers of food outlets are typically greater in more deprived areas (31, 288-290). Since online food delivery services are a complementary way that food prepared out-of-home can be accessed, online food outlet access compounds access through other purchasing formats. This might mean that populations living in areas with the highest number of food outlets accessible online and the highest number of physically accessible food outlets experience a `double-burden` of disadvantage. Indeed, purchasing food prepared outof-home might be an increasingly natural response to enduring exposure across both the physical and digital food environment (141, 146).

The number of unique cuisine types accessible online was inversely associated with deprivation and highest in the least deprived areas. However, these areas also had the lowest absolute number of food outlets accessible online. As a result, the number of food outlets available *within* each unique cuisine category would be lower than elsewhere. This is particularly important to consider in the context of more deprived areas where the number of food outlets accessible online was highest. When registering to accept orders online, food outlets self-select the cuisine category that reflects the food they sell. To gain an advantage in a saturated market, food outlets may select a cuisine believed to differentiate them from others or one that would mean they appear in customer search results more frequently. My evidence from Chapter 4 suggested that the number of unique types of cuisine was not associated with online food delivery service use (214). However, food purchasing practices are complex and influenced by multiple factors (50), and frequent online food delivery service customers indicated that they were aware of the opportunity to access a variety of food online, which they felt was advantageous (242) (see Chapter 3). As such, it remains plausible that access to different types of cuisine is one of many factors that influences the use of online food delivery services.

5.6.3 Possible implications for public health and future research

As with physical access to food outlets that predominantly sell food prepared out-of-home, online food outlet access was greatest in the most deprived areas of England. For populations living in these areas, greater access to food prepared out-of-home through online food delivery services could be cause for public health concern since they may use this purchasing format more frequently. Indeed, in Chapter 4, I demonstrated that a higher number of food outlets accessible online was positively associated with online food delivery service use (214). In turn, this could contribute to existing diet-related inequalities. Exploring variation in levels of online food delivery service use based on area measures of socioeconomic position would extend the research in this chapter and complement the findings I presented in Chapter 4.

5.6.4 Strengths and limitations

As I understand it, this is the first study in the international published literature to investigate online food outlet access on a national scale. Nonetheless, my study is not without limitations. I

108

generated a novel dataset through an automated data collection approach. If food outlets were registered to accept orders through the online food delivery service but not returned in searches, it is possible that the data were incomplete. However, according to annual reports published by the data source, around 30,000 food outlets were registered to accept orders at the time of data collection (181). This number was similar to the number I identified, increasing confidence in the completeness of the data.

I used postcode districts as the unit of analysis. As such, my analyses may be subject to the modifiable areal unit problem (MAUP), whereby findings are sensitive to the shape of the spatial units or the scale of analysis, which has the potential to introduce bias (305). Although the MAUP is not unique to my study (26), in the context of my research, aggregating measures at the postcode district level would mean that I am unable to identify variation *within* studied areas studied. However, postcode districts were used by the online food delivery service so that food outlets could delineate their delivery areas, which helps to justify my approach. Using postcode districts as the unit of analysis also meant that I was limited to using boundary data from 2012. Although boundaries are subject to change over time, the inferences and conclusions drawn from my study are based on contemporaneous exposure and outcome data collected in 2019.

Finally, I operationalised neighbourhoods of given populations as a 1600 m radius around the geographic centroid of postcode districts. My use of a 1600 m radius may have influenced the magnitude of physical food outlet access in the neighbourhood. Although neighbourhoods are not consistently operationalised in this research area (25), a 1600 m radius has been shown to reflect the spatial extent of an individual's typical shopping practices, and this distance could be reasonably walked by an adult in around 20 minutes (265).

5.7 Conclusions

As far as I know, my study is the first to investigate online food outlet access on a national scale. Around one-third of food outlets that predominantly sell food prepared out-of-home were registered to accept orders online in November 2019. The number of food outlets accessible online was highest in the most deprived areas of England. Online food delivery services do not replace the opportunity to use other purchasing formats meaning that they increase overall access to food prepared out-of-home. This increased access could drive more frequent purchasing of food prepared out-of-home and widen existing diet and health inequalities. Despite having a lower number of food outlets accessible online, the number of unique cuisine types accessible online was highest in the least deprived areas.

Chapter 6:

Changes in online access to food prepared out-of-home in England over time and associations with socioeconomic position

6.1 Details of author contributions, publication status and dissemination

For the research in this chapter, I developed the research questions with guidance from JA and TB. I was responsible for data collection, preparation and analysis, and led the interpretation of findings with support from JA and TB. All authors read and provided critical comments on the initial manuscript and agreed to the version that is currently under peer review at JMIR Public Health and Surveillance. For my thesis, I present an extended version of the manuscript that is currently under peer review.

I presented the findings from this chapter during an oral presentation at the 2022 International Medical Geography Symposium (held in-person, Edinburgh, United Kingdom), and a poster presentation at the 2022 UK Public Health Science National Conference (held in-person, Glasgow, United Kingdom). For the latter, the conference abstract is published as: Keeble, M., Adams, J., Burgoine, T. (2022), Changes in online food access during the COVID-19 pandemic and associations with deprivation: a longitudinal analysis. The Lancet. https://doi.org/10.1016/S0140-6736(22)02264-4. I will also share the findings at the 2023 American Association of Geographers Annual Meeting during an oral presentation (held inperson, Denver, United States of America).

6.2 Abstract

Background

Food prepared out-of-home is typically energy-dense and nutrient-poor. Online food delivery services have become a popular way to purchase this food. The number of food outlets that can be accessed online (i.e. ordered from through these services) can influence how frequently they are used. Online food access reportedly increased in England between 2020 and 2022, in the context of the Coronavirus Disease 2019 (COVID-19) pandemic. However, the extent to which this changed is not currently clear. I aimed to investigate changes in online access to food prepared out-of-home over time, compared with a time before the onset of the COVID-19 pandemic. I also aimed to understand the extent to which any changes were associated with deprivation.

Methods

In November 2019, and monthly between June 2020 and March 2022, I used automated data collection to construct a dataset containing information about all food outlets in England registered to accept orders through the leading online food delivery service. Across postcode districts (n=2118), I identified the number and percentage of food outlets registered to accept orders online and the number of food outlets accessible online. I used generalised estimating equations (adjusted for population density, the number of food outlets in the physical food environment and rural urban classification) to investigate changes in each outcome compared to levels from before the COVID-19 pandemic (November 2019). I stratified analyses by area-level socioeconomic position (deprivation quintile (Q)).

Results

Across England, the number of food outlets registered to accept orders online increased from 29,232 in November 2019 to 49,752 in March 2022. The median percentage of food outlets registered to accept orders online increased from 14.3% (interquartile range (IQR): 3.8, 26.0) in November 2019 to 24.0% (IQR: 6.2, 43.5) in March 2022. The median number of food outlets accessible online decreased from 63.5 outlets (IQR: 16.0, 156.0) in November 2019 to 57.0 outlets (IQR: 11.0, 163.0) in March 2022. However, I observed variation across deprivation quintiles. In March 2022, the median number of outlets accessible online was 175.0 outlets (IQR: 104.0, 292.0) in the most deprived areas (Q5) compared to 27.0 outlets (IQR: 8.5, 60.5) in the least deprived (Q1). In adjusted analyses, I estimated that the number of outlets accessible online in the most deprived areas was 10% higher in March 2022, compared to November 2019 (incidence rate ratios (IRR): 1.10; 95% confidence intervals (CI): 1.07, 1.13). In contrast, in the least deprived areas, I estimated a 19% decrease (IRR: 0.81; 95% CI: 0.79, 0.83).

Conclusions

The number of food outlets accessible online increased only in the most deprived areas of England. Future research might attempt to understand the extent to which changes in online food outlet access that I observed were associated with changes in online food delivery service use and, in turn, possible implications for diet and health.

6.3 Introduction

Purchasing and consuming food prepared out-of-home has become increasingly popular in many countries (29). This food accounted for over 50% of total food expenditure in the United States of America (USA) in 2018 (285), and over one quarter of total food expenditure in the United Kingdom (UK) between 2015 and 2017 (306). Food prepared out-of-home is often high in calories (41, 175), and the majority of items served by large chain restaurants exceed recommended levels for salt, total fat, saturated fat, or sugars (45). More frequent consumption of this food has been positively associated with body weight (110). Online food delivery services such as Uber Eats are now an established way of purchasing food prepared out-of-home (307), and it is plausible that the use of these services has implications for diet and diet-related health (147).

When using online food delivery services, customers receive aggregated information about all food outlets they can order from based on an entered location (i.e. food outlets that are accessible). Customers then select a food outlet and place and pay for their order on a single platform. Orders are forwarded to food outlets where meals are prepared before being delivered by couriers working for them or the online food delivery service (147). As in the physical food environment (77), there is evidence that the number of food outlets accessible online can influence online food delivery service use at the individual level (214) (see Chapter 4). Moreover, before the Coronavirus Disease 2019 (COVID-19) pandemic, the potential for online food delivery service use was not equally distributed across England (213) (see Chapter 5). The estimated mean number of food outlets accessible online in 2019 was over 100 in the most deprived areas, compared with 70 in the least deprived areas (213), which could contribute to known inequalities in overall dietary patterns and diet-related health (308).

6.3.1 Access to food prepared out-of-home during the first two years of the COVID-19 pandemic

In March 2020, the UK government published guidance stating that individuals must remain at home if they had symptoms of COVID-19, or had tested positive for COVID-19 or been in contact with others who had. This was followed by a national `lockdown` that restricted non-essential travel and limited physical and social contact (309). Similar action was taken in other countries. Specifically with respect to out-of-home food retail, the Supplemental Nutrition Assistance Program was extended to include digital food retail in the USA (310), whilst rules around the delivery of food prepared out-of-home and alcohol were relaxed in Australia (311). As part of the national lockdown in the UK, the UK government forced the closure of outlets

serving food for on-premises consumption (312, 313). This meant that bars, cafés, pubs and restaurants could no longer operate with their 'primary' function (the primary function of a pub is a drinking establishment, for example). However, at the same time, the UK government introduced emergency regulations that allowed bars, cafés, pubs and restaurants to temporarily operate in a manner that was not their primary function by offering an expanded hot food takeaway service. Before the introduction of these emergency regulations, a pub would have been able to offer only a limited hot food takeaway service in addition to their primary function. Under the emergency regulations, however, they could primarily offer a hot food takeaway service (314). Emergency regulations ended in March 2022. Therefore, after March 2022, bars, cafés, pubs and restaurants should have reverted to their primary function and stopped offering an expanded hot food takeaway service. However, planning professionals from one region in the North East of England reported that due to limited resources during the early stages of the COVID-19 pandemic, the number of outlets that operated with an expanded hot food takeaway service under the emergency regulations was unclear (315). Figure 6.1 provides a timeline of the aforementioned events and information about periods of time when on-premises consumption of food prepared out-of-home would and would not have been possible.

Pre-March 2020		Food outlets in the physical food environment operated with a `primary use` based on their core business operations. The primary use of hot food takeaway outlets would be to serve food prepared out-of-home for off-premises consumption, after previously receiving Local Authority ^a planning permission to operate. Other establishments (bars, cafés, pubs and restaurants) could offer a hot food takeaway service in addition to their primary use, but would need Local Authority planning permission to do so. All food outlets with a customer-facing premises in the physical food environment could feasibly register to accept orders online regardless of their primary use, without Local Authority planning permises in the physical food environment could						
2020	March	Bars, cafés, pubs and restaurants forced to close for on-premises food consumption as part of first national lockdown ^c . Emergency regulations to allow bars, cafés, pubs and restaurants to offer a hot food takeaway service in addition to their primary use introduced.						
	July	Bars, cafés, pubs and restaurants allowed to reopen for on-premises food consumption, but only with table service and with restricted capacity.						
	August	`Eat Out to Help Out` scheme that offered a 50% discount on meals, up to £10 per person every Monday, Tuesday and Wednesday between 3 and 31 August introduced (for on-premises food consumption only).						
	November	Bars, cafés, pubs and restaurants forced to close for on-premises food consumption as part of second national lockdown ^c .						
	December	Bars, cafés, pubs and restaurants allowed to reopen for on-premises food consumption, but only with table service and with restricted capacity.						
	January	Bars, cafés, pubs and restaurants forced to close for on-premises food consumption as part of third national lockdown ^c .						
2021	April	Bars, cafés, pubs and restaurants allowed to reopen for on-premises food consumption, but only with table service and with restricted capacity.						
	July	End of restrictions necessitating table service only and capacity limits for on-premises food consumption inside bars, cafés, pubs and restaurants.						
2022	March	End of emergency regulations introduced in March 2020.						
Post-March 2022		 Bars, cafés, pubs and restaurants <i>should</i> revert to their primary use, and <i>should</i> stop offering a hot food takeaway service if adopted as part of emergency regulations. To continue offering an additional hot food takeaway service, Local Authority planning permission required. All food outlets with customer-facing premises in the physical food environment can feasibly register to accept orders online regardless of their primary use, without Local Authority planning permission. 						

Figure 6.1: Details of selected emergency regulations introduced, action taken, and broader events in England in relation to the COVID-19 pandemic and food outlets selling food prepared out-of-home.

Notes: ^a Local Authorities are administrative bodies that operate at a sub-national and sub-regional level. ^b Registration to accept orders online through online food delivery services remained viable regardless of any subsequent action or change.

^c As part of `lockdown` orders in England, individuals were instructed to `remain at home`.

6.3.2 Possible implications for access to food prepared out-of-home through online food delivery services

It is plausible that food outlets operating with an additional hot food takeaway service under the aforementioned emergency regulations registered to accept orders through online food delivery services. From a business perspective, on-premises food consumption was restricted at times throughout 2020 and 2021 (see Figure 6.1), and therefore, accepting orders through these services could have been an important source of income. Reflective of this, each of Just Eat, Deliveroo and Uber Eats reported an increase in the number of food outlets registered with their services or increased revenue in the UK between 2020 and 2021 (316-318). Focusing on Just Eat as the market leader in the UK, the number of food outlets registered to accept orders across the UK and Ireland in 2021 reportedly increased by 21%, compared with 2020, exceeding 60,000 in total (316). The extent to which these increases were driven by the uptake of emergency regulations is unclear. It is equally possible that the reported increase reflects existing hot food takeaway outlets that were not yet registered to accept orders online choosing to do so. Nevertheless, the reported increase suggests that online access to food prepared out-of-home increased over this period of time. Food sold through online food delivery services is typically prepared inside of existing customer-facing food outlets located in the physical food environment (147). As I discussed in Chapter 1 (see Section 1.2.5.3), the number of food outlets located in the physical food environment is typically higher in more deprived areas of England (31, 319). As such, there was greater potential for existing food outlets to operate under the emergency regulations introduced in March 2022 and to register to accept orders through online food delivery services in more deprived areas compared with less deprived areas. Furthermore, the research I presented in Chapter 5 demonstrated that even before the COVID-19 pandemic, the number of food outlets accessible online was highest in the most deprived areas of England (213). If food outlets that could have potentially registered to accept orders online did so, the number of accessible outlets would plausibly increase further. Given that the number of food outlets accessible online was associated with online food delivery service use amongst adults living across Great Britain (214) (see Chapter 4), greater access might lead to more frequent use amongst existing customers and prompt others to become customers. In turn, this could widen diet-related inequalities.

6.3.3 Study aims

My primary aim was to investigate changes in online access to food prepared out-of-home in England between 2020 and 2022, within the context of the COVID-19 pandemic. Second to this, I aimed to identify the extent to which any changes were associated with area-level socioeconomic position.

6.4 Methods

In this chapter, I build on the research from Chapter 5, where I investigated cross-sectional associations between deprivation and online food outlet access in England.

6.4.1 Study setting, period, and analytic scale

The study setting was England. The study period was November 2019 to March 2022, which coincided with the end of the emergency regulations introduced in March 2020 (see Section 6.3.1: Figure 6.1). Data were not collected between December 2019 and May 2020, meaning that the first available data after November 2019 were from June 2020.

I completed analyses at the postcode district level. This analytic scale reflected the delivery areas of food outlets registered to accept orders online in November 2019 (213, 214) (see Chapters 4 and Chapter 5). Postcode districts are the first half of full postcodes. For the *postcode* `CB2 0QQ`, the *postcode district* is `CB2`. Postcode districts have an average size of 33 mi² (255), and a median population of 23,610 (interquartile range (IQR): 13,320, 34,560) (298).

I used boundary data from 2012, sourced from the UK data service (296), to map postcode districts in a geographic information system (GIS) (ArcGIS version 10.7.1; ESRI Inc., Redlands, CA). Like Chapter 5, I included 2118 postcode districts, which were those entirely within the border of England, as well as those in Scotland or Wales with an intersecting boundary (food outlets could be located in these countries yet deliver to locations in England).

6.4.2 Exposure measure

The exposure was time. I characterised the exposure as `month` because I collected data for outcome measures on a monthly basis, as I will discuss in the next sections.

6.4.3 Outcome measures: data source and collection

I collected data from an online food delivery service widely recognised as the UK market leader (Just Eat) to construct each outcome measure. In 2019, around 30,000 food outlets in the UK

were registered to accept orders through Just Eat, and there were almost 170 million orders placed by customers (252, 292). In pilot work (see Appendix C: Figure S1), only a minority of food outlets registered to accept orders through Deliveroo (the next largest online food delivery service in the UK) were not also registered to accept orders through Just Eat in 2020 (213, 214) (see Chapters 4 and 5). Therefore, I used data from Just Eat as a proxy for online food delivery services. I refer to my data source as `the online food delivery service` hereafter.

I used the same automated data collection approach as outlined in Chapters 4 and 5. Briefly, monthly between June 2020 and March 2022, I used a web-browser extension to identify and collect information about all food outlets registered to accept orders through the online food delivery service across England, Scotland and Wales (299). I completed data collection at the same time each month, with the process taking around 72 hours.

6.4.3.1 Number of food outlets registered to accept orders online

I used Doogal (a free web-based resource) to geocode the postcode of each food outlet identified during data collection (263). When geocoding through Doogal was not successful, I used the GeoConvert tool (maintained by the UK Data Service) (264). I excluded food outlets that could not be geocoded (monthly range: 0.1 to 1.4%). I then mapped the locations of food outlets in a GIS based on supplied coordinates and counted the number located inside each postcode district boundary.

6.4.3.2 Number of food outlets accessible online

In November 2019, June 2020, and July 2020, food outlets registered to accept orders online published the postcode districts that they would deliver to as their `delivery areas`. I collected this information during the data collection described in Section 6.4.3. After July 2020, the online food delivery service no longer published this information. Therefore, between August 2020 and March 2022, I manually identified and counted the number of food outlets that were accessible for each postcode district. Although I used two approaches, the data used in outcome measure construction were the same.

6.4.3.3 Percentage of food outlets registered to accept orders online

I compared the number of food outlets registered to accept orders online to the number located in the physical food environment of the same postcode district. In doing so, I calculated the percentage of food outlets registered to accept orders online.

For the denominator (the number of outlets located in the physical food environment within postcode districts), I used Ordnance Survey Points of Interest (OS POI) data. This is commercial data containing information about retailers across multiple sectors, collated from over 170 suppliers (262). I extracted information for the following food outlet classifications: `Fast food and takeaway outlets` (food outlets selling food prepared out-of-home for off-premises consumption), 'Fast food delivery services' (food outlets selling food prepared out-of-home for delivery, not explicitly through online food delivery services), 'Fish and Chip shops' (food outlets selling a traditional British cuisine, prepared out-of-home, typically for off-premises consumption), 'Restaurants' (food outlets selling food prepared out-of-home for on-premises consumption), 'Pubs, Bars, Inns' (establishments primarily serving alcohol, that can also sell prepared out-of-home for on-premises consumption), and `Cafe, Snack Bars & Tea Rooms` (food outlets selling food prepared out-of-home with no distinguishable consumption location). I selected these classifications based on *a priori* knowledge that they included food outlets typically registered to accept orders online, and to reflect the types of food outlets that emergency regulations introduced in March 2020 allowed to operate with an additional hot food takeaway service (see Section 6.3.1: Figure 6.1). I used coordinates supplied in OS POI data to map the locations of food outlets in the GIS. These coordinates are reported to be accurate up to one metre (262).

I matched monthly data from the online food delivery service with OS POI data that is collected quarterly (see Appendix E: Table S1). I did not match individual food outlets listed in each dataset, meaning that this outcome is the number of food outlets registered to accept orders online (based on data from the online food delivery service) as a percentage of the number of food outlets in the physical food environment (based on OS POI data). Although I report a percentage, I acknowledge that in the strictest sense it has not been calculated as such. As the number of food outlets in the physical food environment, I bounded this measure to between 0% and 100%. When the percentage exceeds 100% it represents that retailers not classified as a food outlet in OS POI data were registered to accept orders online. I excluded postcode districts when this occurred (n=3; 0.14%).

6.4.4 Covariates

The following covariates were `time invariant`, meaning that they did not change over the study period.

6.4.4.1 Index of multiple deprivation

I used data from the 2019 Index of Multiple Deprivation (IMD) to measure relative deprivation. This is a compound measure of deprivation that includes metrics across seven domains: income deprivation, employment deprivation, crime levels, health deprivation and disability, education, skills and training deprivation, barriers to housing and associated services and living environment deprivation (297). Deprivation values are available for lower super output areas (LSOAs) in England, which are administrative boundaries with a mean residential population of 1500 people (298). As LSOAs are typically geographically smaller than postcode districts, I aggregated LSOAs within and intersecting the boundary of each postcode district and calculated the mean IMD score. For analyses, I split postcode districts into quintiles (Q) of deprivation, where Q5 contained the most deprived.

6.4.4.2 Rural Urban classification and population density

I used data from the 2011 rural urban classification to categorise postcode districts as: `rural`, when LSOAs within or intersecting their boundary were most frequently rural (populations less than 10,000 people within combined settlements, the majority of which live in rural-related areas) or `urban` when intersecting LSOAs were most frequently urban (populations greater than 10,000 people within combined settlements, the majority of which live in urban-related areas) (300). Data for rural urban classification were available for 2097 postcode districts (99.0%).

I used data from the 2011 UK census (301) to identify the usual residential population of each postcode district. This is the number of individuals that usually reside in a postcode district, including students and schoolchildren not living away from home during term-time. Data for population density were available for 2088 (98.6%) postcode districts.

6.4.5 Statistical analysis

I used the longitudinal-analysis ('xt') suite of tools in Stata version 16.1 (StataCorp LLC., College Station, TX, USA) to complete statistical analysis (320). I report findings from the start (November 2019) and end (March 2022) of the study period in the Results. Findings for all time points (November 2019, and then monthly between June 2020 and March 2022) are available in Appendix E.

6.4.5.1 Descriptive statistics

As data for each outcome were not normally distributed, I calculated the median and interquartile range (IQR). I also calculated the percent change from baseline (i.e. November 2019) at each time point.

6.4.5.2 Inferential statistics

I used generalised estimating equations (GEE) to investigate changes in each outcome over time. Generalised estimating equations fit generalised linear regression models that estimate a population (i.e. postcode district) average (320). Models were first unadjusted and then adjusted for covariates. I report findings from adjusted models in the Results and unadjusted models in Appendix E. For each set of respective analyses, I included postcode districts with complete data on all relevant measures. Moreover, I specified an `exchangeable` correlation structure to assess monthly changes compared with baseline (November 2019). Due to non-data collection between December 2019 and May 2020, I was unable to use other correlation structures.

Data for count-based outcomes (the number of food outlets registered to accept orders online and the number of food outlets accessible online) were not normally distributed and were overdispersed. I used negative binomial GEE to account for this. Negative binomial GEE report incidence rate ratios (IRR) and 95% confidence intervals (CI). In the context of my research, IRR are the expected change in the outcome at each time point compared with baseline (November 2019). For the percentage of food outlets registered to accept orders online, I rescaled the data to between 0 and 1 and specified a binomial distribution (321). Model coefficients for this outcome are the change in percentage at each time point compared with baseline. For each outcome, I used the `margins` command to estimate the marginal mean count from IRR and the mean percentage from coefficients and report these in the Results. I report the respective IRR and coefficients in Appendix E.

6.4.5.3 Associations with deprivation

For each outcome, I included an interaction term (time x deprivation) in the adjusted GEE to investigate how changes over time varied by level of deprivation and completed a post-hoc test for significance (with statistical significance set at p < 0.01 to account for multiple testing). When interaction terms were significant, I completed analyses stratified by deprivation quintile.

In November 2019, there were inequalities in access to food outlets selling food prepared outof-home in the digital food environment (213) (see Chapter 5). For each outcome, I calculated a slope index of inequality measure at baseline (November 2019) and at the end of the study period (March 2022) to investigate how inequalities changed over time. This measure of inequality is the difference in the respective outcome between the least and most deprived areas, estimated using linear regression (322, 323).

6.5 Results

6.5.1 Number of food outlets registered to accept orders online

Figure 6.2 reports changes in the number of food outlets registered to accept orders online in England. I observed an increase from 29,232 in November 2019 to 49,752 in March 2022, equating to 70.2% growth.



Figure 6.2: Count of food outlets registered to accept orders online in England between baseline (November 2019) and the end of the study period (March 2022). Notes: shaded bars represent time points reported in the Results. Data for all time points are reported in Appendix E. No data were available from December 2019 to May 2020.

Table 6.1 and Tables S2-S9 (see Appendix E) summarise online access to food prepared out-ofhome in England. The median number of food outlets registered to accept orders online per postcode district was 7.0 (IQR: 1.0, 21.0) in November 2019 and 13.0 (IQR: 3.0, 34.0) in March 2022. The median percent change from baseline (November 2019) per postcode district was 65.4 (IQR: 33.3, 100.0) in March 2022 (see Table 6.1).

The increase in the number of food outlets registered to accept orders online across England as a whole was significant at each time point (see Appendix E: Table S11), and there was a significant effect modification by deprivation quintile on time (p<0.001).

Figure 6.3 shows the estimated mean number of food outlets registered to accept orders online, calculated from the IRR of the adjusted negative binomial GEE. At each level of deprivation, I observed that the estimated mean number of food outlets registered to accept orders online had initially increased from baseline levels (November 2019), decreased after June 2020, and then consistently increased, except between November 2020 and December 2020. The estimated mean number was highest in the most deprived areas (Q5) and lowest in the least deprived (Q1) throughout the study period. Absolute growth over time was highest in the most deprived areas, where the estimated mean number was 39.1 outlets at the end of the study period compared with 24.6 outlets at baseline. For the least deprived areas, the number was 6.6 and 3.5 outlets, respectively. The slope index of inequality between the least and most deprived areas was 5.0 outlets (95% CI: 4.5, 4.5) at baseline, and 7.8 outlets (95% CI: 7.0, 8.6) in March 2022.

	Deprivation quintile								
	1 (least deprived)	2	3	4	5 (most deprived)	England			
Number of food outlets registered to accept orders online ^{a b}									
Count									
November 2019	3.0 (1.0, 8.0)	4.0 (1.0, 12.0)	6.0 (1.0, 18.0)	13.0 (3.0, 25.0)	24.0 (12.0, 39.0)	7.0 (1.0, 21.0)			
March 2022	5.0 (1.0, 15.0)	8.0 (1.0, 21.0)	10.0 (2.0, 29.0)	21.0 (4.0, 41.0)	35.0 (20.0, 59.0)	13.0 (3.0, 34.0)			
<i>Change from baseline</i> (%) ^c									
March 2022	80.0 (22.2, 120.0)	69.0 (33.3, 120.0)	66.7 (30.0, 106.3)	62.8 (33.3, 100.0)	57.9 (34.0, 87.5)	65.4 (33.3, 100.0)			
Number of food outlets accessible online ^d									
Count									
November 2019	37.0 (14.0, 70.5)	38.0 (10.0, 96.0)	62.0 (8.5, 134.5)	86.0 (12.0, 190.0)	164.0 (87.0, 273.0)	63.5 (16.0, 156.0)			
March 2022	27.0 (8.5, 60.5)	27.0 (6.0, 103.0)	50.0 (6.5, 133.5)	95.0 (13.0, 217.0)	175.0 (104.0, 292.0)	57.0 (11.0, 163.0)			
<i>Change from baseline</i> (%) ^c									
March 2022	-12.7 (-48.6, 20.0)	-7.5 (-48.8, 28.6)	-1.1 (-35.0, 33.2)	13.8 (-18.2, 52.5)	13.1 (-7.9, 33.8)	0.0 (-32.0, 33.3)			
Percentage of food outlets registered to accept orders online ^e									
Percent (%)									
November 2019	7.9 (2.2, 14.9)	8.7 (1.5, 19.0)	12.5 (1.9, 23.5)	20.4 (6.5, 30.8)	27.8 (19.7, 37.4)	14.3 (3.8, 26.0)			
March 2022	13.2 (4.7, 25.9)	14.9 (4.1, 33.3)	20.5 (5.3, 36.6)	30.8 (11.3, 46.6)	41.9 (30.7, 52.4)	24.0 (7.7, 41.0)			
<i>Change from baseline</i> (%) ^c									
March 2022	70.7 (16.7, 116.0)	62.0 (25.0, 106.2)	55.1 (24.0, 98.9)	53.0 (24.8, 88.9)	44.8 (22.4, 75.8)	55.3 (23.0, 96.1)			

Table 6.1: Descriptive summary of online access to food prepared out-of-home across postcode districts in England (n=2118), at baseline (November 2019) and the end of the study period (March 2022), stratified by deprivation quintile.

Notes: ^a Data are reported as Median (IQR). Postcode districts are small geographical units used for mail and delivery routing in England.

^b Food outlets in the physical food environment registered to accept orders through the UK market leading online food delivery service.

^c Baseline = November 2019.

^d Food outlets registered to accept orders online through the UK market leading online food delivery service that would deliver to a given postcode district.

• Calculated as the number of food outlets registered to accept orders online through the UK market leading online food delivery service compared with the number in the physical food environment, expressed as a percentage.



Figure 6.3: Estimated mean number (count) of food outlets registered to accept orders online in England between November 2019 and March 2022, stratified by deprivation quintile. Notes: 2067 postcode districts included. Quintile (Q) 5 are the most deprived postcode districts. Dashed lines are the baseline (November 2019) number. No data were available from December 2019 to May 2020. Estimated mean number calculated from Incidence Rate Ratios (IRR) of a negative binomial generalised estimated equation adjusted for population density, rural urban classification and the number of food outlets in the physical food environment.

6.5.2 Number of food outlets accessible online

The median number of food outlets accessible online per postcode district was 63.5 (IQR: 16.0, 156.0) in November 2019 and 57.0 (IQR: 11.0, 163.0) in March 2022. The median percent change from baseline per postcode district was 0.0 (IQR: -32.0, 33.3) in March 2022 (see Table 6.1).

The decrease in the number of food outlets accessible online across England as a whole was significant at each time point (see Appendix E: Table S13), and there was a significant effect modification by deprivation quintile on time (p<0.001).

Figure 6.4 reports estimated means calculated from the IRR of the adjusted negative binomial GEE. At each level of deprivation, I observed that the estimated number of food outlets accessible online had decreased from baseline in June 2020. Although this was followed by an upward trajectory, the estimated number remained lower than baseline in less deprived areas (Q1-Q3), but surpassed baseline in areas in the two upper quintiles of deprivation (Q4 and Q5).

This increase was significant for the most deprived areas only (see Appendix E: Table S13). For the most deprived areas, the estimated number increased from 181.9 outlets in November 2019 to 200.0 outlets in March 2022, which contributed to an increasing gap in online food outlet access between the least and most deprived areas. The slope index of inequality between the least and most deprived areas (95% CI: 28.1, 35.9) at baseline, and 37.3 outlets (95% CI: 31.8, 42.9) in March 2022.



Figure 6.4: Estimated mean number (count) of food outlets accessible online in England between November 2019 and March 2022, stratified by deprivation quintile.

Notes: 2067 postcode districts included. Quintile (Q) 5 are the most deprived postcode districts. Dashed lines are the baseline (November 2019) numbers. No data were available from December 2019 to May 2020. Estimated mean number calculated from the Incidence Rate Ratios (IRR) of a negative binomial generalised estimated equation adjusted for population density, rural urban classification and the number of food outlets in the physical food environment.
6.5.3 Percentage of food outlets registered to accept orders online

The median number of food outlets registered to accept orders online as a percentage of the number of food outlets in the physical food environment per postcode district was 14.3 (IQR: 3.8, 26.0) in November 2019 and 24.0 (IQR: 7.7, 41.0) in March 2022. The median percent change from baseline per postcode district was 55.3 (IQR: 23.0, 96.1) in March 2022 (see Table 6.1).

The increase in the percentage of food outlets registered to accept orders online across England as a whole was significant at each time point (see Appendix E: Table S15), and there was a significant effect modification by deprivation quintile on time (p<0.001).

Figure 6.5 reports estimated means calculated from the coefficients of the adjusted GEE. At each level of deprivation, I observed an initial increase from baseline in June 2020 that was followed by a decline, a second increase that equalled or surpassed previous levels, and then another decline before a somewhat more stable increase. Although this trend was evident across all levels of deprivation and the estimated percentage was significantly increased at each level of deprivation by the end of the study period, the magnitude varied. Nevertheless, the estimated mean percentage was highest in the most deprived areas (40.0% in March 2022 compared with 27.4% in November 2019), with these areas also having a higher growth over time in absolute terms, compared with less deprived areas. The slope index of inequality between the least and most deprived areas was 4.5% (95% CI: 4.1, 4.9) at baseline, and 5.9% (95% CI: 5.4, 6.5) in March 2022.



Figure 6.5: Estimated mean percentage (%) of food outlets registered to accept orders online in England between November 2019 and March 2022, stratified by derivation quintile. Note: 2065 postcode districts included. Quintile (Q) 5 are the most deprived postcode districts. Dashed lines are the baseline (November 2019) percentages. No data were available from December 2019 to May 2020. Estimated mean percentage calculated from the coefficients of a generalised estimating equation adjusted for population density and rural urban classification.

6.6 Discussion

6.6.1 Summary of findings

As far as I am aware, my study was the first to examine changes in online access to food prepared out-of-home over time. For England as a whole, I identified an increase in the number of food outlets registered to accept orders online, with this reaching around 50,000 by March 2022. There was a parallel increase in the percentage of food outlets in the physical food environment registered to accept orders online. In contrast, the number of food outlets accessible online (i.e. those that could be ordered from by a given population in a postcode district), was on average lower in March 2022 than before the COVID-19 pandemic in November 2019. However, the magnitude of change for all measures of online access to food prepared out-of-home varied by level of deprivation. The most deprived postcode districts consistently had the highest number and percentage of food outlets registered to accept orders online. Importantly, contrary to the overall trend for England, the number of food outlets accessible online was higher in March 2022 than at baseline in the most deprived areas. I did not observe this for less deprived areas, where the number decreased in some instances. As a result, the inequalities in the opportunity to use online food delivery services to purchase food prepared out-of-home that I reported in Chapter 5 widened over time.

6.6.2 Interpretation of findings

My finding that an increased number of food outlets had registered to accept orders online over the first two years of the COVID-19 pandemic is aligned with reports from major online food delivery services operating in England, as well as from business and news media (317, 318). Moreover, the decrease I observed in June 2020 and July 2020, accords with contemporaneous reports of food outlet closures and decreased order volume through online food delivery services (324). After July 2020, I observed a more consistent increase in the number of food outlets registered to accept orders online. As opportunities for on-premises food consumption were limited by national stay-at-home orders imposed in March 2020, followed by periods of time with restrictions that limited the number of customers consuming purchase food inside of outlets, food business owners perhaps made a strategic business decision to register to accept orders online as a way to generate revenue.

For England as a whole, I identified a decline in the number of food outlets *accessible* online between November 2019 and March 2022. This is in contrast to the increased number of food outlets *registered* to accept orders. The decline I observed was particularly pronounced in the early stages of the COVID-19 pandemic, which might have reflected a period of transition amongst food outlets that had only recently registered to accept orders online. Those that were previously unaccustomed to food delivery may have been operating with a limited delivery radius to ensure they could fulfil customer orders. Moreover, there was broader workforce instability during a time when self-isolation rules were in place, unclear restrictions on maximum travel distances legally allowed and concerns for online food delivery service courier safety (325). In combination, these factors might have contributed to the implementation of limited delivery distances and a reduced number of accessible food outlets.

The initial decline in the number of food outlets accessible online was followed by an increase at all levels of deprivation. However, as of March 2022, it was only amongst more deprived areas, and particularly the most deprived, that the number eventually surpassed levels from November 2019. As I reported in Chapter 5, the number of food outlets accessible online was highest in the most deprived areas of England in November 2019, before the COVID-19 pandemic (213). In the current study, I found evidence of an increasing divergence between the least and most

deprived areas, suggesting that inequalities in online access to food prepared out-of-home widened during the first two-years of the COVID-19 pandemic. Food sold through online food delivery services is commonly delivered by couriers on bicycles (169). As such, the distance that can be travelled whilst maintaining food quality has a natural limit. Due to the spatial distribution of food outlets in England, which is socioeconomically patterned (21, 31, 319), it is likely that the distance between food outlets and customers is shorter (and thus within any natural limit) in more deprived areas compared with less deprived areas. Regardless, given that the number of food outlets accessible online is positively associated with online food delivery service use, as I reported in Chapter 4 (214), it is possible that food purchasing practices were negatively influenced during the COVID-19 pandemic, especially amongst populations living in more deprived areas. Any changes to purchasing practices, could have implications for overall dietary patterns and diet-related health in the longer-term, however, these are unlikely to be immediately observable (326). Further research is required to help better understand how changes in online food outlet access influenced online food delivery service use, and the possible longer-term implications of this on health.

The number of food outlets registered to accept orders online as a percentage of the number of food outlets in the physical food environment had increased by March 2022. However, this increase was only apparent after a period of early instability. Food business owners with a customer-facing premises in the physical food environment reported that being registered to accept orders online was a way for customers to access their food when on-premises food consumption and travel were restricted (325). However, when restrictions ended, simultaneously managing orders placed in-person and online was difficult (325). As a result, it is possible that food business owners who only registered to accept orders online out of necessity when on-premises food consumption was restricted later de-registered when restrictions ended. This scenario would partly explain the successive increases and decreases in the percentage of food outlets registered to accept orders online, which coincided with the start and end of restrictions (see Figure 6.5).

The relative change over time for the percentage of food outlets registered to accept orders online was consistent across all levels of deprivation. However, the absolute change was highest in the most deprived areas, which led to a widening of inequalities in access to food prepared out-of-home. Food sold through online food delivery services is typically prepared inside of existing customer-facing food outlets located in the physical food environment (147). As such, my finding reflects existing urban form in the most deprived areas of England, which has a high density and concentration of food outlets (327, 328), and a concomitant increase in the number of food outlets registered to accept orders online. Although professionals in practice recognise uncertainty due to a lack of monitoring (315), my findings provide evidence to suggest that the introduction of emergency regulations at least partly contributed to a widening of inequalities in online food access.

As of March 2022, the number of food outlets registered to accept orders online as a percentage of the number of food outlets in the physical food environment per postcode district was 24%. As such, there remains considerable scope for food outlets in the physical food environment to register to accept orders online. The scope for a further increase emphasises the coexisting nature of both the physical and digital food environment, leading to overlapping opportunities to purchase food prepared out-of-home (329). On balance, however, there are natural limits to growth since not all business owners necessarily need or want to register to accept orders online. The relative stability of each outcome from June 2021 onwards suggests a plateau might have already occurred. An apparent plateau might help to explain the decision from online food delivery services to withdraw from countries with poor revenue or growth prospects (211).

6.6.3 Possible implications for public health and future research

Over half of the Local Authorities in England have used urban planning to attempt to create healthier physical food environments by preventing new hot food takeaway outlets from opening (330). Although not yet fully clear, like the physical food environment, interventions to restrict access to food prepared out-of-home through online food delivery services might be necessary in the future (145, 212). However, using urban planning to directly target restrictions at online food delivery services might not be possible due to a lack of regulatory mechanisms (144). Accordingly, new public health interventions that account for the business model of online food delivery services may need to be developed.

The emergency regulations introduced in March 2020 ended in March 2022 (314). As a result, bars, cafés, pubs and restaurants *should* have reverted to their primary use and stopped offering an additional hot food takeaway service if they did not previously have permission. However, this does not necessarily mean that they would have had to stop accepting orders online. It is plausible that the levels of online food access I observed in March 2022 will be sustained in the future. In turn, this new baseline of online food access could influence the purchase and consumption of food prepared out-of-home. It would be interesting for future work to investigate the extent to which food outlets that operated within emergency regulations up until

March 2022 remained registered to accept orders online, even if they should have legally stopped offering an additional hot food takeaway service.

6.6.4 Strengths and limitations

I collected data over a period of two years, for a whole country. My data collection allowed me to monitor trends in online access to food prepared out-of-home on a monthly basis. An important limitation of my research, however, is that I do not have substantial pre-pandemic data that would allow any existing trends to be accounted for in analyses. Nevertheless, despite a period of missing data from before the onset of the COVID-19 pandemic in March 2020, and then between March 2020 and May 2020, I present new baseline levels of online access to food prepared out-of-home that can be used for future assessment.

I used postcode districts as the unit of analysis and acknowledge the possibility that my findings are subject to the modifiable areal unit problem (MAUP). Although the spatial unit I used for analyses has the potential to introduce bias (305), my approach reflected information published by the online food delivery service and allowed consistency with the research I presented in Chapter 5 (213).

Food outlets newly registered to accept orders online might have operated with an additional hot food takeaway service under the emergency regulations that were in place between March 2020 and March 2022. Although I have provided evidence about the total number of food outlets registered to accept orders at a given time point, I am unable to comment on whether those newly registered were operating under the emergency regulations. In turn, this limits my ability to conclude that the emergency regulations increased online access to food prepared out-of-home.

6.7 Conclusions

I investigated if and how online access to food prepared out-of-home changed over time, compared with November 2019. The number of food outlets in England that were registered to accept orders online increased. In parallel, the number of food outlets registered to accept orders online as a percentage of the number of food outlets in the physical food environment increased. Although the number of food outlets accessible online decreased for the whole of England, trends differed by level of deprivation. The number of food outlets accessible online decreased in the least deprived areas. However, the number of food outlets accessible online was maintained in more deprived areas, and increased to the largest extent in the most deprived areas. Overall, then, I identified that online access to food prepared out-of-home changed over time, in the context of the first two-years of the COVID-19 pandemic. During this period of time, onpremises consumption of food prepared out-of-home was often restricted and it is possible that sociocultural norms about accessing this food were influenced. Data from March 2022 represent a new baseline from which future changes in access to food prepared out-of-home through online food delivery services can be assessed. Future research might attempt to understand the extent to which the changes I identified were associated with changes in online food delivery service use and in turn, longer-term implications for overall dietary patterns and diet-related health.

Chapter 7: Discussion

I start this chapter by summarising and interpreting the principal findings of the research presented in previous chapters. After this, I discuss the overarching methodological considerations of my research and outline possible implications for public health and future research.

7.1 Summary and interpretation of principal findings

Returning to the aims of my thesis, I wanted to provide an improved understanding of the use of online food delivery services to access food prepared out-of-home, specifically with regard to the:

- 1. Prevalence of online food delivery service use and sociodemographic characteristics of customers.
- 2. Experiences of online food delivery service use from the perspective of customers.
- Associations between measures of access to food prepared out-of-home through online food delivery services and online food delivery service use and body weight.
- 4. Extent to which access to food prepared out-of-home through online food delivery services is socioeconomically patterned, and how this access and any inequality changed over time.

I addressed my first aim in Chapter 2. I found that 15% of respondents living across Australia, Canada, Mexico, the United Kingdom (UK), and the United States of America (USA) reported use of an online food delivery service in the week prior to online survey completion, with the highest prevalence in Mexico. Online food delivery service customers were often male, younger, lived with children, and more highly educated. Furthermore, they often identified with an ethnic minority. Patterns were largely similar across countries. My findings for online food delivery service customers were consistent with evidence on the sociodemographic characteristics of adults reported to consume food prepared out-of-home in Australia (84), Canada (219), the UK (40, 111), and the USA (331). The consistency between my findings and this previous research suggests that population groups with certain sociodemographic characteristics purchase and consume food prepared out-of-home. To some extent, therefore, it might be possible to transfer findings from the physical food environment to the context of the digital food environment. The similarities in sociodemographic characteristics also make the findings from my qualitative research particularly useful because they help to understand possible reasons for using one purchasing format over another.

In Chapter 3, where I addressed my second aim, frequent online food delivery service customers recognised that these services facilitate access to food prepared out-of-home. Indeed, participants reported that purchasing this food was one reason for using these services. A growing body of evidence indicates that the foods sold through online food delivery services are high in calories, total fat, saturated fat and salt (159, 162, 165, 168), which was aligned with participant expectations about `takeaway food`. This insight into sociocultural norms about food prepared out-of-home has relevance for multiple purchasing formats.

Each domain of food access was discussed in some way by the frequent online food delivery service customers in my research. For instance, participants discussed how the type and price of food (i.e. acceptability and affordability) and estimated delivery times (i.e. accommodation) influenced their online food delivery service use. Additionally, participants reported that the number of food outlets they could order from (i.e. availability and accessibility) was influential because it was higher and more varied than in the physical food environment. This perspective was supported by quantitative evidence in Chapter 4, where I addressed my third aim. In this chapter, I reported that respondents with the highest number of food outlets accessible online (between 182 and 879) had greater odds of online food delivery service use compared to those with the lowest number. Together, these findings provide confidence that the multiple domains of food access are relevant to the use of online food delivery services. Although I did not identify an association between the number of food outlets accessible online food delivery service use and body weight might be better investigated through other measures of exposure such as food purchasing data. I will discuss this further in Section 7.4.3.

I proposed that online food access precedes the use of online food delivery services and that individual-level sociodemographic characteristics might influence the decision to seek out these services. A notable finding from Chapter 4 was that the highest number of food outlets accessible online (i.e. the greatest exposure) was positively associated with online food delivery service use amongst individuals with the highest level of education, yet not those who were less well educated. A higher level of education can be protective against the influence of food outlet exposure in the physical food environment in terms of purchasing available food (109, 332), which is in contrast to my finding. Individuals of a higher socioeconomic position (for example, those who are more highly educated) are often early-adopters of new technology (333). This might mean that they are more inclined to use online food delivery services to purchase food prepared out-of-home because it is aligned with broader digital technology use. Moreover, level of education contributes to differences in the types of food purchased out-of-home (111). Although my qualitative research in Chapter 3 indicated that online food delivery services were used to purchase energy-dense and nutrient-poor food, individuals with a higher level of education may identify and purchase relatively healthier foods due to having greater health literacy (189). This might be especially possible when a higher number of food outlets are accessible.

I addressed my fourth aim in Chapter 5, where I found that the number of food outlets accessible online was highest in the most deprived areas of England, and that levels were almost 50% greater than in the least deprived areas (106 outlets versus 70 outlets). I also addressed my fourth aim in Chapter 6, where I found that the number of food outlets accessible online significantly increased between November 2019 and March 2022 in the most deprived areas of England, but not in less deprived areas. Moreover, the number and percentage of food outlets in the physical food environment registered to accept orders online also increased over time and were consistently highest in the most deprived areas of England. Although these changes in online access to food prepared out-of-home cannot be directly attributed to the Coronavirus Disease 2019 (COVID-19) pandemic, my findings indicate that existing differences widened over this period of time. My findings are consistent with evidence that increases in the prevalence of childhood obesity were greatest in the most deprived areas of England between 2020 and 2021 (334).

7.2 Methodological considerations

I have considered the strengths and limitations of my research in each respective chapter. In this section, I discuss broader methodological considerations that are relevant to my research.

7.2.1 Confounding and reverse causality

The research in my thesis had either a cross-sectional or repeat cross-sectional study design. For many reasons, it is not generally possible to make strong causal inferences from the findings of research with these study designs (62). First, like other study designs, it is possible that the findings are subject to bias from unmeasured confounding (62). An unmeasured confounder is a variable associated with both an exposure and an outcome of interest that is not subsequently accounted for in analyses (335). In Chapter 4, I investigated the relationship between the number of food outlets accessible online (exposure) and online food delivery service use

(outcome). In this instance, the outcome was proximal and specific to the measured exposure and on a plausible pathway, which helps to minimise (although not eliminate) the risk of bias from unmeasured confounding (257). These considerations are also outlined in the Bradford Hill `criteria` commonly used to assess the causal nature of research findings (336, 337). Nevertheless, it is plausible that the positive association I reported was subject to unobserved factors. Online food delivery service customers may have selected to live in a certain area to gain access to better travel infrastructure and local amenities, which in turn might lead to a higher number of food outlets in the physical food environment and greater online food outlet access. As such, the association I observed may be attributable to lifestyle preferences that I did not account for (338, 339). Furthermore, the findings that I presented in Chapter 3 suggested that the price of food and marketing influenced online food delivery service use. However, these were not captured in my exposure measure.

Second, the findings from research with a cross-sectional or repeat cross-sectional study design, including my own, are vulnerable to bias from reverse causality. This bias refers to instances where an outcome precedes and influences the exposure rather than vice versa, which hampers the opportunity to make strong causal inferences (337, 340). It is possible that bias from reverse causality was present in the aforementioned positive association between the number of food outlets accessible online and online food delivery service use. For example, the use of online food delivery services by customers might influence the number of food outlets that register to accept orders online and become accessible in a given area, rather than exposure to food outlets through online food delivery services influencing online food delivery service use. Elsewhere, in Chapter 2, for example, it is unlikely that bias from reverse causality is a concern in associations between sociodemographic characteristics such as age and sex and online food delivery service use (62).

7.2.2 Generalisability

The research in my thesis was mostly conducted in England; a high-income, westernised country, where the consumption of food prepared out-of-home is popular (40, 341). Online food delivery services have been available in England since around 2006 (163, 316), and are used to a greater extent than in some other high-income countries (183, 211). Online food delivery services have only become available in some low- or middle-income countries and other high-income countries more recently than England (144). Access to and use of these services may be limited in these countries because it has not yet become normal to purchase food prepared out-of-home in this manner. However, online food delivery services are becoming increasingly available worldwide, which might influence sociocultural norms (147). Moreover, a global

137

nutrition transition has been underway for several years, with worldwide purchasing practices increasingly reflecting `western` culture (8, 206, 342). This global nutrition transition might mean that there is an increased demand for food prepared out-of-home through online food delivery services in the future. Therefore, although my findings might have limited generalisability to some countries as it stands, they provide insight into the mechanisms through which future demand would be met.

7.2.3 Investigating multiple domains of food access

Previous research in the physical food environment has tended to investigate the availability and accessibility domains of food access through measures of food outlet presence, density, or proximity (27, 77, 90). In part, the focus on these domains (rather than affordability, acceptability or accommodation) reflects the use of secondary data that are publicly available (26, 291). Moreover, to a limited extent, food outlet availability and accessibility precede the remaining domains of food access (343). Nevertheless, investigating domains of food access in isolation does not capture the multiple and complex factors that influence food purchasing practices (76, 89). Although I too investigated the availability and accessibility domains of food access in Chapters 4, 5, and 6, I also used qualitative research to investigate the remaining domains in Chapter 3. Doing so was important because it provided a comprehensive account of online food delivery service use. It also provided evidence about domains of food access that are not necessarily spatially dependent, such as the opportunity to reallocate time while waiting for meal delivery (i.e. accommodation).

7.2.4 Approach to, and scale of, data collection

I used web scraping to automate data collection from the leading online food delivery service in the UK. My use of web scraping was a considerable strength because it allowed me to conduct analyses on a national scale. In turn, in Chapters 5 and 6, I provided evidence about variation in measures of online food access across England as a whole. Doing so expanded the scope and scale of previous research that was conducted within single cities, which might have limited variability in terms of exposures and outcomes measured. Moreover, my use of web scraping allowed me to maintain a consistent data collection schedule and construct a national dataset spanning 22 months that will now be difficult to reconstruct retrospectively. While there are some concerns about the use of web scraping, including copyright infringement issues, these have not become fully established (344). Moreover, this data collection approach has been used in previous food retail research (345), and by the Office for National Statistics (the executive office of the UK Statistics Authority), who published a `web scraping policy` containing data

collection principles (346). I adhered to these principles throughout my data collection, for example, by including an appropriate delay between searches on the online food delivery service website so as not to overwhelm their servers.

Although the use of web scraping was a considerable strength of my research, there were also limitations. Collecting data from a single online food delivery service and not collecting data from small-scale services plausibly underestimated access to food prepared out-of-home through this type of service. To investigate the extent to which data collection from a single online food delivery service would lead to exposure misclassification, I compared information about food outlets identified through my data source with information collected from the next largest online food delivery service operating in England. For one region in England, over 90% of food outlets appeared in both datasets (see Appendix C: Figure S1). This comparison was useful because it provided confidence in data completeness as any undercounting was likely to have been small. Nevertheless, future research should aim to include data from multiple online food delivery services might differ. Any differences might mean that the accessible food is not consistent in terms of its energy and nutrient content.

7.2.5 Purposeful exposure to online food delivery services

To become fully aware of the number of food outlets that can be ordered from (i.e. that are accessible), individuals must enter a delivery location on the platform of a given online food delivery service. It is only at this point that 'exposure' to food outlets, as I conceived it in my research, truly occurs. As such, it is possible that the findings I reported in Chapter 4 are subject to bias that is comparable to selective daily mobility bias in studies investigating physical food environment exposures. This bias describes instances where exposure to food outlets, for example, is a result of individuals intentionally visiting an outlet to purchase a meal. In the context of my research, the presence of this bias would make it difficult to determine the extent to which personal preferences influence interactions with online food delivery services and subsequent exposure to food outlets that can be ordered from (347, 348). Relatedly, in Chapters 5 and 6, I provided evidence on the potential to use online food delivery services. For this potential to be realised, given populations living in the studied areas would need to be online food delivery service customers.

7.3 Possible implications for public health and policy

My empirical research has not established a definitive need for public health intervention. Nevertheless, the European region of the World Health Organization (WHO) have raised concerns about online food delivery service use because the foods available to purchase are typically high in calories, total fat, saturated fat and salt. In turn, they have indicated that public health intervention may be required in the future (212). My research also suggests, subject to further investigation, that this may be required. In the next sections, I will discuss opportunities for public health intervention from the perspective that it may be necessary in the future.

7.3.1 The interdependency of the physical and digital food environments

The business model of online food delivery services is predicated on the interdependency of the physical and digital food environments, especially the location of customer-facing food outlets. As a result, public health interventions adopted in the physical food environment are likely to influence food retail in the digital food environment. For instance, urban planning determines the spatial structures of an area (327), and public health planning documents outline how professionals in practice might prevent new outlets that would be selling food prepared out-ofhome from opening (54, 55, 349, 350). As of 2018, half of Local Authorities in England (n=164, 51%) had adopted an urban planning policy to prevent new hot food takeaway outlets from opening in their boundary area (330), with a similar approach adopted in the USA (351). Implementation of this intervention would transfer to the digital food environment since a new food outlet would not have been allowed to open, thereby preventing it from registering to accept orders through online food delivery services. Given the coexisting nature of access to food prepared out-of-home in the physical and digital food environments, and despite the apparent increase in the popularity of using online food delivery services, public health interventions that include aspects of the physical food environment continue to be relevant. This seems especially the case since almost two-thirds of respondents in Chapter 2 had purchased food prepared out-of-home in the past week, but had not used an online food delivery service.

Other public health interventions might seek to address online food delivery service marketing in the physical food environment, which can include advertising inside of food outlets and on bus shelters, delivery couriers with branded clothing and transport, and food arriving in online food delivery service (rather than individual food outlet) branded packaging. Although being an online food delivery service customer may have led to greater awareness (101), the participants in my qualitative research recognised that the aforementioned marketing was present in the physical food environment (see Chapter 3). Moreover, given that the number of food outlets registered to accept orders online was highest in the most deprived areas of England (see Chapters 5 and 6), it is reasonable to suggest that online food delivery service marketing is more prevalent in these areas. As marketing influences the purchase and consumption of food prepared out-of-home (149, 352), interventions could aim to limit exposure to this in the physical food environment. There is precedent for such approaches, with advertising restriction on the Transport for London network successfully reducing the purchase of energy, sugar and fat from foods high in fat, salt and sugar amongst populations living in the intervention area (353).

7.3.2 A public health intervention involving multiple stakeholders

In the previous section, I described interventions that seek to improve public health by restricting exposure to customer-facing food outlets and online food delivery service marketing in the physical food environment. However, frequent online food delivery service customers in my qualitative research reported no intention to discontinue their online food delivery service use (see Chapter 3). In part, this was because the food available through these services met their expectations about what takeaway food `should` be in terms of its type and energy and nutrient content. Public health interventions that attempt to guide choice towards foods that are lower in calories and nutrients of public health concern, rather than necessarily attempting to prevent online food delivery service use, are likely to be most aligned with my findings (354, 355).

An example of such an approach is the introduction of a food healthiness rating score for individual food businesses registered to accept orders online. These scores would be determined and assigned based on the composition of menus and the energy and nutrient content of foods sold, with businesses that have a healthier rating then *automatically* appearing first in customer search results (157). There is evidence that this type of intervention was acceptable to online food delivery service customers who had the opportunity to personalise how search results were displayed to them (for example, based on food healthiness) (356). Moreover, this intervention could influence purchasing practices since it is aligned with formative evidence that prioritising food businesses selling lower energy items when presenting search results on a simulated online food delivery service website led to the purchase of 12% fewer calories (166 kcal) compared with a control set of search results (357). Elsewhere, one systematic review reported that prominent placement of healthier items (according to the Eatwell Guide published by Public Health England) in supermarkets was associated with an increased purchase of displayed foods (358).

Introducing a food healthiness rating score and automatically presenting food businesses with a healthier rating first in customer search results would be multi-faceted and require the input of multiple stakeholders. Thus, it is useful to assess the mechanisms through which such an intervention might influence the purchase and consumption of food prepared out-of-home. Automatically presenting businesses selling healthier items first in customer search results might encourage retailers to reformulate their current menu items or to develop new menus according to nutritional standards required to achieve a healthier score. In the USA, mandatory calorie labelling on menus was associated with a reduction in the energy and salt content of food sold by large chain restaurants (359, 360). A food healthiness rating score could arguably operate in a similar manner. Additionally, food sold through online food delivery services is typically prepared inside of existing customer-facing premises in the physical food environment (39, 147). Public health interventions that aimed to improve the nutritional guality of food sold out-ofhome by having businesses adapt their cooking practices through recipe reformulation and changing ingredients were acceptable and implementable (278, 361). If successful, these improvements would transfer to the food sold through online food delivery services. Finally, as online food delivery services already dictate the content and style of information displayed to customers (362), there is an opportunity for them to ensure that information about food healthiness rating scores of businesses (as well as information about their menu, hygiene rating, and food prices and pictures) are presented in a manner that maximises customer engagement. This might include the allocation of a colour based on food healthiness rating scores, which would be similar to the use of traffic light schemes on food packaging that can favourably influence food purchasing (363).

7.4 Future research agenda

Figure 7.1 is an amended version of the proposed causal pathway that I presented in Figure 1.6 (see Section 1.4.4). Solid lines in Figure 7.1 represent evidence generated from the research I presented in Chapters 2 to 6. Despite my contribution to knowledge, further research is required to understand the need for public health intervention and the consistency of the associations I observed. The latter is outlined in the Bradford Hill `criteria` I referred to in Section 7.2.1 (337). This further research might also investigate distal outcomes of online food delivery service use, such as those linked to the consumption of food prepared out-of-home, including cardiovascular diseases (364). In the next sections, I outline a possible agenda for future research.





7.4.1 Cumulative access to food prepared out-of-home

In Section 7.3.1, I referred to the interdependency of the physical and digital food environments, which is particularly important for future research. In the past, the outcome of food outlet exposure in the physical food environment was the purchase of food from customer-facing premises. However, this is no longer necessarily the case (365, 366). For instance, a given individual could pass a food outlet in the physical food environment, which might prime them for later use of an online food delivery service rather than purchasing food directly from the food outlet. As it stands, however, the physical food environment and the digital food environment are mostly viewed independently (144). Doing so might create a false dichotomy that fails to capture exposures that attenuate theoretical and plausible associations with outcomes (148). Future research should seek to consider how the physical food environment and the digital food environment provide cumulative access to food prepared out-of-home across multiple purchasing formats, and how this is associated with the purchase and consumption of food prepared out-of-home. Data that allows exposures and outcomes in both the physical and digital food environment to be considered would be required. It will be essential for this future research to consider factors that are not necessarily captured within conceptualisations of food environments. For example, social and economic factors form part of the socioecological model and operate at multiple levels (see Section 1.2.5.1: Figure 1.6). Within

these factors, and beyond the location of food outlets in the physical food environment, other socioeconomic and environmental conditions such as neighbourhood walkability and perceived safety, digital connectivity and road networks might influence food access and food purchasing practices (69, 72).

7.4.2 Supplementation or substitution of alternative purchasing formats and food sources

One aim of the online food delivery service business model is to become the preferred way that food prepared out-of-home is purchased (367). However, the potential for online food delivery service use to *supplement* the use of other purchasing formats is a public health concern (141). Although it is untested, if online food delivery services were used in addition to other purchasing formats without displacement, then overall consumption of food prepared out-ofhome would increase. It is also possible that an increased frequency of online food delivery service use supplements the consumption of other types of food (i.e. not necessarily food prepared out-of-home) without displacement. In both scenarios, overall energy intake might increase, which could have health implications if sustained.

Another possibility is that the use of online food delivery services *substitutes* alternative ways of purchasing food prepared out-of-home and the consumption of other types of food (368). In this scenario, overall food consumption might be unchanged. Importantly, however, even if the use of online food delivery services does not change the overall consumption of food prepared out-of-home, the `problem` of consuming this food remains, with it now being accessed in a different manner. Awareness of this is important in the context of the possible need for public health intervention.

In line with the aforementioned scenarios, future research should seek to investigate the extent to which the use of online food delivery services contributes to overall food consumption, particularly for food prepared out-of-home. This knowledge will help provide a better understanding about the possible implications for overall dietary patterns and body weight. Longitudinal cohort studies that include data on overall food consumption and the purchasing format used to access food prepared out-of-home would be well suited for this research. In addition, cross-sectional research that measures energy intake between groups who use online food delivery services more and less frequently could provide useful insight.

7.4.3 Moving beyond the frequency of online food delivery service use

I found that the number of food outlets accessible online was associated with online food delivery service use in Chapter 4, however, I found no evidence that this exposure was

associated with body weight after controlling for potential confounders. As I discussed in Section 1.2.4.1, more frequent consumption of food prepared out-of-home is associated with poorer overall dietary patterns and living with obesity. Furthermore, the European region of the WHO have raised concerns about the increasing popularity of using online food delivery services due to the types of food sold (212). Indeed, there is evidence from Australia, New Zealand, and Canada that the food sold through online food delivery services in these countries is high in calories, total fat, saturated fat and salt (159, 165, 166). However, the food purchased through online food delivery services has not yet been established. A possible disconnect between the food available and the food purchased means that without validation, the implications of online food delivery service use on overall dietary patterns and body weight cannot be assumed (90, 148). Future research should aim to move beyond outcomes related to the frequency of online food delivery service use. This could be achieved by investigating the energy and nutrient content of food purchased. Transaction data linked with nutritional composition or menu data, offers a potential solution. Although I acknowledge that this information is not easily attainable, it may be accessible from food retailers, as demonstrated elsewhere (369, 370).

7.4.4 Online food delivery service diversification

Since the research presented in my thesis was conceived, online food delivery services in the UK have diversified in terms of the type of food accessible through their services. Although facilitating access to food prepared out-of-home remains their core function, they now also provide access to supermarkets and convenience stores selling food that would not typically be regarded as prepared out-of-home (254, 317). To some extent, this diversification reflects a digital society and sociocultural norms about receiving items rapidly and on-demand (146). Online food delivery services will plausibly be encouraged to further diversify in the future to meet customer demand for different types of food (371). Moreover, at the time of writing, despite growth in terms of revenue and the number of orders processed, many online food delivery services are not yet profitable. Operating across multiple areas of food retail may be perceived as a way to increase the potential to become profitable (39, 325, 372). As described by the Food Standards Agency, monitoring the operations of online food delivery services will be essential (373), especially in light of the COVID-19 pandemic, which is reported to have transformed food retail (374). Given that the public health implications of online food delivery service diversification are unknown, future research could seek to understand how these services contribute to overall food retail by providing access to different types of food, possibly by building on the dataset I generated for my research in Chapter 6.

7.4.5 Changes to the location of food preparation

Finally, alongside new and different types of food being available to purchase, there has been evolution in the preparation location for the food sold through online food delivery services. This is best demonstrated by the development of facilities known as `dark kitchens` (375). These facilities allow food businesses to register to accept orders through online food delivery services and prepare meals for delivery without the financial costs of having a customer-facing premises (376). In one London Borough, three dark kitchens, hosting 124 individual food businesses were identified in 2021 (376). This finding underlines the potential for the dark kitchen business model to increase access to food prepared out-of-home through online food delivery services, as the number of food businesses (i.e. 124) was greater than the number of premises approved to function (i.e. 3). Future research might seek to understand the public health implications of the dark kitchen business model, including how it uniquely contributes to the number of food outlets accessible online.

Chapter 8: Conclusions

Previous research on access to food prepared out-of-home and the relationship between consumption of this food, and diet and body weight has typically been conducted in the context of the physical food environment. However, food prepared out-of-home can now also be accessed through online food delivery services, which are an aspect of food retail within the digital food environment. Despite these services contributing to public health concerns about the consumption of food prepared out-of-home, knowledge about the extent of their use, by whom, and the factors that might influence this, was limited. My thesis aimed to contribute to the existing evidence base by providing an improved understanding about these aspects of online food delivery service use.

First, I provided evidence that online food delivery services had been used by around one in six adults living across five upper-middle or high-income countries in the week prior to online survey completion in 2018. I found that online food delivery service customers tended to be male, younger, more highly educated, often lived with children, and identified with an ethnic minority. I also contributed to the current evidence base by providing knowledge about factors that influence online food delivery service use. Frequent online food delivery service customers living in the UK reported that these services provide streamlined purchasing processes and sell food that meets their expectations about 'takeaway food'. Moreover, these frequent customers informed me that they could access a higher number of food outlets through online food delivery services compared with alternative purchasing formats. In turn, this influenced their decision to purchase food prepared out-of-home in this manner. Indeed, the number of food outlets accessible online emerged as being particularly important in the chapters that followed my qualitative research. Adults living in Great Britain with the highest number of food outlets accessible online had greater odds of online food delivery service use in the week prior to online survey completion in 2019. Taken together, this is cause for public health concern as the number of food outlets accessible online was highest in the most deprived areas of England in November 2019. Furthermore, the number increased in the most deprived areas but not elsewhere between 2020 and 2022, which suggests that inequalities in online access to food prepared out-of-home widened over time. In turn, this could lead to greater overall consumption of food prepared out-of-home amongst populations that had barriers to having healthier overall dietary patterns even before the emergence of online food delivery services.

Within an increasingly digital society, online food delivery services will likely continue to be used to purchase food prepared out-of-home. Public health interventions have been adopted to address concerns about access to and consumption of this food. However, these interventions do not necessarily address access to food prepared out-of-home through online food delivery services. Although further research is required to understand how food purchased through online food delivery services contributes to overall dietary patterns and health, there may be a need to develop new public health interventions, or to adapt and extend existing ones, to better address online access to food prepared out-of-home.

List of references

1. Swinburn B, Sacks G, Vandevijvere S, Kumanyika S, Lobstein T, Neal B, Barquera S, Friel S, Hawkes C, Kelly B, L'Abbé M, Lee A, Ma J, Macmullan J, Mohan S, Monteiro C, Rayner M, Sanders D, Snowdon W, Walker C, INFORMAS. Informas (international network for food and obesity/non-communicable diseases research, monitoring and action support): Overview and key principles. Obesity Reviews. 2013; 14: 1-12. https://doi.org/10.1111/obr.12087.

2. Swinburn B, Egger G, Raza F. Dissecting obesogenic environments: The development and application of a framework for identifying and prioritizing environmental interventions for obesity. Preventive Medicine. 1999; 29: 563-70. https://doi.org/10.1006/pmed.1999.0585.

3. Branca F, Lartey A, Oenema S, Aguayo V, Stordalen GA, Richardson R, Arvelo M, Afshin A. Transforming the food system to fight non-communicable diseases. BMJ. 2019; 364: I296. https://doi.org/10.1136/bmj.I296.

4. Mytton OT, Nnoaham K, Eyles H, Scarborough P, Ni Mhurchu C. Systematic review and metaanalysis of the effect of increased vegetable and fruit consumption on body weight and energy intake. BMC Public Health. 2014; 14: 886. https://doi.org/10.1186/1471-2458-14-886.

5. Schwingshackl L, Hoffmann G, Kalle-Uhlmann T, Arregui M, Buijsse B, Boeing H. Fruit and vegetable consumption and changes in anthropometric variables in adult populations: A systematic review and meta-analysis of prospective cohort studies. PLOS ONE. 2015; 10: e0140846. https://doi.org/10.1371/journal.pone.0140846.

6. Rao M, Afshin A, Singh G, Mozaffarian D. Do healthier foods and diet patterns cost more than less healthy options? A systematic review and meta-analysis. BMJ Open. 2013; 3: e004277. https://doi.org/10.1136/bmjopen-2013-004277.

7. Shangguan S, Afshin A, Shulkin M, Ma W, Marsden D, Smith J, Saheb-Kashaf M, Shi P, Micha R, Imamura F, Mozaffarian D. A meta-analysis of food labeling effects on consumer diet behaviors and industry practices. American Journal of Preventive Medicine. 2019; 56: 300-14. https://doi.org/10.1016/j.amepre.2018.09.024.

8. Popkin BM. Nutrition transition and the global diabetes epidemic. Current diabetes reports. 2015; 15: 64-78. https://doi.org/10.1007/s11892-015-0631-4.

9. Mozaffarian D. Dietary and policy priorities to reduce the global crises of obesity and diabetes. Nature Food. 2020; 1: 38-50. https://doi.org/10.1038/s43016-019-0013-1.

10. Jia P, Lakerveld J, Wu J, Stein A, Root Elisabeth D, Sabel Clive E, Vermeulen R, Remais Justin V, Chen X, Brownson Ross C, Amer S, Xiao Q, Wang L, Verschuren WMM, Wu T, Wang Y, James P. Top 10 research priorities in spatial lifecourse epidemiology. Environmental Health Perspectives. 2019; 127: 074501-7. https://doi.org/10.1289/EHP4868.

11. Jones NRV, Tong TYN, Monsivais P. Meeting uk dietary recommendations is associated with higher estimated consumer food costs: An analysis using the national diet and nutrition survey and consumer expenditure data, 2008–2012. Public Health Nutrition. 2017; 21: 948-56. https://doi.org/10.1017/S1368980017003275.

12. The GBD 2015 Obesity Collaborators. Health effects of overweight and obesity in 195 countries over 25 years. New England Journal of Medicine. 2017; 377: 13-27. https://doi.org/10.1056/NEJMoa1614362.

13. NHS Digital. Health survey for england, 2019: Data tables. 2020. First accessed: 07.01.2022. Available from: https://digital.nhs.uk/data-and-information/publications/statistical/health-survey-for-england/2019/health-survey-for-england-2019-data-tables.

14. Bambra C. Placing intersectional inequalities in health. Health & Place. 2022; 75: 102761. https://doi.org/10.1016/j.healthplace.2022.102761.

15. McGowan VJ, Buckner S, Mead R, McGill E, Ronzi S, Beyer F, Bambra C. Examining the effectiveness of place-based interventions to improve public health and reduce health inequalities: An umbrella review. BMC Public Health. 2021; 21: 1888. https://doi.org/10.1186/s12889-021-11852-z.

16. National Food Strategy. National food strategy: An independent review for the government. 2022. First accessed: 06.07.2022. Available from: https://www.nationalfoodstrategy.org/.

17. Wang MC, Kim S, Gonzalez AA, MacLeod KE, Winkleby MA. Socioeconomic and food-related physical characteristics of the neighbourhood environment are associated with body mass index. J Epidemiol Community Health. 2007; 61: 491-8. https://doi.org/10.1136/jech.2006.051680.

18. Olstad DL, Kirkpatrick SI. Planting seeds of change: Reconceptualizing what people eat as eating practices and patterns. International Journal of Behavioral Nutrition and Physical Activity. 2021; 18: 32. https://doi.org/10.1186/s12966-021-01102-1.

19. Venn D, Banwell C, Dixon J. Australia's evolving food practices: A risky mix of continuity and change. Public Health Nutrition. 2016; 20: 2549-58. https://doi.org/10.1017/S136898001600255X.

20. Lowe MR, Butryn ML. Hedonic hunger: A new dimension of appetite? Physiology & Behavior. 2007; 91: 432-9. https://doi.org/10.1016/j.physbeh.2007.04.006.

21. Lake AA, O'Malley C, Moore HJ. Environmental drivers of obesity: Individual willpower versus societal responsibility. Nutrition Bulletin. 2022; 47: 277-81. https://doi.org/10.1111/nbu.12574.

22. Lake AA, Burgoine T, Stamp E, Grieve R. The foodscape: Classification and field validation of secondary data sources across urban/rural and socio-economic classifications in england. International Journal of Behavioral Nutrition and Physical Activity. 2012; 9: 1-12. https://doi.org/10.1186/1479-5868-9-37.

23. Pinho MGM, Mackenbach JD, Oppert JM, Charreire H, Bárdos H, Rutter H, Compernolle S, Beulens JWJ, Brug J, Lakerveld J. Exploring absolute and relative measures of exposure to food environments in relation to dietary patterns among european adults. Public Health Nutrition. 2018; 22: 1037-47. https://doi.org/10.1017/S1368980018003063.

24. Jackson P, Viehoff V. Reframing convenience food. Appetite. 2016; 98: 1-11. https://doi.org/10.1016/j.appet.2015.11.032.

25. Gesteiro E, García-Carro A, Aparicio-Ugarriza R, González-Gross M. Eating out of home: Influence on nutrition, health, and policies: A scoping review. Nutrients. 2022; 14: 1265. https://doi.org/10.3390/nu14061265.

26. Wilkins E, Radley D, Morris M, Hobbs M, Christensen A, Marwa WL, Morrin A, Griffiths C. A systematic review employing the geofern framework to examine methods, reporting quality and associations between the retail food environment and obesity. Health & Place. 2019; 57: 186-99. https://doi.org/10.1016/j.healthplace.2019.02.007.

27. Charreire H, Casey R, Salze P, Simon C, Chaix B, Banos A, Badariotti D, Weber C, Oppert JM. Measuring the food environment using geographical information systems: A methodological review. Public Health Nutr. 2010; 13: 1773-85. https://doi.org/10.1017/s1368980010000753.

28. Lake A, Burgoine T, Greenhalgh F, Stamp E, Tyrrell R. The foodscape: Classification and field validation of secondary data sources. Health & Place. 2010; 16: 666-73. https://doi.org/10.1016/j.healthplace.2010.02.004.

29. Wellard-Cole L, Davies A, Allman-Farinelli M. Contribution of foods prepared away from home to intakes of energy and nutrients of public health concern in adults: A systematic review. Crit Rev Food Sci Nutr. 2022; 62: 5511-22. https://doi.org/10.1080/10408398.2021.1887075.

30. Murcott A. Introducing the sociology of food and eating: Bloomsbury Publishing; 2019.

31. Maguire ER, Burgoine T, Monsivais P. Area deprivation and the food environment over time: A repeated cross-sectional study on takeaway outlet density and supermarket presence in norfolk, uk, 1990–2008. Health & Place. 2015; 33: 142-7. https://doi.org/10.1016/j.healthplace.2015.02.012.

32. Pinho MGM, Mackenbach JD, den Braver NR, Beulens JJW, Brug J, Lakerveld J. Recent changes in the dutch foodscape: Socioeconomic and urban-rural differences. International Journal of Behavioral Nutrition and Physical Activity. 2020; 17: 43. https://doi.org/10.1186/s12966-020-00944-5.

33. Ramírez-Toscano Y, Pérez-Ferrer C, Bilal U, Auchincloss AH, Barrientos-Gutierrez T. Socioeconomic deprivation and changes in the retail food environment of mexico from 2010 to 2020. Health & Place. 2022; 77: 102865. https://doi.org/10.1016/j.healthplace.2022.102865.

34. Needham C, Orellana L, Allender S, Sacks G, Blake MR, Strugnell C. Food retail environments in greater melbourne 2008–2016: Longitudinal snalysis of intra-city variation in density and healthiness of food outlets. International Journal of Environmental Research and Public Health. 2020; 17: 1321. https://doi.org/10.3390/ijerph17041321.

35. Hobbs M, Mackenbach JD, Wiki J, Marek L, McLeod GFH, Boden JM. Investigating change in the food environment over 10 years in urban new zealand: A longitudinal and nationwide geospatial study. Social Science & Medicine. 2021; 269: 113522. https://doi.org/10.1016/j.socscimed.2020.113522.

36. Lachat C, Nago E, Verstraeten R, Roberfroid D, Van Camp J, Kolsteren P. Eating out of home and its association with dietary intake: A systematic review of the evidence. Obesity reviews : an official journal of the International Association for the Study of Obesity. 2012; 13: 329-46. https://doi.org/10.1111/j.1467-789X.2011.00953.x.

37. Harnack LJ, Jeffery RW, Boutelle KN. Temporal trends in energy intake in the united states: An ecologic perspective. The American journal of clinical nutrition. 2000; 71: 1478-84. https://doi.org/10.1093/ajcn/71.6.1478.

38. Statista. Restaurant delivery and takeaway market in the uk. 2022. First accessed: 08.06.2022. Available from: https://www.statista.com/study/36742/takeaways-in-the-united-kingdom-uk-statista-dossier/.

39. Allen J, Piecyk M, Piotrowska M. An analysis of online shopping and home delivery in the uk. 2017. First accessed: 29.04.2021. Available from:

https://westminsterresearch.westminster.ac.uk/item/q16z5/analysis-of-online-shopping-and-home-delivery-in-the-uk.

40. Adams J, Goffe L, Brown T, Lake A, Summerbell C, White M, Wrieden W, Adamson A. Frequency and socio-demographic correlates of eating meals out and take-away meals at home: Cross-sectional analysis of the uk national diet and nutrition survey, waves 1–4 (2008–12). Int J Behav Nutr Phys Act. 2015; 12: 1-9. https://doi.org/10.1186/s12966-015-0210-8.

41. Robinson E, Jones A, Whitelock V, Mead BR, Haynes A. (over)eating out at major uk restaurant chains: Observational study of energy content of main meals. BMJ. 2018; 363: 1-8. https://doi.org/10.1136/bmj.k4982.

42. Prentice AM, Jebb SA. Fast foods, energy density and obesity: A possible mechanistic link. Obesity Reviews. 2003; 4: 187-94. https://doi.org/10.1046/j.1467-789X.2003.00117.x.

43. Theis DRZ, Adams J. Differences in energy and nutritional content of menu items served by popular uk chain restaurants with versus without voluntary menu labelling: A cross-sectional study. PLOS ONE. 2019; 14: e0222773. https://doi.org/10.1371/journal.pone.0222773.

44. Scrinis G, Monteiro C. From ultra-processed foods to ultra-processed dietary patterns. Nature Food. 2022; 3: 671-3. https://doi.org/10.1038/s43016-022-00599-4.

45. Huang Y, Burgoine T, Theis D, Adams J. Differences in energy and nutrient content of menu items served by large chain restaurants in the us and the uk. Public Health Nutrition. 2022; 25: 2671-9. https://doi.org/10.1017/S1368980022001379.

46. Jaworowska A, Blackham T, Stevenson L, Davies IG. Determination of salt content in hot takeaway meals in the united kingdom. Appetite. 2012; 59: 517-22. https://doi.org/10.1016/j.appet.2012.06.018.

47. Wellard-Cole L, Hooper A, Watson WL, Hughes C. Nutrient composition of australian fastfood and fast-casual children's meals available in 2016 and changes in fast-food meals between 2010 and 2016. Public Health Nutrition. 2019; 22: 2981-8. https://doi.org/10.1017/S1368980019001897.

48. Bauer KW, Hearst MO, Earnest AA, French SA, Oakes JM, Harnack LJ. Energy content of u.S. Fast-food restaurant offerings: 14-year trends. Am J Prev Med. 2012; 43: 490-7. https://doi.org/10.1016/j.amepre.2012.06.033.

49. Wellard L, Glasson C, Chapman K. Sales of healthy choices at fast food restaurants in australia. Health promotion journal of Australia : official journal of Australian Association of Health Promotion Professionals. 2012; 23: 37-41. https://doi.org/10.1071/he12037.

50. Janssen HG, Davies IG, Richardson LD, Stevenson L. Determinants of takeaway and fast food consumption: A narrative review. Nutr Res Rev. 2018; 31: 16-34. https://doi.org/10.1017/s0954422417000178.

51. Gordon-Larsen P. Food availability/convenience and obesity. Advances in nutrition (Bethesda, Md). 2014; 5: 809-17. https://doi.org/10.3945/an.114.007070.

52. Bivoltsis A, Cervigni E, Trapp G, Knuiman M, Hooper P, Ambrosini GL. Food environments and dietary intakes among adults: Does the type of spatial exposure measurement matter? A systematic review. International Journal of Health Geographics. 2018; 17: 19-39. https://doi.org/10.1186/s12942-018-0139-7. 53. Fleischhacker SE, Evenson KR, Rodriguez DA, Ammerman AS. A systematic review of fast food access studies. Obesity reviews : an official journal of the International Association for the Study of Obesity. 2011; 12: e460-e71. https://doi.org/10.1111/j.1467-789X.2010.00715.x.

54. Public Health England. Using the planning system to promote healthy weight environments. 2020. First accessed: 07.10.2020. Available from:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/ file/863821/PHE_Planning_healthy_weight_environments_guidance__1_.pdf.

55. Public Health England. Health matters: Obesity and the food environment. 2017. First accessed: 04.03.2019. Available from: https://www.gov.uk/government/publications/health-matters-obesity-and-the-food-environment/health-matters-obesity-and-the-food-environment--2#resources.

 Public Health England. Obesity and the environment. Density of fast food outlets at 31/12/2017. 2017. First accessed: 15.10.2018. Available from: https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/ file/741555/Fast_Food_map.pdf.

57. Gil Á, Martinez de Victoria E, Olza J. Indicators for the evaluation of diet quality. Nutricion hospitalaria. 2015; 31: 128-44. https://doi.org/10.3305/nh.2015.31.sup3.8761.

58. Barnes TL, French SA, Mitchell NR, Wolfson J. Fast-food consumption, diet quality and body weight: Cross-sectional and prospective associations in a community sample of working adults. Public Health Nutr. 2016; 19: 885-92. https://doi.org/10.1017/s1368980015001871.

59. Taher AK, Evans N, Evans CEL. The cross-sectional relationships between consumption of takeaway food, eating meals outside the home and diet quality in british adolescents. Public Health Nutrition. 2018; 22: 63-73. https://doi.org/10.1017/S1368980018002690.

60. Moore LV, Diez Roux AV, Nettleton JA, Jacobs DR, Franco M. Fast-food consumption, diet quality, and neighborhood exposure to fast food: The multi-ethnic study of atherosclerosis. American Journal of Epidemiology. 2009; 170: 29-36. https://doi.org/10.1093/aje/kwp090.

61. Cobb LK, Appel LJ, Franco M, Jones-Smith JC, Nur A, Anderson CA. The relationship of the local food environment with obesity: A systematic review of methods, study quality, and results. Obesity. 2015; 23: 1331-44. https://doi.org/10.1002/oby.21118.

62. Shahar E, Shahar DJ. Causal diagrams and the cross-sectional study. Clin Epidemiol. 2013; 5: 57-65. https://doi.org/10.2147/clep.s42843.

63. Rosenheck R. Fast food consumption and increased caloric intake: A systematic review of a trajectory towards weight gain and obesity risk. Obesity Reviews. 2008; 9: 535-47. https://doi.org/10.1111/j.1467-789X.2008.00477.x.

64. Giskes K, van Lenthe F, Avendano-Pabon M, Brug J. A systematic review of environmental factors and obesogenic dietary intakes among adults: Are we getting closer to understanding obesogenic environments? Obesity Reviews. 2011; 12: e95-e106. https://doi.org/10.1111/j.1467-789X.2010.00769.x.

65. Atanasova P, Kusuma D, Pineda E, Frost G, Sassi F, Miraldo M. The impact of the consumer and neighbourhood food environment on dietary intake and obesity-related outcomes: A systematic review of causal impact studies. Social Science & Medicine. 2022; 299: 114879. https://doi.org/10.1016/j.socscimed.2022.114879.

66. Story M, Kaphingst KM, Robinson-O'Brien R, Glanz K. Creating healthy food and eating environments: Policy and environmental approaches. Annual Review of Public Health. 2008; 29: 253-72. https://doi.org/10.1146/annurev.publhealth.29.020907.090926.

67. Dahlgren G, Whitehead M. The dahlgren-whitehead model of health determinants: 30 years on and still chasing rainbows. Public Health. 2021; 199: 20-4. https://doi.org/10.1016/j.puhe.2021.08.009.

68. Robinson T. Applying the socio-ecological model to improving fruit and vegetable intake among low-income african americans. J Community Health. 2008; 33: 395-406. https://doi.org/10.1007/s10900-008-9109-5.

69. Downs SM, Ahmed S, Fanzo J, Herforth A. Food environment typology: Advancing an expanded definition, framework, and methodological approach for improved characterization of wild, cultivated, and built food environments toward sustainable diets. Foods. 2020; 9: 532. https://doi.org/10.3390/foods9040532.

70. Glanz K, Sallis JF, Saelens BE, Frank LD. Healthy nutrition environments: Concepts and measures. American journal of health promotion : AJHP. 2005; 19: 330-3. https://doi.org/10.4278/0890-1171-19.5.330.

71. Winkler MR, Zenk SN, Baquero B, Steeves EA, Fleischhacker SE, Gittelsohn J, Leone LA, Racine EF. A model depicting the retail food environment and customer interactions: Components, outcomes, and future directions. International journal of environmental research and public health. 2020; 17: 7591. https://doi.org/10.3390/ijerph17207591.

72. Turner C, Aggarwal A, Walls H, Herforth A, Drewnowski A, Coates J, Kalamatianou S, Kadiyala S. Concepts and critical perspectives for food environment research: A global framework with implications for action in low- and middle-income countries. Global Food Security. 2018; 18: 93-101. https://doi.org/10.1016/j.gfs.2018.08.003.

73. Penchansky R, Thomas JW. The concept of access: Definition and relationship to consumer satisfaction. Medical Care. 1981; 19: 127-40. https://doi.org/10.1097/00005650-198102000-00001

74. Clary C, Matthews SA, Kestens Y. Between exposure, access and use: Reconsidering foodscape influences on dietary behaviours. Health & Place. 2017; 44: 1-7. https://doi.org/10.1016/j.healthplace.2016.12.005.

75. Ayala GX, Rogers M, Arredondo EM, Campbell NR, Baquero B, Duerksen SC, Elder JP. Awayfrom-home food intake and risk for obesity: Examining the influence of context. Obesity. 2008; 16: 1002-8. https://doi.org/10.1038/oby.2008.34.

76. Turner G, Green R, Alae-Carew C, Dangour AD. The association of dimensions of fruit and vegetable access in the retail food environment with consumption; a systematic review. Global Food Security. 2021; 29: 100528. https://doi.org/10.1016/j.gfs.2021.100528.

77. Caspi CE, Sorensen G, Subramanian SV, Kawachi I. The local food environment and diet: A systematic review. Health & Place. 2012; 18: 1172-87. https://doi.org/10.1016/j.healthplace.2012.05.006.

78. Lytle LA, Sokol RL. Measures of the food environment: A systematic review of the field, 2007–2015. Health & Place. 2017; 44: 18-34. https://doi.org/10.1016/j.healthplace.2016.12.007.

79. Lam TM, Vaartjes I, Grobbee DE, Karssenberg D, Lakerveld J. Associations between the built environment and obesity: An umbrella review. International Journal of Health Geographics. 2021; 20: 1-24. https://doi.org/10.1186/s12942-021-00260-6.

80. Needham C, Sacks G, Orellana L, Robinson E, Allender S, Strugnell C. A systematic review of the australian food retail environment: Characteristics, variation by geographic area, socioeconomic position and associations with diet and obesity. Obesity Reviews. 2020; 21: e12941. https://doi.org/10.1111/obr.12941.

81. Jeffery RW, Baxter J, McGuire M, Linde J. Are fast food restaurants an environmental risk factor for obesity? International Journal of Behavioral Nutrition and Physical Activity. 2006; 3: 1-6. https://doi.org/10.1186/1479-5868-3-2.

82. Paquet C, Daniel M, Knäuper B, Gauvin L, Kestens Y, Dubé L. Interactive effects of reward sensitivity and residential fast-food restaurant exposure on fast-food consumption. The American journal of clinical nutrition. 2010; 91: 771-6. https://doi.org/10.3945/ajcn.2009.28648.

83. Thornton LE, Bentley RJ, Kavanagh AM. Fast food purchasing and access to fast food restaurants: A multilevel analysis of viclanes. International Journal of Behavioral Nutrition and Physical Activity. 2009; 6: 28-38. https://doi.org/10.1186/1479-5868-6-28.

84. Turrell G, Giskes K. Socioeconomic disadvantage and the purchase of takeaway food: A multilevel analysis. Appetite. 2008; 51: 69-81. https://doi.org/10.1016/j.appet.2007.12.002.

85. Allman-Farinelli M, Rahman H, Nour M, Wellard-Cole L, Watson LW. The role of supportive food environments to enable healthier choices when eating meals prepared outside the home: Findings from focus groups of 18 to 30-year-olds. Nutrients. 2019; 11: 1-11. https://doi.org/10.3390/nu11092217.

86. Neve KL, Isaacs A. How does the food environment influence people engaged in weight management? A systematic review and thematic synthesis of the qualitative literature. Obesity Reviews. 2021; 23: 1-14. https://doi.org/10.1111/obr.13398.

87. Higgs S, Thomas J. Social influences on eating. Current Opinion in Behavioral Sciences. 2016; 9: 1-6. https://doi.org/10.1016/j.cobeha.2015.10.005.

88. Isaacs A, Halligan J, Neve K, Hawkes C. From healthy food environments to healthy wellbeing environments: Policy insights from a focused ethnography with low-income parents' in england. Health & Place. 2022; 77: 102862. https://doi.org/10.1016/j.healthplace.2022.102862.

89. Shannon J, Reese AM, Ghosh D, Widener MJ, Block DR. More than mapping: Improving methods for studying the geographies of food access. American Journal of Public Health. 2021; 111: 1418-22. https://doi.org/10.2105/ajph.2021.306339.

90. Lytle LA. Measuring the food environment: State of the science. Am J Prev Med. 2009; 36: 134-44. https://doi.org/10.1016/j.amepre.2009.01.018.

91. Lucan SC. Concerning limitations of food-environment research: A narrative review and commentary framed around obesity and diet-related diseases in youth. Journal of the Academy of Nutrition and Dietetics. 2015; 115: 205-12. https://doi.org/10.1016/j.jand.2014.08.019.

92. Lake A. Neighbourhood food environments: Food choice, foodscapes and planning for health. Proceedings of the Nutrition Society. 2018; 77: 1-8. https://doi.org/10.1017/S0029665118000022.

93. Day PL, Pearce J. Obesity-promoting food environments and the spatial clustering of food outlets around schools. American Journal of Preventive Medicine. 2011; 40: 113-21. https://doi.org/10.1016/j.amepre.2010.10.018.

94. Fraser LK, Edwards KL, Cade J, Clarke GP. The geography of fast food outlets: A review. International Journal of Environmental Research and Public Health. 2010; 7: 2290-308. https://doi.org/10.3390/ijerph7052290.

95. Black C, Moon G, Baird J. Dietary inequalities: What is the evidence for the effect of the neighbourhood food environment? Health & Place. 2014; 27: 229-42. https://doi.org/10.1016/j.healthplace.2013.09.015.

96. Lamb KE, Thornton LE, Cerin E, Ball K. Statistical approaches used to assess the equity of access to food outlets: A systematic review. AIMS public health. 2015; 2: 358-401. https://doi.org/10.3934/publichealth.2015.3.358.

97. Ribeiro AI, Fraga S, Severo M, Kelly-Irving M, Delpierre C, Stringhini S, Kivimaki M, Joost S, Guessous I, Severi G, Giles G, Sacerdote C, Vineis P, Barros H, Alberts J, Alenius H, Avendano M, Baglietto L, Baltar V, Bartley M, Barros H, Bellone M, Berger E, Blane D, Bochud M, Candiani G, Carmeli C, Carra L, Castagne R, Chadeau-Hyam M, Cima S, Costa G, Courtin E, Delpierre C, Donkin A, D'Errico A, Dugue P-A, Elliot P, Fagherazzi G, Fiorito G, Fraga S, Gandini M, Gares V, Gerbouin-Rerolle P, Giles G, Goldberg M, Greco D, Hodge A, Kelly-Irving M, Karimi M, Karisola P, Kivimaki M, Laine J, Lang T, Laurent A, Layte R, Lepage B, Lorsch D, Machell G, Mackenbach J, de Mestral C, McCrory C, Miller C, Milne R, Muennig P, Nusselder W, Petrovic D, Pilapil L, Polidoro S, Preisig M, Ribeiro AI, Ricceri F, Recalcati P, Reinhard E, Robinson O, Valverde JR, Saba S, Santegoets F, Simmons T, Severi G, Stringhini S, Tabak A, Terhi V, Tieulent J, Vaccarella S, Vigna-Taglianti F, Vineis P, Vollenweider P, Zins M. Association of neighbourhood disadvantage and individual socioeconomic position with all-cause mortality: A longitudinal multicohort analysis. The Lancet Public Health. 2022; 7: e447-e57. https://doi.org/10.1016/S2468-2667(22)00036-6.

98. Gallo R, Barrett L, Lake A. The food environment within the primary school fringe. British Food Journal. 2014; 116: 1259-75. https://doi.org/10.1108/BFJ-04-2013-0091.

99. Bodicoat DH, Carter P, Comber A, Edwardson C, Gray LJ, Hill S, Webb D, Yates T, Davies MJ, Khunti K. Is the number of fast-food outlets in the neighbourhood related to screen-detected type 2 diabetes mellitus and associated risk factors? Public Health Nutrition. 2014; 18: 1698-705. https://doi.org/10.1017/S1368980014002316.

100. Public Health England. Fast food outlets: Density by local authority in england. 2018. First accessed: 02.11.2022. Available from: https://www.gov.uk/government/publications/fast-food-outlets-density-by-local-authority-in-england.

101. van Rongen S, Poelman MP, Thornton L, Abbott G, Lu M, Kamphuis CBM, Verkooijen K, de Vet E. Neighbourhood fast food exposure and consumption: The mediating role of neighbourhood social norms. International Journal of Behavioral Nutrition and Physical Activity. 2020; 17: 61-70. https://doi.org/10.1186/s12966-020-00969-w.

102. Galobardes B, Shaw M, Lawlor DA, Lynch JW, Davey SG. Indicators of socioeconomic position (part 1). J Epidemiol Community Health. 2006; 60: 7-12. https://doi.org/10.1136/jech.2004.023531.

103. Galobardes B, Shaw M, Lawlor DA, Lynch JW, Davey SG. Indicators of socioeconomic position (part 2). J Epidemiol Community Health. 2006; 60: 95-101. https://doi.org/10.1136/jech.2004.028092. 104. Inglis V, Ball K, Crawford D. Socioeconomic variations in women's diets: What is the role of perceptions of the local food environment? Journal of Epidemiology and Community Health. 2008; 62: 191-7. https://doi.org/10.1136/jech.2006.059253.

105. Galobardes B, Smith GD, Lynch J. Measuring socioeconomic position in health research. British Medical Bulletin. 2007; 81-82: 21-37. https://doi.org/10.1093/bmb/ldm001.

106. Sacks G, Robinson E, Cameron AJ. Issues in measuring the healthiness of food environments and interpreting relationships with diet, obesity and related health outcomes. Current obesity reports. 2019; 8: 98-111. https://doi.org/10.1007/s13679-019-00342-4.

107. Mackenbach JD, Nelissen KGM, Dijkstra SC, Poelman MP, Daams JG, Leijssen JB, Nicolaou M. A systematic review on socioeconomic differences in the association between the food environment and dietary behaviors. Nutrients. 2019; 11: 2215. https://doi.org/10.3390/nu11092215.

108. Miura K, Giskes K, Turrell G. Socioeconomic differences in takeaway food consumption and their contribution to inequalities in dietary intakes. Journal of Epidemiology and Community Health. 2009; 63: 820-6. https://doi.org/10.1136/jech.2008.086504.

109. Burgoine T, Forouhi NG, Griffin SJ, Brage S, Wareham NJ, Monsivais P. Does neighborhood fast-food outlet exposure amplify inequalities in diet and obesity? A cross-sectional study. The American journal of clinical nutrition. 2016; 103: 1540-7. https://doi.org/10.3945/ajcn.115.128132.

110. Pereira MA, Kartashov AI, Ebbeling CB, Van Horn L, Slattery ML, Jacobs DR, Jr., Ludwig DS. Fast-food habits, weight gain, and insulin resistance (the cardia study): 15-year prospective analysis. Lancet. 2005; 365: 36-42. https://doi.org/10.1016/s0140-6736(04)17663-0.

111. Mills S, Adams J, Wrieden W, White M, Brown H. Sociodemographic characteristics and frequency of consuming home-cooked meals and meals from out-of-home sources: Cross-sectional analysis of a population-based cohort study. Public Health Nutrition. 2018; 21: 2255-66. https://doi.org/10.1017/S1368980018000812.

112. Miura K, Giskes K, Turrell G. Socio-economic differences in takeaway food consumption among adults. Public Health Nutrition. 2011; 15: 218-26. https://doi.org/10.1017/S136898001100139X.

113. Cohen DA, Babey SH. Contextual influences on eating behaviours: Heuristic processing and dietary choices. Obesity reviews: an official journal of the International Association for the Study of Obesity. 2012; 13: 766-79. https://doi.org/10.1111/j.1467-789X.2012.01001.x.

114. Sobal J, Bisogni CA. Constructing food choice decisions. Annals of Behavioral Medicine. 2009; 38: 37-46. https://doi.org/10.1007/s12160-009-9124-5.

115. Grunseit AC, Cook AS, Conti J, Gwizd M, Allman-Farinelli M. "Doing a good thing for myself": A qualitative study of young adults' strategies for reducing takeaway food consumption. BMC Public Health. 2019; 19: 525-37. https://doi.org/10.1186/s12889-019-6731-3.

116. Thornton LE, Crawford DA, Ball K. Who is eating where? Findings from the socioeconomic status and activity in women (sesaw) study. Public Health Nutr. 2011; 14: 523-31. https://doi.org/10.1017/s1368980010003022.

117. Ball K, Crawford D. Socioeconomic status and weight change in adults: A review. Social Science & Medicine. 2005; 60: 1987-2010. https://doi.org/10.1016/j.socscimed.2004.08.056.

118. Pechey R, Monsivais P. Socioeconomic inequalities in the healthiness of food choices: Exploring the contributions of food expenditures. Preventive Medicine. 2016; 88: 203-9. https://doi.org/10.1016/j.ypmed.2016.04.012.

119. Dave JM, An LC, Jeffery RW, Ahluwalia JS. Relationship of attitudes toward fast food and frequency of fast-food intake in adults. Obesity. 2009; 17: 1164-70. https://doi.org/10.1038/oby.2009.26.

120. Wills WJ, Danesi G, Kapetanaki AB. Lunchtime food and drink purchasing: Young people's practices, preferences and power within and beyond the school gate. Cambridge Journal of Education. 2016; 46: 195-210. https://doi.org/10.1080/0305764X.2015.1110114.

121. Caraher M, Lloyd S, Mansfield M, Alp C, Brewster Z, Gresham J. Secondary school pupils' food choices around schools in a london borough: Fast food and walls of crisps. Appetite. 2016; 103: 208-20. https://doi.org/10.1016/j.appet.2016.04.016.

122. Jabs J, Devine CM, Bisogni CA, Farrell TJ, Jastran M, Wethington E. Trying to find the quickest way: Employed mothers' constructions of time for food. J Nutr Educ Behav. 2007; 39. https://doi.org/10.1016/j.jneb.2006.08.011.

123. Kant AK, Graubard BI. Eating out in america, 1987–2000: Trends and nutritional correlates. Preventive Medicine. 2004; 38: 243-9. https://doi.org/10.1016/j.ypmed.2003.10.004.

124. Meah A. Fathers, food practices and the circuits of intimacy in families in northern england. Gender, Place & Culture. 2017; 24: 1145-64. https://doi.org/10.1080/0966369X.2017.1372387.

125. Pinho MGM, Mackenbach JD, Charreire H, Oppert J-M, Bárdos H, Rutter H, Compernolle S, Beulens JWJ, Brug J, Lakerveld J. Spatial access to restaurants and grocery stores in relation to frequency of home cooking. International Journal of Behavioral Nutrition and Physical Activity. 2018; 15: 6. https://doi.org/10.1186/s12966-017-0640-6.

126. Adams J, White M. Prevalence and socio-demographic correlates of time spent cooking by adults in the 2005 uk time use survey. Cross-sectional analysis. Appetite. 2015; 92: 185-91. https://doi.org/10.1016/j.appet.2015.05.022.

127. Clifford Astbury C, Foley L, Penney TL, Adams J. How does time use differ between individuals who do more versus less foodwork? A compositional data analysis of time use in the united kingdom time use survey 2014–2015. Nutrients. 2020; 12: 2280. https://doi.org/10.3390/nu12082280.

128. Meah A, Watson M. Saints and slackers: Challenging discourses about the decline of domestic cooking. Sociological Research Online. 2011; 16: 6.html. https://doi.org/10.5153%2Fsro.2341.

129. Grzymisławska M, Puch EA, Zawada A, Grzymisławski M. Do nutritional behaviors depend on biological sex and cultural gender? Advances in clinical and experimental medicine : official organ Wroclaw Medical University. 2020; 29: 165-72. https://doi.org/10.17219/acem/111817.

130. Clayton JA, Tannenbaum C. Reporting sex, gender, or both in clinical research? JAMA. 2016; 316: 1863-4. https://doi.org/10.1001/jama.2016.16405.

131. Kress AC, Asberry A, Taillepierre JD, Johns MM, Tucker P. Collection of data on sex, sexual orientation, and gender identity by u.S. Public health data and monitoring systems, 2015-2018. International Journal of Environmental Research and Public Health. 2021; 18. https://doi.org/10.3390/ijerph182212189.

132. Gilbert PA, Khokhar S. Changing dietary habits of ethnic groups in europe and implications for health. Nutrition Reviews. 2008; 66: 203-15. https://doi.org/10.1111/j.1753-4887.2008.00025.x.

133. Bhopal RS. Research agenda for tackling inequalities related to migration and ethnicity in europe. Journal of Public Health. 2012; 34: 167-73. https://doi.org/10.1093/pubmed/fds004.

134. Sharma S, Cade J, Landman J, Cruickshank JK. Assessing the diet of the british africancaribbean population: Frequency of consumption of foods and food portion sizes. International Journal of Food Sciences and Nutrition. 2002; 53: 439-44. https://doi.org/10.1080/0963748021000044787.

135. Jang M, Vorderstrasse A. Socioeconomic status and racial or ethnic differences in participation: Web-based survey. JMIR research protocols. 2019; 8: e11865. https://doi.org/10.2196/11865.

136. Dekker LH, Nicolaou M, van Dam RM, de Vries JHM, de Boer EJ, Brants HAM, Beukers MH, Snijder MB, Stronks K. Socio-economic status and ethnicity are independently associated with dietary patterns: The helius-dietary patterns study. Food & Nutrition Research. 2015; 59: 26317. https://doi.org/10.3402/fnr.v59.26317.

137. Shareck M, Benmarhnia T, Berger N, Smith NR, Lewis D, Cummins S. Does the neighborhood food environment contribute to ethnic inequalities in fast-food intake? Findings from the oriel study. Preventive Medicine Reports. 2019; 16: 1-7. https://doi.org/10.1016/j.pmedr.2019.100998.

138. Lawrence JM, Devlin E, Macaskill S, Kelly M, Chinouya M, Raats MM, Barton KL, Wrieden WL, Shepherd R. Factors that affect the food choices made by girls and young women, from minority ethnic groups, living in the uk. Journal of human nutrition and dietetics : the official journal of the British Dietetic Association. 2007; 20: 311-9. https://doi.org/10.1111/j.1365-277X.2007.00766.x.

139. Statista. Internet usage worldwide. 2022. First accessed: 23.08.2022. Available from: https://www.statista.com/study/12322/global-internet-usage-statista-dossier/.

140. Goodman MK, Jaworska S. Mapping digital foodscapes: Digital food influencers and the grammars of good food. Geoforum. 2020; 117: 183-93. https://doi.org/10.1016/j.geoforum.2020.09.020.

141. Stephens J, Miller H, Militello L. Food delivery apps and the negative health impacts for americans. Frontiers in Nutrition. 2020; 7: 1-2. https://doi.org/10.3389/fnut.2020.00014.

142. Maimaiti M, Zhao X, Jia M, Ru Y, Zhu S. How we eat determines what we become: Opportunities and challenges brought by food delivery industry in a changing world in china. European Journal of Clinical Nutrition. 2018; 72: 1282-6. https://doi.org/10.1038/s41430-018-0191-1. 143. United Nations System Standing Committee on Nutrition. Unscn nutrition 45 – nutrition in a digital world. 2020. First accessed: 15.09.2020. Available from: https://www.unscn.org/uploads/web/news/UNSCN-Nutrition-45-WEB.pdf.

144. Bates S, Reeve B, Trevena H. A narrative review of online food delivery in australia: Challenges and opportunities for public health nutrition policy. Public Health Nutrition. 2020; Epub ahead of print: 1-11. https://doi.org/10.1017/S1368980020000701.

145. World Health Organization: Regional Office for Europe. Digital food environments: Factsheet. 2021. First accessed: 07.03.2022. Available from: https://www.euro.who.int/en/healthtopics/disease-prevention/nutrition/publications/2021/digital-food-environments-factsheet-2021.

146. Feldman Z, Goodman MK. Digital food culture, power and everyday life. European Journal of Cultural Studies. 2021; 24: 1227-42. https://doi.org/10.1177/13675494211055501.

147. Granheim SI, Løvhaug AL, Terragni L, Torheim LE, Thurston M. Mapping the digital food environment: A systematic scoping review. Obesity Reviews. 2022; 23: e13356. https://doi.org/10.1111/obr.13356.

148. Fernandez MA, Raine KD. Digital food retail: Public health opportunities. Nutrients. 2021; 13: 3789. https://doi.org/10.3390/nu13113789.

149. Boyland E, Thivel D, Mazur A, Ring-Dimitriou S, Frelut ML, Weghuber D. Digital food marketing to young people: A substantial public health challenge. Annals of Nutrition and Metabolism. 2020; 76: 6-9. https://doi.org/10.1159/000506413.

150. Sadeghirad B, Duhaney T, Motaghipisheh S, Campbell NRC, Johnston BC. Influence of unhealthy food and beverage marketing on children's dietary intake and preference: A systematic review and meta-analysis of randomized trials. Obesity Reviews. 2016; 17: 945-59. https://doi.org/10.1111/obr.12445.

151. Jia SS, Gibson AA, Ding D, Allman-Farinelli M, Phongsavan P, Redfern J, Partridge SR. Perspective: Are online food delivery services emerging as another obstacle to achieving the 2030 united nations sustainable development goals? Frontiers in Nutrition. 2022; 9: 1-8. https://doi.org/10.3389/fnut.2022.858475.

152. Richardson S, Lawrence K, Schoenthaler AM, Mann D. A framework for digital health equity. npj Digital Medicine. 2022; 5: 1-6. https://doi.org/10.1038/s41746-022-00663-0.

153. Khandpur N, Zatz LY, Bleich SN, Taillie LS, Orr JA, Rimm EB, Moran AJ. Supermarkets in cyberspace: A conceptual framework to capture the influence of online food retail environments on consumer behavior. International Journal of Environmental Research and Public Health. 2020; 17: 8639. https://doi.org/10.3390/ijerph17228639.

154. Meemken E-M, Bellemare MF, Reardon T, Vargas CM. Research and policy for the fooddelivery revolution. Science. 2022; 377: 810-3. https://doi.org/10.1126/science.abo2182.

155. Backman P. Delivery the disruptor. 2020. First accessed: 29.03.2020. Available from: https://www.peterbackmanfs.com/delivery-the-disruptor.

156. Poelman MP, Thornton L, Zenk SN. A cross-sectional comparison of meal delivery options in three international cities. European Journal of Clinical Nutrition. 2020; 74: 1465-73. https://doi.org/10.1038/s41430-020-0630-7. 157. Goffe L, Uwamahoro NS, Dixon CJ, Blain AP, Danielsen J, Kirk D, Adamson AJ. Supporting a healthier takeaway meal choice: Creating a universal health rating for online takeaway fast-food outlets. International Journal of Environmental Research and Public Health. 2020; 17: 9260. https://doi.org/10.3390/ijerph17249260.

158. Hossain F, Adelaja AO. Consumers' interest in alternative food delivery systems: Results from a consumer survey in new jersey. Journal of Food Distribution Research. 2000; 31: 49-67. https://doi.org/10.22004/ag.econ.27754.

159. Wang C, Korai A, Jia SS, Allman-Farinelli M, Chan V, Roy R, Raeside R, Phongsavan P, Redfern J, Gibson AA, Partridge SR. Hunger for home delivery: Cross-sectional analysis of the nutritional quality of complete menus on an online food delivery platform in australia. Nutrients. 2021; 13: 905. https://doi.org/10.3390/nu13030905.

160. Akram U, Ansari AR, Fu G, Junaid M. Feeling hungry? Let's order through mobile! Examining the fast food mobile commerce in china. Journal of Retailing and Consumer Services. 2020; 56: 102142. https://doi.org/10.1016/j.jretconser.2020.102142.

161. Freeman B, Kelly B, Baur L, Chapman K, Chapman S, Gill T, King L. Digital junk: Food and beverage marketing on facebook. Am J Public Health. 2014; 104: e56-64. https://doi.org/10.2105/ajph.2014.302167.

162. Horta PM, Souza JdPM, Rocha LL, Mendes LL. Digital food environment of a brazilian metropolis: Food availability and marketing strategies used by delivery apps. Public Health Nutrition. 2020; 24: 544-8. https://doi.org/10.1017/S1368980020003171.

163. Just Eat. Our markets. 2022. First accessed: 31.10.2022. Available from: https://www.justeattakeaway.com/our-markets.

164. Wong BLH, Maaß L, Vodden A, van Kessel R, Sorbello S, Buttigieg S, Odone A. The dawn of digital public health in europe: Implications for public health policy and practice. The Lancet Regional Health - Europe. 2022; 14: 100316. https://doi.org/10.1016/j.lanepe.2022.100316.

165. Partridge SR, Gibson AA, Roy R, Malloy JA, Raeside R, Jia SS, Singleton AC, Mandoh M, Todd AR, Wang T, Halim NK, Hyun K, Redfern J. Junk food on demand: A cross-sectional analysis of the nutritional quality of popular online food delivery outlets in australia and new zealand. Nutrients. 2020; 12: 3107. https://doi.org/10.3390/nu12103107.

166. Brar K, Minaker LM. Geographic reach and nutritional quality of foods available from mobile online food delivery service applications: Novel opportunities for retail food environment surveillance. BMC Public Health. 2021; 21: 458. https://doi.org/10.1186/s12889-021-10489-2.

167. Norriss D, Crossin R, Curl A, Bidwell S, Clark E, Pocock T, Gage R, McKerchar C. Food outlet access and the healthiness of food available 'on-demand' via meal delivery apps in new zealand. Nutrients. 2022; 14: 4228. https://doi.org/10.3390/nu14204228.

168. Horta PM, Matos JDP, Mendes LL. Food promoted on an online food delivery platform in a brazilian metropolis during the coronavirus disease (covid-19) pandemic: A longitudinal analysis. Public Health Nutrition. 2022; 25: 1336-45. https://doi.org/10.1017/S1368980022000489.

169. Li C, Mirosa M, Bremer P. Review of online food delivery platforms and their impacts on sustainability. Sustainability. 2020; 12: 5528. https://doi.org/10.3390/su12145528.

170. Jia P, Luo M, Li Y, Zheng J-S, Xiao Q, Luo J. Fast-food restaurant, unhealthy eating, and childhood obesity: A systematic review and meta-analysis. Obesity Reviews. 2019; Advance online publication: 1-27. https://doi.org/10.1111/obr.12944.

171. World Health Organization. Global health observatory (gho) data. 2020. First accessed: 29.03.2020. Available from: https://www.who.int/gho/ncd/risk_factors/overweight_text/en/.

172. Imamura F, Micha R, Khatibzadeh S, Fahimi S, Shi P, Powles J, Mozaffarian D. Dietary quality among men and women in 187 countries in 1990 and 2010: A systematic assessment. The Lancet Global Health. 2015; 3: 132-42. https://doi.org/10.1016/S2214-109X(14)70381-X.

173. Finegood DT, Merth TDN, Rutter H. Implications of the foresight obesity system map for solutions to childhood obesity. Obesity. 2010; 18: S13-S6. https://doi.org/10.1038/oby.2009.426.

174. Rutter H. The single most important intervention to tackle obesity.... International Journal of Public Health. 2012; 57: 657-8. https://doi.org/10.1007/s00038-012-0385-6.

175. Jaworowska A, Toni MB, Rachel L, Catherine T, Matthew A, Leonard S, Ian GD. Nutritional composition of takeaway food in the uk. Nutrition & Food Science. 2014; 44: 414-30. https://doi.org/10.1108/NFS-08-2013-0093.

176. Scourboutakos MJ, L'Abbé MR. Restaurant menus: Calories, caloric density, and serving size. American Journal of Preventive Medicine. 2012; 43: 249-55. https://doi.org/10.1016/j.amepre.2012.05.018.

177. Todd JE, Mancino L, Lin B-H. The impact of food away from home on adult diet quality. USDA-ERS Economic Research Report Paper. 2010; 2020: 1-24. https://doi.org/10.22004/ag.econ.58298.

178. Yeo VCS, Goh S-K, Rezaei S. Consumer experiences, attitude and behavioral intention toward online food delivery (ofd) services. Journal of Retailing and Consumer Services. 2017; 35: 150-62. https://doi.org/10.1016/j.jretconser.2016.12.013.

179. Deliveroo. About deliveroo. 2019. First accessed: 06.10.2020. Available from: https://deliveroo.co.uk/about-us.

180. Uber Eats. When and where is uber eats available? 2019. First accessed: 17.12.19. Available from: https://help.uber.com/ubereats/article/when-and-where-is-uber-eats-available-?nodeId=3f8de61e-09dd-4844-afb2-749c9ffc65a8.

181. Just Eat. 2019 half year results. 2019. First accessed: 23.09.2019. Available from: https://jeweb-11431-s3.s3.eu-west2.amazonaws.com/application/files/3615/6455/1992/Just_Eat_plc__2019_Half_Year_Results_RNS.pdf.

182. Grubhub. What is grubhub? 2020. First accessed: 06.04.2020. Available from: https://about.grubhub.com/about-us/what-is-grubhub/default.aspx.

183. Statista. Online food delivery report 2019. 2019. First accessed: 21.02.2020. Available from: https://www.statista.com/study/40457/food-delivery/.

184. Hammond D, White CM, Rynard VL, Vanderlee L. International food policy study: Technical report - 2018 survey (wave 2). 2019. First accessed: 01.10.2019. Available from: https://foodpolicystudy.com/methods/.
185. Kant AK, Whitley MI, Graubard BI. Away from home meals: Associations with biomarkers of chronic disease and dietary intake in american adults, nhanes 2005-2010. International Journal of Obesity. 2015; 39: 820-7. https://doi.org/10.1038/ijo.2014.183.

186. Wiggers D, Vanderlee L, White CM, Reid JL, Minaker L, Hammond D. Food sources among young people in five major canadian cities. Canadian Journal of Public Health. 2018; 109: 506-15. https://doi.org/10.17269/s41997-018-0083-0.

187. Meah A, Jackson P. Convenience as care: Culinary antinomies in practice. Environment and Planning A: Economy and Space. 2017; 49: 2065-81. https://doi.org/10.1177/0308518X17717725.

188. Nielsen MW, Stefanick ML, Peragine D, Neilands TB, Ioannidis JPA, Pilote L, Prochaska JJ, Cullen MR, Einstein G, Klinge I, LeBlanc H, Paik HY, Schiebinger L. Gender-related variables for health research. Biology of Sex Differences. 2021; 12: 23. https://doi.org/10.1186/s13293-021-00366-3.

189. Galobardes B, Morabia A, Bernstein MS. Diet and socioeconomic position: Does the use of different indicators matter? International Journal of Epidemiology. 2001; 30: 334-40. https://doi.org/10.1093/ije/30.2.334.

190. World Health Organization. Body mass index - bmi. 2010. First accessed: 20.05.2019. Available from: https://www.euro.who.int/en/health-topics/disease-prevention/nutrition/a-healthy-lifestyle/body-mass-index-bmi.

191. Pedersen AB, Mikkelsen EM, Cronin-Fenton D, Kristensen NR, Pham TM, Pedersen L, Petersen I. Missing data and multiple imputation in clinical epidemiological research. Clin Epidemiol. 2017; 9: 157-66. https://doi.org/10.2147/CLEP.S129785.

192. Langkamp DL, Lehman A, Lemeshow S. Techniques for handling missing data in secondary analyses of large surveys. Academic Pediatrics. 2010; 10: 205-10. https://doi.org/10.1016/j.acap.2010.01.005.

193. Sperandei S. Understanding logistic regression analysis. Biochemia Medica. 2014; 24: 12-8. https://doi.org/10.11613/BM.2014.003.

194. Greenland S. Modeling and variable selection in epidemiologic analysis. Am J Public Health. 1989; 79: 340-9. https://doi.org/10.2105/ajph.79.3.340.

195. McGowan L, Caraher M, Raats M, Lavelle F, Hollywood L, McDowell D, Spence M, McCloat A, Mooney E, Dean M. Domestic cooking and food skills: A review. Crit Rev Food Sci Nutr. 2017; 57: 2412-31. https://doi.org/10.1080/10408398.2015.1072495.

196. Virudachalam S, Long J, Harhay M, Polsky D, Feudtner C. Prevalence and patterns of cooking dinner at home in the USA: National health and nutrition examination survey (nhanes) 2007–2008. Public Health Nutr. 2014; 17: 1022-30. https://doi.org/10.1017/s1368980013002589.

197. Taillie LS. Who's cooking? Trends in us home food preparation by gender, education, and race/ethnicity from 2003 to 2016. Nutrition Journal. 2018; 17: 41-50. https://doi.org/10.1186/s12937-018-0347-9.

198. Clifford Astbury C, Penney TL, Adams J. Comparison of individuals with low versus high consumption of home-prepared food in a group with universally high dietary quality: A cross-sectional analysis of the uk national diet & nutrition survey (2008–2016). International Journal of

Behavioral Nutrition and Physical Activity. 2019; 16: 9-24. https://doi.org/10.1186/s12966-019-0768-7.

199. Ross PT, Hart-Johnson T, Santen SA, Zaidi NLB. Considerations for using race and ethnicity as quantitative variables in medical education research. Perspectives on Medical Education. 2020; 9: 318-23. https://doi.org/10.1007/s40037-020-00602-3.

200. Kern DM, Auchincloss AH, Stehr MF, Roux AVD, Moore LV, Kanter GP, Robinson LF. Neighborhood prices of healthier and unhealthier foods and associations with diet quality: Evidence from the multi-ethnic study of atherosclerosis. Int J Environ Res Public Health. 2017; 14: 1-14. https://doi.org/10.3390/ijerph14111394.

201. Liese AD, Ma X, Hutto B, Sharpe PA, Bell BA, Wilcox S. Food shopping and acquisition behaviors in relation to bmi among residents of low-income communities in south carolina. Int J Environ Res Public Health. 2017; 14: 1-15. https://doi.org/10.3390/ijerph14091075.

202. Chen K, Chan AHS. A review of technology acceptance by older adults. Gerontechnology. 2011; 10: 1-12. https://doi.org/10.4017/gt.2011.10.01.006.00.

203. Jabs J, Devine CM. Time scarcity and food choices: An overview. Appetite. 2006; 47: 196-204. https://doi.org/10.1016/j.appet.2006.02.014.

204. Warin M, Jay B, Zivkovic T. "Ready-made" assumptions: Situating convenience as care in the australian obesity debate. Food and Foodways. 2019; 27: 273-95. https://doi.org/10.1080/07409710.2019.1673004.

205. Penney TL, Jones NRV, Adams J, Maguire ER, Burgoine T, Monsivais P. Utilization of awayfrom-home food establishments, dietary approaches to stop hypertension dietary pattern, and obesity. American Journal of Preventive Medicine. 2017; 53: 155-63. https://doi.org/10.1016/j.amepre.2017.06.003.

206. Rivera JA, Barquera S, Gonzalez-Cossio T, Olaiz G, Sepulveda J. Nutrition transition in mexico and in other latin american countries. Nutrition Reviews. 2004; 62: S149-S57. https://doi.org/10.1111/j.1753-4887.2004.tb00086.x.

207. PEW Research Center. Internet penetration rates are high in north america, europe and parts of the asia-pacific 2018. 2018. First accessed: 24.01.2020. Available from: https://www.pewresearch.org/global/2018/06/19/social-media-use-continues-to-rise-in-developing-countries-but-plateaus-across-developed-ones/pg_2018-06-19_global-tech_0-01/.

208. Krumpal I. Determinants of social desirability bias in sensitive surveys: A literature review. Quality & Quantity. 2013; 47: 2025-47. https://doi.org/10.1007/s11135-011-9640-9.

209. Cook A, Pryer J, Shetty P. The problem of accuracy in dietary surveys. Analysis of the over 65 uk national diet and nutrition survey. Journal of Epidemiology and Community Health. 2000; 54: 611-6. https://doi.org/10.1136/jech.54.8.611.

210. Burningham K, Venn S. "Two quid, chicken and chips, done": Understanding what makes for young people's sense of living well in the city through the lens of fast food consumption. Local Environment. 2021; 27: 1-17. https://doi.org/10.1080/13549839.2021.2001797.

211. Statista. Eservices report 2022 - online food delivery. 2022. First accessed: 10.08.2022. Available from: https://www.statista.com/study/40457/food-delivery/.

212. World Health Organization: Regional Office for Europe. Slide to order: A food systems approach to meals delivery apps: Who european office for the prevention and control of noncommunicable diseases. 2021. First accessed: 08.03.2022. Available from: https://apps.who.int/iris/handle/10665/350121.

213. Keeble M, Adams J, Bishop TRP, Burgoine T. Socioeconomic inequalities in food outlet access through an online food delivery service in england: A cross-sectional descriptive analysis. Applied Geography. 2021; 133: 102498. https://doi.org/10.1016/j.apgeog.2021.102498.

214. Keeble M, Adams J, Vanderlee L, Hammond D, Burgoine T. Associations between online food outlet access and online food delivery service use amongst adults in the uk: A cross-sectional analysis of linked data. BMC Public Health. 2021; 21: 1968. https://doi.org/10.1186/s12889-021-11953-9.

215. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (coreq): A 32-item checklist for interviews and focus groups. International Journal for Quality in Health Care. 2007; 19: 349-57. https://doi.org/10.1093/intqhc/mzm042.

216. Milne J, Oberle K. Enhancing rigor in qualitative description: A case study. Journal of wound, ostomy, and continence nursing : official publication of The Wound, Ostomy and Continence Nurses Society. 2005; 32: 413-20. https://doi.org/10.1097/00152192-200511000-00014.

217. Bradshaw C, Atkinson S, Doody O. Employing a qualitative description approach in health care research. Global Qualitative Nursing Research. 2017; 4: 1-8. https://doi.org/10.1177/2333393617742282.

218. Keeble M, Adams J, Sacks G, Vanderlee L, White CM, Hammond D, Burgoine T. Use of online food delivery services to order food prepared away-from-home and associated sociodemographic characteristics: A cross-sectional, multi-country analysis. International Journal of Environmental Research and Public Health. 2020; 17: 5190. https://doi.org/10.3390/ijerph17145190.

219. Dana LM, Hart E, McAleese A, Bastable A, Pettigrew S. Factors associated with ordering food via online meal ordering services. Public Health Nutrition. 2021; 24: 5704-9. https://doi.org/10.1017/S1368980021001294.

220. O'Connor A, Jackson L, Goldsmith L, Skirton H. Can i get a retweet please? Health research recruitment and the twittersphere. Journal of Advanced Nursing. 2014; 70: 599-609. https://doi.org/10.1111/jan.12222.

221. Hokke S, Hackworth NJ, Bennetts SK, Nicholson JM, Keyzer P, Lucke J, Zion L, Crawford SB. Ethical considerations in using social media to engage research participants: Perspectives of australian researchers and ethics committee members. Journal of Empirical Research on Human Research Ethics. 2020; 15: 12-27. https://doi.org/10.1177/1556264619854629.

222. Gelinas L, Pierce R, Winkler S, Cohen IG, Lynch HF, Bierer BE. Using social media as a research recruitment tool: Ethical issues and recommendations. The American Journal of Bioethics. 2017; 17: 3-14. https://doi.org/10.1080/15265161.2016.1276644.

223. Office for National Statistics. 2011 census: Key statistics for england and wales: March 2011. 2011. First accessed: 20.01.2021. Available from:

https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populatione stimates/bulletins/2011censuskeystatisticsforenglandandwales/2012-12-11#qualifications.

224. Okumus B. A qualitative investigation of millennials' healthy eating behavior, food choices, and restaurant selection. Food, Culture & Society. 2021; 24: 509-24. https://doi.org/10.1080/15528014.2021.1882168.

225. Braun V, Clarke V. To saturate or not to saturate? Questioning data saturation as a useful concept for thematic analysis and sample-size rationales. Qualitative Research in Sport, Exercise and Health. 2019; 13: 1-16. https://doi.org/10.1080/2159676X.2019.1704846.

226. Saunders B, Sim J, Kingstone T, Baker S, Waterfield J, Bartlam B, Burroughs H, Jinks C. Saturation in qualitative research: Exploring its conceptualization and operationalization. Quality and Quantity. 2018; 52: 1893-907. https://doi.org/10.1007/s11135-017-0574-8.

227. Keeble M, Burgoine T, White M, Summerbell C, Cummins S, Adams J. Planning and public health professionals' experiences of using the planning system to regulate hot food takeaway outlets in england: A qualitative study. Health & Place. 2021; 67: 102305. https://doi.org/10.1016/j.healthplace.2020.102305.

228. Braun V, Clarke V. Can i use ta? Should i use ta? Should i not use ta? Comparing reflexive thematic analysis and other pattern-based qualitative analytic approaches. Counselling and Psychotherapy Research. 2020; 21: 1-11. https://doi.org/10.1002/capr.12360.

229. Braun V, Clarke V. Reflecting on reflexive thematic analysis. Qualitative Research in Sport, Exercise and Health. 2019; 11: 589-97. https://doi.org/10.1080/2159676X.2019.1628806.

230. Willis DG, Sullivan-Bolyai S, Knafl K, Cohen MZ. Distinguishing features and similarities between descriptive phenomenological and qualitative description research. Western Journal of Nursing Research. 2016; 38: 1185-204. https://doi.org/10.1177/0193945916645499.

231. Barbour RS. Checklists for improving rigour in qualitative research: A case of the tail wagging the dog? BMJ. 2001; 322: 1115-7. https://doi.org/10.1136/bmj.322.7294.1115.

232. Bazeley P. Integrating analyses in mixed methods research. London: SAGE Publications; 2017.

233. Devine CM. A life course perspective: Understanding food choices in time, social location, and history. Journal of Nutrition Education and Behavior. 2005; 37: 121-8. https://doi.org/10.1016/S1499-4046(06)60266-2.

234. Verain MCD, van den Puttelaar J, Zandstra EH, Lion R, de Vogel-van den Bosch J, Hoonhout HCM, Onwezen MC. Variability of food choice motives: Two dutch studies showing variation across meal moment, location and social context. Food Quality and Preference. 2021; 98: 1-12. https://doi.org/10.1016/j.foodqual.2021.104505.

235. McPhail D, Chapman GE, Beagan BL. "Too much of that stuff can't be good": Canadian teens, morality, and fast food consumption. Social Science & Medicine. 2011; 73: 301-7. https://doi.org/10.1016/j.socscimed.2011.05.022.

236. Riesenberg D, Backholer K, Zorbas C, Sacks G, Paix A, Marshall J, Blake MR, Bennett R, Peeters A, Cameron AJ. Price promotions by food category and product healthiness in an australian supermarket chain, 2017–2018. American Journal of Public Health. 2019; 109: 1434-9. https://doi.org/10.2105/AJPH.2019.305229.

237. Hawkes C. Sales promotions and food consumption. Nutrition Reviews. 2009; 67: 333-42. https://doi.org/10.1111/j.1753-4887.2009.00206.x.

238. Widener MJ, Ren L, Astbury CC, Smith LG, Penney TL. An exploration of how meal preparation activities relate to self-rated time pressure, stress, and health in canada: A time use approach. SSM - Population Health. 2021; 15: 100818. https://doi.org/10.1016/j.ssmph.2021.100818.

239. Ziauddeen N, Almiron-Roig E, Penney TL, Nicholson S, Kirk SFL, Page P. Eating at food outlets and "on the go" is associated with less healthy food choices in adults: Cross-sectional data from the uk national diet and nutrition survey rolling programme (2008-2014). Nutrients. 2017; 9: 1315-30. https://doi.org/10.3390/nu9121315.

240. UK Government. Restricting promotions of products high in fat, sugar and salt by location and by price: Government response to public consultation. 2020. First accessed: 04.03.2022. Available from: https://www.gov.uk/government/consultations/restricting-promotions-of-foodand-drink-that-is-high-in-fat-sugar-and-salt/outcome/restricting-promotions-of-products-highin-fat-sugar-and-salt-by-location-and-by-price-government-response-to-publicconsultation#references.

241. Bennett R, Zorbas C, Huse O, Peeters A, Cameron AJ, Sacks G, Backholer K. Prevalence of healthy and unhealthy food and beverage price promotions and their potential influence on shopper purchasing behaviour: A systematic review of the literature. Obesity Reviews. 2020; 21: e12948. https://doi.org/10.1111/obr.12948.

242. Keeble M, Adams J, Burgoine T. Investigating experiences of frequent online food delivery service use: A qualitative study in uk adults. BMC Public Health. 2022; 22: 1365. https://doi.org/10.1186/s12889-022-13721-9.

243. Nowell LS, Norris JM, White DE, Moules NJ. Thematic analysis: Striving to meet the trustworthiness criteria. International Journal of Qualitative Methods. 2017; 16: 1609406917733847. https://doi.org/10.1177/1609406917733847.

244. Krefting L. Rigor in qualitative research: The assessment of trustworthiness. American Journal of Occupational Therapy. 1991; 45: 214-22. https://doi.org/10.5014/ajot.45.3.214.

245. Burgoine T, Forouhi NG, Griffin SJ, Wareham NJ, Monsivais P. Associations between exposure to takeaway food outlets, takeaway food consumption, and body weight in cambridgeshire, uk: Population based, cross sectional study. BMJ. 2014; 348: g1464. https://doi.org/10.1136/bmj.g1464.

246. Granheim SI, Opheim E, Terragni L, Torheim LE, Thurston M. Mapping the digital food environment: A scoping review protocol. BMJ Open. 2020; 10: e036241. https://doi.org/10.1136/bmjopen-2019-036241.

247. Halkier B. Normalising convenience food? Food, Culture & Society. 2017; 20: 133-51. https://doi.org/10.1080/15528014.2016.1243768.

248. Egger G, Swinburn B. An "ecological" approach to the obesity pandemic. BMJ. 1997; 315: 477-80. https://doi.org/10.1136/bmj.315.7106.477.

249. Kremers SP, de Bruijn G-J, Visscher TL, van Mechelen W, de Vries NK, Brug J. Environmental influences on energy balance-related behaviors: A dual-process view. International Journal of Behavioral Nutrition and Physical Activity. 2006; 3: 9. https://doi.org/10.1186/1479-5868-3-9.

250. Aggarwal A, Rehm CD, Monsivais P, Drewnowski A. Importance of taste, nutrition, cost and convenience in relation to diet quality: Evidence of nutrition resilience among us adults using

national health and nutrition examination survey (nhanes) 2007-2010. Prev Med. 2016; 90: 184-92. https://doi.org/10.1016/j.ypmed.2016.06.030.

251. Hammond D, White CM, Rynard VL, Vanderlee L. International food policy study technical report - 2019 survey (wave 3). 2020. First accessed: 29.04.2021. Available from: http://foodpolicystudy.com/METHODS/.

252. Statista. Food delivery and takeaway market in the united kingdom (uk) - statistics & facts. 2020. First accessed: 20.02.2021. Available from: https://www.statista.com/topics/4679/food-delivery-and-takeaway-market-in-the-united-kingdom-uk/.

253. Just Eat. Just eat q1 2020 trading update. 2020. First accessed: 13.04.2021. Available from: https://corporate.takeaway.com/media/press-releases/.

254. Just Eat. Just eat takeaway.Com annual report 2020. 2021. First accessed: 20.04.2021. Available from: https://www.justeattakeaway.com/investors/annual-reports/.

255. Beacon Dodsworth. All you need to know about postcodes but were afraid to ask. 2020. First accessed: 11.07.2022. Available from: https://beacon-dodsworth.co.uk/blog/all-you-need-to-know-about-postcodes-but-were-afraid-to-ask/.

256. Burke MA, Carman KG. You can be too thin (but not too tall): Social desirability bias in selfreports of weight and height. Economics & Human Biology. 2017; 27: 198-222. https://doi.org/10.1016/j.ehb.2017.06.002.

257. Rummo PE, Guilkey DK, Ng SW, Meyer KA, Popkin BM, Reis JP, Shikany JM, Gordon-Larsen P. Does unmeasured confounding influence associations between the retail food environment and body mass index over time? The coronary artery risk development in young adults (cardia) study. International Journal of Epidemiology. 2017; 46: 1456-64. https://doi.org/10.1093/ije/dyx070.

258. Penney LT, Burgoine T, Monsivais P. Relative density of away from home food establishments and food spend for 24,047 households in england: A cross-sectional study. International Journal of Environmental Research and Public Health. 2018; 15: 2821. https://doi.org/10.3390/ijerph15122821.

259. Burgoine T, Harrison F. Comparing the accuracy of two secondary food environment data sources in the uk across socio-economic and urban/rural divides. International Journal of Health Geographics. 2013; 12: 1-8. https://doi.org/10.1186/1476-072X-12-2.

260. Ordnance Survey. Points of interest. 2020. First accessed: 20.11.2020. Available from: https://www.ordnancesurvey.co.uk/business-government/products/points-of-interest.

261. PointX. Pointx classification scheme. 2006. First accessed: 20.11.2020. Available from: http://www.pointx.co.uk/downloads/Classification2.0.pdf.

262. Ordnance Survey. Points of interest - technical specification. 2019. First accessed: 11.07.2022. Available from: https://www.ordnancesurvey.co.uk/documents/product-support/tech-spec/points-of-interest-technical-specification.pdf.

263. Doogal. Doogal. 2022. First accessed: 24.05.2022. Available from: https://www.doogal.co.uk/.

264. GeoConvert. Geoconvert. 2022. First accessed: 24.05.2022. Available from: http://geoconvert.mimas.ac.uk/.

265. Smith G, Gidlow C, Davey R, Foster C. What is my walking neighbourhood? A pilot study of english adults' definitions of their local walking neighbourhoods. International Journal of Behavioral Nutrition and Physical Activity. 2010; 7: 34-46. https://doi.org/10.1186/1479-5868-7-34.

266. Liu JL, Han B, Cohen DA. Beyond neighborhood food environments: Distance traveled to food establishments in 5 us cities, 2009-2011. Preventing chronic disease. 2015; 12: 1-9. https://doi.org/10.5888/pcd12.150065.

267. Dare S, Mackay DF, Pell JP. Relationship between smoking and obesity: A cross-sectional study of 499,504 middle-aged adults in the uk general population. PloS one. 2015; 10: e0123579. https://doi.org/10.1371/journal.pone.0123579.

268. Elliston KG, Ferguson SG, Schuz N, Schuz B. Situational cues and momentary food environment predict everyday eating behavior in adults with overweight and obesity. Health psychology : official journal of the Division of Health Psychology, American Psychological Association. 2017; 36: 337-45. https://doi.org/10.1037/hea0000439.

269. Boelsen-Robinson T, Backholer K, Peeters A. Digital marketing of unhealthy foods to australian children and adolescents. Health promotion international. 2015; 31: 523-33. https://doi.org/10.1093/heapro/dav008.

270. Brown JA, Ferdinands AR, Prowse R, Reynard D, Raine KD, Nykiforuk CIJ. Seeing the food swamp for the weeds: Moving beyond food retail mix in evaluating young people's food environments. SSM - Population Health. 2021: 100803. https://doi.org/10.1016/j.ssmph.2021.100803.

271. Stamatakis E, Coombs N, Rowlands A, Shelton N, Hillsdon M. Objectively-assessed and selfreported sedentary time in relation to multiple socioeconomic status indicators among adults in england: A cross-sectional study. BMJ Open. 2014; 4: e006034. https://doi.org/10.1136/bmjopen-2014-006034.

272. Yau A, Adams J, Boyland EJ, Burgoine T, Cornelsen L, de Vocht F, Egan M, Er V, Lake AA, Lock K, Mytton O, Petticrew M, Thompson C, White M, Cummins S. Sociodemographic differences in self-reported exposure to high fat, salt and sugar food and drink advertising: A cross-sectional analysis of 2019 uk panel data. BMJ Open. 2021; 11: e048139. https://doi.org/10.1136/bmjopen-2020-048139.

273. Fulkerson JA. Fast food in the diet: Implications and solutions for families. Physiology & Behavior. 2018; 193: 252-6. https://doi.org/10.1016/j.physbeh.2018.04.005.

274. Johns N, Edwards JSA, Hartwell HJ. Menu choice: Satisfaction or overload? Journal of Culinary Science & Technology. 2013; 11: 275-85. https://doi.org/10.1080/15428052.2013.798564.

275. Scheibehenne B, Greifeneder R, Todd PM. Can there ever be too many options? A metaanalytic review of choice overload. Journal of Consumer Research. 2010; 37: 409-25. https://doi.org/10.1086/651235.

276. Yau A, Adams J, White M, Nicolaou M. Differences in diet quality and socioeconomic patterning of diet quality across ethnic groups: Cross-sectional data from the helius dietary patterns study. European Journal of Clinical Nutrition. 2020; 74: 387-96. https://doi.org/10.1038/s41430-019-0463-4. 277. Bagwell S. Healthier catering initiatives in london, uk: An effective tool for encouraging healthier consumption behaviour? Critical Public Health. 2014; 24: 35-46. https://doi.org/10.1080/09581596.2013.769670.

278. Hillier-Brown F, Lloyd S, Muhammad L, Summerbell C, Goffe L, Hildred N, Adams J. Feasibility and acceptability of a takeaway masterclass aimed at encouraging healthier cooking practices and menu options in takeaway food outlets. Public Health Nutrition. 2019; 22: 2268-78. https://doi.org/10.1017/s1368980019000648.

279. Wilkins E, Morris M, Radley D, Griffiths C. Methods of measuring associations between the retail food environment and weight status: Importance of classifications and metrics. SSM - Population Health. 2019; 8: 100404. https://doi.org/10.1016/j.ssmph.2019.100404.

280. Ni Mhurchu C, Vandevijvere S, Waterlander W, Thornton LE, Kelly B, Cameron AJ, Snowdon W, Swinburn B, INFORMAS. Monitoring the availability of healthy and unhealthy foods and nonalcoholic beverages in community and consumer retail food environments globally. Obesity Reviews. 2013; 14: 108-19. https://doi.org/10.1111/obr.12080.

281. Kestens Y, Lebel A, Chaix B, Clary C, Daniel M, Pampalon R, Theriault M, p Subramanian SV. Association between activity space exposure to food establishments and individual risk of overweight. PLOS ONE. 2012; 7: e41418. https://doi.org/10.1371/journal.pone.0041418.

282. Elliott P, Wartenberg D. Spatial epidemiology: Current approaches and future challenges. Environmental Health Perspectives. 2004; 112: 998-1006. https://doi.org/10.1289/ehp.6735.

283. Just Eat. Just eat plc annual report and accounts 2018. 2019. First accessed: 06.05.2021. Available from: https://www.justeattakeaway.com/download/53d527a0-499e-4d16-b545-6ce83c1135c6.

284. Christensen A, Griffiths C, Hobbs M, Gorse C, Radley D. Accuracy of buffers and self-drawn neighbourhoods in representing adolescent gps measured activity spaces: An exploratory study. Health & Place. 2021; 69: 102569. https://doi.org/10.1016/j.healthplace.2021.102569.

285. United States Department of Agriculture Economic Research Service. Food prices and spending. 2020. First accessed: 16.09.2020. Available from: https://www.ers.usda.gov/data-products/ag-and-food-statistics-charting-the-essentials/food-prices-and-spending/.

286. Robinson E, Marty L, Jones A, White M, Smith R, Adams J. Will calorie labels for food and drink served outside the home improve public health? BMJ. 2021; 372: n40. https://doi.org/10.1136/bmj.n40.

287. Lake A, Townshend T. Obesogenic environments: Exploring the built and food environments. Journal of the Royal Society for the Promotion of Health. 2006; 126: 262-7. https://doi.org/10.1177/1466424006070487.

288. Block JP, Scribner RA, DeSalvo KB. Fast food, race/ethnicity, and income: A geographic analysis. American Journal of Preventive Medicine. 2004; 27: 211-7. https://doi.org/10.1016/j.amepre.2004.06.007.

289. Simon PA, Kwan D, Angelescu A, Shih M, Fielding JE. Proximity of fast food restaurants to schools: Do neighborhood income and type of school matter? Prev Med. 2008; 47: 284-8. https://doi.org/10.1016/j.ypmed.2008.02.021.

290. Smoyer-Tomic KE, Spence JC, Raine KD, Amrhein C, Cameron N, Yasenovskiy V, Cutumisu N, Hemphill E, Healy J. The association between neighborhood socioeconomic status and exposure to supermarkets and fast food outlets. Health & Place. 2008; 14: 740-54. https://doi.org/10.1016/j.healthplace.2007.12.001.

291. Wilkins EL, Morris MA, Radley D, Griffiths C. Using geographic information systems to measure retail food environments: Discussion of methodological considerations and a proposed reporting checklist (geo-fern). Health & Place. 2017; 44: 110-7. https://doi.org/10.1016/j.healthplace.2017.01.008.

292. Just Eat. Just eat takeaway.Com q4 2020 trading update. 2021. First accessed: 15.01.2021. Available from: https://www.justeattakeaway.com/media/press-releases/.

293. Swinburn BA, Sacks G, Hall KD, McPherson K, Finegood DT, Moodie ML, Gortmaker SL. The global obesity pandemic: Shaped by global drivers and local environments. The Lancet. 2011; 378: 804-14. https://doi.org/10.1016/S0140-6736(11)60813-1.

294. Furst T, Connors M, Bisogni CA, Sobal J, Falk LW. Food choice: A conceptual model of the process. Appetite. 1996; 26: 247-65. https://doi.org/10.1006/appe.1996.0019.

295. Bishop TRP, von Hinke S, Hollingsworth B, Lake AA, Brown H, Burgoine T. Automatic classification of takeaway food outlet cuisine type using machine (deep) learning. Machine learning with applications. 2021; 6: 100106. https://doi.org/10.1016/j.mlwa.2021.100106.

296. UK Data Service. Census boundary data. 2020. First accessed: 15.09.2020. Available from: https://census.ukdataservice.ac.uk/get-data/boundary-data.aspx.

297. Ministry of Housing Communities and Local Government. English indices of deprivation 2019. 2019. First accessed: 11.07.2022. Available from: https://www.gov.uk/government/statistics/english-indices-of-deprivation-2019.

298. Office for National Statistics. Census geography. 2016. First accessed: 10.05.2022. Available from: https://www.ons.gov.uk/methodology/geography/ukgeographies/censusgeography.

299. Webscraper IO. Webscraper io. 2022. First accessed: 24.05.2022. Available from: https://www.webscraper.io/.

300. Office for National Statistics. 2011 rural/urban classification. 2016. First accessed: 11.07.2022. Available from:

https://www.ons.gov.uk/methodology/geography/geographicalproducts/ruralurbanclassification s/2011ruralurbanclassification.

301. Office for National Statistics. Usual resident population. 2013. First accessed: 11.07.2022. Available from: https://www.nomisweb.co.uk/census/2011/ks101ew.

302. Office for National Statistics. Workday population: Population density. 2014. First accessed: 15.09.2020. Available from: https://www.nomisweb.co.uk/census/2011/wd102ew.

303. Ellaway A, Macdonald L, Lamb K, Thornton L, Day P, Pearce J. Do obesity-promoting food environments cluster around socially disadvantaged schools in glasgow, scotland? Health & Place. 2012; 18: 1335-40. https://doi.org/10.1016/j.healthplace.2012.06.001.

304. Mazidi M, Speakman JR. Higher densities of fast-food and full-service restaurants are not associated with obesity prevalence. The American journal of clinical nutrition. 2017; 106: 603-13. https://doi.org/10.3945/ajcn.116.151407.

305. Fotheringham AS, Wong DWS. The modifiable areal unit problem in multivariate statistical analysis. Environment and Planning A: Economy and Space. 1991; 23: 1025-44. https://doi.org/10.1068/a231025.

306. Cornelsen L, Berger N, Cummins S, Smith RD. Socio-economic patterning of expenditures on 'out-of-home' food and non-alcoholic beverages by product and place of purchase in britain. Social Science & Medicine. 2019; 235: 1-11. https://doi.org/10.1016/j.socscimed.2019.112361.

307. Mahawar N, Jia SS, Korai A, Wang C, Allman-Farinelli M, Chan V, Raeside R, Phongsavan P, Redfern J, Gibson AA, Partridge SR, Roy R. Unhealthy food at your fingertips: Cross-sectional analysis of the nutritional quality of restaurants and takeaway outlets on an online food delivery platform in new zealand. Nutrients. 2022; 14: 4567. https://doi.org/10.3390/nu14214567.

308. Peeters A, Blake MRC. Socioeconomic inequalities in diet quality: From identifying the problem to implementing solutions. Current Nutrition Reports. 2016; 5: 150-9. https://doi.org/10.1007/s13668-016-0167-5.

309. UK Government. Prime minister's statement on coronavirus (covid-19): 23 march 2020. 2020. First accessed: 31.05.2022. Available from: https://www.gov.uk/government/speeches/pm-address-to-the-nation-on-coronavirus-23-march-2020.

310. Foster IS, Liu SY, Hoffs CT, LeBoa C, Chen AS, Rummo PE. Disparities in snap online grocery delivery and implementation: Lessons learned from california during the 2020-21 covid pandemic. Health & Place. 2022; 76: 102811. https://doi.org/10.1016/j.healthplace.2022.102811.

311. Colbert S, Wilkinson C, Thornton L, Richmond R. Covid-19 and alcohol in australia: Industry changes and public health impacts. Drug Alcohol Rev. 2020; 39: 435-40. https://doi.org/10.1111/dar.13092.

312. UK Government. New regulations created by secretary of state for business closure (covid-19). 2020. First accessed: 31.05.2022. Available from: https://www.gov.uk/government/news/new-regulations-created-by-secretary-of-state-forbusiness-closure-covid-19.

313. UK Government. Government to grant permission for pubs and restaurants to operate as takeaways as part of coronavirus response. 2020. First accessed: 31.05.2022. Available from: https://www.gov.uk/government/news/government-to-grant-permission-for-pubs-and-restaurants-to-operate-as-takeaways-as-part-of-coronavirus-response.

314. Town and Country Planning. 2020 no. 1243. 2020. First accessed: 31.05.2022. Available from: https://www.legislation.gov.uk/uksi/2020/1243/made/data.pdf.

315. Moore H, Lake A, O'Malley C, Bradford C, Gray N, Chang M, Mathews C, Townshend T. The impact of covid-19 on the hot food takeaway planning regulatory environment: Perspectives of local authority professionals in the north east of england. Perspectives in Public Health. 2022; Epub ahead of print: 1-9. https://doi.org/10.1177/17579139221106343.

316. Just Eat. Annual report 2021. 2022. First accessed: 31.05.2022. Available from: https://www.justeattakeaway.com/annual-reports.

317. Deliveroo. Q1 trading update. 2022. First accessed: 31.05.2022. Available from: https://corporate.deliveroo.co.uk/investors/results-reports-presentations/.

318. Nicolle E. Uber eats orders skyrocket 160 per cent in lockdown. 2020. First accessed: 31.05.2022. Available from: https://www.cityam.com/uber-eats-orders-in-the-uk-skyrocket-as-lockdown-powers-up-delivery-apps/.

319. Townshend T, Lake A. Obesogenic environments: Current evidence of the built and food environments. Perspectives in Public Health. 2017; 137: 38-44. https://doi.org/10.1177/1757913916679860.

320. StataCorp LLC. Longitudinal-data / panel-data reference manual release 17. 2021. First accessed: 11.02.2022. Available from: https://www.stata.com/manuals/xt.pdf.

321. UCLA: Statistical Consulting Group. How does one do regression when the dependent variable is a proportion? 2021. First accessed: 10.05.2022. Available from: https://stats.oarc.ucla.edu/stata/faq/how-does-one-do-regression-when-the-dependent-variable-is-a-proportion/.

322. Khang Y-H, Lim D, Bahk J, Kim I, Kang H-Y, Chang Y, Jung-Choi K. A publicly well-accepted measure versus an academically desirable measure of health inequality: Cross-sectional comparison of the difference between income quintiles with the slope index of inequality. BMJ Open. 2019; 9: e028687. https://doi.org/10.1136/bmjopen-2018-028687.

323. Renard F, Devleesschauwer B, Speybroeck N, Deboosere P. Monitoring health inequalities when the socio-economic composition changes: Are the slope and relative indices of inequality appropriate? Results of a simulation study. BMC Public Health. 2019; 19: 662. https://doi.org/10.1186/s12889-019-6980-1.

324. Economics Observatory. How has coronavirus affected pubs, cafes and restaurants? 2020. First accessed: 06.06.2022. Available from: https://www.economicsobservatory.com/how-has-coronavirus-affected-pubs-cafes-and-restaurants.

325. Ecker Y, Strüver A. Towards alternative platform futures in post-pandemic cities? A case study on platformization and changing socio-spatial relations in on-demand food delivery. Digital Geography and Society. 2022; 3: 100032. https://doi.org/10.1016/j.diggeo.2022.100032.

326. Dubois L, Girard M, Bergeron N. The choice of a diet quality indicator to evaluate the nutritional health of populations. Public Health Nutrition. 2007; 3: 357-65. https://doi.org/10.1017/S136898000000409.

327. Townshend T, Lake AA. Obesogenic urban form: Theory, policy and practice. Health & Place. 2009; 15: 909-16. https://doi.org/10.1016/j.healthplace.2008.12.002.

328. Townshend TG. Toxic high streets. Journal of Urban Design. 2017; 22: 167-86. https://doi.org/10.1080/13574809.2015.1106916.

329. Lord C, Bates O, Friday A, McLeod F, Cherrett T, Martinez-Sykora A, Oakey A. The sustainability of the gig economy food delivery system (deliveroo, ubereats and just-eat): Histories and futures of rebound, lock-in and path dependency. International Journal of Sustainable Transportation. 2022; Epub ahead of print: 1-13. https://doi.org/10.1080/15568318.2022.2066583.

330. Keeble M, Burgoine T, White M, Summerbell C, Cummins S, Adams J. How does local government use the planning system to regulate hot food takeaway outlets? A census of current practice in england using document review. Health & Place. 2019; 57: 171-8. https://doi.org/10.1016/j.healthplace.2019.03.010. 331. Bowman SA, Vinyard BT. Fast food consumption of u.S. Adults: Impact on energy and nutrient intakes and overweight status. Journal of the American College of Nutrition. 2004; 23: 163-8. https://doi.org/10.1080/07315724.2004.10719357.

332. Vogel C, Lewis D, Ntani G, Cummins S, Cooper C, Moon G, Baird J. The relationship between dietary quality and the local food environment differs according to level of educational attainment: A cross-sectional study. PLoS One. 2017; 12: e0183700. https://doi.org/10.1371/journal.pone.0183700.

333. Azzopardi-Muscat N, Sørensen K. Towards an equitable digital public health era: Promoting equity through a health literacy perspective. Eur J Public Health. 2019; 29: 13-7. https://doi.org/10.1093/eurpub/ckz166.

334. NHS Digital. National child measurement programme, england 2020/21 school year. 2021. First accessed: 28.10.2022. Available from: https://digital.nhs.uk/data-and-information/publications/statistical/national-child-measurement-programme/2020-21-school-year#.

335. Ananth CV, Schisterman EF. Hidden biases in observational epidemiology: The case of unmeasured confounding. BJOG : an international journal of obstetrics and gynaecology. 2018; 125: 644-6. https://doi.org/10.1111/1471-0528.14960.

336. Hill AB. The environment and disease: Association or causation? Proceedings of the Royal Society of Medicine. 1965; 58: 295-300. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1898525/

337. Cox LA. Modernizing the bradford hill criteria for assessing causal relationships in observational data. Critical Reviews in Toxicology. 2018; 48: 682-712. https://doi.org/10.1080/10408444.2018.1518404.

338. Lamb KE, Thornton LE, King TL, Ball K, White SR, Bentley R, Coffee NT, Daniel M. Methods for accounting for neighbourhood self-selection in physical activity and dietary behaviour research: A systematic review. International Journal of Behavioral Nutrition and Physical Activity. 2020; 17: 45. https://doi.org/10.1186/s12966-020-00947-2.

339. Besser LM, Brenowitz WD, Meyer OL, Hoermann S. Methods to address self-selection and reverse causation in studies of neighborhood environments and brain health. International Journal of Environmental Research and Public Health. 2021; 18: 6484. https://doi.org/10.3390/ijerph18126484.

340. Flegal KM, Graubard BI, Williamson DF, Cooper RS. Reverse causation and illness-related weight loss in observational studies of body weight and mortality. American Journal of Epidemiology. 2010; 173: 1-9. https://doi.org/10.1093/aje/kwq341.

341. Food Standards Agency. Food and you 2: Wave 4 key findings. 2021. First accessed: 28.06.2022. Available from: https://doi.org/10.46756/sci.fsa.zdt530.

342. Popkin BM, Adair LS, Ng SW. Global nutrition transition and the pandemic of obesity in developing countries. Nutrition Reviews. 2012; 70: 3-21. https://doi.org/10.1111/j.1753-4887.2011.00456.x.

343. Herforth A, Ahmed S. The food environment, its effects on dietary consumption, and potential for measurement within agriculture-nutrition interventions. Food Security. 2015; 7: 505-20. https://doi.org/10.1007/s12571-015-0455-8.

344. Hillen J. Web scraping for food price research. British Food Journal. 2019; 121: 3350-61. https://doi.org/10.1108/BFJ-02-2019-0081.

345. Harrington RA, Adhikari V, Rayner M, Scarborough P. Nutrient composition databases in the age of big data: Fooddb, a comprehensive, real-time database infrastructure. BMJ Open. 2019; 9: 1-10. https://doi.org/10.1136/bmjopen-2018-026652.

346. Office for National Statistics. Web scraping policy. 2020. First accessed: 15.09.2020. Available from:

https://www.ons.gov.uk/aboutus/transparencyandgovernance/datastrategy/datapolicies/webscr apingpolicy#:~:text=During%20web%20scraping%2C%20ONS%20will,allowed%20for%20access %20or%20not.

347. Kwan M-P. The limits of the neighborhood effect: Contextual uncertainties in geographic, environmental health, and social science research. Annals of the American Association of Geographers. 2018; 108: 1482-90. https://doi.org/10.1080/24694452.2018.1453777.

348. Campbell M, Marek L, Hobbs M. Reconsidering movement and exposure: Towards a more dynamic health geography. Geography Compass. 2021; 15: 1-17. https://doi.org/10.1111/gec3.12566.

349. Public Health England. Healthy people, healthy places briefing obesity and the environment: Regulating the growth of fast food outlets 2014. First accessed: 01.06.2018. Available from:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/296248/Obesity_and_environment_March2014.pdf

350. Public Health England. Healthy high streets challenge and tasters fried chicken shop. 2018. First accessed: 11.07.2022. Available from: https://www.gov.uk/government/case-studies/healthy-high-streets-challenge-and-tasters-fried-chicken-shop.

351. Sturm R, Cohen DA. Zoning for health? The year-old ban on new fast-food restaurants in south la. Health Affairs. 2009; 28: w1088-w97. https://doi.org/10.1377/hlthaff.28.6.w1088.

352. Boyland EJ, Whalen R. Food advertising to children and its effects on diet: Review of recent prevalence and impact data. Pediatric Diabetes. 2015; 16: 331-7. https://doi.org/10.1111/pedi.12278.

353. Yau A, Berger N, Law C, Cornelsen L, Greener R, Adams J, Boyland EJ, Burgoine T, de Vocht F, Egan M, Er V, Lake AA, Lock K, Mytton O, Petticrew M, Thompson C, White M, Cummins S. Changes in household food and drink purchases following restrictions on the advertisement of high fat, salt, and sugar products across the transport for london network: A controlled interrupted time series analysis. The Lancet. 2021; 398: S15. https://doi.org/10.1016/S0140-6736(21)02558-7.

354. Lakerveld J, Mackenbach J. The upstream determinants of adult obesity. Obes Facts. 2017; 10: 216-22. https://doi.org/10.1159/000471489.

355. Nuffield Council on Bioethics. Public health: Ethical issues. London2007.

356. Goffe L, Chivukula SS, Bowyer A, Bowen S, Toombs AL, Gray CM. Web augmentation for well-being: The human-centred design of a takeaway food ordering digital platform. Interacting with Computers. 2022; 33: 335-52. https://doi.org/10.1093/iwc/iwac015.

357. NESTA. Reordering food options on delivery apps could help to reduce obesity. 2022. First accessed: 09.08.2022. Available from:

https://media.nesta.org.uk/documents/Nesta_BIT_AHL_Food_delivery_apps_July_2022_Final_pdf _aWtrcHp.pdf.

358. Shaw SC, Ntani G, Baird J, Vogel CA. A systematic review of the influences of food store product placement on dietary-related outcomes. Nutrition reviews. 2020; 78: 1030-45. https://doi.org/10.1093/nutrit/nuaa024.

359. Bleich SN, Wolfson JA, Jarlenski MP. Calorie changes in large chain restaurants: Declines in new menu items but room for improvement. Am J Prev Med. 2016; 50: e1-e8. https://doi.org/10.1016/j.amepre.2015.05.007.

360. Wolfson JA, Moran AJ, Jarlenski MP, Bleich SN. Trends in sodium content of menu items in large chain restaurants in the u.S. American Journal of Preventive Medicine. 2018; 54: 28-36. https://doi.org/10.1016/j.amepre.2017.08.018.

361. Hillier-Brown FC, Summerbell CD, Moore HJ, Wrieden WL, Adams J, Abraham C, Adamson A, Araújo-Soares V, White M, Lake AA. A description of interventions promoting healthier readyto-eat meals (to eat in, to take away, or to be delivered) sold by specific food outlets in england: A systematic mapping and evidence synthesis. BMC Public Health. 2017; 17: 93-110. https://doi.org/10.1186/s12889-016-3980-2.

362. Goffe L, Chivukula SS, Bowyer A, Bowen S, Toombs AL, Gray CM. Appetite for disruption: Designing human-centred augmentations to an online food ordering platform. 34th British HCI Conference 34. 2021: 155-67. https://doi.org/10.14236/ewic/HCI2021.16.

363. Cecchini M, Warin L. Impact of food labelling systems on food choices and eating behaviours: A systematic review and meta-analysis of randomized studies. Obesity Reviews. 2016; 17: 201-10. https://doi.org/10.1111/obr.12364.

364. Bahadoran Z, Mirmiran P, Azizi F. Fast food pattern and cardiometabolic disorders: A review of current studies. Health promotion perspectives. 2015; 5: 231-40. https://doi.org/10.15171/hpp.2015.028.

365. Chaix B. How daily environments and situations shape behaviors and health: Momentary studies of mobile sensing and smartphone survey data. Health & Place. 2020; 61: 102241. https://doi.org/10.1016/j.healthplace.2019.102241.

366. Mason TB, Do B, Wang S, Dunton GF. Ecological momentary assessment of eating and dietary intake behaviors in children and adolescents: A systematic review of the literature. Appetite. 2020; 144: 104465. https://doi.org/10.1016/j.appet.2019.104465.

367. Mills S. Finding the 'nudge' in hypernudge. Technology in Society. 2022; 71: 102117. https://doi.org/10.1016/j.techsoc.2022.102117.

368. Shen Y, Ta N, Kwan M-P, Zhen F. Chapter 12: Icts and changes in activity–travel behavior in urban china: Existing research and future directions. Cheltenham, UK: Edward Elgar Publishing; 2020 02 Mar. 2020.

369. Clark SD, Shute B, Jenneson V, Rains T, Birkin M, Morris MA. Dietary patterns derived from uk supermarket transaction data with nutrient and socioeconomic profiles. Nutrients. 2021; 13. https://doi.org/10.3390/nu13051481.

370. Jenneson V, Clarke GP, Greenwood DC, Shute B, Tempest B, Rains T, Morris MA. Exploring the geographic variation in fruit and vegetable purchasing behaviour using supermarket transaction data. Nutrients. 2021; 14: 177. https://doi.org/10.3390/nu14010177.

371. Grabher G, van Tuijl E. Uber-production: From global networks to digital platforms. Environment and Planning A: Economy and Space. 2020; 52: 1005-16. https://doi.org/10.1177/0308518X20916507.

372. Jacobides MG, Lianos I. Regulating platforms and ecosystems: An introduction. Industrial and Corporate Change. 2021; 30: 1131-42. https://doi.org/10.1093/icc/dtab060.

373. Food Standards Agency. Food in the digital platform economy – making sense of a dynamic ecosystem. 2022. First accessed: 04.07.2022. Available from: https://www.food.gov.uk/research/emerging-challenges-and-opportunities/food-in-the-digital-platform-economy-making-sense-of-a-dynamic-ecosystem.

374. Chang M, Green L, Cummins S. All change. Has covid-19 transformed the way we need to plan for a healthier and more equitable food environment? URBAN DESIGN International. 2020; 26: 291-5. https://doi.org/10.1057/s41289-020-00143-5.

375. Shapiro A. Platform urbanism in a pandemic: Dark stores, ghost kitchens, and the logisticalurban frontier. 2022; 1: 1-20. https://doi.org/10.1177/14695405211069983.

376. Rinaldi C, D'Aguilar M, Egan M. Understanding the online environment for the delivery of food, alcohol and tobacco: An exploratory analysis of 'dark kitchens' and rapid grocery delivery services. International Journal of Environmental Research and Public Health. 2022; 19: 5523. https://doi.org/10.3390/ijerph19095523.

Supplementary material

Appendix A: Supplementary material for Chapter 2

Variable and category ^a	Au	ustralia	C	anada	N	/lexico		UK		USA	Т	otal
	(n:	=3578)	(n	=3698)	(n	=3515)	(n	=4694)	(n	=3893)	(n=	19378)
Sex												
Male	1732	(48.4)	1835	(49.6)	1690	(48.1)	2249	(47.9)	1860	(47.8)	9367	(48.3)
Female	1847	(51.6)	1863	(50.4)	1824	(51.9)	2444	(52.1)	2032	(52.2)	10010	(51.7)
Ethnicity												
Majority	2752	(76.9)	2972	(80.4)	2828	(80.5)	4211	(89.7)	2975	(76.4)	15738	(81.2)
Minority	826	(23.1)	726	(19.6)	687	(19.5)	483	(10.3)	918	(23.6)	3640	(18.8)
Age (years): median (IQR)	48	(33-61)	51	(34-62)	38	(28-50)	50	(34-64)	49	(32-62)	47 ((32-61)
Education												
Low	1512	(42.3)	1514	(40.9)	666	(18.9)	2269	(48.3)	2261	(58.1)	8222	(42.4)
Medium	1171	(32.7)	1256	(34.0)	423	(13.2)	1091	(23.2)	390	(10.0)	4370	(22.6)
High	895	(25.0)	928	(25.1)	2387	(67.9)	1333	(28.4)	1242	(31.9)	6785	(35.0)
BMI (kg/m²)												
Not overweight (≤ 24.9)	1422	(39.7)	1383	(37.4)	1551	(44.1)	1802	(38.4)	1364	(35.0)	7522	(38.8)
Overweight (25.0-29.9)	990	(27.7)	1107	(29.9)	1099	(31.2)	1306	(27.8)	1160	(29.8)	5661	(29.2)
Obesity (<u>></u> 30.0)	780	(21.8)	928	(25.1)	562	(16.0)	818	(17.4)	1092	(28.1)	4180	(21.6)
Missing	386	(10.8)	280	(7.6)	304	(8.7)	767	(16.3)	277	(7.1)	2015	(10.4)
Child <18 years in home												
No	2670	(74.6)	2931	(79.2)	1723	(49.0)	3564	(75.9)	2963	(76.1)	13850	(71.5)
Yes	909	(25.4)	767	(20.8)	1792	(51.0)	1130	(24.1)	930	(23.9)	5528	(28.5)

Table S1: Sociodemographic characteristics of analytic sample (n=19,378) from the 2018 International Food Policy Study.

Note: a data presented as n (%) unless stated, may not always equate to corresponding total due to rounding.

Variable and category	Au	stralia	Ca	nada	M	lexico		UK	ι ι	JSA	Т	otal	p value for
	(n=	=3578)	(n=	3698)	(n=	=3515)	(n=	4694)	(n=	3893)	(n=1	19378)	difference ^b
Non-online food delivery	2188	(61.2)	2420	(65.4)	2396	(68.2)	2439	(52.0)	2721	(69.9)	12163	(62.8)	p>0.0001
service customers													
Sex													p>0.0001
Male	1011	(46.2)	1236	(51.1)	1174	(49.0)	1167	(47.9)	1278	(47.0)	5866	(48.2)	
Female	1177	(53.8)	1185	(48.9)	1222	(51.0)	1271	(52.1)	1443	(53.0)	6298	(51.8)	
Ethnicity													p>0.0001
Majority	1706	(77.9)	1930	(79.8)	1985	(82.8)	2167	(90.1)	2130	(78.3)	9947	(81.8)	
Minority	483	(22.1)	490	(20.2)	411	(17.2)	242	(9.9)	591	(21.7)	2216	(18.2)	
Age (years): median (IQR)	48 ((34-61)	48 (33-61)	40	(28-52)	50 (36-64)	50 (33-61)	47 (3	32-60)	
Education													p>0.0001
Low	915	(41.8)	953	(39.4)	508	(21.2)	1075	(44.1)	1591	(58.5)	5042	(41.5)	
Medium	725	(33.1)	824	(34.0)	326	(13.6)	591	(24.2)	285	(10.5)	2751	(22.6)	
High	548	(25.0)	644	(26.6)	1562	(65.2)	773	(31.7)	844	(31.0)	4370	(35.9)	
BMI (kg/m²)													p>0.0001
Not overweight (≤ 24.9)	855	(39.1)	884	(36.5)	1035	(43.2)	910	(37.3)	927	(34.1)	4611	(38.8)	
Overweight (25.0-29.9)	601	(27.5)	736	(30.4)	758	(31.6)	718	(29.4)	827	(30.4)	3460	(29.2)	
Obesity (≤ 30.0)	502	(23.0)	623	(25.7)	385	(16.1)	429	(17.6)	788	(29.0)	2727	(21.6)	
Missing	230	(10.5)	178	(7.3)	218	(9.1)	382	(15.7)	178	(6.5)	1185	(10.4)	
Child <18 years in home													
No	1646	(75.2)	1893	(78.2)	1302	(54.3)	1879	(77.0)	2108	(77.5)	8828	(72.6)	p>0.0001
Yes	542	(24.8)	528	(21.8)	1094	(45.7)	460	(23.0)	612	(22.5)	3335	(27.4)	

Table S2: Sociodemographic characteristics of non-online food delivery service customers (n=12,163) amongst the analytic sample (n=19,378), from the 2018 International Food Policy Study ^a.

Notes: ^a Non-online food delivery service customers had purchased at least one meal prepared out-of-home directly from a food outlet, but not through an online food delivery service, in the past week. Unless stated data reported as n (%). May not always equate to corresponding total due to rounding.

^b p values from Pearson's χ2 test, represents difference between categories within variable.



Figure S1: Frequency of online food delivery service use in the past week amongst online food delivery service customers (n=2929). Data are from the 2018 International Food Policy Study.

		Model 0 ^a		Model 1 ^a
Variable and category	OR ^b	95% CI	OR ^b	95% CI
Sex (male)	1.41	1.28, 1.55	1.54	1.38, 1.71
Age (years)	0.94	0.94, 0.95	0.95	0.94, 0.95
Education				
Low	ref	-	ref	-
Medium	1.23	1.06, 1.42	-	-
High	2.57	2.28, 2.90	-	-
Ethnicity (minority)	2.35	2.10, 2.63	1.64	1.45, 1.86
BMI (kg/m ²)				
Not overweight (≤ 24.9)	ref	-	ref	-
Overweight (25.0-29.9)	0.68	0.61, 0.76	0.90	0.79, 1.03
Obesity (≥ 30.0)	0.53	0.46, 0.61	0.87	0.75, 1.01
Missing	1.05	0.90, 1.23	1.01	0.85, 1.20
Child <18 years in home (yes)	3.84	3.48, 4.23	2.85	2.57, 3.17
Country				
Australia	0.85	0.74, 0.99	0.74	0.61, 0.85
Canada	0.51	0.44, 0.60	0.46	0.39, 0.55
Mexico	1.80	1.58, 2.06	1.01	0.87, 1.18
UK	ref	-	ref	-
USA	0.71	0.61, 0.82	0.61	0.52, 0.72

Table S3: Associations between prevalence of any online food delivery service use in the past week and sociodemographic characteristics amongst the analytic sample (n=19,378) from the 2018 International Food Policy Study, analysed using logistic regression.

Notes: ^a Model 0 was unadjusted. Model 1 was adjusted for all independent variables except education.

^b Odds Ratio.

^c 95% confidence intervals.

Interaction term ^a	OR	95% CI
Sex x country (female = reference)		
Male x Australia	1.26	0.91, 1.76
Male x Canada	0.96	0.67, 1.37
Male x Mexico	0.66	0.50, 0.88
Male x USA	1.12	0.80, 1.55
Ethnicity x country (majority = reference)		
Minority x Australia	0.72	0.48, 1.08
Minority x Canada	0.90	0.60, 1.36
Minority x Mexico	0.92	0.62, 1.37
Minority x USA	1.13	0.77, 1.65
Age x country (continuous)		
Age x Australia	1.00	0.99, 1.02
Age x Canada	1.01	0.99, 1.02
Age x Mexico	1.03	1.02, 1.05
Age x USA	1.00	0.99, 1.02
Child <18 years at home x country (no = reference)		
Yes x Australia	0.82	0.59, 1.14
Yes x Canada	0.80	0.56, 1.14
Yes x Mexico	1.19	0.87, 1.61
Yes x USA	1.22	0.88, 1.69
Education x country (low = reference)		
Medium x Australia	1.47	0.94, 2.30
Medium x Canada	1.71	1.05, 2.77
Medium x Mexico	2.00	1.21, 3.31
Medium x USA	2.04	1.30, 3.20
High x Australia	2.10	1.40, 3.13
High x Canada	1.66	1.06, 2.60
High x Mexico	2.42	1.66, 3.55
High x USA	2.84	1.97, 4.08
BMI x country (not overweight = reference)		
Overweight x Australia	1.01	0.67, 1.54
Overweight x Canada	0.97	0.61, 1.53
Overweight x Mexico	1.15	0.81, 1.63
Overweight x USA	1.12	0.75, 1.66
Obesity x Australia	0.56	0.33, 0.93
Obesity x Canada	0.71	0.43, 1.16
Obesity x Mexico	0.93	0.60, 1.43
Obesity x USA	0.57	0.36, 0.91

Table S4. Odds ratio (OR) and 95% confidence intervals (CI) from two-way interaction termadded to separate, maximally adjusted, logistic regression models. Data are from the 2018International Food Policy Study.

Note: ^a Interaction term = independent variable x country, reference country throughout = UK. Interaction term added to separate maximally adjusted logistic regression models (Model 2).

Appendix B: Supplementary material for Chapter 3

UK cities
Birmingham
Bradford
Brighton
Bristol
Cornwall
Coventry
Edinburgh
Glasgow
Leeds
Leicester
Liverpool
London
Manchester
Newcastle-upon-Tyne
Sheffield
UK or food based
AskUK
GhostKitchens
JustEatUK
Unitedkingdom

Box S1: Names of Subredd	its used to	post study	information.

Box S2: Telephone interview topic guide with examples of questions used during data collection.

PRE-AMBLE AND INTRODUCTION

- Participant age, gender, marital status, and occupation confirmed.
- Definitions for `online food delivery services` (online takeaways) and `food prepared out-of-home` (takeaway food) discussed and confirmed.

PATTERNS OF TAKEAWAY FOOD PURCHASING

- What are the different ways that you have purchased takeaway food in the past 12 months?
- What do you think about when you are going to buy takeaway food? Maybe you could touch on things like how you choose the way that you are going to buy it, and the types of things you think about.
- Thinking about before March, when and why would you typically buy takeaway food?
- What is it about these days or times?

ONLINE FOOD DELIVERY SERVICE ADOPTION AND USE

(If not discussed during conversation so far)

- Which online takeaways have you used before?
- When did you start using `them`? [Note: `them` from this point refers to the name of the online takeaway]
- What made you first start using them?
- When might you use them instead of another purchasing format?
- How do you choose which food outlet you are going to buy from when you use them?

• Do you think using an online takeaway changes the food outlets you buy from? Followed up with, how?

PERCEIVED BENEFITS AND DRAWBACKS OF ONLINE TAKEAWAYS

- What do you think are the most important features of online takeaways? What are you expecting when you use them?
- How do you think these features might be different compared with other ways of buying takeaway food?
- Still thinking about before March, in your opinion, what are the main benefits of using online takeaways?
- What is not so good about using online takeaways?
- You mentioned some things that might not be so good about using online takeaways, so why do you keep using them what is it about online takeaways?

BROADER BEHAVIOURS AND PERCEPTIONS

- What do you think about when you hear the term `takeaway food`?
- What do you think about the types of food you can buy through online takeaways?
- Is the food you can buy online any different from the food you can buy in other ways?
- Thinking about all of the different places that you can buy takeaway food from near where you live, how does that compare to online takeaways?
- How do you think using an online takeaway has changed other ways of buying takeaway food?
- How do you think you will you buy takeaway food in the future?
- How does using online takeaways fit in with other things like cooking food at home?

SUMMARY

- Important points identified throughout interview reframed, and accuracy confirmed.
- Participant asked to discuss any points in more detail or if anything had not been asked but should have been.

Appendix C: Supplementary material for Chapter 4



Comparison of food outlets registered with Just Eat and Deliveroo in one region in England (April 2020).

(A) = Food outlets registered to accept orders through Just Eat. N = 1673

(B) = Food outlets registered to accept orders through Deliveroo. N = 334

(C) = Food outlets exclusively registered to accept orders through Deliveroo. N = 91

Figure S1: Comparison between food outlets registered with Just Eat and Deliveroo in one region in England. Data used in comparisons collected in April 2020.

Variable and category	Ful	I sample	Ir	ncluded	Excluded		
	(n	1=4139)	(r	(n=3337)		(n=/32)	
Sex							
Male	2022	(49.0)	1648	(49.4)	341	(46.6)	
Female	2117	(51.0)	1689	(50.6)	391	(53.4)	
Ethnicity							
Minority	460	(1.0)	333	(10.0)	118	(16.2)	
Majority	3648	(88.0)	2988	(89.6)	599	(81.8)	
Not available	31	(11.0)	15	(0.5)	15	(2.0)	
Age							
Mean (SD)	48.2	(16.8)	49.4	(16.9)	43.4	(15.7)	
18-29 years	803	(19.4)	570	(17.1)	208	(28.4)	
30-44 years	1020	(24.6)	799	(24.0)	201	(27.4)	
45-59 years	1069	(25.8)	875	(26.2)	180	(24.6)	
>60 years	1247	(30.1)	1093	1093 (32.8)		(19.6)	
Education level							
Low	2095	(50.6)	1688	(50.6)	364	49.6)	
Medium	848	(20.5)	708	(21.2)	132	(18.1)	
High	1166	(28.2)	932	(27.9)	215	(29.3)	
Not available	31	(0.7)	9	(0.3)	21	(2.9)	
Ability to make ends meet							
Not easy	2340	(56.5)	1992	(59.7)	507	69.3	
Easy	1799	(43.5)	1345	(40.3)	225	30.7	
BMI (kg/m²): mean (SD)	26.5	(5.3)	26.6	(5.3)	26.2	(5.5)	
	(3329 r	espondents)	(2863 ו	respondents)	(457 respondents)		
Weight Status							
Not overweight (BMI≤ 24.9)	1415	(34.2)	1170	(35.1)	227	(31.0)	
Overweight (BMI 25-29.9)	1097	(26.5)	945	(28.3)	136	(18.6)	
Obesity (BMI≥ 30)	727	(17.6)	624	(18.7)	94	(12.9)	
Not available	900	(21.7)	598	(17.9)	275	(37.5)	
Child at home							
No	2975	(71.9)	2433	(72.9)	490	(66.7)	
Yes	1158	(27.9)	901	(27.0)	239	(32.6)	
Not available	6	(0.1)	3	(0.1)	3	(0.4)	
Regular smoker		•					
No	3206	(77.4)	2599	(77.9)	557	(76.1)	
Yes	926	(22.4)	735	(22.0)	170	(23.3)	
Not available	7	(0.2)	3	(0.1)	5	(0.6)	

Table S1: Sociodemographic characteristics of UK respondents from the 2019 International Food Policy Study ^a.

Notes: ^a Data presented as weighted number of respondents (%) unless stated. May not always equate to corresponding total due to rounding. **Full sample** = all respondents. **Included** = respondents with exposure data, but no covariate data. **Excluded** = respondents with no exposure data.

Table S2: Online food delivery ser	vice use in the past we	ek amongst UK respon	dents from the 2019
International Food Policy Study ^a .			

Online food delivery service use	Full sample (n=4139)		In (n	Included (n=3337)		Excluded (n=732)	
Not applicable	1185	(28.6)	902	(27.0)	213	(29.1)	
Refuse to Answer	17	(0.4)	7	(0.2)	10	(13.1)	
Don't Know	53	(1.3)	45	(1.3)	9	(11.8)	
0	2169	(52.4)	1830	(54.9)	339	(46.3)	
1+	715	(17.3)	552	(16.6)	162	(22.2)	

Notes: ^a Data presented as weighted number of respondents (%), unless stated. May not always equate to corresponding total due to rounding. **Not applicable** = respondents that had not purchased a meal prepared out-of-home in the past week. **Full sample** = all respondents. **Included** = respondents with exposure data, but no covariate data. **Excluded** = respondents with no exposure data.

Table S3: Odds of online food delivery service use in the past week per quarter (Q) of online food outlet access amongst the analytic sample (n=3067) from the 2019 International Food Policy Study, analysed using unadjusted binomial logistic regression.

Food outlet number ^a	OR ^b	95% CI	c
Q1 (0-34)	ref	-	-
Q2 (35-85)	1.35	0.91	2.03
Q3 (86-181)	1.60	1.07	2.37
Q4 (182-879)	3.11	2.17	4.45

Notes: a Bracketed number is the number of food outlets accessible online for each Q. b Odds ratio.

^c 95% confidence intervals.

Interaction term ^a	OR	95	% CI
Food outlet access x education (Low = reference)			
Q2 x Medium	0.75	0.26	2.14
Q2 x High	0.59	0.22	1.59
Q3 x Medium	1.19	0.42	3.35
Q3 x High	0.78	0.30	2.04
Q4 x Medium	0.58	0.21	1.60
Q4 x High	1.22	0.49	3.01
Food outlet access x age (18-29 = reference)			
Q2 x 30-44	1.05	0.38	2.90
Q2 x 45-59	0.75	0.20	2.90
Q2 x ≥60	0.77	0.12	5.07
Q3 x 30-44	0.71	0.26	1.93
Q3 x 45-59	0.70	0.19	2.60
Q3 x ≥60	0.53	0.09	2.99
Q4 x 30-44	0.68	0.27	1.70
Q4 x 45-59	0.57	0.17	1.87
Q4 x ≥60	1.86	0.40	8.65
Food outlet access x sex (Male = reference)			
Q2 x Female	3.64	1.54	8.57
Q3 x Female	4.78	2.05	11.16
Q4 x Female	3.33	1.50	7.46
Food outlet access x child at home (No = reference)			
Q2 x Yes	1.26	0.55	2.90
Q3 x Yes	1.01	0.44	2.29
Q4 x Yes	0.81	0.38	1.72

Table S4: Odds ratio (OR) and 95% confidence intervals (CI) from multiplicative interaction terms added to separate adjusted binomial logistic regression models. Data are from the 2019 International Food Policy Study.

Notes: ^a Quarter (Q) 1 was the reference group throughout. Analyses adjusted for the following potential confounders: neighbourhood food outlet access, sex, age, education level, perceived income adequacy, living with children, and ethnicity. The number of food outlets accessible online for each Q were: Q1 (0-34), Q2 (35-85), Q3 (86-181), Q4 (182-879).

Table S5: Sensitivity analyses. Odds of online food delivery service use in the past week per quarter
(Q) of online food outlet access amongst the analytic sample (n=3067) from the 2019 International
Food Policy Study, analysed using adjusted binomial logistic regression ^a .

Food outlet number	OR	95% CI	
Neighbourhood supermarket ^b			
Q1 (0-34)	ref	-	-
Q2 (35-85)	1.22	0.77	1.92
Q3 (86-181)	1.26	0.80	2.00
Q4 (182-879)	1.79	1.14	2.83
Broader food outlet type ^b			
Q1 (0-34)	ref	-	-
Q2 (35-85)	1.22	0.77	1.92
Q3 (86-181)	1.26	0.79	2.00
Q4 (182-879)	1.75	1.10	2.76

Notes: ^a Broader food outlet types = nine categories included from 2019 Ordnance Survey Points of Interest data ('Fast food and takeaway outlets', 'Fast food delivery services', 'Fish and Chip shops', 'Restaurants', 'Cafés, snack bars and tea rooms', 'Convenience stores', 'Supermarkets', 'Bakeries' and 'Delicatessens'). Analyses adjusted for the following potential confounders: neighbourhood food outlet access, sex, age, education level, perceived income adequacy, presence of children at home, and ethnicity.

^b Bracketed number = number of food outlets accessible online for each Q.

Table S6: Odds of online food delivery service use in the past week per quarter (Q) of online access to unique types of cuisine, amongst the analytic sample (n=3067) from the 2019 International Food Policy Study, analysed using unadjusted binomial logistic regression.

Unique type of cuisine number ^a	OR	95% CI		
Q1 (0-64)	ref	-	-	
Q2 (65-85)	0.60	0.41	0.86	
Q3 (86-108)	0.91	0.65	1.29	
Q4 (109-148)	1.18	0.86	1.62	

Note: ^a Bracketed number = number unique types of cuisine accessible online for each Q.

Table S7: Secondary analyses. Coefficients (coef.) and 95% confidence intervals (CI) for the association between online food outlet access and body mass index and weight status amongst the analytic sample (n=3067) from the 2019 International Food Policy Study, analysed using multinomial linear or multinomial logistic regression.

	Model 0 ª			Model 1 ª		
Variable and category	coef.	95% CI		coef.	95% CI	
Body mass index (kg/m ²)						
Q1 (0-34) ^b	ref	-		ref	-	-
Q2 (35-85)	-0.04	-0.74	0.67	0.03	-0.65	0.72
Q3 (86-181)	-0.17	-0.85	0.51	0.14	-0.54	0.81
Q4 (182-879)	-1.08	-1.74	-0.42	-0.11	-0.84	0.61
Weight status ^c	OR ^d	95% CI		OR	95% CI	
Overweight						
Q1 (0-34)	ref	-	-	ref	-	-
Q2 (35-85)	0.88	0.66	1.18	0.88	0.65	1.18
Q3 (86-181)	0.96	0.72	1.28	1.01	0.75	1.36
Q4 (182-879)	0.80	0.60	1.07	1.02	0.72	1.43
Obesity						
Q1 (0-34)	ref	-	-	ref	-	-
Q2 (35-85)	0.97	0.69	1.34	0.99	0.70	1.39
Q3 (86-181)	0.82	0.59	1.14	0.93	0.66	1.32
Q4 (182-879)	0.62	0.44	0.88	0.92	0.62	1.38
<i>Not available</i> ^e						
Q1 (0-34)	ref	-	-	ref	-	-
Q2 (35-85)	1.27	0.88	1.83	1.23	0.85	1.78
Q3 (86-181)	1.08	0.74	1.58	1.10	0.75	1.62
Q4 (182-879)	1.39	0.98	1.97	1.42	0.95	2.12

Notes: ^a Model 0 was unadjusted. Model 1 was adjusted for the following potential confounders: neighbourhood food outlet access, sex, age, education level, perceived income adequacy, living with children, ethnicity, and smoking status.

^b Bracketed number = number of food outlets accessible online for each quarter (Q).

^c `Not overweight` category used as reference group throughout.

^d Odds Ratio. 95% confidence intervals.

^e Not available category = no BMI (n=517).

Appendix D: Supplementary material for Chapter 5

Table S1: counts of food outlets registered to accept orders online and food outlets with a customerfacing premises in the physical food environment of postcode districts in England (n=2118) in November 2019.

Measure	Number
Online food delivery service	
Food outlets registered	29232
Physical food environment	
Food outlets within postcode district ^a	82455
Food outlets within neighbourhood $^{\mathrm{b}}$	376513

Notes: ^a Food outlet categories included: Fast food and takeaway outlets; Fast food delivery services; Fish and Chip shops; Restaurants. Data collected June 2019.

^b `Neighbourhood` = 1600 m Euclidean radius `neighbourhood` buffer of postcode district geographic centroid.

Table S2: Coefficients (coef.) and 95% confidence intervals (CI) for the association between deprivation and the percentage of food outlets registered to accept orders online amongst postcode districts in England. Estimated using unadjusted and adjusted general linear models.

		Model 0	а		Model 1 ^a			
Percentage registered ^b (%)	coef.	95% CI		coef.	9	5% CI		
IMD score (deciles)								
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-		
2 (10.22-12.08)	-0.96	-4.63	2.70	-0.84	-3.86	2.18		
3 (12.09-14.00)	0.49	-3.16	4.13	1.27	-1.75	4.30		
4 (14.01-15.91)	4.40	0.74	8.05	2.72	-0.30	5.74		
5 (15.92-18.18)	3.92	0.26	7.58	2.29	-0.73	5.31		
6 (18.19-20.60)	5.39	1.74	9.04	2.93	-0.09	5.95		
7 (20.61-23.54)	8.10	4.44	11.75	4.18	1.14	7.22		
8 (23.55-27.06)	16.15	12.49	19.80	9.75	6.68	12.82		
9 (27.07-32.89)	21.80	18.14	25.45	11.12	8.01	14.23		
10 (32.90-69.51); most deprived	29.03	25.37	32.69	20.02	16.93	23.10		

Notes: ^a Model 0 was unadjusted. 2113 postcode districts included. Model 1 was adjusted for population density and rural urban classification. 2084 postcode districts included.

^b Percentage registered = percentage of food outlets in postcode district registered to accept orders online.

Table S3: Association between deprivation and online food outlet access amongst postcode districts
in England. Estimated using unadjusted and adjusted negative binomial regression.

	Model 0 ^a				Model 1 ^a		
Food outlets accessible online (count)	IRR ^b	95	% CI ^c	IRR ^b	95	% CI ^c	
IMD score (deciles)							
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-	
2 (10.22-12.08)	1.02	0.81	1.29	0.97	0.81	1.17	
3 (12.09-14.00)	1.32	1.04	1.67	0.97	0.81	1.18	
4 (14.01-15.91)	1.57	1.24	1.99	1.01	0.84	1.23	
5 (15.92-18.18)	1.51	1.19	1.90	0.94	0.78	1.13	
6 (18.19-20.60)	1.68	1.33	2.12	0.87	0.72	1.06	
7 (20.61-23.54)	2.16	1.71	2.72	0.99	0.81	1.20	
8 (23.55-27.06)	2.64	2.09	3.33	1.12	0.92	1.36	
9 (27.07-32.89)	3.08	2.44	3.89	1.19	0.97	1.45	
10 (32.90-69.51); most deprived	3.51	2.78	4.44	1.51	1.24	1.83	

Notes: ^a Model 0 was unadjusted. 2118 postcode districts included. Model 1 was adjusted for rural urban classification, population density, and the number of food outlets within their boundary. 2088 postcode districts included.

^b Incidence Rate Ratios (IRR) represent expected difference of outcome at each level of deprivation, compared with the reference group.

^c95% confidence intervals.



Figure S1: Deciles of the predicted number (count) of food outlets accessible online across postcode districts (n=2088) in England, in November 2019. Number estimated from a negative binomial regression model adjusted for the number of food outlets within a postcode district boundary, and rural urban classification and population density

Table S4: Association between	deprivation and onl	ine unique cuisine ty	pe access amongst postcode
districts in England. Estimated	using unadjusted an	d adjusted negative	binomial regression.

	Model 0 ^a			Model 1 ^a				
Unique cuisine types accessible (count)	IRR ^b	IRR ^b 95% CI ^c		IRR ^b	95% CI ^c			
IMD score (deciles)								
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-		
2 (10.22-12.08)	0.96	0.80	1.14	0.90	0.82	0.99		
3 (12.09-14.00)	1.05	0.88	1.25	0.86	0.78	0.95		
4 (14.01-15.91)	1.18	1.00	1.41	0.81	0.73	0.89		
5 (15.92-18.18)	1.12	0.94	1.33	0.76	0.69	0.84		
6 (18.19-20.60)	1.21	1.02	1.44	0.75	0.68	0.83		
7 (20.61-23.54)	1.39	1.17	1.66	0.67	0.61	0.74		
8 (23.55-27.06)	1.62	1.37	1.93	0.73	0.66	0.81		
9 (27.07-32.89)	1.87	1.58	2.23	0.80	0.72	0.88		
10 (32.90-69.51); most deprived	2.08	1.75	2.48	0.83	0.75	0.92		

Notes: ^a Model 0 was unadjusted. 2118 postcode districts included. Model 1 was adjusted for rural urban classification, population density, the number of food outlets within the postcode district boundary and the number of food outlets accessible online. 2088 postcode districts included.

^b Incidence Rate Ratios (IRR) represent expected difference of outcome at each level of deprivation, compared with the reference group.

^c 95% confidence intervals.

Table S5: Coefficients (coef.) and 95% confidence intervals (CI) for the association between deprivation and the percentage of neighbourhood food outlets accessible online amongst postcode districts in England. Estimated using unadjusted and adjusted general linear models.

		Model 0 ^a	l	Model 1 ^a			
Percentage accessible online ^b (%)	coef. 95% CI ^c		% CI ^c	coef.	959	% CI °	
IMD score (deciles)							
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-	
2 (10.22-12.08)	-9.72	-20.63	1.20	-8.17	-18.80	2.46	
3 (12.09-14.00)	-16.41	-27.28	-5.55	-12.19	-22.85	-1.53	
4 (14.01-15.91)	-14.35	-25.21	-3.48	-14.28	-24.90	-3.65	
5 (15.92-18.18)	-20.05	-30.92	-9.17	-19.32	-29.95	-8.69	
6 (18.19-20.60)	-21.84	-32.67	-11.00	-21.11	-31.72	-10.51	
7 (20.61-23.54)	-30.34	-41.21	-19.48	-29.40	-40.10	-18.69	
8 (23.55-27.06)	-16.54	-27.41	-5.68	-14.79	-25.58	-3.99	
9 (27.07-32.89)	-10.59	-21.43	0.25	-11.06	-21.97	-0.16	
10 (32.90-69.51); most deprived	0.05	-10.80	10.90	-3.38	-14.22	7.45	

Notes: ^a Model 0 was unadjusted. 2104 postcode districts included. Model 1 was adjusted for population density and rural urban classification. 2076 postcode districts included.

^b Percentage accessible online = the number of food outlets accessible online as a percentage of the number physically accessible in the neighbourhood, based on Ordnance Survey Points of Interest data. `Neighbourhood` = 1600 m Euclidean radius `neighbourhood` buffer of postcode district geographic centroid.

^c 95% confidence intervals.

Table S6: Sensitivity analyses: Coefficients (coef.) and 95% confidence intervals (CI) for the association between deprivation and the percentage of food outlets registered to accept orders online amongst postcode districts in England. Estimated using unadjusted and adjusted general linear models and nine categories of food outlets from Ordnance Survey Points of Interest data ^a.

		Model 0 ^b		Model	1 ^b
Percentage registered ^c (%)	coef.	95% CI	coe	f. 95% (CI
IMD score (deciles)					
1 (4.28-10.21); least deprived	ref		ref	-	-
2 (10.22-12.08)	-0.05	-1.83 1.72	0.03	3 -1.44	1.50
3 (12.09-14.00)	0.37	-1.40 2.15	0.77	7 -0.70	2.24
4 (14.01-15.91)	2.17	0.40 3.95	1.37	7 -0.10	2.84
5 (15.92-18.18)	2.59	0.82 4.37	1.83	3 0.36	3.30
6 (18.19-20.60)	2.51	0.74 4.29	1.35	5 -0.12	2.82
7 (20.61-23.54)	4.24	2.47 6.01	2.36	5 0.88	3.84
8 (23.55-27.06)	7.73	5.95 9.50	4.58	3.09	6.07
9 (27.07-32.89)	11.35	9.58 13.1	2 6.18	3 4.67	7.69
10 (32.90-69.51); most deprived	14.45	12.67 16.2	2 9.99	9 8.49	11.49

Notes: ^a Food outlet categories included: Fast food and takeaway outlets, Fast food delivery services, Fish and Chip shops, Restaurants, Cafés, snack bars and tea rooms, Convenience stores, Supermarkets, Bakeries, Delicatessens. ^b Model 0 was unadjusted. 2118 postcode districts included. Model 1 was adjusted for population density and rural urban classification. 2088 postcode districts included.

^c Percentage registered = the number of food outlets registered to accept orders online as a percentage of the number of food outlets with a customer-facing premises, based on Ordnance Survey Points of Interest data percentage of food outlets in postcode district registered to accept orders online based on Ordnance Survey Points of Interest data.

Table S7: Sensitivity analyses: Association between deprivation and online food outlet access amongst postcode districts in England. Estimated using unadjusted and adjusted negative binomial regression and nine categories of food outlets from Ordnance Survey Points of Interest data^a.

	Model 0 ^b			Model 1 ^b			
Food outlets accessible online	IRR ^c	95% CI	d	IRR ^c	95% CI ^d		
IMD score (deciles)							
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-	
2 (10.22-12.08)	1.02	0.81	1.29	0.98	0.81	1.18	
3 (12.09-14.00)	1.32	1.04	1.67	0.98	0.81	1.18	
4 (14.01-15.91)	1.57	1.24	1.99	1.01	0.84	1.22	
5 (15.92-18.18)	1.51	1.19	1.90	0.94	0.78	1.14	
6 (18.19-20.60)	1.68	1.33	2.12	0.87	0.72	1.05	
7 (20.61-23.54)	2.16	1.71	2.72	0.99	0.81	1.20	
8 (23.55-27.06)	2.64	2.09	3.33	1.12	0.92	1.36	
9 (27.07-32.89)	3.08	2.44	3.89	1.19	0.97	1.45	
10 (32.90-69.51); most deprived	3.51	2.78	4.44	1.51	1.24	1.83	

Notes: ^a Food outlet categories included: Fast food and takeaway outlets, Fast food delivery services, Fish and Chip shops, Restaurants, Cafés, snack bars and tea rooms, Convenience stores, Supermarkets, Bakeries, Delicatessens. ^b Model 0 was unadjusted. 2118 postcode districts included. Model 1 was adjusted for population density and rural urban classification. 2087 postcode districts included.

^c Incidence Rate Ratios represent expected difference of outcome at each level of deprivation, compared with the reference group.

^d 95% confidence intervals.

Table S8 : Sensitivity analyses: Association between deprivation and online unique cuisine type access
amongst postcode districts in England. Estimated using unadjusted and adjusted negative binomial
regression and nine categories of food outlets from Ordnance Survey Points of Interest data ^a .

	Model 0 ^b			Model 1 ^b		
Unique cuisine types accessible count)	IRR ^c 95% CI ^d		IRR ^c	95	5% CI ^d	
IMD score (deciles)						
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-
2 (10.22-12.08)	0.96	0.80	1.14	0.90	0.82	0.99
3 (12.09-14.00)	1.05	0.88	1.25	0.86	0.78	0.95
4 (14.01-15.91)	1.18	1.00	1.41	0.81	0.73	0.89
5 (15.92-18.18)	1.12	0.94	1.33	0.77	0.69	0.84
6 (18.19-20.60)	1.21	1.02	1.44	0.75	0.68	0.83
7 (20.61-23.54)	1.39	1.17	1.66	0.67	0.61	0.75
8 (23.55-27.06)	1.62	1.37	1.93	0.73	0.66	0.81
9 (27.07-32.89)	1.87	1.58	2.23	0.80	0.73	0.89
10 (32.90-69.51); most deprived	2.08	1.75	2.48	0.83	0.75	0.92

Notes: ^a Food outlet categories included: Fast food and takeaway outlets, Fast food delivery services, Fish and Chip shops, Restaurants, Cafés, snack bars and tea rooms, Convenience stores, Supermarkets, Bakeries, Delicatessens.

^b Model 0 was unadjusted. 2118 postcode districts included. Model 1 was adjusted for rural urban classification, population density, the number of food outlets in the postcode district boundary, based on Ordnance Survey Points of Interest data and the number of food outlets accessible online. 2088 postcode districts included.

^c Incidence Rate Ratios represent expected difference of outcome at each level of deprivation, compared with the reference group.

^d 95% confidence intervals

Table S9: Sensitivity analyses: Coefficients (coef.) and 95% confidence intervals (CI) for the association between deprivation and the percentage of neighbourhood food outlets accessible online amongst postcode districts in England. Estimated using unadjusted and adjusted general linear models and nine categories of food outlets from Ordnance Survey Points of Interest data ^a.

	Model 0 ^b			Model 1 ^b		
Percentage accessible online ^c (%)	coef.	95% CI		coef.	95% CI	
IMD score (deciles)						
1 (4.28-10.21); least deprived	ref	-	-	ref	-	-
2 (10.22-12.08)	-0.44	-7.09	6.22	0.59	-6.00	7.19
3 (12.09-14.00)	-2.08	-8.71	4.55	-0.09	-6.70	6.53
4 (14.01-15.91)	-6.66	-13.31	-0.01	-6.69	-13.30	-0.08
5 (15.92-18.18)	-7.07	-13.72	-0.42	-6.62	-13.22	-0.02
6 (18.19-20.60)	-9.49	-16.13	-2.85	-9.13	-15.73	-2.53
7 (20.61-23.54)	-12.40	-19.04	-5.75	-12.07	-18.72	-5.41
8 (23.55-27.06)	-6.08	-12.73	0.57	-5.38	-12.09	1.33
9 (27.07-32.89)	-1.99	-8.63	4.65	-2.31	-9.10	4.48
10 (32.90-69.51); most deprived	3.08	-3.57	9.73	1.18	-5.57	7.92

Notes: ^a Food outlet categories included: Fast food and takeaway outlets, Fast food delivery services, Fish and Chip shops, Restaurants, Cafés, snack bars and tea rooms, Convenience stores, Supermarkets, Bakeries, Delicatessens. ^b Model 0 was unadjusted. 2104 postcode districts included. Model 1 was adjusted for population density and rural urban classification. 2087 postcode districts included.

^c Percentage accessible online = the number of food outlets accessible online as a percentage of the number physically accessible in the neighbourhood, based on Ordnance Survey Points of Interest data. `Neighbourhood` = 1600 m Euclidean radius `neighbourhood` buffer of postcode district geographic centroid.

Appendix E: Supplementary material for Chapter 6

Month	Quarter ^b
November 2019	Q4 2019
June 2020	Q3 2020
July 2020	Q3 2020
August 2020	Q3 2020
September 2020	Q4 2020
October 2020	Q4 2020
November 2020	Q4 2020
December 2020	Q1 2021
January 2021	Q1 2021
February 2021	Q1 2021
March 2021	Q2 2021
April 2021	Q2 2021
May 2021	Q2 2021
June 2021	Q3 2021
July 2021	Q3 2021
August 2021	Q3 2021
September 2021	Q4 2021
October 2021	Q4 2021
November 2021	Q4 2021
December 2021	Q1 2022
January 2022	Q1 2022
February 2022	Q1 2022
March 2022	Q1 2022

Table S1: Operationalisation of matching monthly and quarterly data ^a.

Notes: ^a Monthly data from the online food delivery service.

^b Quarterly data from Ordnance Survey Points of Interest.

Deprivation quintile								
	1 (least deprived)	2	3	4	5 (most deprived)	England		
Month								
Nov 2019	3.0 (1.0-8.0)	4.0 (1.0-12.0)	6.0 (1.0-18.0)	13.0 (3.0-25.0)	24.0 (12.0-39.0)	7.0 (1.0-21.0)		
Jun 2020	4.0 (1.0-9.0)	5.0 (1.0-15.0)	7.0 (1.0-19.5)	15.0 (3.0-28.0)	25.0 (14.0-42.0)	8.0 (2.0-24.0)		
Jul 2020	3.0 (1.0-9.0)	5.0 (0.0-15.0)	6.0 (1.0-17.0)	12.0 (2.0-26.0)	24.0 (10.0-39.0)	7.0 (1.0-22.0)		
Aug 2020	4.0 (1.0-10.0)	5.0 (1.0-16.0)	7.0 (1.0-20.0)	14.0 (3.0-29.0)	27.0 (13.0-44.0)	8.0 (2.0-25.0)		
Sept 2020	4.0 (1.0-11.0)	5.0 (1.0-16.0)	7.0 (1.0-21.0)	15.0 (4.0-30.0)	28.0 (15.0-46.0)	9.0 (2.0-26.0)		
Oct 2020	4.0 (1.0-11.0)	6.0 (1.0-16.0)	7.0 (1.0-22.0)	16.0 (4.0-32.0)	29.0 (16.0-48.0)	10.0 (2.0-27.0)		
Nov 2020	5.0 (1.0-13.0)	7.0 (1.0-19.0)	9.0 (2.0-24.5)	17.0 (4.0-34.0)	30.0 (17.0-50.0)	11.0 (3.0-29.0)		
Dec 2020	4.0 (1.0-11.5)	6.0 (1.0-17.0)	8.0 (1.0-23.0)	16.0 (4.0-33.0)	30.0 (16.0-48.0)	10.0 (2.0-28.0)		
Jan 2021	5.0 (1.0-13.0)	7.0 (1.0-20.0)	10.0 (2.0-24.0)	18.0 (4.0-35.0)	31.0 (18.0-51.0)	11.0 (3.0-30.0)		
Feb 2021	5.0 (2.0-14.0)	7.0 (1.0-20.0)	10.0 (2.0-25.0)	18.5 (5.0-38.0)	33.0 (18.0-54.0)	12.0 (3.0-31.0)		
Mar 2021	5.0 (1.5-13.0)	7.0 (1.0-21.0)	9.0 (2.0-25.0)	18.0 (4.0-38.0)	33.0 (18.0-54.0)	12.0 (3.0-32.0)		
Apr 2021	5.0 (1.0-14.0)	7.0 (1.0-20.0)	9.5 (2.0-25.0)	18.5 (4.0-38.0)	33.0 (18.0-55.0)	12.0 (3.0-32.0)		
May 2021	5.0 (1.0-14.0)	7.0 (1.0-20.0)	9.5 (2.0-25.5)	19.0 (5.0-39.0)	34.0 (18.0-56.0)	12.0 (3.0-32.0)		
Jun 2021	5.0 (1.0-14.0)	7.0 (1.0-19.0)	10.0 (2.0-25.5)	20.0 (5.0-39.0)	35.0 (19.0-56.0)	12.0 (3.0-32.0)		
Jul 2021	5.0 (1.0-14.0)	8.0 (1.0-19.0)	10.0 (2.0-26.0)	19.5 (5.0-39.0)	35.0 (19.0-56.0)	12.0 (3.0-33.0)		
Aug 2021	5.0 (1.0-13.5)	8.0 (1.0-20.0)	10.0 (2.0-27.0)	19.0 (4.0-41.0)	34.0 (19.0-56.0)	12.0 (3.0-33.0)		
Sept 2021	5.0 (1.0-14.0)	8.0 (1.0-20.0)	10.0 (2.0-27.0)	20.0 (4.0-41.0)	35.0 (19.0-58.0)	13.0 (3.0-33.0)		
Oct 2021	5.0 (1.0-14.0)	8.0 (1.0-21.0)	10.0 (2.0-28.0)	20.0 (5.0-41.0)	35.0 (20.0-60.0)	13.0 (3.0-33.0)		
Nov 2021	5.0 (1.0-14.0)	8.0 (1.0-21.0)	10.0 (2.0-28.0)	20.0 (5.0-41.0)	36.0 (20.0-59.0)	13.0 (3.0-34.0)		
Dec 2021	5.0 (1.0-14.0)	8.0 (1.0-20.0)	10.0 (2.0-27.5)	20.0 (5.0-41.0)	35.0 (20.0-59.0)	13.0 (3.0-34.0)		
Jan 2022	5.0 (1.0-14.5)	8.0 (1.0-21.0)	10.0 (2.0-27.5)	20.0 (4.0-41.0)	36.0 (20.0-59.0)	13.0 (3.0-34.0)		
Feb 2022	5.0 (1.0-15.0)	8.0 (1.0-21.0)	10.0 (1.5-28.0)	21.0 (5.0-41.0)	36.0 (19.0-59.0)	13.0 (3.0-34.0)		
Mar 2022	5.0 (1.0-15.0)	8.0 (1.0-21.0)	10.0 (2.0-29.0)	21.0 (4.0-41.0)	35.0 (20.0-59.0)	13.0 (3.0-34.0)		

Table S2: number (count) of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation ^{a b}.

Notes: ^a data are reported as Median (IQR).

^b food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data, registered to accept orders through the online food delivery service.
	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England													
	1 (least deprived)	2	3	4	5 (most deprived)	England								
Month														
Nov 2019 (baseline)	-	-	-	-	-	-								
Jun 2020	14.3 (0.0-36.7)	14.0 (0.0-35.4)	11.1 (0.0-30.8)	12.0 (0.0-25.0)	11.1 (0.0-23.1)	12.0 (0.0-29.5)								
Jul 2020	13.3 (0.0-39.4)	15.4 (0.0-40.0)	13.2 (0.0-33.3)	8.3 (-5.6-26.3)	10.7 (0.0-23.8)	11.8 (0.0-33.3)								
Aug 2020	20.0 (0.0-50.0)	23.7 (0.0-50.0)	19.3 (0.0-46.2)	20.0 (0.0-40.9)	17.0 (3.7-33.3)	20.0 (0.0-42.9)								
Sept 2020	25.0 (0.0-54.9)	25.0 (0.0-50.0)	22.2 (0.0-50.0)	24.8 (3.8-41.7)	22.2 (8.0-36.7)	23.1 (0.0-46.8)								
Oct 2020	33.3 (0.0-66.7)	28.6 (0.0-60.0)	26.7 (4.9-55.6)	28.6 (8.3-50.0)	26.7 (12.0-41.7)	28.1 (6.3-50.0)								
Nov 2020	50.0 (9.1-100.0)	50.0 (20.0-85.7)	43.7 (18.2-83.3)	35.9 (16.7-66.7)	32.1 (16.9-50.0)	41.5 (16.7-75.0)								
Dec 2020	36.8 (0.0-75.0)	37.7 (0.0-68.6)	34.9 (0.0-66.7)	30.8 (9.1-56.3)	28.6 (14.3-47.5)	33.3 (7.1-61.5)								
Jan 2021	58.3 (14.3-100.0)	50.0 (25.0-100.0)	50.0 (20.0-87.5)	41.3 (17.6-74.2)	36.1 (21.4-56.3)	45.1 (20.0-83.3)								
Feb 2021	66.7 (25.0-100.0)	59.7 (30.3-100.0)	50.0 (23.1-100.0)	45.2 (26.2-77.8)	41.3 (26.7-63.6)	50.0 (25.2-91.8)								
Mar 2021	60.0 (20.0-100.0)	56.8 (28.6-100.0)	50.0 (19.4-87.5)	45.6 (21.4-80.0)	42.9 (27.3-65.5)	50.0 (24.0-87.5)								
Apr 2021	63.2 (20.0-100.0)	55.8 (27.3-100.0)	50.0 (20.0-91.7)	49.2 (24.0-83.3)	44.4 (28.6-67.5)	50.0 (25.0-91.7)								
May 2021	66.7 (25.0-100.0)	60.0 (30.0-100.0)	50.0 (22.2-95.7)	51.8 (24.0-87.5)	47.5 (30.2-72.4)	52.0 (26.7-95.2)								
Jun 2021	66.7 (20.0-100.0)	57.1 (28.6-100.0)	50.0 (23.1-100.0)	51.8 (23.1-89.5)	50.0 (33.3-72.4)	54.5 (27.3-100.0)								
Jul 2021	66.7 (20.0-100.0)	59.0 (30.8-100.0)	50.0 (23.1-100.0)	51.9 (25.0-89.5)	50.0 (32.7-74.5)	54.5 (28.0-100.0)								
Aug 2021	68.8 (20.0-100.0)	60.0 (31.6-100.0)	52.8 (20.0-100.0)	52.8 (27.3-96.0)	50.0 (33.3-80.8)	54.5 (28.0-100.0)								
Sept 2021	68.4 (22.2-109.1)	62.5 (33.3-100.0)	56.3 (25.0-100.0)	56.3 (31.3-97.1)	51.6 (33.3-83.3)	57.9 (30.8-100.0)								
Oct 2021	71.4 (21.4-110.0)	63.2 (33.3-100.0)	56.9 (28.0-100.0)	60.0 (32.1-96.8)	52.8 (34.5-83.3)	60.0 (30.7-100.0)								
Nov 2021	75.0 (25.0-114.3)	63.8 (33.3-104.3)	56.5 (26.2-100.0)	61.3 (33.3-100.0)	55.6 (36.6-87.5)	61.1 (33.3-100.0)								
Dec 2021	66.7 (20.0-116.7)	63.8 (33.3-106.3)	58.6 (27.3-100.0)	59.2 (33.3-100.0)	55.6 (35.7-84.4)	60.0 (33.0-100.0)								
Jan 2022	71.4 (20.0-120.0)	66.1 (34.5-107.1)	60.0 (28.1-106.3)	62.5 (32.1-100.0)	57.1 (35.0-87.5)	62.5 (33.3-100.0)								
Feb 2022	71.4 (22.2-116.7)	66.7 (33.3-114.3)	61.6 (28.6-109.7)	62.0 (31.8-100.0)	55.6 (36.2-85.7)	62.5 (33.3-100.0)								
Mar 2022	80.0 (22.2-120.0)	69.0 (33.3-120.0)	66.7 (30.0-106.3)	62.8 (33.3-100.0)	57.9 (34.0-87.5)	65.4 (33.3-100.0)								

Table S3: change (%) for the number of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation ^{a b}.

^b food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data, registered to accept orders through the online food delivery service.

	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England													
	1 (least deprived)	2	3	4	5 (most deprived)	England								
Month														
Nov 2019	37.0 (14.0-70.5)	38.0 (10.0-96.0)	62.0 (8.5-134.5)	86.0 (12.0-190.0)	164.0 (87.0-273.0)	63.5 (16.0-156.0)								
Jun 2020	19.0 (6.0-43.0)	18.0 (5.0-68.0)	36.5 (4.0-95.0)	64.5 (9.0-152.0)	134.0 (72.0-223.0)	41.0 (9.0-120.0)								
Jul 2020	17.0 (4.0-42.5)	16.0 (3.0-63.0)	30.0 (2.0-85.0)	38.5 (1.0-122.0)	115.0 (45.0-207.0)	32.0 (4.0-104.0)								
Aug 2020	20.0 (6.0-47.0)	17.0 (3.0-68.0)	33.0 (3.0-97.0)	58.5 (3.0-156.0)	137.0 (67.0-235.0)	38.0 (6.0-122.0)								
Sept 2020	20.0 (6.0-48.0)	19.0 (5.0-74.0)	40.0 (4.0-103.5)	70.0 (11.0-166.0)	145.0 (76.0-242.0)	45.0 (9.0-131.0)								
Oct 2020	21.0 (6.0-49.0)	20.0 (5.0-80.0)	41.0 (5.0-106.5)	71.5 (11.0-171.0)	149.0 (79.0-254.0)	45.5 (9.0-135.0)								
Nov 2020	24.5 (20.0-51.0)	23.0 (11.0-86.0)	42.5 (10.0-110.0)	76.0 (20.0-177.0)	157.0 (84.0-261.0)	48.0 (20.0-141.0)								
Dec 2020	22.0 (7.0-52.5)	21.0 (5.0-83.0)	42.5 (5.0-111.5)	74.5 (12.0-179.0)	156.0 (81.0-260.0)	47.0 (10.0-141.0)								
Jan 2021	24.0 (19.0-53.5)	23.0 (10.0-84.0)	43.5 (10.0-114.5)	77.0 (17.0-185.0)	164.0 (84.0-273.0)	49.0 (18.0-146.0)								
Feb 2021	26.0 (17.0-58.0)	23.0 (11.0-92.0)	48.0 (10.0-127.0)	83.5 (19.0-193.0)	171.0 (89.0-286.0)	53.0 (18.0-155.0)								
Mar 2021	27.0 (9.0-59.0)	24.0 (6.0-95.0)	49.0 (5.5-131.0)	86.5 (14.0-204.0)	175.0 (93.0-298.0)	55.0 (11.0-161.0)								
Apr 2021	26.5 (9.0-59.0)	25.0 (6.0-97.0)	50.5 (6.0-132.0)	89.5 (14.0-204.0)	178.0 (94.0-300.0)	54.5 (11.0-164.0)								
May 2021	27.0 (9.0-59.0)	24.0 (6.0-97.0)	49.5 (6.0-133.0)	88.5 (15.0-205.0)	178.0 (97.0-299.0)	55.0 (11.0-165.0)								
Jun 2021	26.0 (9.0-59.0)	25.0 (7.0-99.0)	49.5 (6.0-133.5)	89.5 (14.0-212.0)	181.0 (97.0-301.0)	54.0 (11.0-167.0)								
Jul 2021	27.0 (9.0-58.5)	25.0 (7.0-99.0)	48.0 (6.0-133.0)	89.5 (13.0-207.0)	180.0 (100.0-299.0)	55.0 (11.0-166.0)								
Aug 2021	27.0 (8.0-59.5)	26.0 (6.0-100.0)	48.5 (5.5-133.0)	89.5 (13.0-208.0)	176.0 (98.0-294.0)	56.0 (11.0-164.0)								
Sept 2021	27.0 (9.0-58.5)	26.0 (7.0-98.0)	48.5 (6.0-136.0)	91.0 (13.0-214.0)	180.0 (102.0-299.0)	56.0 (12.0-167.0)								
Oct 2021	26.0 (9.0-59.5)	27.0 (6.0-100.0)	49.0 (6.0-136.5)	90.0 (14.0-213.0)	181.0 (100.0-301.0)	56.0 (12.0-166.0)								
Nov 2021	26.0 (8.5-59.5)	27.0 (7.0-102.0)	49.0 (6.0-136.0)	93.0 (14.0-211.0)	185.0 (103.0-305.0)	57.0 (12.0-168.0)								
Dec 2021	26.0 (8.0-58.5)	26.0 (7.0-98.0)	50.0 (6.5-132.0)	93.5 (13.0-211.0)	180.0 (104.0-303.0)	57.0 (11.0-164.0)								
Jan 2022	26.5 (8.0-60.0)	26.0 (6.0-100.0)	50.0 (6.0-133.5)	95.5 (15.0-210.0)	178.0 (106.0-303.0)	57.0 (11.0-165.0)								
Feb 2022	26.5 (8.0-60.0)	25.0 (6.0-100.0)	50.0 (7.0-134.5)	97.0 (13.0-214.0)	174.0 (104.0-297.0)	57.0 (12.0-164.0)								
Mar 2022	27.0 (8.5-60.5)	27.0 (6.0-103.0)	50.0 (6.5-133.5)	95.0 (13.0-217.0)	175.0 (104.0-292.0)	57.0 (11.0-163.0)								

Table S4: number (count) of food outlets accessible online in postcode districts in England, stratified by deprivation ^{a b}.

^b food outlets registered to accept orders online that would deliver to a given postcode district.

	1 (least deprived)2345 (most deprived)England													
Month	·			·										
Nov 2019 (baseline)	_	_	_	_	_	_								
	277(628 120)	222 (571 124)	286 (522 / 1)	100(262.00)	172 (276 / 2)	25.5 (50.0 6.7)								
Juli 2020	-37.7 (-02.0 - 13.0)	-33.3(-37.1-12.4)	-20.0(-33.2-4.1)	-19.0 (-30.2-0.0)	-17.5 (-27.04.2)	-23.3(-30.00.7)								
Jui 2020	-40.5 (-71.414.3)	-33.3 (-64.310.7)	-31.3 (-66.1 - 6.1)	-21.4 (-60.0-0.0)	-17.0 (-32.02.7)	-27.6 (-60.55.9)								
Aug 2020	-34.1 (-64.310.0)	-33.3 (-62.57.7)	-27.7 (-56.1-0.0)	-15.6 (-38.7-10.0)	-12.6 (-27.1-2.2)	-23.5 (-50.0-0.0)								
Sept 2020	-33.3 (-62.58.3)	-29.9 (-55.62.3)	-22.3 (-49.1-0.0)	-11.5 (-31.9-12.2)	-8.8 (-23.1-6.2)	-19.6 (-46.0-0.0)								
Oct 2020	-30.9 (-60.05.7)	-28.6 (-53.8-0.0)	-20.1 (-47.6-2.6)	-8.3 (-30.2-16.7)	-5.8 (-20.6-10.0)	-16.9 (-43.7-5.3)								
Nov 2020	-11.1 (-37.3-21.1)	-10.1 (-40.0-24.8)	-10.9 (-36.5-20.8)	0.6 (-22.2-25.8)	-0.8 (-16.2-14.4)	-5.5 (-30.9-20.8)								
Dec 2020	-28.6 (-55.6-0.0)	-24.2 (-55.0-3.6)	-17.3 (-44.2-6.6)	0.0 (-23.9-22.4)	-1.8 (-17.6-12.9)	-13.0 (-41.9-11.1)								
Jan 2021	-7.1 (-36.5-24.8)	-8.3 (-38.9-30.8)	-6.0 (-33.3-27.2)	5.6 (-19.6-33.6)	2.2 (-14.6-17.5)	-1.5 (-29.1-25.0)								
Feb 2021	-4.8 (-33.3-29.0)	0.0 (-32.0-33.3)	0.0 (-30.2-30.7)	11.0 (-13.8-41.1)	8.0 (-9.0-24.0)	2.4 (-23.8-30.0)								
Mar 2021	-14.8 (-47.5-14.7)	-10.1 (-48.1-21.6)	-5.9 (-36.8-24.3)	11.8 (-14.4-40.4)	10.5 (-7.5-28.4)	0.0 (-31.6-27.4)								
Apr 2021	-14.5 (-48.4-14.8)	-8.3 (-49.2-25.0)	-3.7 (-37.0-26.5)	12.1 (-12.5-43.6)	12.6 (-5.5-30.8)	0.0 (-31.5-28.6)								
May 2021	-14.0 (-47.1-17.4)	-8.3 (-46.0-24.3)	-3.7 (-37.0-26.6)	13.9 (-12.9-43.4)	12.8 (-4.3-31.5)	0.0 (-31.3-29.3)								
Jun 2021	-12.5 (-48.4-20.0)	-8.2 (-45.5-24.1)	-2.4 (-35.3-25.5)	13.1 (-14.6-43.3)	12.7 (-5.5-32.2)	0.0 (-30.8-28.7)								
Jul 2021	-15.2 (-50.0-18.2)	-8.3 (-44.7-25.0)	-4.6 (-37.5-25.2)	14.0 (-14.2-41.4)	12.9 (-6.4-32.6)	0.0 (-31.6-28.4)								
Aug 2021	-16.3 (-50.0-15.4)	-11.3 (-50.0-20.4)	-4.0 (-39.2-25.0)	10.6 (-13.9-43.5)	11.8 (-7.2-31.7)	0.0 (-32.7-28.3)								
Sept 2021	-15.2 (-50.0-18.4)	-8.3 (-49.2-25.0)	-1.8 (-36.8-28.9)	12.8 (-14.7-45.5)	13.9 (-4.5-34.1)	0.0 (-31.0-30.6)								
Oct 2021	-14.5 (-50.0-18.5)	-8.7 (-47.4-25.9)	-2.3 (-38.4-31.4)	12.3 (-14.3-49.2)	14.3 (-4.6-33.6)	0.0 (-31.8-31.7)								
Nov 2021	-14.5 (-49.4-14.3)	-8.8 (-50.0-27.7)	0.0 (-37.0-31.0)	14.6 (-14.8-51.7)	14.7 (-5.8-35.2)	0.0 (-32.3-32.3)								
Dec 2021	-16.1 (-50.0-15.0)	-9.5 (-50.0-26.5)	-4.0 (-36.1-30.9)	13.6 (-15.6-50.0)	14.0 (-6.3-33.4)	0.0 (-32.7-31.7)								
Jan 2022	-13.5 (-48.6-16.7)	-9.3 (-50.0-25.0)	-1.4 (-37.3-31.4)	15.3 (-13.8-52.7)	13.3 (-5.7-34.6)	0.0 (-31.8-31.6)								
Feb 2022	-14.5 (-50.0-19.0)	-9.4 (-50.0-27.3)	-1.8 (-36.0-32.3)	11.8 (-16.7-50.0)	13.2 (-7.9-33.1)	0.0 (-33.3-32.0)								
Mar 2022	-12.7 (-48.6-20.0)	-7.5 (-48.8-28.6)	-1.1 (-35.0-33.2)	13.8 (-18.2-52.5)	13.1 (-7.9-33.8)	0.0 (-32.0-33.3)								

Table S5: change (%) for the number of food outlets accessible online in postcode districts in England, stratified by deprivation ^{a b}.

Notes: ^a data are reported as Median (IQR). ^b food outlets registered to accept orders online that would deliver to a given postcode district.

			Deprivation quint	tile		
	1 (least deprived)	2	3	4	5 (most deprived)	England
Month						
Nov 2019	7.9 (2.2-14.9)	8.7 (1.5-19.0)	12.5 (1.9-23.5)	20.4 (6.5-30.8)	27.8 (19.7-37.4)	14.3 (3.8-26.0)
Jun 2020	11.8 (3.3-21.5)	13.6 (3.0-30.4)	19.0 (3.7-36.2)	31.9 (8.8-50.0)	48.6 (34.5-63.5)	22.2 (6.2-43.5)
Jul 2020	10.5 (2.8-21.1)	13.0 (0.0-31.0)	17.5 (2.4-35.0)	26.7 (5.3-47.6)	47.1 (29.8-61.3)	20.0 (4.3-41.2)
Aug 2020	12.5 (3.9-23.3)	13.6 (2.6-32.8)	19.2 (4.0-39.2)	31.6 (7.7-53.7)	50.9 (36.0-66.7)	22.7 (6.0-46.0)
Sept 2020	9.7 (2.9-18.8)	11.1 (2.5-25.0)	15.5 (3.4-28.6)	23.7 (7.7-37.8)	33.3 (23.6-43.7)	17.4 (5.2-32.7)
Oct 2020	10.5 (3.5-19.6)	11.8 (2.9-26.0)	15.9 (3.8-29.8)	24.0 (7.7-38.5)	34.2 (24.7-45.5)	18.2 (5.6-33.8)
Nov 2020	12.5 (4.2-22.9)	14.3 (3.2-29.7)	18.1 (4.9-33.8)	25.8 (9.1-41.5)	36.0 (26.6-46.7)	20.8 (6.8-36.6)
Dec 2020	10.8 (3.7-20.5)	12.1 (2.6-27.0)	16.4 (3.8-31.4)	24.2 (7.7-39.3)	35.1 (25.7-45.5)	18.6 (5.7-34.8)
Jan 2021	12.5 (4.8-23.5)	14.9 (3.0-29.8)	18.3 (4.7-33.8)	26.3 (8.5-42.2)	37.4 (27.2-48.1)	21.1 (6.9-37.5)
Feb 2021	13.5 (5.5-25.0)	14.9 (3.4-32.3)	19.0 (5.0-36.0)	27.9 (9.4-43.6)	39.2 (28.6-50.0)	22.0 (7.4-39.1)
Mar 2021	12.5 (5.2-23.8)	14.8 (3.4-31.3)	18.8 (4.3-34.3)	27.0 (9.2-44.6)	39.5 (28.4-50.0)	21.6 (7.1-38.6)
Apr 2021	12.5 (4.9-23.9)	14.8 (3.0-31.3)	18.8 (4.6-34.0)	27.7 (8.8-45.5)	40.2 (28.7-50.0)	21.6 (7.1-39.2)
May 2021	12.4 (5.0-23.7)	14.3 (3.4-32.6)	19.3 (4.9-34.4)	28.4 (9.3-46.2)	40.9 (29.3-50.9)	22.1 (7.2-40.0)
Jun 2021	12.5 (4.8-23.7)	14.3 (3.9-32.4)	19.5 (4.9-34.6)	27.9 (9.6-45.6)	40.7 (29.3-51.8)	22.2 (7.1-40.0)
Jul 2021	12.5 (4.5-24.8)	14.3 (4.0-32.6)	20.0 (4.9-34.6)	28.1 (9.6-46.2)	40.7 (29.1-51.3)	22.2 (7.1-40.0)
Aug 2021	12.5 (4.5-23.8)	14.0 (3.9-33.3)	19.8 (4.9-35.3)	29.8 (9.5-45.7)	41.0 (29.3-52.8)	22.6 (7.1-40.3)
Sept 2021	13.0 (4.8-24.8)	14.3 (3.9-32.9)	20.0 (4.9-35.0)	29.4 (9.7-46.1)	40.9 (29.5-51.5)	22.9 (7.2-40.3)
Oct 2021	13.0 (4.9-24.5)	14.8 (3.9-33.3)	20.2 (5.3-35.8)	29.5 (9.7-46.1)	41.2 (29.8-52.0)	23.0 (7.5-40.8)
Nov 2021	13.1 (5.1-24.9)	14.9 (4.0-33.3)	20.0 (5.3-35.9)	30.1 (9.7-46.2)	41.5 (30.8-53.1)	23.2 (7.4-40.9)
Dec 2021	12.5 (4.8-25.2)	14.8 (4.0-33.3)	20.1 (5.3-35.6)	30.0 (9.4-46.2)	40.9 (30.2-53.1)	23.1 (7.2-40.8)
Jan 2022	12.7 (4.8-25.5)	14.8 (4.0-33.0)	20.4 (5.0-36.2)	30.1 (10.6-46.4)	41.5 (30.8-53.2)	23.4 (7.4-40.9)
Feb 2022	12.8 (4.8-25.0)	14.8 (4.2-33.3)	20.7 (5.3-36.6)	31.0 (11.0-46.2)	41.5 (30.2-53.1)	23.3 (7.7-40.8)
Mar 2022	13.2 (4.7-25.9)	14.9 (4.1-33.3)	20.5 (5.3-36.6)	30.8 (11.3-46.6)	41.9 (30.7-52.4)	24.0 (7.7-41.0)

Table S6: percentage of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation ^{ab}.

^b percentage calculated as the number of food outlets registered to accept orders online compared with the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data.

			Deprivation quintile			
	1 (least deprived)	2	3	4	5 (most deprived)	England
Month						
Nov 2019 (baseline)	-	-	-	-	-	-
Jun 2020	44.7 (17.4-81.0)	54.1 (28.3-79.2)	53.5 (30.0-82.9)	62.9 (33.3-92.4)	75.3 (49.2-99.5)	58.1 (31.1-88.8)
Jul 2020	45.0 (14.8-83.9)	56.6 (26.3-84.1)	55.3 (22.2-84.6)	55.5 (20.9-88.5)	69.6 (40.6-99.7)	57.2 (24.0-90.5)
Aug 2020	56.5 (20.8-95.7)	63.6 (31.7-101.6)	66.1 (37.5-101.3)	74.1 (34.0-111.1)	85.1 (52.8-115.3)	69.2 (35.0-107.2)
Sept 2020	22.1 (0.0-54.6)	21.3 (2.3-52.3)	19.9 (2.3-44.0)	22.1 (3.6-40.0)	19.4 (3.4-36.6)	20.5 (2.2-44.0)
Oct 2020	28.7 (0.0-62.5)	26.8 (5.7-59.1)	25.6 (5.2-55.2)	25.4 (6.4-46.3)	23.1 (7.7-39.4)	25.2 (5.2-51.0)
Nov 2020	50.0 (9.0-100.0)	46.9 (16.7-79.3)	40.0 (13.9-76.5)	34.3 (13.6-59.2)	29.5 (14.1-50.7)	38.3 (13.8-69.7)
Dec 2020	33.8 (1.8-71.4)	33.5 (6.2-64.5)	31.9 (5.9-60.9)	27.3 (7.5-52.3)	24.7 (8.0-43.8)	29.2 (6.2-57.5)
Jan 2021	52.9 (10.2-105.7)	50.9 (20.9-89.7)	44.6 (17.1-82.9)	38.2 (13.1-66.7)	33.3 (15.2-53.8)	41.3 (15.5-77.3)
Feb 2021	63.1 (23.3-113.5)	57.1 (26.5-99.0)	46.8 (20.0-88.6)	41.2 (20.8-71.6)	37.9 (20.6-59.5)	46.3 (21.9-85.8)
Mar 2021	55.6 (19.3-102.3)	51.6 (21.7-94.0)	45.1 (14.5-79.8)	43.5 (17.8-70.9)	37.6 (20.2-61.3)	44.6 (18.7-79.2)
Apr 2021	58.7 (20.3-100.0)	50.3 (23.0-94.9)	45.6 (17.9-83.8)	44.4 (19.0-76.3)	39.5 (20.6-63.0)	45.6 (20.0-82.8)
May 2021	61.4 (19.3-103.9)	53.0 (23.1-100.0)	45.2 (19.1-86.6)	47.5 (19.9-81.4)	41.6 (22.8-66.1)	48.4 (20.8-85.7)
Jun 2021	62.5 (15.4-105.6)	51.8 (22.5-100.0)	46.8 (18.6-88.7)	45.0 (20.0-79.7)	43.8 (23.8-65.0)	48.6 (20.4-86.7)
Jul 2021	62.7 (12.8-103.0)	53.0 (22.8-100.0)	46.8 (18.6-89.7)	46.5 (19.7-81.5)	43.1 (23.8-66.7)	49.2 (20.4-87.4)
Aug 2021	63.9 (13.3-105.8)	54.0 (23.0-100.0)	46.7 (14.3-89.6)	47.0 (22.2-84.7)	43.9 (24.2-67.3)	50.0 (20.8-89.6)
Sept 2021	64.8 (16.7-109.8)	54.2 (25.0-105.3)	49.5 (17.2-89.3)	46.7 (22.6-83.0)	43.4 (24.3-70.2)	50.5 (21.8-90.0)
Oct 2021	66.7 (15.3-109.1)	55.9 (25.6-105.3)	50.0 (21.0-90.9)	50.0 (23.4-86.2)	44.0 (25.4-73.0)	51.9 (22.9-91.4)
Nov 2021	67.6 (18.4-112.3)	56.9 (28.0-106.5)	50.0 (19.1-90.9)	51.8 (26.2-86.7)	46.2 (24.4-76.3)	53.0 (23.3-93.8)
Dec 2021	62.3 (14.5-111.9)	58.2 (23.5-102.2)	51.4 (22.6-92.0)	49.1 (26.2-86.8)	44.8 (23.4-74.5)	51.9 (22.5-92.2)
Jan 2022	63.0 (15.3-112.3)	58.9 (26.1-102.2)	53.1 (23.0-100.0)	53.6 (22.0-87.9)	46.3 (24.0-76.3)	53.8 (22.7-93.8)
Feb 2022	66.6 (16.7-113.5)	58.6 (25.0-104.5)	54.3 (21.9-100.0)	54.2 (22.7-84.2)	44.8 (25.0-74.8)	54.2 (23.1-93.9)
Mar 2022	70.7 (16.7-116.0)	62.0 (25.0-106.2)	55.1 (24.0-98.9)	53.0 (24.8-88.9)	44.8 (22.4-75.8)	55.3 (23.0-96.1)

Table S7: change (%) for the percentage of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation ^{a b}.

^b percentage calculated as the number of food outlets registered to accept orders online compared with the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data.

			Deprivation qui	ntile		
	1 (least deprived)	2	3	4	5 (most deprived)	England
Month						
Nov 2019	52.3 (42.0)	59.5 (42.7)	69.3 (65.4)	83.3 (77.1)	100.5 (82.1)	71.9 (65.3)
Jun 2020	40.6 (29.9)	44.6 (29.3)	50.1 (42.9)	56.7 (47.9)	63.0 (48.5)	50.5 (40.8)
Jul 2020	40.6 (29.9)	44.6 (29.3)	50.1 (42.9)	56.7 (47.9)	63.0 (48.5)	50.5 (40.8)
Aug 2020	40.6 (29.9)	44.6 (29.3)	50.1 (42.9)	56.7 (47.9)	63.0 (48.5)	50.5 (40.8)
Sept 2020	52.8 (42.5)	60.2 (43.2)	70.4 (66.4)	84.9 (77.7)	102.6 (82.1)	73.1 (65.9)
Oct 2020	52.8 (42.5)	60.2 (43.2)	70.4 (66.4)	84.9 (77.7)	102.6 (82.1)	73.1 (65.9)
Nov 2020	52.8 (42.5)	60.2 (43.2)	70.4 (66.4)	84.9 (77.7)	102.6 (82.1)	73.1 (65.9)
Dec 2020	53.0 (42.9)	60.6 (43.5)	70.9 (66.9)	85.2 (78.1)	103.1 (82.2)	73.5 (66.2)
Jan 2021	53.0 (42.9)	60.6 (43.5)	70.9 (66.9)	85.2 (78.1)	103.1 (82.2)	73.5 (66.2)
Feb 2021	53.0 (42.9)	60.6 (43.5)	70.9 (66.9)	85.2 (78.1)	103.1 (82.2)	73.5 (66.2)
Mar 2021	53.5 (43.2)	61.2 (43.7)	71.4 (67.0)	86.1 (78.5)	104.3 (83.0)	74.2 (66.7)
Apr 2021	53.5 (43.2)	61.2 (43.7)	71.4 (67.0)	86.1 (78.5)	104.3 (83.0)	74.2 (66.7)
May 2021	53.5 (43.2)	61.2 (43.7)	71.4 (67.0)	86.1 (78.5)	104.3 (83.0)	74.2 (66.7)
Jun 2021	53.8 (43.5)	61.4 (43.9)	71.6 (67.3)	86.6 (78.3)	105.3 (83.2)	74.6 (66.9)
Jul 2021	53.8 (43.5)	61.4 (43.9)	71.6 (67.3)	86.6 (78.3)	105.3 (83.2)	74.6 (66.9)
Aug 2021	53.8 (43.5)	61.4 (43.9)	71.6 (67.3)	86.6 (78.3)	105.3 (83.2)	74.6 (66.9)
Sept 2021	53.8 (43.7)	61.8 (44.4)	72.1 (68.1)	87.7 (80.0)	106.5 (84.1)	75.3 (67.8)
Oct 2021	53.8 (43.7)	61.8 (44.4)	72.1 (68.1)	87.7 (80.0)	106.5 (84.1)	75.3 (67.8)
Nov 2021	53.8 (43.7)	61.8 (44.4)	72.1 (68.1)	87.7 (80.0)	106.5 (84.1)	75.3 (67.8)
Dec 2021	54.1 (43.9)	62.1 (44.6)	72.4 (68.3)	88.1 (80.3)	106.9 (84.3)	75.6 (68.0)
Jan 2022	54.1 (43.9)	62.1 (44.6)	72.4 (68.3)	88.1 (80.3)	106.9 (84.3)	75.6 (68.0)
Feb 2022	54.1 (43.9)	62.1 (44.6)	72.4 (68.3)	88.1 (80.3)	106.9 (84.3)	75.6 (68.0)
Mar 2022	54.1 (43.9)	62.1 (44.6)	72.4 (68.3)	88.1 (80.3)	106.9 (84.3)	75.6 (68.0)

Table S8: number (count) of food outlets in the physical food environment in postcode districts in England, stratified by deprivation ^{a b}.

Notes: ^a data are reported as Mean (standard deviation).

^b the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data, published quarterly.

			Deprivation qui	ntile		
	1 (least deprived)	2	3	4	5 (most deprived)	England
Month						
Nov 2019 (baseline)	-	-	-	-	-	-
Jun 2020	-18.4 (10.5)	-20.9 (11.0)	-22.8 (11.4)	-26.8 (13.1)	-33.4 (11.6)	-24.2 (12.6)
Jul 2020	-18.4 (10.5)	-20.9 (11.0)	-22.8 (11.4)	-26.8 (13.1)	-33.4 (11.6)	-24.2 (12.6)
Aug 2020	-18.4 (10.5)	-20.9 (11.0)	-22.8 (11.4)	-26.8 (13.1)	-33.4 (11.6)	-24.2 (12.6)
Sept 2020	1.0 (9.0)	1.1 (8.0)	1.7 (9.6)	2.5 (10.1)	4.0 (11.7)	2.0 (9.7)
Oct 2020	1.0 (9.0)	1.1 (8.0)	1.7 (9.6)	2.5 (10.1)	4.0 (11.7)	2.0 (9.7)
Nov 2020	1.0 (9.0)	1.1 (8.0)	1.7 (9.6)	2.5 (10.1)	4.0 (11.7)	2.0 (9.7)
Dec 2020	1.4 (9.5)	1.7 (8.2)	2.6 (9.6)	2.9 (11.3)	4.7 (11.7)	2.6 (10.1)
Jan 2021	1.4 (9.5)	1.7 (8.2)	2.6 (9.6)	2.9 (11.3)	4.7 (11.7)	2.6 (10.1)
Feb 2021	1.4 (9.5)	1.7 (8.2)	2.6 (9.6)	2.9 (11.3)	4.7 (11.7)	2.6 (10.1)
Mar 2021	2.2 (10.3)	2.8 (9.0)	3.3 (9.9)	4.4 (12.7)	6.0 (12.4)	3.7 (10.9)
Apr 2021	2.2 (10.3)	2.8 (9.0)	3.3 (9.9)	4.4 (12.7)	6.0 (12.4)	3.7 (10.9)
May 2021	2.2 (10.3)	2.8 (9.0)	3.3 (9.9)	4.4 (12.7)	6.0 (12.4)	3.7 (10.9)
Jun 2021	2.6 (10.6)	3.1 (9.1)	3.6 (10.0)	5.1 (12.9)	7.3 (13.2)	4.3 (11.3)
Jul 2021	2.6 (10.6)	3.1 (9.1)	3.6 (10.0)	5.1 (12.9)	7.3 (13.2)	4.3 (11.3)
Aug 2021	2.6 (10.6)	3.1 (9.1)	3.6 (10.0)	5.1 (12.9)	7.3 (13.2)	4.3 (11.3)
Sept 2021	2.5 (11.2)	3.6 (10.2)	4.1 (10.6)	6.1 (13.3)	8.5 (13.9)	4.9 (12.0)
Oct 2021	2.5 (11.2)	3.6 (10.2)	4.1 (10.6)	6.1 (13.3)	8.5 (13.9)	4.9 (12.0)
Nov 2021	2.5 (11.2)	3.6 (10.2)	4.1 (10.6)	6.1 (13.3)	8.5 (13.9)	4.9 (12.0)
Dec 2021	3.1 (11.6)	4.1 (10.3)	4.5 (10.7)	6.6 (13.5)	9.0 (14.1)	5.3 (12.2)
Jan 2022	3.1 (11.6)	4.1 (10.3)	4.5 (10.7)	6.6 (13.5)	9.0 (14.1)	5.3 (12.2)
Feb 2022	3.1 (11.6)	4.1 (10.3)	4.5 (10.7)	6.6 (13.5)	9.0 (14.1)	5.3 (12.2)
Mar 2022	3.1 (11.6)	4.1 (10.3)	4.5 (10.7)	6.6 (13.5)	9.0 (14.1)	5.3 (12.2)

Table S9: change (%) for the number of food outlets in the physical food environment in postcode districts in England, stratified by deprivation ^{a b}.

Notes: ^a data are reported as Mean (standard deviation).

^b the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data, published quarterly.

	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England																	
	1 (least d	deprived)		2			3			4			5 (most o	deprived)		Englan	d	
	IRR	9	5% CI	IRR	95	5% CI	IRR	9	5% CI	IRR	95	5% CI	IRR	9	5% CI	IRR	9	5% CI
Month																		
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref
Jun 2020	1.18	1.16	1.20	1.18	1.16	1.20	1.17	1.15	1.19	1.16	1.13	1.18	1.13	1.11	1.16	1.15	1.14	1.16
Jul 2020	1.18	1.16	1.20	1.17	1.15	1.19	1.10	1.08	1.12	1.03	1.01	1.06	1.06	1.04	1.08	1.09	1.08	1.10
Aug 2020	1.29	1.27	1.32	1.27	1.25	1.30	1.23	1.21	1.25	1.22	1.19	1.25	1.19	1.17	1.22	1.23	1.22	1.24
Sept 2020	1.32	1.30	1.35	1.31	1.29	1.34	1.29	1.27	1.32	1.29	1.26	1.32	1.25	1.22	1.27	1.28	1.27	1.29
Oct 2020	1.38	1.36	1.41	1.38	1.35	1.40	1.36	1.33	1.38	1.35	1.32	1.38	1.29	1.27	1.32	1.34	1.32	1.35
Nov 2020	1.57	1.54	1.60	1.55	1.52	1.58	1.50	1.47	1.52	1.44	1.41	1.48	1.36	1.33	1.39	1.45	1.43	1.46
Dec 2020	1.45	1.42	1.47	1.45	1.42	1.48	1.43	1.40	1.46	1.40	1.37	1.43	1.33	1.31	1.36	1.39	1.38	1.40
Jan 2021	1.61	1.58	1.64	1.59	1.56	1.62	1.53	1.50	1.55	1.49	1.45	1.52	1.41	1.38	1.44	1.49	1.48	1.50
Feb 2021	1.69	1.66	1.72	1.66	1.63	1.69	1.62	1.59	1.65	1.55	1.52	1.59	1.46	1.43	1.50	1.56	1.54	1.57
Mar 2021	1.68	1.65	1.71	1.67	1.64	1.70	1.61	1.58	1.64	1.56	1.53	1.60	1.49	1.46	1.52	1.57	1.56	1.58
Apr 2021	1.69	1.66	1.72	1.67	1.64	1.70	1.63	1.60	1.66	1.59	1.56	1.63	1.52	1.48	1.55	1.59	1.58	1.60
May 2021	1.71	1.68	1.74	1.69	1.66	1.73	1.65	1.62	1.68	1.62	1.58	1.66	1.53	1.50	1.57	1.61	1.60	1.62
Jun 2021	1.72	1.69	1.75	1.70	1.67	1.73	1.65	1.62	1.68	1.63	1.60	1.67	1.55	1.52	1.59	1.62	1.61	1.64
Jul 2021	1.73	1.70	1.76	1.71	1.67	1.74	1.66	1.63	1.69	1.64	1.60	1.68	1.56	1.52	1.59	1.63	1.62	1.64
Aug 2021	1.75	1.72	1.78	1.72	1.69	1.75	1.67	1.63	1.70	1.66	1.62	1.70	1.56	1.53	1.60	1.64	1.63	1.65
Sept 2021	1.77	1.74	1.80	1.74	1.71	1.77	1.68	1.65	1.71	1.68	1.65	1.72	1.58	1.55	1.62	1.66	1.65	1.68
Oct 2021	1.78	1.75	1.81	1.75	1.72	1.78	1.70	1.66	1.73	1.70	1.66	1.74	1.59	1.56	1.63	1.67	1.66	1.69
Nov 2021	1.80	1.77	1.83	1.76	1.73	1.80	1.71	1.68	1.74	1.71	1.67	1.75	1.61	1.58	1.65	1.69	1.67	1.70
Dec 2021	1.79	1.76	1.82	1.75	1.72	1.78	1.71	1.68	1.74	1.70	1.66	1.74	1.61	1.57	1.64	1.68	1.67	1.70
Jan 2022	1.81	1.78	1.84	1.76	1.73	1.80	1.73	1.69	1.76	1.71	1.67	1.75	1.62	1.58	1.65	1.70	1.68	1.71
Feb 2022	1.82	1.79	1.85	1.76	1.73	1.80	1.73	1.70	1.76	1.71	1.68	1.76	1.61	1.57	1.64	1.69	1.68	1.71
Mar 2022	1.84	1.81	1.88	1.78	1.75	1.81	1.75	1.71	1.78	1.72	1.68	1.76	1.61	1.57	1.64	1.70	1.69	1.72

Table S10: Incidence Rate Ratios (IRR) and 95% confidence intervals (CI) for the number (count) of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation, estimated using an unadjusted negative binomial generalised estimating equation.

	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England																		
	1 (least o	leprived)		2			3			4			5 (most	deprived)		Englar	ıd		
	IRR	95	5% CI	IRR	9	5% CI	IRR	9	5% CI	IRR	9	5% CI	IRR	9	5% CI	IRR	95	5% CI	
Month																			
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	
Jun 2020	1.22	1.17	1.28	1.24	1.19	1.28	1.21	1.16	1.26	1.19	1.14	1.23	1.14	1.11	1.17	1.18	1.16	1.19	
Jul 2020	1.19	1.14	1.24	1.22	1.18	1.27	1.16	1.11	1.21	1.08	1.04	1.12	1.07	1.05	1.10	1.11	1.10	1.13	
Aug 2020	1.32	1.26	1.38	1.31	1.27	1.36	1.28	1.23	1.34	1.25	1.20	1.30	1.20	1.16	1.23	1.24	1.22	1.26	
Sept 2020	1.30	1.24	1.35	1.30	1.25	1.34	1.27	1.22	1.33	1.27	1.22	1.31	1.24	1.21	1.27	1.26	1.25	1.28	
Oct 2020	1.38	1.32	1.44	1.36	1.32	1.41	1.34	1.29	1.40	1.32	1.27	1.37	1.28	1.25	1.31	1.32	1.31	1.34	
Nov 2020	1.61	1.55	1.68	1.58	1.53	1.64	1.55	1.48	1.61	1.41	1.36	1.47	1.35	1.31	1.38	1.47	1.45	1.50	
Dec 2020	1.43	1.38	1.50	1.40	1.35	1.45	1.40	1.34	1.46	1.34	1.30	1.39	1.31	1.28	1.35	1.36	1.34	1.38	
Jan 2021	1.66	1.59	1.73	1.60	1.54	1.65	1.59	1.53	1.66	1.45	1.40	1.51	1.39	1.35	1.42	1.51	1.49	1.53	
Feb 2021	1.74	1.67	1.82	1.68	1.62	1.74	1.64	1.58	1.71	1.52	1.47	1.57	1.44	1.41	1.48	1.57	1.55	1.60	
Mar 2021	1.70	1.63	1.77	1.65	1.59	1.71	1.59	1.53	1.66	1.52	1.46	1.57	1.47	1.43	1.50	1.56	1.54	1.58	
Apr 2021	1.70	1.63	1.77	1.64	1.58	1.69	1.60	1.54	1.67	1.55	1.49	1.60	1.49	1.45	1.53	1.57	1.55	1.60	
May 2021	1.71	1.64	1.79	1.67	1.61	1.73	1.62	1.55	1.68	1.57	1.51	1.62	1.51	1.47	1.55	1.59	1.57	1.62	
Jun 2021	1.71	1.64	1.78	1.67	1.61	1.73	1.62	1.56	1.69	1.58	1.52	1.63	1.53	1.49	1.57	1.60	1.58	1.63	
Jul 2021	1.71	1.64	1.78	1.67	1.61	1.73	1.64	1.58	1.71	1.59	1.53	1.65	1.53	1.49	1.57	1.61	1.59	1.63	
Aug 2021	1.72	1.65	1.80	1.67	1.62	1.73	1.64	1.58	1.71	1.61	1.56	1.67	1.54	1.50	1.57	1.62	1.60	1.64	
Sept 2021	1.74	1.67	1.81	1.69	1.63	1.75	1.66	1.60	1.73	1.64	1.58	1.70	1.56	1.52	1.60	1.64	1.62	1.66	
Oct 2021	1.74	1.67	1.81	1.71	1.65	1.77	1.68	1.62	1.75	1.66	1.60	1.72	1.57	1.53	1.61	1.65	1.63	1.68	
Nov 2021	1.76	1.69	1.84	1.72	1.66	1.78	1.70	1.63	1.77	1.67	1.61	1.73	1.59	1.55	1.63	1.67	1.65	1.69	
Dec 2021	1.75	1.68	1.82	1.69	1.63	1.76	1.70	1.63	1.77	1.66	1.60	1.72	1.59	1.55	1.62	1.66	1.64	1.69	
Jan 2022	1.77	1.69	1.84	1.71	1.65	1.77	1.72	1.65	1.79	1.66	1.60	1.72	1.59	1.56	1.63	1.67	1.65	1.70	
Feb 2022	1.77	1.70	1.85	1.73	1.67	1.79	1.72	1.65	1.79	1.68	1.62	1.74	1.59	1.55	1.63	1.68	1.66	1.70	
Mar 2022	1.80	1.73	1.88	1.73	1.67	1.80	1.73	1.66	1.80	1.68	1.62	1.74	1.59	1.55	1.63	1.69	1.66	1.71	

Table S11: Incidence Rate Ratios (IRR) and 95% confidence intervals (CI) for the number (count) of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation, estimated using an adjusted negative binomial generalised estimating equation ^a.

Note: ^a adjusted for population density, rural urban classification and the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data. 2067 postcode districts included.

	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England																	
	1 (least o	deprived)		2			3			4			5 (most	deprived)		Englar	nd	
	IRR	9	5% CI	IRR	9	5% CI	IRR	95% CI		IRR	9	5% CI	IRR	9	5% CI	IRR	9	5% CI
Month																		
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref
Jun 2020	0.71	0.70	0.73	0.82	0.80	0.84	0.77	0.76	0.79	0.84	0.82	0.86	0.82	0.80	0.85	0.80	0.80	0.81
Jul 2020	0.68	0.67	0.69	0.75	0.74	0.77	0.72	0.70	0.73	0.68	0.66	0.70	0.71	0.69	0.73	0.71	0.70	0.71
Aug 2020	0.78	0.76	0.79	0.87	0.85	0.89	0.82	0.80	0.83	0.89	0.86	0.91	0.86	0.83	0.88	0.85	0.84	0.86
Sept 2020	0.81	0.79	0.82	0.96	0.94	0.98	0.89	0.87	0.91	0.97	0.94	1.00	0.91	0.89	0.94	0.92	0.91	0.92
Oct 2020	0.84	0.82	0.85	1.01	0.99	1.03	0.94	0.92	0.96	1.02	0.99	1.05	0.95	0.92	0.97	0.96	0.95	0.97
Nov 2020	0.95	0.93	0.96	1.09	1.07	1.11	1.00	0.98	1.02	1.08	1.05	1.11	0.98	0.95	1.01	1.02	1.01	1.03
Dec 2020	0.89	0.88	0.91	1.07	1.05	1.10	1.00	0.98	1.02	1.08	1.05	1.11	0.98	0.95	1.01	1.01	1.00	1.02
Jan 2021	0.96	0.94	0.98	1.12	1.09	1.14	1.04	1.01	1.06	1.12	1.09	1.15	1.02	0.99	1.05	1.05	1.04	1.06
Feb 2021	1.01	0.99	1.03	1.18	1.15	1.20	1.09	1.07	1.11	1.17	1.14	1.20	1.08	1.04	1.11	1.11	1.10	1.12
Mar 2021	1.00	0.98	1.02	1.19	1.16	1.21	1.11	1.09	1.13	1.20	1.17	1.23	1.11	1.08	1.14	1.13	1.12	1.14
Apr 2021	1.02	1.00	1.04	1.21	1.18	1.23	1.13	1.11	1.15	1.22	1.19	1.26	1.13	1.10	1.17	1.15	1.14	1.16
May 2021	1.02	1.01	1.04	1.22	1.20	1.25	1.14	1.12	1.17	1.24	1.21	1.27	1.13	1.10	1.17	1.16	1.15	1.17
Jun 2021	1.01	0.99	1.03	1.19	1.16	1.21	1.12	1.10	1.14	1.21	1.18	1.25	1.13	1.09	1.16	1.14	1.13	1.15
Jul 2021	1.02	1.00	1.04	1.19	1.17	1.22	1.12	1.10	1.15	1.22	1.19	1.25	1.13	1.10	1.16	1.14	1.13	1.15
Aug 2021	1.01	1.00	1.03	1.19	1.16	1.21	1.11	1.09	1.13	1.21	1.18	1.25	1.12	1.09	1.15	1.14	1.12	1.15
Sept 2021	1.03	1.01	1.05	1.21	1.19	1.24	1.14	1.12	1.17	1.24	1.21	1.28	1.14	1.10	1.17	1.16	1.15	1.17
Oct 2021	1.03	1.01	1.05	1.20	1.18	1.23	1.14	1.11	1.16	1.24	1.20	1.27	1.14	1.11	1.17	1.16	1.14	1.17
Nov 2021	1.03	1.01	1.05	1.21	1.18	1.24	1.14	1.12	1.16	1.24	1.21	1.27	1.14	1.11	1.18	1.16	1.15	1.17
Dec 2021	1.02	1.00	1.04	1.20	1.17	1.22	1.13	1.10	1.15	1.22	1.19	1.26	1.13	1.10	1.16	1.15	1.13	1.16
Jan 2022	1.02	1.00	1.04	1.20	1.18	1.23	1.13	1.11	1.16	1.23	1.20	1.26	1.13	1.10	1.17	1.15	1.14	1.16
Feb 2022	1.01	0.99	1.03	1.18	1.15	1.21	1.11	1.09	1.13	1.21	1.18	1.24	1.11	1.08	1.14	1.13	1.12	1.14
Mar 2022	1.02	1.00	1.04	1.18	1.16	1.21	1.11	1.09	1.14	1.21	1.18	1.24	1.11	1.08	1.15	1.13	1.12	1.15

Table S12: Incidence Rate Ratios (IRR) and 95% confidence intervals (CI) for the number (count) of food outlets accessible online in postcode districts in England, stratified by deprivation, estimated using an unadjusted negative binomial generalised estimating equation.

	Deprivation quintile 1 (least deprived) 2 3 4 5 (most deprived) England																	
	1 (least o	leprived)		2			3			4			5 (most d	leprived)		Englar	nd	
	IRR	9	5% CI	IRR	95	% CI	CI IRR 95% CI IRR 95% CI		5% CI	IRR	95% CI		IRR	95% CI				
Month																		
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref
Jun 2020	0.57	0.55	0.59	0.58	0.56	0.60	0.66	0.64	0.68	0.74	0.71	0.77	0.84	0.81	0.87	0.67	0.66	0.68
Jul 2020	0.55	0.54	0.57	0.55	0.53	0.57	0.62	0.60	0.64	0.61	0.58	0.63	0.73	0.70	0.75	0.60	0.59	0.61
Aug 2020	0.61	0.59	0.63	0.61	0.59	0.63	0.69	0.67	0.71	0.77	0.74	0.80	0.87	0.84	0.90	0.70	0.69	0.71
Sept 2020	0.63	0.61	0.65	0.64	0.62	0.66	0.75	0.73	0.77	0.82	0.79	0.85	0.90	0.87	0.93	0.73	0.72	0.75
Oct 2020	0.65	0.64	0.67	0.67	0.65	0.69	0.78	0.76	0.81	0.86	0.83	0.89	0.93	0.91	0.96	0.77	0.75	0.78
Nov 2020	0.86	0.83	0.88	0.82	0.80	0.85	0.93	0.90	0.96	0.97	0.93	1.01	0.97	0.94	1.00	0.91	0.90	0.92
Dec 2020	0.69	0.68	0.71	0.70	0.68	0.73	0.83	0.80	0.85	0.91	0.87	0.94	0.97	0.94	1.00	0.81	0.79	0.82
Jan 2021	0.85	0.83	0.87	0.83	0.81	0.86	0.97	0.95	1.00	0.99	0.96	1.03	1.01	0.98	1.04	0.93	0.91	0.94
Feb 2021	0.88	0.85	0.90	0.86	0.84	0.89	1.01	0.98	1.04	1.04	1.00	1.08	1.06	1.03	1.10	0.96	0.95	0.98
Mar 2021	0.79	0.77	0.81	0.79	0.76	0.82	0.93	0.90	0.95	1.02	0.98	1.06	1.10	1.06	1.13	0.91	0.90	0.92
Apr 2021	0.80	0.78	0.82	0.81	0.78	0.83	0.95	0.92	0.97	1.03	1.00	1.07	1.12	1.08	1.15	0.92	0.91	0.94
May 2021	0.81	0.79	0.83	0.81	0.79	0.84	0.95	0.92	0.98	1.05	1.01	1.09	1.12	1.09	1.16	0.93	0.92	0.95
Jun 2021	0.80	0.78	0.83	0.80	0.78	0.83	0.94	0.91	0.97	1.02	0.99	1.06	1.11	1.08	1.15	0.92	0.91	0.93
Jul 2021	0.80	0.78	0.82	0.80	0.78	0.83	0.94	0.91	0.96	1.03	0.99	1.06	1.11	1.08	1.15	0.92	0.91	0.93
Aug 2021	0.80	0.78	0.82	0.79	0.77	0.82	0.92	0.90	0.95	1.02	0.98	1.06	1.10	1.07	1.14	0.91	0.90	0.93
Sept 2021	0.81	0.79	0.83	0.81	0.78	0.84	0.96	0.93	0.98	1.04	1.00	1.08	1.12	1.09	1.16	0.93	0.92	0.95
Oct 2021	0.81	0.79	0.83	0.81	0.78	0.83	0.95	0.93	0.98	1.04	1.00	1.08	1.12	1.09	1.16	0.93	0.92	0.94
Nov 2021	0.81	0.79	0.83	0.81	0.78	0.84	0.96	0.93	0.99	1.04	1.00	1.08	1.13	1.09	1.16	0.93	0.92	0.95
Dec 2021	0.80	0.78	0.82	0.80	0.77	0.83	0.95	0.92	0.98	1.03	0.99	1.07	1.11	1.08	1.15	0.92	0.91	0.94
Jan 2022	0.80	0.78	0.82	0.80	0.78	0.83	0.95	0.93	0.98	1.03	1.00	1.07	1.12	1.08	1.15	0.93	0.91	0.94
Feb 2022	0.80	0.78	0.82	0.80	0.77	0.82	0.95	0.92	0.98	1.02	0.98	1.06	1.10	1.06	1.13	0.92	0.90	0.93
Mar 2022	0.81	0.79	0.83	0.80	0.78	0.83	0.95	0.92	0.98	1.03	0.99	1.07	1.10	1.07	1.13	0.92	0.91	0.93

Table S13: Incidence Rate Ratios (IRR) and 95% confidence intervals (CI) for the number (count) of food outlets accessible online in postcode districts in England, stratified by deprivation, estimated using an adjusted negative binomial generalised estimating equation ^a.

Note: ^a adjusted for population density, rural urban classification and the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data. 2067 postcode districts included.

	Deprivation quintile 1 (least deprived) 2																	
	1 ^{(least d}	leprived)		2			3			4			5 (most d	leprived)		Englan	d	
	coef.	95	5% CI	coef.	9	5% CI	coef.	95	5% CI	coef.	95	5% CI	coef.	9	5% CI	coef.	9	5% CI
Month																		
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref
Jun 2020	0.47	0.43	0.51	0.54	0.50	0.58	0.54	0.51	0.58	0.64	0.60	0.69	0.83	0.79	0.88	0.60	0.58	0.62
Jul 2020	0.45	0.41	0.50	0.52	0.47	0.57	0.48	0.43	0.53	0.52	0.45	0.58	0.69	0.62	0.76	0.53	0.50	0.55
Aug 2020	0.56	0.52	0.60	0.63	0.57	0.68	0.61	0.56	0.66	0.69	0.63	0.75	0.92	0.86	0.97	0.67	0.65	0.69
Sept 2020	0.28	0.24	0.32	0.30	0.26	0.33	0.26	0.22	0.29	0.25	0.22	0.29	0.23	0.20	0.26	0.25	0.23	0.26
Oct 2020	0.34	0.31	0.38	0.35	0.31	0.39	0.31	0.27	0.35	0.30	0.26	0.33	0.28	0.25	0.32	0.30	0.29	0.32
Nov 2020	0.51	0.46	0.55	0.51	0.46	0.56	0.45	0.41	0.49	0.39	0.35	0.43	0.36	0.33	0.40	0.42	0.40	0.44
Dec 2020	0.39	0.35	0.44	0.40	0.35	0.44	0.35	0.32	0.39	0.33	0.29	0.38	0.31	0.28	0.35	0.34	0.32	0.36
Jan 2021	0.54	0.48	0.59	0.53	0.48	0.58	0.47	0.42	0.51	0.42	0.38	0.47	0.40	0.36	0.44	0.44	0.42	0.46
Feb 2021	0.60	0.55	0.65	0.59	0.54	0.64	0.53	0.49	0.57	0.49	0.45	0.54	0.47	0.43	0.50	0.51	0.49	0.53
Mar 2021	0.56	0.52	0.61	0.57	0.52	0.61	0.50	0.46	0.55	0.48	0.43	0.53	0.47	0.43	0.51	0.49	0.47	0.51
Apr 2021	0.58	0.53	0.62	0.57	0.52	0.62	0.52	0.48	0.57	0.51	0.46	0.56	0.50	0.46	0.54	0.51	0.49	0.53
May 2021	0.59	0.54	0.63	0.59	0.54	0.64	0.54	0.49	0.58	0.53	0.48	0.58	0.52	0.48	0.56	0.53	0.51	0.55
Jun 2021	0.58	0.54	0.63	0.58	0.54	0.63	0.54	0.50	0.58	0.53	0.48	0.58	0.52	0.48	0.56	0.53	0.51	0.55
Jul 2021	0.59	0.54	0.63	0.59	0.54	0.63	0.54	0.50	0.59	0.54	0.49	0.58	0.53	0.49	0.57	0.53	0.51	0.55
Aug 2021	0.61	0.56	0.66	0.60	0.55	0.64	0.55	0.50	0.59	0.56	0.51	0.61	0.53	0.49	0.58	0.54	0.52	0.56
Sept 2021	0.62	0.57	0.66	0.59	0.55	0.64	0.55	0.51	0.60	0.56	0.51	0.62	0.54	0.49	0.58	0.55	0.53	0.57
Oct 2021	0.62	0.57	0.67	0.61	0.56	0.65	0.57	0.52	0.62	0.58	0.53	0.63	0.55	0.51	0.59	0.56	0.54	0.58
Nov 2021	0.63	0.59	0.68	0.62	0.57	0.66	0.58	0.53	0.63	0.58	0.53	0.63	0.58	0.53	0.62	0.57	0.55	0.59
Dec 2021	0.62	0.57	0.67	0.60	0.55	0.65	0.58	0.53	0.62	0.57	0.52	0.62	0.56	0.52	0.61	0.56	0.54	0.58
Jan 2022	0.63	0.58	0.68	0.61	0.57	0.66	0.59	0.54	0.63	0.57	0.52	0.62	0.57	0.53	0.62	0.57	0.55	0.59
Feb 2022	0.63	0.59	0.68	0.62	0.58	0.67	0.59	0.54	0.64	0.58	0.53	0.63	0.56	0.51	0.60	0.57	0.55	0.59
Mar 2022	0.65	0.60	0.70	0.63	0.58	0.68	0.60	0.55	0.65	0.58	0.53	0.63	0.56	0.52	0.61	0.58	0.56	0.60

Table S14: Coefficients (coef.) and 95% confidence intervals (CI) for the percentage of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation, calculated using an unadjusted generalised estimating equation ^a.

Note: ^a percentage calculated as the number of food outlets registered to accept orders online compared with the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data. 2115 postcode districts included.

	Deprivation quintile																		
	1 (least o	1 (least deprived)			2			3			4			5 (most deprived)			England		
	coef.	95% CI		coef.	95% CI		coef.	95% CI		coef.	95% CI		coef.	95% CI		coef.	95% CI		
Month																			
Nov 2019	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	ref	
Jun 2020	0.50	0.46	0.54	0.59	0.54	0.63	0.59	0.55	0.64	0.72	0.67	0.78	0.85	0.81	0.90	0.67	0.64	0.69	
Jul 2020	0.48	0.44	0.53	0.57	0.51	0.62	0.53	0.47	0.59	0.58	0.50	0.66	0.71	0.64	0.78	0.58	0.55	0.61	
Aug 2020	0.60	0.55	0.64	0.68	0.62	0.74	0.67	0.61	0.72	0.77	0.70	0.85	0.94	0.88	1.00	0.74	0.72	0.77	
Sept 2020	0.30	0.26	0.34	0.32	0.28	0.36	0.28	0.24	0.31	0.28	0.24	0.32	0.23	0.20	0.27	0.27	0.26	0.29	
Oct 2020	0.37	0.33	0.41	0.38	0.34	0.43	0.33	0.29	0.37	0.33	0.29	0.37	0.29	0.26	0.32	0.33	0.31	0.35	
Nov 2020	0.54	0.49	0.59	0.56	0.50	0.61	0.48	0.44	0.53	0.43	0.39	0.48	0.37	0.33	0.40	0.46	0.44	0.48	
Dec 2020	0.42	0.37	0.46	0.43	0.38	0.49	0.38	0.34	0.43	0.37	0.32	0.42	0.32	0.28	0.36	0.37	0.35	0.39	
Jan 2021	0.57	0.52	0.62	0.58	0.52	0.63	0.50	0.45	0.55	0.47	0.42	0.53	0.41	0.37	0.44	0.49	0.47	0.51	
Feb 2021	0.63	0.58	0.69	0.65	0.60	0.71	0.58	0.53	0.62	0.55	0.50	0.60	0.48	0.44	0.51	0.56	0.54	0.58	
Mar 2021	0.60	0.55	0.65	0.62	0.57	0.67	0.55	0.50	0.59	0.54	0.48	0.59	0.48	0.44	0.52	0.54	0.52	0.57	
Apr 2021	0.61	0.56	0.66	0.63	0.57	0.68	0.57	0.52	0.62	0.57	0.51	0.63	0.51	0.46	0.55	0.56	0.54	0.58	
May 2021	0.62	0.57	0.67	0.65	0.60	0.70	0.58	0.53	0.63	0.59	0.53	0.65	0.53	0.49	0.57	0.58	0.56	0.60	
Jun 2021	0.62	0.57	0.66	0.64	0.59	0.69	0.59	0.54	0.63	0.59	0.54	0.65	0.53	0.49	0.58	0.58	0.56	0.60	
Jul 2021	0.62	0.57	0.67	0.64	0.59	0.70	0.59	0.54	0.64	0.60	0.54	0.66	0.54	0.49	0.58	0.59	0.56	0.61	
Aug 2021	0.64	0.59	0.69	0.65	0.60	0.70	0.60	0.54	0.65	0.62	0.57	0.68	0.54	0.50	0.59	0.60	0.58	0.62	
Sept 2021	0.65	0.60	0.70	0.65	0.60	0.70	0.60	0.55	0.65	0.63	0.57	0.69	0.55	0.50	0.59	0.60	0.58	0.63	
Oct 2021	0.66	0.61	0.71	0.66	0.61	0.72	0.62	0.57	0.67	0.64	0.59	0.70	0.56	0.52	0.60	0.62	0.59	0.64	
Nov 2021	0.67	0.62	0.72	0.67	0.62	0.73	0.63	0.58	0.68	0.65	0.59	0.71	0.59	0.54	0.63	0.63	0.61	0.65	
Dec 2021	0.66	0.61	0.71	0.66	0.60	0.71	0.63	0.57	0.68	0.63	0.57	0.70	0.57	0.53	0.62	0.62	0.59	0.64	
Jan 2022	0.67	0.62	0.72	0.66	0.61	0.71	0.64	0.59	0.69	0.64	0.58	0.70	0.58	0.54	0.63	0.63	0.60	0.65	
Feb 2022	0.67	0.62	0.72	0.68	0.63	0.73	0.64	0.59	0.70	0.65	0.59	0.71	0.57	0.52	0.61	0.63	0.60	0.65	
Mar 2022	0.69	0.64	0.74	0.69	0.63	0.74	0.66	0.60	0.71	0.65	0.59	0.71	0.57	0.52	0.62	0.64	0.61	0.66	

Table S15: Coefficients (coef.) and 95% confidence intervals (CI) for the percentage of food outlets registered to accept orders online in postcode districts in England, stratified by deprivation, calculated using an adjusted generalised estimating equation ^a.

Note: ^a percentage calculated as the number of food outlets registered to accept orders online compared with the number of food outlets with a customer-facing premises in a postcode district, based on Ordnance Survey Points of Interest data. Adjusted for population density and rural urban classification. 2065 postcode districts included.

Page intentionally-left-blank